Treatment of Primary Influent Wastewater during Wet Weather Flow by Powdered Activated Carbon – Catalyzed Ozonation: From Fundamentals to Application

by Mirna Alameddine

A thesis submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

in

Environmental Science

Department of Civil and Environmental Engineering University of Alberta

© Mirna Alameddine, 2020

ABSTRACT

Cities equipped with combined sewers to accommodate domestic and industrial wastewater as well as storm water runoffs, encounter the need to prevent seasonal overflows into water bodies. An extensive literature review pertaining to the treatment of primary influent (PI) during wet weather conditions determined that a coagulation/ flocculation/ sedimentation process was required for at least the removal of suspended solids. Ozonation was deemed promising especially for the oxidation of micropollutants (MPs) and disinfection. Yet, although ozone (O_3) is a renowned strong oxidant, several MPs are not sufficiently reactive with O₃ which calls for a process upgrade. Powdered activated carbon (PAC) was favourable to catalyze the ozonation because of its adsorbing capacity. There has been limited and conflicting knowledge pertaining to PAC-catalyzed ozonation especially in the ambiguously described catalytic mechanisms. The objective of this study was to propose a complete treatment for PI during wet weather flow by optimizing the operational parameters in conventional primary treatment and exploring single or PAC-catalyzed ozonation as a supplementary step for MPs' degradation and disinfection. This was addressed through a meticulous investigation of the catalytic mechanisms using a mixture of pharmaceuticals, personal care products, herbicides and perfluorinated compounds which have different affinities towards O₃.

Initially, the enhanced primary treatment of municipal PI during wet weather conditions was studied through a comprehensive approach from bench to full scale. Three metal-based coagulants were tested in a series of jar tests. Aluminum sulfate (alum) outperformed other coagulants since 1 mg of aluminum added as alum with low mixing was able to remove 22 Nephelometric Turbidity Units (NTUs), 19 mg of oxygen as chemical oxygen demand (COD) and 0.8 mg of orthophosphate (ortho-P). The removal of total suspended solids (TSS) depended mostly on rapid

mixing while COD and ortho-P removals depended on slow mixing and coagulant dose. In bench and full-scale operations, the addition of polymer did not lead to any pronounced improvements. Turbidity and percent ultraviolet transmittance showed good correlation with TSS and ortho-P which evoked their use as surrogates for online process control.

PAC was then used to catalyze the ozonation of a mixture of seventeen MPs at near environmental concentrations in clean water matrix. Thirteen of those MPs were studied for the first time by PAC-catalyzed ozonation. The latter improved the removal of O₃-resistant MPs by up to 27% and promoted faster specific degradation rates. Radical probing experiments showed that scavenging hydroxyl radical ('OH) did not have a significant impact on MPs removals, while scavenging other reactive oxygen species was more influential. A detailed study by Electron Paramagnetic Resonance (EPR) spectroscopy ascertained that the decomposition of O₃ in presence of PAC at neutral pH did not boost the generation of free 'OH. Instead, it likely produced adsorbed 'OH as PAC'-HO and other strong oxidizing species like $[^{1}O_{2}+PAC]$. Thus, free 'OH did not play any significant role in PAC-catalyzed ozonation. The effect of varying the O₃ specific dose as g O₃/g DOC (dissolved organic carbon) on the removal of those MPs by single and PAC-catalyzed ozonation was also investigated. Most MPs were more prone for removal by catalytic ozonation where faster removal kinetics were obtained. Carbamazepine, an anticonvulsant, and Atrazine, an herbicide, were found to be good surrogates for fast and slow reacting compounds, respectively. The removal kinetics were strongly dependent on the level of DOC.

After establishing a better understanding of the catalytic mechanisms, single and PACcatalyzed ozonation were explored for the first time during wet weather flow for organics removal, disinfection, MPs' reduction and toxicity abatement. Overall, applying either ozonation process after the primary treatment was more efficient than applying them simultaneously. PAC-catalyzed ozonation was particularly favourable for the removal of organics and O₃-resistant MPs. Both single and PAC-catalyzed ozonation achieved 4 log removal of *E. coli*, and reduced the acute and genetic toxicity and the estrogenic activity of PI. A detailed cost analysis revealed that in post treatment, single ozonation would cost 0.06 to $0.32 \,\text{/m}^3$ while PAC-catalyzed ozonation would cost 0.32 to $0.63 \,\text{/m}^3$ for a flow rate ranging between 100 and 600 MLD. A comprehensive performance assessment demonstrated that PAC-catalyzed ozonation was promising for PI treatment during wet weather flow with one drawback related to the disposal of PAC.

PREFACE

The research described in this thesis is an original work that I, Mirna Alameddine, planned, designed, performed, interpreted and analyzed under the supervision of Dr. Mohamed Gamal El-Din in the Department of Civil and Environmental Engineering at the University of Alberta. The thesis was designed in a paper-based format with chapters (2-6) representing stand-alone papers that have been published, submitted or will be submitted for publication. I conducted the experimental work and prepared the manuscript. Post-doctoral fellows in our research group contributed to the analysis of samples and editing of manuscripts. All the contributions are further detailed below:

- Chapter 2: A version of this chapter will be submitted to Journal of Environmental Management as "A Review of Coagulation and Advanced Oxidation Processes for the Treatment of Primary Influent Wastewater during Wet Weather Flow Conditions". Mr. Abdulrahim Al Umairi will be the co-author of the manuscript. Dr. Mohamed Gamal El-Din contributed to the research planning and manuscript revision. Dr. Pamela Chelme-Ayala, Dr. Zuo Tong How and Dr. Shailesh Sable contributed to the revisions and edits.
- Chapter 3: A version of this chapter was published as "Alameddine, M., Al Umairi, A. R., Shaikh, M. Z., & Gamal El-Din, M. (2020). Bench to Full-Scale Enhanced Primary Treatment of Municipal Wastewater under Wet Weather Flow for Minimized Pollution Load: Evaluation of Chemical Addition and Process Control Indicators. *Can. J. of Civ. Eng.* doi: https://doi.org/10.1139/cjce-2019-0515". Mr. Abdulrahim Al Umairi and Mr. Mohammed Zakee Shaikh were the co-author of the manuscript, they contributed to the experimental work and write-up. Dr. Mohamed Gamal El-Din contributed to the research planning and manuscript

revision. Dr. Pamela Chelme-Ayala and Dr Selamawit Ashaghre Messele contributed to the manuscript revision and edit. Mr. Abdul Mohammed contributed to the manuscript revision.

- Chapter 4: A version of this chapter was submitted to Applied Catalysis B: Environmental (June 30, 2020) as: "Alameddine, M., Siraki, A., Tonoyan, L., & Gamal El-Din, M. Treatment of a Mixture of Pharmaceuticals, Herbicides and Perfluorinated Compounds by PAC-Catalyzed Ozonation: Insight into the Non-Free 'OH Contingent Mechanisms''. Dr. Mohamed Gamal El-Din contributed to the research planning and manuscript revision. Dr Arno Siraki and Dr Lusine Tonoyan contributed to the EPR analysis. Dr. Shailesh Sable contributed to the experimental planning and manuscript revision. Dr. Zuo Tong How contributed to the samples' analysis and manuscript revision. Dr. Pamela Chelme Ayala contributed to the manuscript revision and editing. Mrs. Bing Lin from EPCOR Water Services contributed to the manuscript revision.
- Chapter 5: A version of this chapter will be combined with a version of Chapter 6 and submitted to the Journal of Hazardous Materials. Dr. Mohamed Gamal El-Din contributed to the research planning and manuscript revision. Dr. Zuo Tong How contributed to the samples analysis and manuscript revision. Dr. Pamela Chelme-Ayala contributed to the revision and editing.
- Chapter 6 will be combined and jointly submitted with a version of Chapter 5 to Journal of Hazardous Materials (September, 2020) as: "Alameddine, M., How, Z.T., & Gamal El-Din, M. Advancing the Treatment of Primary Influent Wastewater during Wet Weather Flow by Single versus PAC-catalyzed Ozonation for Micropollutants Removal, Disinfection and Toxicity Abatement". Dr. Mohamed Gamal El-Din contributed to the research planning and manuscript revision. Mr. Abdulrahim Al Umairi contributed to the wastewater samples collection. Dr. Zuo Tong How contributed to the experimental planning, samples analysis and manuscript revision. Dr. Pamela Chelme-Ayala contributed to the manuscript revision and editing. Mrs. Bing Lin from EPCOR Water Services contributed to the revision and editing.

ACKNOWLEDGMENTS

All the work presented hereafter was made possible under the supervision of Dr. Mohamed Gamal El-Din to whom I am genuinely thankful. Beyond the knowledge that he generously shared, Dr. Gamal El-Din never refrained from throwing new challenges and raising the standards. He deliberately instilled in me strength, resilience and determination. I am also tremendously thankful to Dr. Pamela Chelme-Ayala who facilitated my research by following up, reviewing, editing, and working selflessly to provide multiple insights for improvement. I am likewise deeply appreciative of the mentorship of Dr Shailesh Sable who generously offered knowledge, guidance, feedback and experimental help. I equally recognize Dr. Zuo Tong How for his valuable input to my thesis and for the analysis of my numerous samples, and Dr. Selamawit Ashaghre Messele for her wisdom and keen support. I am also grateful to Dr. Arno Siraki and Dr. Lusine Tonoyan for their contribution to the EPR analysis and to my colleague Mr. Abdulrahim Al Umairi for every assistance he kindly provided. Last but not least, I would like to acknowledge the financial support received from EPCOR water services, the Helmholtz-Alberta Initiative (HAI) through the Alberta Environment and Parks' EcoTrust Program, and the Natural Sciences and Engineering Research Council of Canada (NSERC) through their research grant.

Besides, I would like to express my sincere gratitude towards my in-laws for their love and support, and to my two distinctive brothers and incredible sister for constantly modeling hard work and determination. My utmost gratitude goes to my parents who against all the uncertainties we have been through, managed to raise a strong devoted family with care and compassion. Finally, I owe everything to my husband and daughter, my ultimate strength and my weakness who readily made several adjustments to cope with my PhD journey. I cherish them above all and recognize that any accomplishment happens only with them and for them.

DEDICATION

To the soul of a beautiful person watching over me for the past seven years,

To my amazing parents and siblings,

To the most wonderful husband who always has my back,

And to the most precious little girl who cannot wait to flip through these pages...

TABLE OF CONTENT

CHAPTER 1	: Introduction and Research Objectives1
1.1. Back	sground1
1.1.1.	Combined Sewer Overflow (CSO): occurrence and environmental concerns1
1.1.2. I	Micropollutants (MPs) in wastewater effluents: occurrence, and health and environmental concerns
1.1.3. (Ozonation for MPs degradation and wastewater treatment4
1.2. Rese	arch scope and hypotheses5
1.3. Rese	arch objectives7
1.4. Thes	sis organization
References	
CHAPTER 2	Literature Review
2.1. Trea	tment of Combined Sewer Overflow17
2.1.1. I	Introduction17
2.1.2.	Treatment options for the removal of macropollutants from CSO18
2.1.3.	Treatment options for the removal of micropollutants from CSO and wastewater effluents and for disinfection
2.1.4. I	Bridging the gaps for improved CSO treatment25
2.1.5.	Conclusions
2.2. Micr	opollutants degradation by PAC-catalyzed ozonation27
2.2.1. I	Introduction
2.2.2. I	Fundamentals of the catalytic mechanisms in PAC-catalyzed ozonation: approaches and main findings
2.2.3.	Bridging the gaps for better understanding and application of PAC-catalyzed ozonation
2.2.3.1.	Catalyst characterization
2.2.3.2.	Dosage
2.2.3.3.	Catalytic mechanisms and pathways – Concept validation44
2.2.3.4.	Intermediates, by-products and toxicity tests
2.2.3.5.	Matrix effect
2.2.3.6.	Experimental limitations46
2.2.3.7.	Fate of the catalyst: recovery and reuse or disposal
2.2.4	Conclusions

Refer	ences		49
CHAPT Wastew Chemics 3.1.	ER 3 Bench to Full-s ater under Wet Weather al Addition and Process Con Introduction	cale Enhanced Primary Tree Flow for Minimized Pollution trol Indicators	eatment of Municipal n Load: Evaluation of 66
3.2.	Material and methods		
3.2.	1. Wastewater		
3.2.	2. Coagulants and polymer		70
3.2.	3. Jar test		70
3.2.	4. Full-scale application of	alum/polymer	73
3.2.	5. Analytical methods		74
3.3.	Results and discussion		74
3.3.	1. Wastewater quality		74
3.3.	2. Performance assessment	of coagulants	75
3.3.	3. Optimization of the alun	dose	
3.3.4	4. Effect of polymer type of during wet weather flow	n the performance of alum in the	treatment of wastewater
3.3.	5. Effect of alum and polyr	ner addition during full-scale appl	ication88
3.3.	6. Effective parameters for	online process control	
3.3.	7. Implications of these rem	ovals on the monitoring of micro	pollutants94
3.3.	8. Recommended future int	erventions	96
3.4.	Conclusions		
Refer	ences		
CHAPTER 4: Treatment of a Mixture of Pharmaceuticals, Herbicides and Perfluorinated Compounds by PAC-Catalyzed Ozonation: Insight into the Non-Free 'OH Contingent Mechanisms 108			
4.2	Material and methods		111
4.2.	1. Chemicals and standards		
4.2.1	2. Catalyst characterization		
4.2.	3. Experimental procedures		
4.2	4. Analytical methods		
4.3.	Results and discussion		
4.3.	1. Catalyst characterization		

4.3.2.	Ozone decomposition in single and catalytic ozonation	119
4.3.3.	Removal of micropollutants	124
4.3.4.	Formation of reactive oxygen species (ROS)	
4.3.4.1	1. Catalyst surface changes	
4.3.4.2	2. Competitive radical test	
4.3.4.3	3. Identification of ROS by EPR spectroscopy	141
4.3.5.	Proposed catalytic ozonation mechanisms	145
4.4. Co	onclusions	149
Reference	es	150
CHAPTER Micropollu and Kinetic 5.1. Int	R 5: Assessment of the Effect of Ozone Specific Dose on the Itants in a Mixture by Single and PAC-catalyzed Ozonation through cs	Removal of Benchmarks 158 158
5.2. Ma	aterial and methods	160
5.2.1.	Micropollutants and reagents	160
5.2.2.	Target mixture	161
5.2.3.	Experimental procedures	163
5.2.3.1	1. Ozonation experiments	163
5.2.3.2	2. Kinetics study	163
5.2.4.	Analytical methods	164
5.2.5.	Detection of intermediates	164
5.3. Res	esults and discussion	164
5.3.1.	Effect of O ₃ specific dose on the removal of MPs by single ozonation.	164
5.3.2.	Effect of O ₃ specific dose on the decomposition of O ₃ in single ozonation	and catalytic
5.3.3.	Benchmarking the removal of MPs by single and PAC-catalyzed ozor	nation171
5.3.4.	Kinetics of MPs removal in a mixture by single and catalytic ozonatic	»n177
5.3.5.	Formation of intermediates in single and PAC-catalyzed ozonation	
5.4. Co	onclusions	184
Reference	ces	
CHAPTER during We Ozonation f 6.1. Int	R 6: Advancing the Treatment of Primary Influent and Effluent et Weather Flow by Single versus Powdered Activated Carl for Micropollutants removal, Disinfection and Toxicity Abatement troduction	t Wastewater bon-catalyzed 197 197

6.2.	Ma	terial and methods	
6.2.	.1.	Sample collection and preparation	
6.2.	.2.	Experimental procedures	
6.2.	.3.	Wastewater spiking with MPs	
6.2.	.4.	Chemicals	
6.2.	.5.	Analytical methods	
6.2.	.6.	Toxicity tests	
6.3.	Res	ults and discussion	
6.3.	.1.	Wastewater characterization and IOD	
6.3.	.2.	Dose screening for O ₃ and PAC	
6.3.	.3.	Simultaneous versus post treatment of PI during wet weather by catalyzed ozonation	y single and PAC212
6.3.	.4.	Toxicity assays and fate of PAC	216
6.3.	.5.	Cost Analysis	
6.3.	.6.	Overall performance assessment	221
6.4.	Сог	nclusions	
Refer	ence	28	
СНАРТ	ΓER	7: Conclusions and Recommendations	
7.1.	The	esis overview	
7.2.	Сог	nclusions	234
7.3.	Rec	commendations	237
BIBLIOGRAPHY			
APPEN	DIX		
1.	Sup	plementary Tables and Figures	
2.	Car	nadian Science Publishing – License Terms and Conditions	

LIST OF TABLES

Table 2.1 Percent removal and effluent concentrations of selected MPs in typical WWTPs
(Adapted from:(Margot et al., 2015) and (Das et al., 2017))24
Table 2.2 The advantages and disadvantages of applying ozonation for the disinfection of CSC
(Adapted from (Wojtenko et al., 2001b))25
Table 2.3 Soluble and insoluble catalysts applied in HoCO and HeCO and available in literature
Table 2.4 Summary of the key findings in literature pertaining to ozonation catalyzed by activated
carbon41
Table 3.1 Details of the 2 ³ factorial design used for the coagulants assessment
Table 3.2 Calculations of metal-equivalent coagulant doses
Table 3.3 Conversion of mixing conditions from rpm, min to Gt (Based on B-KER ² – Phipps &
Bird Specifications)
Table 3.4 Water quality parameters of the untreated diluted primary influent (PI) used in the
bench-scale experiment and the influent wastewater to the WWTP in the full-scale runs75
Table 3.5 Results of the Three-Factor ANOVA test showing the interactions between
contaminants percent removals and the three factors: rapid mixing, slow mixing, and coagulan
dose at 95% confidence level
Table 3.6 ANOVA models. 82
Table 3.7 Chemical and physical properties of the polymers used in this study
Table 3.8 Water quality parameters of untreated and treated samples during full-scale application
of Alum and Alum + Polymer A at the WWTP89
Table 4.1 List of Micropollutants included in this study
Table 4.2 Monitored mass (m/z) and detection mode for the 17 micropollutants
Table 4.3 Percent removal in single and catalytic ozonation before and after the addition of radica
scavengers139
Table 5.1 Characteristics and occurrence of the studied MPs. 162

 Table 5.3 Heatmap of MPs' removal by single and PAC catalyzed ozonation at 1 minute and 10 minutes of treatment and different O3 specific doses.

 173

Table 6.1 Average characteristics of primary influent samples used in the first stage of the study,
after dilution with deionized water (50:50) to simulate primary influent during wet weather
conditions208
Table 6.2 Characteristics of the primary influent (PI) used in the performance assessment of single
and PAC-catalyzed ozonation as simultaneous or post treatments in comparison with the
characteristics of secondary and final effluents209
Table 6.3 Assumptions for the cost estimates. 220
Table 6.4 CapEx ^a and OpEx ^b analyses for the different treatment options with single and PAC-
catalyzed ozonation, assuming the total period of operation is 15 years221
Table 6.5 Color-coded performance assessment of the different treatment options in comparison
with primary treatment (CFS) only
Table A.1 Predicted half-lives of MPs in a reaction with molecular O_3 only under different O_3
specific doses
Table A.2 Predicted half-lives of MPs in a reaction with hydroxyl radicals 'OH only under
different O ₃ specific doses
Table A.3 Oxidation intermediates and by-products for the 17 studied MPs reported in literature.
Table A.4 Cost analysis for single and PAC-catalyzed ozonation as ST and as PT 278

LIST OF FIGURES

Figure 3.1 Simplified representation of the EPT system
Figure 3.2 Mass of contaminant removed by 1 mg of active metal at high and low coagulant doses.
The effect of mixing and settling on the removal was excluded in the calculation. In untreated
wastewater, the parameters ranged as follows: Turbidity 47.2-106.0 NTU; TSS 99-167 mg/L;
COD 160-361 mg O ₂ /L; and Ortho-P 2.10-6.35 mg P/L78
Figure 3.3 Effect of alum dosage on the removal of TSS from wastewater during wet weather
conditions
Figure 3.4 Effect of polymer addition on turbidity, TSS, ortho-P and %UVT (applied alum dose
is 75 mg /L and polymer dose is 0.75 mg/L)85
Figure 3.5 Impact of alum dose applied with anionic polymer A in a fixed dose ratio of 100:1 (as
alum: polymer in mg/L) on the removal of turbidity, TSS and Ortho-P87
Figure 3.6 Removal of turbidity, TSS and ortho-P during the full-scale application of alum and
alum + polymer A91
Figure 3.7 Comparison between bench-scale and full-scale applications of 75 mg/L alum only and
75 mg/L alum + 0.75 mg/L polymer for the removal of turbidity, TSS and ortho-P92
Figure 3.8 Correlations between contaminants measured at full-sale level
Figure 3.9 Correlations between contaminants measured at bench-scale level
Figure 4.1 Assessed PAC characteristics: (A) pH _{pzc} by Solid Addition Method; (B) XPS spectra
for untreated PAC119
Figure 4.2 Comparison between the MPs removal (Ln C_0/C) in a mixture with two O_3 doses: 5
mg/L and 10 mg/L after 10 minutes of single ozonation (in borate buffer pH 7; Mixture of 17
MPs with 500 ppb each)
Figure 4.3 O ₃ decomposition kinetics in clean matrix using 10 mg/L O ₃ in borate buffer at pH 7
and different doses of PAC. (x axis represents time in min; y axis represents ln [O ₃]/[O ₃] ₀ where
$[O_3]_0$ is the residual O_3 at time = 12 seconds)
Figure 4.4 Effect of PAC dose on (A) the initial O ₃ demand (IOD) in the first 12 seconds; and (B)
the first-order rate constant of O_3 decomposition in borate buffer at pH 7 (O_3 dose: 10 mg/L –
PAC doses: 0; 25; 50; 75; 100; 250; 500 mg/L – Borate Buffer, pH 7)122

- Figure 4.5 Changes in the percent removal of MPs after 1 min of catalytic ozonation using different PAC doses: 25 mg/L PAC vs 100 mg/L PAC. The rates of O₃ decomposition were 1.74 L mg⁻¹ min⁻¹ with 25 mg/L PAC and 0.14 L mg⁻¹ min⁻¹ with 100 mg/L and (O₃: 10 mg/L; PAC: 25 mg/L 100 mg/L; borate buffer pH 7; Mixture of 17 MPs with 500 ppb each).....123
- **Figure 4.6** MPs that were completely removed within the first 1 minute of oxidation in a buffered mixture of 17 MPs: adsorption (blue circle), single ozonation (orange, square), catalytic ozonation (green, triangle). (MPs mixture: 500 ppb each; O₃: 10 mg/L; PAC: 100 mg/L)...125
- Figure 4.7 (A) Removal of MPs in a buffered mixture of 17 MPs by different treatments: Adsorption (blue circle), Single Ozonation (orange, square), Catalytic ozonation (green, triangle); (B) Comparison between the 1 minute percent removal by synergistic effect (Adsorption + Single Ozonation) versus catalytic effect (catalytic ozonation) where C₀ represents the initial concentration of the MPs and C represents their concentration at 1 minute; (C) Comparison between the apparent degradation rate (k in min⁻¹) calculated for the first minute of the treatment.
- Figure 4.8 Specific degradation rate for each MP in the mixture as μM of MP removed per μM O₃ utilized in 1 minute (MPs mixture: 500 ppb each; O₃: 10 mg/L; PAC: 100 mg/L, borate buffer pH 7).
- Figure 4.9 FT-IR Spectra for PAC samples recovered after 30 minutes exposure to O₃ and/or the mixture of MPs (with 4 mM borate buffer pH 7, 10 mg/L O₃, 100 mg/L PAC, mixture of 17 MPs at 500 ppb each).
- Figure 4.11 Measured residual O₃ in buffered mixture of 17 MPs pH 7, during single ozonation (O₃) and catalytic ozonation (O₃ + PAC) with and without radical scavengers. Mixture of 17 MPs: 500 ppb each; O₃: 10 mg/L; PAC: 100 mg/L PAC; TBA: 2 mM; p-BZQ: 2 mM; NaN₃: 2 mM.

- Figure 4.15 EPR spectra obtained in clean water with (A) 100 mM DMPO, featuring a spectrum for DMPO-OH (quartet signal) in single ozonation, and a signal with 7 lines representing DMPO-X in catalytic ozonation; (B) 5.9 M TEMP, featuring a signal with 3 lines at equal intensity in both single and catalytic ozonation; (C) 20 mM MNP, featuring no spectrum in single ozonation, and low-intensity signal in catalytic ozonation; (D) 2.5 M PBN, featuring a spectrum for PBN-OH (triplet of doublets) in single ozonation, and no signal in catalytic ozonation.
- Figure 4.16 EPR spectra obtained with 100 mM DMPO in clean water under different experimental conditions: Single ozonation with 10 mg O₃/L; Catalytic Ozonation with 10 mg O₃/L and 100 mg PAC/L; Catalytic Ozonation in presence of TBA with 10 mg O₃/L and 100 mg PAC/L and 2 mM TBA; in (A) Borate buffer, and (B) Phosphate buffer......144

- Figure 5.2 Residual O₃ and MPs specific degradation rates (mg MP removed after 1 minute/mg O₃ applied) in single and PAC catalyzed ozonation with different O₃ specific doses: Mixture of MPs at 0.05 mg/L, Mixture of MPs at 0.5 mg/L, Mixture of MPs at 4 mg/L each in borate

buffer; O ₃ dose: 10 mg/L, PAC dose: 100 mg/L, specific O ₃ doses: (A) 0.27g O ₃ /g DOC; (B)
2.17 g O ₃ /g DOC; (C) 21.64 g O ₃ /g DOC170
Figure 5.3 Comparison between the removal as $Ln C_0/C$ by single ozonation, catalytic ozonation
and adsorption for the different categories of MPs at three O_3 specific doses: (A) 0.27 g O_3/g
DOC; (B) 2.17 g O ₃ /g DOC, and (C) 21.64 g O ₃ /g DOC during a 10 minutes reaction time:
blue: Group I, red: Group II, green: Group III, yellow: Group IV
Figure 5.4 Correlation between the removal (Ln C/C_0) of CBZ and the removal of MPs in group
I, the removal of ATZ and the removal of MPs in groups II, III and IV under three different O ₃
specific doses during single ozonation175
Figure 5.5 Correlation between the removal (Ln C/C_0) of CBZ and the removal of MPs in group
I, the removal of ATZ and the removal of MPs in groups II, III and IV under three different O ₃
specific doses during catalytic ozonation176
Figure 5.6 Removal of 2,4-D, ATZ and FCZ by (A) Single ozonation and (B) Catalytic ozonation
as a function of time
Figure 5.7 Degradation kinetics of MPs by single ozonation (Mixture of MPs at 0.05 mg/L each,
borate buffer: 4 mM, O ₃ dose: 10 mg/L)180
Figure 5.8 Degradation kinetics of MPs by catalytic ozonation divided into three different zones
(Mixture of MPs at 0.05 mg/L each, borate buffer: 4 mM, O ₃ dose: 10 mg/L, PAC: 100 mg/L).
Figure 5.9 Comparison between the peak areas corresponding to the detected intermediates in
single ozonation and in PAC-catalyzed ozonation at 10 minutes reaction time184

Figure 6.1 Schematic representation of the performed single and catalytic ozonation experiments
as simultaneous and post treatments
Figure 6.2 Effect of O3 dose on the removal of turbidity, COD and DOC after 30 minutes of single
ozonation as (A) Simultaneous Treatment, and (B) Post Treatment (a dose of 0 g O_3 /g DOC
corresponds to CFS only)
Figure 6.3 Effect of PAC dose on the removal of COD and DOC after 30 minutes of adsorption
as a control in (A) Simultaneous Treatment and (B) Post Treatment (a dose of 0 mg/L PAC
corresponds to CFS only)

Figure 6.4 Comparison between Simultaneous and Post treatments for the removal of turbidity,
COD _d , DOC, E. coli and enhancement of %UVT during the treatment of PI during wet weather
flow (O3 specific dose: 0.5 g O3/g DOC; PAC dose: 100 mg/L). Results correspond to samples
collected after 1 hour settling in ST and 1 minute treatment in PT213
Figure 6.5 Effect of the treatment on the removal of CBZ, ATZ and DOC in PI during wet weather
that was spiked with 2 mg/L CBZ and 2 mg/L ATZ. Results correspond to samples collected
after 1 hour settling in ST and 1 minute treatment in PT
Figure 6.6 Effect of the treatment on the acute toxicity and genotoxicity of the primary effluent:

Figure A.1 Graphical summary of Chapter 3.	
Figure A.2 Graphical summary of Chapter 4.	
Figure A.3 Graphical summary of Chapters 5 and 6	

ABBREVIATIONS

Alum	Aluminum sulfate
ANOVA	Analysis of Variance
CapEx	Capital Expenditure
CFS	Coagulation/ Flocculation/ Sedimentation
COD	Chemical Oxygen Demand
CSO	Combined Sewer Overflow
DOC	Dissolved Organic Carbon
EPR	Electron Paramagnetic Resonance
EPT	Enhanced Primary Treatment
FeCl ₃	Ferric chloride
FT-IR	Fourier-transform infrared spectroscopy
М	Molar
mM	milli Molar
MLD	Mega liters per day
MPN	Most Probable Number
MP	Micropollutant
NaN ₃	Sodium azide
OpEx	Operational and Maintenance Expenditure
O ₃	Ozone
¹ O ₂	Single Oxygen
O2 ^{-•}	Superoxide
•ОН	Hydroxyl radical

OH-	Hydroxyl ion
Ortho-P	Orthophosphate
PAC	Powdered Activated Carbon
PACl	Polyaluminum chloride
pBZQ	Para-benzoquinone
PI	Primary Influent
PT-CO	Post treatment – Catalytic ozonation
PT-SO	Post treatment – Single ozonation
pH _{pzc}	Point of zero charge
ppb	Parts per billion
ppm	Parts per million
rpm	Rotations per minutes
ST-CO	Simultaneous treatment – Catalytic ozonation
ST-SO	Simultaneous treatment – Single ozonation
SS	Suspended solids
TBA	Tertiary butyl-alcohol
TOC	Total Organic Carbon
TSS	Total Suspended Solids
UVA	Ultra-violet absorbance
UPLC-TOF-MS	Ultra-performance liquid chromatography time-of-flight mass
	spectrometry
WWTP	Wastewater Treatment Plant
XPS	X-ray photoelectron spectroscopy
%UVT	Percent Ultra-violet Transmittance

CHAPTER 1: Introduction and Research Objectives

1.1. Background

1.1.1. Combined Sewer Overflow (CSO): occurrence and environmental concerns

Combined sewer overflows (CSOs) occur seasonally when the capacity of a wastewater network is exceeded due to increased volumes from heavy rainfalls and snow melts. In such cases, wastewater treatment plants (WWTPs) become unable to accommodate the excess wastewater that is usually of domestic and industrial type diluted by storm water runoffs. This excess ends up overflowing to the closest water stream as per the combined sewer system design. The volume of untreated or poorly treated CSO and the frequency of its discharge will essentially contribute to the concentration of pollutants including pathogens, organic and inorganic compounds in the receiving water body (Kay et al., 2016; Tondera et al., 2016). It will consequently affect its suitability for drinking or recreational purposes. Problems with irresponsible and unsafe CSO discharge can further transcend, impacting the aquatic ecosystem and public health with detrimental waterborne diseases (Jalliffier-Verne et al., 2016). In Canada, it is reported that 87% of the population is connected via a wastewater collection network to certain type of treatment, and only few treatment plants undertake the initiative to store and treat CSO. As such, local direct discharges from CSO have not been routinely monitored (Canada, 2019; CWN, 2018). Treatment options to mitigate CSO include physical and chemical processes applied on primary influent wastewater. As such, coagulation/ flocculation/ sedimentation has been tested and applied for the elimination of conventional parameters such as turbidity, suspended solids and chemical oxygen demand (El Samrani et al., 2008a; Gibson et al., 2016; Jung et al., 2015; Wang et al., 2015b). Also, tertiary reed beds were used to maximize the removal of biochemical oxygen demand, total suspended solids, and ammonia to match target values in tertiary treatment systems (Green et al.,

1999). Ultraviolet irradiation (Tondera et al., 2015; Wojtenko et al., 2001a), ozone (Tondera et al., 2015; Wojtenko et al., 2001b), performic acid (Chhetri et al., 2015; Chhetri et al., 2014; Tondera et al., 2016), and peracetic acid (Chhetri et al., 2016; Chhetri et al., 2014; Coyle et al., 2014) were reportedly effective methods to address disinfection, a prominent topic for alleviating the environmental impacts of CSO. While each treatment provides part of the solution, there is an arising need for a comprehensive approach towards the treatment of primary influent during wet weather conditions to prevent the discharge of raw CSO. This approach should be set to target three areas of concern: conventional macropollutants (contaminants present at high concentrations), pathogens, and micropollutants (natural or manmade contaminants present at very low concentrations).

1.1.2. Micropollutants (MPs) in wastewater effluents: occurrence, and health and environmental concerns

Micropollutants (MPs) are natural or anthropogenic compounds present in the environment at very low concentrations. They exist in different classes including: pharmaceuticals, personal care products, herbicides, pesticides, industrial products and surfactants (Stamm et al., 2016). It was reported that the number of known and registered MPs in Europe exceeded 100,000, many of which can find their way to the aquatic environment through point and non-point sources (Eggen et al., 2014). They can provoke undesirable impacts on the environmental and public health when inappropriately disposed of. For example, MPs belonging to the group of endocrine disrupting compounds (EDCs) are known to affect the reproductive system in aquatic animals (Eggen et al., 2014). Other MPs have expressed toxic levels for specific aquatic species by virtue of their high biological activity aligned with their intended uses such as antibiotics and insecticides (Eggen et al., 2014; Stamm et al., 2016). The risk to the public health arises when the receiving water body is intended for human use, either through recreation or consumption. Some herbicides and pesticides have been associated with possible carcinogenicity and risk of acute toxicity (Bozoglu, 2011), and some perfluorinated compounds have been linked to cancer, damages to the bones, liver, and immune, reproductive and developmental systems (Khalil et al., 2016; Pelch et al., 2019). Amid all those risks and until now, there are no national standards or guidelines to regulate the permissible concentrations of MPs in the final effluent in North America. Many of those MPs have been registered under a list of Contaminants of Emerging Concerns (CECs) or were allocated a maximum contaminant level goal in drinking water (Health Canada, 2019; USEPA, 2010) to protect the public health, while there is a growing motive for their regulation. In contrast, several European countries, primarily Switzerland, are pioneering the research and application of MPs removal techniques as well their regulations in effluent wastewater. In fact, Waters Protection Ordinance, a Swiss act implemented in 2016 required the removal of at least 80% of the content of influent wastewater in trace organic pollutants by monitoring six MPs in a list of twelve including carbamazepine, mecoprop, diclofenac and others (Joss et al., 2016; WPO, 2020).

CSOs can be as a relevant source of MPs as WWTPs and thus can imply comparable adverse effects (Shu et al., 2016). To better elucidate, it is estimated that conventional WWTPs are more effective for macropollutants removal than for MPs removal with no guarantee for meeting safe and acceptable concentrations in the final effluent. Ryu et al. (2014) alleged that physical, chemical and biological processes in a WWTP can decrease the MPs in the effluent by 28% as compared to 97% decrease in suspended solids and 86% decrease in chemical oxygen demand. Qi et al. (2015) traced the removal of some MPs in WWTPs. The authors confirmed that the effect of the treatment processes on their removal can be inconsistent between WWTPs and dependent on the characteristics of each MP. The researchers also ascertained that coagulation/ flocculation was not very effective for the removal of the majority of the studied MPs.

1.1.3. Ozonation for MPs degradation and wastewater treatment

While upgrading and optimizing the conventional wastewater treatment for enhanced removal of suspended solids and organics can result in better removal of hydrophobic MPs, other hydrophilic MPs difficult to degrade will persist. Subsequently, introducing advanced tertiary treatment is recommended. In particular ozonation and adsorption are praised for their ability to remove persistent compounds and for their cost and energy efficiencies (Das et al., 2017; Margot et al., 2015; von Sonntag & von Gunten, 2012). Ozone (O₃) is widely recognized for its oxidation and disinfection properties. It was deemed effective in wastewater treatment in multiple studies on the ozonation of real wastewater for the removal of MPs in secondary effluents (Altmann et al., 2014; Can & Çakır, 2010; Huber et al., 2005; Ikehata et al., 2008; Jin et al., 2016; Lee et al., 2014b; Nothe et al., 2009; Wildhaber et al., 2015; Zhou et al., 2015). Also, applying ozonation for wastewater disinfection has been known and practised for a while (Xu et al., 2002). In CSO treatment, ozonation was only studied for disinfection.

Single ozonation is one way for applying O₃. It was examined for the removal of several MPs as a standalone treatment for model compounds (Seredyńska-Sobecka et al., 2005), or in combination with other processes such as following secondary treatment of real wastewater (Huber et al., 2005; Lajeunesse et al., 2013; Lee et al., 2013; Nothe et al., 2009). Yet the fact that many MPs exhibit limited reactivity with molecular O₃ prompted the addition of a catalyst in an attempt to improve the removals. This has become the recent trend in O₃-based advanced oxidation processes (AOPs) mainly to promote the degradation of organics through an indirect pathway that involves the generation of radicals. Free radicals are capable of oxidizing more compounds by virtue of their non-selectivity and strong oxidizing potential. Carbon-based catalysts are common in the literature and have been applied in different forms, alone and in combination with other catalysts. The interest in carbon material is mainly associated with its practicality in terms of cost

and ease of application, in addition to its potential dual role as catalyst and adsorbent. Powdered activated carbon (PAC) is a potential good candidate. Studies on PAC-catalyzed ozonation are limited and have focused on only one (Faria et al., 2008a; Faria et al., 2008b; Faria et al., 2009; Goncalves et al., 2012; Gonçalves et al., 2013a; Gonçalves et al., 2014; Guzman-Perez et al., 2011) or a small mixture of four MPs only (Rozas et al., 2017). Those studies were performed using high concentration of target compounds in clean water and they seldom addressed the matrix effect while claiming free hydroxyl radicals as the main oxidants.

1.2. Research scope and hypotheses

This thesis pursued the enhanced treatment of primary influent wastewater during wet weather conditions as an alternative for the discharge of untreated CSO into water bodies. It elaborated on two treatment methods: coagulation/ flocculation/ sedimentation and ozonation processes. The studies were performed at bench and full-scale levels on real wastewater samples and on model compounds to explore practical treatment solutions as well as fundamental knowledge in catalytic mechanisms. The following hypotheses initiated this research and steered the objectives:

- Enhanced primary treatment for the removal of conventional wastewater contaminants from primary influent wastewater during wet weather conditions can be sufficient to meet the target value for effluent discharge.
- The contribution of coagulant aids to the overall removal efficiency of major macropollutants is minimal compared to that of the coagulant.
- The use of PAC as catalyst is effective for the removal of O₃-resistant MPs during ozonation processes. Its dual role as catalyst and adsorbent depends on the matrix composition and concentration of target compounds.

- 4) The mechanism of PAC-catalyzed ozonation is driven by different pathways that rely on reactive oxygen species at the surface or in the interface between PAC and the bulk solution:
 - a. Molecular O₃ and reactive oxygen species oxidize the organic molecules adsorbed onto the surface of the catalyst;
 - b. The catalyst initiates O₃ decomposition into radicals which oxidize the organic MPs adsorbed onto the catalyst surface or in solution;
 - c. O₃ reacts with the surface of the catalyst producing surface radicals which react with organic molecules available in solution or with the adsorbed organic molecules producing organic radicals that will be further oxidized by hydroxyl radicals and molecular O₃ in solution;.
- 5) The apparent (i.e., overall) removal kinetics of MPs in a mixture differ from theoretical kinetics of individual compounds due to the competition between pollutants. However, similar trends in the removal rates are expected in complex mixtures.
- 6) Single and PAC-catalyzed ozonation are beneficial additions to the enhanced primary treatment (EPT) of the primary influent wastewater during wet weather conditions for both disinfection and MPs removal.
- PAC-catalyzed ozonation can demonstrate a superior performance compared to single ozonation by favouring disinfection and toxicity reduction.
- Cost is one considerable limitation for the application of PAC-catalyzed ozonation due to the significant addition implied by the catalyst.

1.3. Research objectives

The main thesis objectives are presented below according to their appearance in the thesis: Chapter 3 addressed hypotheses 1 and 2 and its objectives were:

- To compare the performance of metal-based coagulants and anionic and cationic polymers as coagulant-aids; and
- 2) To define the optimum conditions for coagulation/ flocculation/ sedimentation process for the removal of conventional contaminants from primary influent (PI) wastewater during wet weather flow.

Chapter 4 addressed hypotheses 3 and 4 and its objectives were:

- To determine the optimum catalyst dose to catalyze the decomposition of ozone and maximize the degradation of selected MPs;
- 2) To compare the removal efficiencies of single and catalytic ozonation in a diversified mixture of MPs, from different categories and with different affinities towards ozone; and
- To study the catalytic mechanisms involved in the ozonation and catalytic ozonation of selected MPs in a clean water matrix using radical probing and other available methods.

Chapter 5 addressed hypothesis 5 and its objectives were:

1) To study the matrix effect on the single ozonation and catalytic ozonation of selected MPs and the kinetics of their degradation by single and catalytic ozonation.

Chapter 6 addressed hypotheses 6, 7 and 8 and its objectives were:

- 1. To study the impact of single and catalytic ozonation on the removal of organic contaminants and the disinfection of PI during wet weather;
- 2. To assess the feasibility of single and PAC-catalyzed ozonation for the treatment of PI during wet weather conditions.

1.4. Thesis organization

This thesis consists of seven chapters that were logically organized following the research stages to meet the above-mentioned objectives.

- Chapter 1 provides a general background on the environmental and public health concerns that are associated with the free disposal of untreated wastewater during wet weather conditions. It highlights the growing interest in addressing the removal of micropollutants in particular and tackles the gap in the knowledge pertaining to their treatment as well as that of wastewater during wet weather conditions in particular. Chapter 1 also enlists the research objectives and thesis organization.
- Chapter 2 compiles an extensive literature review on the treatment of wastewater during wet weather flow and the application of ozonation methods for the degradation of micropollutants.
- Chapter 3 reports the optimum chemical dosing for the enhanced primary treatment of PI during wet weather conditions. It also compares data from bench and full-scale applications.
- Chapter 4 explores the mechanisms involved in PAC-catalyzed ozonation through radical probing and electron paramagnetic resonance. The chapter presents the effect of PAC on the decomposition of ozone and how PAC-catalyzed ozonation improves the removal of model compounds. The study uses a diversified mixture at a neutral pH (pH 7) that includes seventeen micropollutants belonging to different categories (pharmaceuticals and personal care products, herbicides and perfluorinated compounds). Eight of those micropollutants are studied for the first time with PAC-catalyzed ozonation, at near environmental concentrations.
- Chapter 5 addresses the effect of ozone specific dose on the removal of MPs in a buffered mixture at neutral pH by single and PAC-catalyzed ozonation. It benchmarks the removals by PAC-catalyzed ozonation compared to single ozonation and determines surrogate MPs. Removal kinetics and formation of intermediates are also tackled in this chapter.

- Chapter 6 provides a comprehensive approach towards the application of single and PAC-catalyzed ozonation for the treatment of PI during wet weather conditions. This approach includes a performance assessment based on the removal of conventional parameters and micropollutants as well as disinfection and toxicity levels. The chapter also comprises a detailed cost analysis to evaluate the feasibility of the studied treatments.
- Chapter 7 wraps up the thesis by summarizing the major findings and conclusions. It depicts the objectives that were met and provides insights for future work.

References

- Altmann, J., Ruhl, A. S., Zietzschmann, F., & Jekel, M. (2014). Direct comparison of ozonation and adsorption onto powdered activated carbon for micropollutant removal in advanced wastewater treatment. Water Res., 55, 185-193. doi: 10.1016/j.watres.2014.02.025
- Bozoglu, F. (2011). Impact of Pesticides as Organic Micro-Pollutants on the Environment and Risks for Mankind. In H. Alpas, S. M. Berkowics & I. Ermakova (Eds.), NATO Science for Peace and Security Series -C: Environmental Security - Environmental Security and Ecoterrorism (pp. 73-82): Springer, Dordrecht. doi: https://doi.org/10.1007/978-94-007-1235-5 6
- Can, Z. S., & Çakır, E. (2010). Treatability of Organic Constituents in the Paşaköy Wastewater Treatment Plant Effluent by O₃and O₃/H₂O₂. Ozone: Sci. & Eng., 32(3), 209-214. doi: 10.1080/01919511003796087
- Canada, E. (2019). Municipal Wastewater Status in Canada, Accessed June 2020 from http://ec.gc.ca/eu-ww/default.asp?lang=En&n=6E4ACEEE-21A9-4A1A-89F1

- Chhetri, R. K., Bonnerup, A., & Andersen, H. R. (2016). Combined Sewer Overflow pretreatment with chemical coagulation and a particle settler for improved peracetic acid disinfection. J. of Ind. & Eng. Chem., 37, 372-379. doi: 10.1016/j.jiec.2016.03.049
- Chhetri, R. K., Flagstad, R., Munch, E. S., Hørning, C., Berner, J., Kolte-Olsen, A., ... Andersen,
 H. R. (2015). Full scale evaluation of combined sewer overflows disinfection using performic acid in a sea-outfall pipe. Chem. Engi. J., 270, 133-139. doi: 10.1016/j.cej.2015.01.136
- Chhetri, R. K., Thornberg, D., Berner, J., Gramstad, R., Ojstedt, U., Sharma, A. K., & Andersen,
 H. R. (2014). Chemical disinfection of combined sewer overflow waters using performic acid or peracetic acids. Sci. Total Environ., 490, 1065-1072. doi: 10.1016/j.scitotenv.2014.05.079
- Coyle, E. E., Ormsbee, L. E., & Brion, G. M. (2014). Peracetic Acid as an Alternative Disinfection
 Technology for Wet Weather Flows. Water Environ. Res., 86(8), 687-697. doi:
 10.2175/106143014x13975035525663
- CWN. (2018). Canada's Challenges and Opportunities to Address Contaminants in Wastewater Supporting Document 2
- Das, S., Ray, N. M., Wan, J., Khan, A., Chakraborty, T., & Ray, M. B. (2017). Micropollutants in Wastewater: Fate and Removal Processes. doi: 10.5772/65644
- Eggen, R. I. L., Hollender, J., Joss, A., Scharer, M., & Stamm[†], C. (2014). Reducing the Discharge of Micropollutants in the Aquatic Environment: The Benefits of Upgrading Wastewater Treatment Plants. Environ. Sci. & Technol., 48, 7683-7689. doi: dx.doi.org/10.1021/es500907n

- El Samrani, A. G., Lartiges, B. S., & Villieras, F. (2008). Chemical coagulation of combined sewer overflow: heavy metal removal and treatment optimization. Water Res., 42(4-5), 951-960. doi: 10.1016/j.watres.2007.09.009
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2008a). Activated carbon catalytic ozonation of oxamic and oxalic acids. Appl. Catal. B: Environ., 79(3), 237-243. doi: 10.1016/j.apcatb.2007.10.021
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2008b). Catalytic ozonation of sulfonated aromatic compounds in the presence of activated carbon. Appl. Catal. B: Environ., 83(1-2), 150-159. doi: 10.1016/j.apcatb.2008.02.010
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2009). Activated carbon and ceria catalysts applied to the catalytic ozonation of dyes and textile effluents. Appl. Catal. B: Environ., 88(3-4), 341-350. doi: 10.1016/j.apcatb.2008.11.002
- Gibson, J., Farnood, R., & Seto, P. (2016). Chemical pretreatment of combined sewer overflows for improved UV disinfection. Water Sci Technol, 73(2), 375-381. doi: 10.2166/wst.2015.447
- Goncalves, A. G., Orfao, J. J., & Pereira, M. F. (2012). Catalytic ozonation of sulphamethoxazole in the presence of carbon materials: catalytic performance and reaction pathways. J. Hazard. Mater., 239-240, 167-174. doi: 10.1016/j.jhazmat.2012.08.057
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2013). Ozonation of bezafibrate promoted
 by carbon materials. Appl. Catal. B: Environ., 140-141, 82-91. doi: 10.1016/j.apcatb.2013.03.034
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2014). Ozonation of erythromycin over carbon materials and ceria dispersed on carbon materials. Chem. Eng. J., 250, 366-376. doi: 10.1016/j.cej.2014.04.012

- Green, M. B., Martin, J. R., & Griffin, P. (1999). Treatment of combined sewer overflows at small wastewater treatment works by constructed reed beds Wat. Sci. Tech, 40(3), 357-364.
- Guzman-Perez, C. A., Soltan, J., & Robertson, J. (2011). Kinetics of catalytic ozonation of atrazine in the presence of activated carbon. Sep. & Purif. Technol., 79, 8-14. doi: 10.1016/j.seppur.2011.02.035
- Health.Canada. (2019). Guidelines for Canadian Drinking Water Quality Summary Table. In H.E. a. C. S. B. Water and Air Quality Bureau (Ed.). Ottawa, Ontario: Health Canada.
- Huber, M. M., Gobel, A., Joss, A., Hermann, N., Loffler, D., Mcardell, C. S., . . . von Gunten, U. (2005). Oxidation of Pharmaceuticals during Ozonation of Municipal Wastewater Effluents: A Pilot Study. Environ. Sci. Technol, 39(11), 4290-4299.
- Ikehata, K., Gamal El-Din, M., & Snyder, S. A. (2008). Ozonation and Advanced Oxidation Treatment of Emerging Organic Pollutants in Water and Wastewater. Ozone: Sci. & Eng., 30(1), 21-26. doi: 10.1080/01919510701728970
- Jalliffier-Verne, I., Heniche, M., Madoux-Humery, A. S., Galarneau, M., Servais, P., Prevost, M.,
 & Dorner, S. (2016). Cumulative effects of fecal contamination from combined sewer overflows: Management for source water protection. J. Environ. Manage., 174, 62-70. doi: 10.1016/j.jenvman.2016.03.002
- Jin, P., Jin, X., Bjerkelund, V. A., Østerhus, S. W., Wang, X. C., & Yang, L. (2016). A study on the reactivity characteristics of dissolved effluent organic matter (EfOM) from municipal wastewater treatment plant during ozonation. Water Res., 88, 643-652. doi: http://dx.doi.org/10.1016/j.watres.2015.10.0600043-1354/©2015
- Joss, A., Schärer, M., & Abegglen, C. (2016). Micropollutants: The Swiss Strategy. Paper presented at the Water 2020.

- Jung, C., Oh, J., & Yoon, Y. (2015). Removal of acetaminophen and naproxen by combined coagulation and adsorption using biochar: influence of combined sewer overflow components. Environ. Sci. Pollut. Res. Int., 22(13), 10058-10069. doi: 10.1007/s11356-015-4191-6
- Kay, P., Hughes, S. R., Ault, J. R., Ashcroft, A. E., & Brown, L. E. (2016). Widespread, routine occurrence of pharmaceuticals in sewage effluent, combined sewer overflows and receiving waters. Environ. Pollut., 220, 1447e1455. doi: http://dx.doi.org/10.1016/j.envpol.2016.10.087
- Khalil, N., Chen, A., Lee, M., Czerwinski, S. A., Ebert, J. R., DeWitt, J. C., & Kannan, K. (2016).
 Association of Perfluoroalkyl Substances, Bone Mineral Density, and Osteoporosis in the U.S. Population in NHANES 2009-2010. Environ. Health Perspectives 124(1), 81-87. doi: doi: 10.1289/ehp.1307909
- Lajeunesse, A., Blais, M., Barbeau, B., Sauvé, S., & Gagnon, a. C. (2013). Ozone oxidation of antidepressants in wastewater-treatment evaluation and characterization of new by-products by LC-QToFMS. Chem. Cent. J., 7(15), 11.
- Lee, Y., Gerrity, D., Lee, M., Bogeat, A. E., Salhi, E., Gamage, S., . . . von Gunten, U. (2013). Prediction of Micropollutant Elimination during Ozonation of Municipal Wastewater Effluents: Use of Kinetic and Water Specific Information. Environ. Sci. & Technol., 47, 5872-5881. doi: dx.doi.org/10.1021/es400781r
- Lee, Y., Kovalova, L., McArdell, C. S., & von Gunten, U. (2014). Prediction of micropollutant elimination during ozonation of a hospital wastewater effluent. Water Res, 64, 134-148. doi: 10.1016/j.watres.2014.06.027

- Margot, J., Rossi, L., Barry, D. A., & Holliger, C. (2015). A review of the fate of micropollutants in wastewater treatment plants. Wiley Interdisciplinary Reviews: Water, 2(5), 457-487. doi: 10.1002/wat2.1090
- Nothe, T., Fahlenkamp, H., & von Sonntag, C. (2009). Ozonation of wastewater: Rate of ozone consumption and hydroxyl radical yield. Environ. Sci. Technol., 43(15), 5990–5995.
- Pelch, K. E., Reade, A., A.M.Wolffe, T., & F.Kwiatkowski, C. (2019). PFAS health effects database: Protocol for a systematic evidence map. Environ. Int., 130. doi: https://doi.org/10.1016/j.envint.2019.05.045
- Qi, W., Singer, H., Berg, M., Muller, B., Pernet-Coudrier, B., Liu, H., & Qu, J. (2015). Elimination of polar micropollutants and anthropogenic markers by wastewater treatment in Beijing, China. Chemosphere, 119, 1054-1061. doi: 10.1016/j.chemosphere.2014.09.027
- Rozas, O., Baeza, C., Nunez, K., Rossner, A., Urrutia, R., & Mansilla, H. D. (2017). Organic micropollutants (OMPs) oxidation by ozone: Effect of activated carbon on toxicity abatement. Sci. of the Total Environ, 590-591, 430-439. doi: 10.1016/j.scitotenv.2016.12.120
- Ryu, J., Oh, J., Snyder, S. A., & Yoon, Y. (2014). Determination of micropollutants in combined sewer overflows and their removal in a wastewater treatment plant (Seoul, South Korea).
 Environ. Monit. Assess, 186(5), 3239-3251. doi: 10.1007/s10661-013-3613-5
- Seredyńska-Sobecka, B., Tomaszewska, M., & W.Morawski, A. (2005). Removal of micropollutants from water by ozonation/biofiltration process. Desalination, 182(1-3), 151-157. doi: https://doi.org/10.1016/j.desal.2005.03.015
- Shu, Z., Singh, A., Klamerth, N., McPhedran, K., Bolton, J. R., Belosevic, M., & Gamal-El-Din,
 M. (2016). Pilot-Scale UV/H₂O₂ Advanced Oxidation Process for Municipal Reuse Water:
 Assessing Micropollutant Degradation and Estrogenic Impacts on Goldfish (Carassius)

 auratus
 L.).
 Water
 Res,
 101,
 157-166.
 doi:

 http://dx.doi.org/10.1016/j.watres.2016.05.0790043-1354/©2016

- Stamm, C., Joss, A., Ackermann, M., & R.I.L.Eggen. (2016). Unravelling the Impacts of Micropollutants in Aquatic Ecosystems: Interdisciplinary Studies at the Interface of Large-Scale Ecology. Adv. in Ecol. Res. (Vol. 55, pp. 183-223). doi: http://dx.doi.org/10.1016/bs.aecr.2016.07.002183
- Tondera, K., Klaer, K., Gebhardt, J., Wingender, J., Koch, C., Horstkott, M., . . . Pinnekamp, J. (2015). Reducing pathogens in combined sewer overflows using ozonation or UV irradiation. Int. J. Hyg. Environ, Health, 218(8), 731-741. doi: 10.1016/j.ijheh.2015.09.002
- Tondera, K., Klaer, K., Koch, C., Hamza, I. A., & Pinnekamp, J. (2016). Reducing pathogens in combined sewer overflows using performic acid. Int J Hyg Environ Health, 219(7 Pt B), 700-708. doi: 10.1016/j.ijheh.2016.04.009
- USEPA. (2010). NPDES Permit Writers' Manual (EPA-833-K-10-001). Washington, DC: Office of Water Washington DC.
- von Sonntag, C., & von Gunten, U. (2012). Chemistry of Ozone in Water and Wastewater Treatment: IWA Publishing
- Wang, W., Xu, Z., Li, H., & Jin, W. (2015). Optimization of coagulation–flocculation process for combined sewer overflow wastewater treatment using response surface methodology. Desalination & Water Treat., 57(32), 14824-14832. doi: 10.1080/19443994.2015.1067832
- Wildhaber, S. Y., Mestankova, H., Scharer, M., Schirmer, K., Salhi, E., & von Gunten, U. (2015).
 Novel test procedure to evaluate the treatability of wastewater with ozone. Water Res, 75, 324-335. doi: 10.1016/j.watres.2015.02.030
- Wojtenko, I., Stinson, M. K., & Field, R. (2001a). Challenges of Combined Sewer Overflow Disinfection by Ultraviolet Light Irradiation. Critical Reviews in Environ. Sci. & Technol., 31(3), 223-239. doi: 10.1080/20016491089217
- Wojtenko, I., Stinson, M. K., & Field, R. (2001b). Performance of Ozone as a Disinfectant for Combined Sewer Overflow. Critical Reviews in Environ. Sci. & Technol., 31(4), 295-309. doi: 10.1080/20016491089235

Waters Protection Ordinance AS 1998 2863 C.F.R. (2020).

- Xu, P., Janex, M.-L., Savoye, P., Cockx, A., & Lazarova, V. (2002). Wastewater disinfection by ozone: main parameters for process design. Water Res., 36, 1043-1055.
- Zhou, H., Liu, J., Xia, H., Zhang, Q., Ying, T., & Hu, T. (2015). Removal and reduction of selected organic micro-pollutants in effluent sewage by the ozone-based oxidation processes. Chem. Eng. J., 269, 245-254. doi: 10.1016/j.cej.2015.01.116

CHAPTER 2: Literature Review

2.1. Treatment of Combined Sewer Overflow

2.1.1. Introduction

Wastewater collection networks represent a fundamental component of a city's infrastructure and its wastewater treatment process. In areas where those networks are designed to accommodate storm water runoffs in addition to regular domestic and industrial wastewaters, they become known as combined sewers. Their operation varies seasonally reaching full capacity during events of heavy rain and intensive snow melts. In such cases, wastewater treatment plants (WWTPs) to which those networks are connected and which are theoretically designed to operate with a small footprint, become unable to contain the excess wastewater during wet weather conditions which ends up overflowing to the closest water stream as per the combined sewer system design. According to the USEPA (2018), this phenomenon is known as combined sewer overflow (CSO) and it consists of regular untreated wastewater diluted by storm water. The volume of poorly treated or untreated CSO and the frequency of its discharge will essentially contribute to the concentration of pollutants (including pathogens, organic and inorganic compounds) in the receiving water body (Kay et al., 2016; Tondera et al., 2016), and consequently affect its suitability for drinking or recreational purposes. Problems with irresponsible and unsafe CSO discharge can further transcend, impacting the aquatic ecosystem and public health with detrimental waterborne diseases (Jalliffier-Verne et al., 2016).

CSO is commonly characterized in terms of its mass loading of macropollutants including, metals, suspended solids (SS), total nitrogen (TN), *Escherichia coli* (*E. coli*), biochemical oxygen demand (BOD), chemical oxygen demand (COD), total phosphorus (TP) and orthophosphate (ortho-P) (Diaz-Fierros et al., 2002; Gasperi et al., 2012; Irvine et al., 2005; Kwon et al., 2014);

Pongmala et al. (2015); (Soonthornnonda & Christensen, 2008). Among the metals studied in CSO, zinc and copper were the most abundant with concentrations reaching up to 3525 μ g/L and 1180 μ g/L, respectively. Other CSO constituents include around 180 mg/L of BOD, around 270 mg/L of COD, 240 mg/L of total solids, 280 mg/L of TP, and 70 mg/L of ortho-P (Gasperi et al., 2008). It is believed that CSO discharges increase the level of *E. coli* in the aquatic environment in a cumulative manner (by 0.5 to 2 log) (Madoux-Humery et al., 2016). It also participates to the mass load of metal contaminants in the receiving environment (Irvine et al., 2005). Recently, there have been a growing interest in studying the contribution of CSO to the release of micropollutants (MPs) into the environment (Evans et al., 2016; Gasperi et al., 2008; Madoux-Humery et al., 2016).

The objective of this section in Chapter 2 is to provide a comprehensive literature review regarding the applied and potential CSO treatment processes for the removal of macro and micropollutants and for disinfection. This section also aims to provide an overview of the gaps and limitations in those processes and some recommendations to overcome them.

2.1.2. Treatment options for the removal of macropollutants from CSO

The term "macropollutants" is used to refer to conventional contaminants that are available at high concentrations, in the range of mg/L. Macropollutants include organics, inorganics, and pathogens (Verlicchi et al., 2012). These compounds are usually regulated in the final wastewater effluent on different governmental levels. To target their removal from CSO, physical and chemical processes that can be applied during primary treatment or for disinfection are the most common in literature. As such, coagulation/ flocculation/ sedimentation processes using aluminum sulfate (alum), ferric chloride (FeCl₃) or polyaluminum chloride (PACl) have been applied to study the elimination of macropollutants (namely: turbidity, SS, COD) (El Samrani et al., 2008a; Gibson

et al., 2016; Jung et al., 2015; Wang et al., 2015b). Also, in a successful attempt, tertiary reed beds were used to maximize the removal of BOD, total suspended solids (TSS), and ammonia (NH₄-N) from CSO and match target values in tertiary treatment systems (Green et al., 1999). The disinfection of CSOs has also been a prominent topic with ultra-violet irradiation (UV) (Tondera et al., 2015; Wojtenko et al., 2001a), ozone (O₃) (Tondera et al., 2015; Wojtenko et al., 2001b), performic acid (Chhetri et al., 2015; Chhetri et al., 2014; Tondera et al., 2016), and peracetic acid (Chhetri et al., 2016; Chhetri et al., 2014; Coyle et al., 2014) proven to be effective. In comparison, studies on the removal of MPs from CSO are still scarce (Jung et al., 2015; Pongmala et al., 2015).

Evidence of full scale applications of the optimum coagulant dose are rarely available in literature where most studies are limited to bench scale applications (Baghvand et al., 2010; El Samrani et al., 2008b; Guida et al., 2007; Wang et al., 2015b). Software models were built for real time reading using sensors which help, to a certain extent, with the prediction of chemical doses especially during wet weather flow (Ratnaweera & Fettig, 2015; Tik & Vanrolleghem, 2017).

Coagulation is the process during which colloids are destabilized and aggregated forming larger particles able to settle out. In wastewater treatment, colloids are particles with a size ranging between 0.001 and 1 µm and are normally in suspension because of Brownian motion, the random movement and collision with water molecules. On the other hand, flocculation is the process of formation of larger particles or flocs due to colloidal aggregation which can be induced by mixing and velocity gradients. In chemical coagulation/ flocculation, a coagulant and a flocculent are added. The coagulant is the chemical (natural or synthetic polymer or a metal salt) that initiates colloidal destabilization. The flocculent is the chemical (typically an organic polymer) that promotes flocculation and strengthens the formed flocs (Metcalf & Eddy, 2014). Polymers act through charge neutralization, polymer bridging or electrostatic patch. During charge neutralization, the polymer reduces the surface charge of the negatively charged colloids in

wastewater diminishing the forces of repulsion between them which keep them in suspension, and amplifying the forces of attraction, which result in the formation of microflocs (Lee et al., 2014a). During polymer bridging, polymers adsorb to the particle surface forming loops and tails making the particle more likely to bind with another one that has similarly adsorbed polymers and thus producing larger and sturdier flocs. In cases where the flocculent is abundantly added, it is expected that the particle surface will become excessively covered with polymers and that polymer bridging will no longer become possible, instead particles re-stabilization would take place. In electrostatic patch, cationic polymers adsorb onto the surface of negatively charged colloids forming cationic or positive "patches" at the adsorption site that attract to the negative sites on other colloids and result in particle aggregation. The resulting flocs are usually stronger than those formed by charge neutralization but weaker than those obtained by polymer bridging (Lee et al., 2014a). While cationic polymers are more prone to function by charge neutralization and electrostatic patch (Lee et al., 2014a), they are less applied than anionic polymers which are widely used and seem to act by polymer bridging (Haydar & Aziz, 2009).

In one study (El Samrani et al., 2008a), two metal-based coagulants FeCl₃ and PACl were tested and compared for the chemical coagulation of CSO without the addition of coagulant-aids. The aluminum-based coagulant (i.e., PACl) was privileged for being more effective in the reduction of turbidity (that was associated with the removal of metals, namely: copper, lead, zinc and chromium), and for having lower optimum dose than FeCl₃. Coagulant although overdosing would result in metals resuspension and release. It was thus recommended to match the necessary coagulant dose with the fluctuating CSO quality parameters that can be predicted by its conductivity. Alum was also applied in the coagulation/ flocculation of CSO where a dose of 100 mg/L was found to maximize UV transmittance to 85%. When coupled with a cationic polymer the latter was recognized for fast-acting, enhancing the formation of larger flocs and increasing

the turbidity removal, although it had no major impact on TSS or UV transmittance (Gibson et al., 2016). In coagulation with aluminum-based salts, it is the salt hydrolysis product that is responsible for particles destabilization. As such, when alum dissociates in water, it forms metal complexes known as coordination compounds which undertake the destabilization process (Metcalf and Eddy, 2003). Some of the important precipitation reactions (during chemical phosphorous removal) that take place when alum is added to wastewater include the reaction between alum and calcium bicarbonate producing aluminum hydroxide precipitate that can enmesh and remove colloids (Reaction 2.1), and between alum and phosphorus producing insoluble aluminum phosphate (Reaction 2.2).

$$Al_{2}(SO_{4})_{3} \cdot 18 H_{2}O + 3 Ca(HCO_{3})_{2} \leftrightarrow 3 CaSO_{4} + 2 Al(OH)_{3} + 6 CO_{2} + 18 H_{2}O \qquad Reaction \ 2.1$$
$$Al^{3+} + PO_{4}^{3-} \leftrightarrow AlPO_{4} \qquad Reaction \ 2.2$$

2.1.3. Treatment options for the removal of micropollutants from CSO and wastewater effluents and for disinfection

Micropollutants (MPs) are natural or anthropogenic compounds present in the environment at very low concentrations that can provoke undesirable impacts on the environmental and public health. There are different classes of MPs including: pharmaceuticals, personal care products, herbicides, pesticides, industrial products and surfactants (Stamm et al., 2016).

As the problem of disposal of MPs without appropriate treatment arose, researchers started shedding more light on their occurrence, sources and eventually treatment options (Kay et al., 2016; Lee et al., 2014b). Unsurprisingly, CSOs can be as a relevant source of MPs (pharmaceuticals, personal care products, herbicides and pesticides) as WWTPs and thus they can imply comparable adverse effects on aquatic and terrestrial fauna and flora (Shu et al., 2016). Nevertheless, until now, there are no national standards or guidelines to regulate the permissible

concentrations of MPs in the final effluent in North America. Most of those MPs have been registered under a list of Contaminants of Emerging Concerns (CECs) or were allocated a maximum contaminant level goal in drinking water (USEPA, 2010) to protect the public health, while there is a growing motive for their regulation. In contrast, several European countries, primarily Switzerland have been leading the way in regulating MPs in effluent wastewater in addition to researching and applying techniques for their removal. Meanwhile there have been screening and modeling approaches to estimate the overall load of MPs in the environment and mitigate their cost-prohibitive and impractical quantification (Arlos et al., 2014; Johnson & Williams, 2004; Launay et al., 2016; Mutzner et al., 2016; Tolouei et al., 2019).

In an attempt to evaluate the overall wastewater treatment efficiency for the removal of MPs, there have been several studies surveying the fate of MPs in conventional WWTPs and the capacity of different advanced oxidation processes (AOPs) for their degradation. Accordingly, it is estimated that conventional WWTPs are less effective for MPs removal compared to macropollutants removal. Interestingly, the removal efficiency of the treatment still does not guarantee meeting safe and acceptable concentrations in the effluent for final disposal. Ryu et al. (2014) alleged that physical, chemical and biological processes in a WWTP can decrease the MPs in the effluent only by 28% as compared to 97% decrease in suspended solids and 86% decrease in COD. Qi et al. (2015) traced the removal of some MPs in WWTPs. The authors confirmed that the effect of the treatment processes on their removal can be inconsistent between WWTPs and dependent on the characteristics of each MP. Overall, coagulation/ flocculation was not very effective for the majority of the studied MPs. It was however confirmed that activated sludge and bioreactors can remove 40% to 60% of pharmaceuticals and personal care products (PPCPs) (Miege et al., 2009; Sui et al., 2011) and generally more than 70% of the non-polar compounds. Their removal mechanism consists mainly of sorption (onto suspended solids or dissolved colloidal matter), biotransformation (by microorganisms as growth material, or by enzymes in side reactions) or volatilization (during aeration). Polar compounds are less removed in those processes. Other researchers (Das et al., 2017; Lee et al., 2011; Margot et al., 2015; Qi et al., 2015; Sui et al., 2010) further confirmed that activated sludge and biodegradation were less efficient compared to oxidation which removed more than 70% of selected PPCPs. Table 2.1 shows the removal efficiency and effluent occurrence of selected MPs from three different categories in conventional WWTPs.

While upgrading and optimizing the conventional wastewater treatment for enhanced removal of TSS, BOD, DOC and ammonium can result in better removal of hydrophobic MPs, other hydrophilic MPs that are difficult to degrade persist. Therefore, introducing tertiary treatment is recommended. In particular ozonation and adsorption are praised for their ability to remove persistent compounds as well as for their cost and energy efficiency (Das et al., 2017; Margot et al., 2015; von Sonntag & von Gunten, 2012).

Tab	le 2.	1	Percent	removal	and	effluent	concentrations	of	selected	MPs	in	typical	WWTPs
(Ada	apted	l fr	om:(Ma	rgot et al.	, 201	5) and (I	Das et al., 2017))					

Mic	cropollutant	Class	Typical WWTP %	Typical Effluent Concentration				
			Removal	(ng/L)				
	Pharmaceuticals and Personal Care Products (PPCPs)							
1	Carbamazepine	Anticonvulsant	16	140-832				
2	Carbendazim	Antifungal	30	81-100				
3	Clindamycin	Antibacterial	10	50-115				
4	Fluconazole	Antifungal	15	108-110				
5	Gemfibrozil	Lipid-regulating agent	39	1-420				
6	Ibuprofen	Anti-inflammatory	80	81-460				
7	Sulfamethoxazole	Antibacterial	44	238-1190				
8	Naproxen	Anti-inflammatory	40	27-462				
9	Trimethoprim	Antibacterial	35	150-482				
10	Ethinylestradiol	Semisynthetic	60	0-5				
		Estrogen						
11	Triclosan	Antiseptic	90	75-200				
Herbicides								
12	Atrazine	Herbicide	23	4-10				
13	2,4-Dichlorophenoxyacetic	Herbicide	-	13				
	Acid							
14	Mecoprop	Herbicide	25	424-500				
15	Diazinon	Insecticide	40	40-173				
Industrial Compounds								
16	Perfluorooctanoic Acid	Surfactant	1	13				
17	Heptadecafluoro-	Surfactant	1	12				
	octanesulfonic acid							

In fact, O₃ was an effective oxidant in wastewater treatment as concluded in multiple studies on the ozonation of real wastewater for the removal of MPs in secondary effluents (Altmann et al., 2014; Can & Çakır, 2010; Huber et al., 2005; Ikehata et al., 2008; Jin et al., 2016; Lee et al., 2014b; Nothe et al., 2009; Wildhaber et al., 2015; Zhou et al., 2015). Similarly, applying ozonation for wastewater disinfection purposes has also been known and practised for a while (Xu et al., 2002).

However, in CSO treatment, ozonation was only studied for disinfection. Wojtenko et al. (2001b) presented a comprehensive review for the application of ozonation in CSO treatment including advantages and disadvantages (Table 2.2). Tondera et al. (2015) used a synthetic CSO by highly diluting the influent wastewater with groundwater (1:7 to 1:3 as influent wastewater:

groundwater). The authors performed the ozonation experiments in semi-batch systems with relatively high specific O_3 doses ranging between 0.6 and 4.8 g O_3 /g DOC applied over 15 minutes contact time to mainly target disinfection. The researchers reported 3.4 log removal of *E. coli* with 8.4 -12.1 mg/L O_3 and concluded that ozonation was efficient for the reduction of bacteria, viruses and parasite.

Table 2.2 The advantages and disadvantages of applying ozonation for the disinfection of CSO (Adapted from (Wojtenko et al., 2001b))

Ozonation Advantages	Ozonation Disadvantages
 Very strong disinfectant 	 Unstable, must be manufactured in-situ
 Does not require purchase, shipment, storage 	 High capital cost, operation and
or handling of chemicals	maintenance cost
 Does not require large space 	 Difficult to quantify in water
 Easily adaptable to existing facilities 	 Requires relatively high equipment maintenance
 Requires short contact time 	 Must be coupled with biological filtration
 Effective for a wide range of microbes 	to minimize potential regrowth
 Low half-life (no toxic residual) 	 Reacts with organics (substrates for microbial growth)
 Improves the receiving water quality by: effluent decolourization, turbidity and odor reduction, increase in dissolved oxygen concentration 	 Safety consideration: O₃ is toxic and corrosive
 No toxicity to aquatic life after ozonation 	 No full-scale application of O₃ for CSO treatment

2.1.4. Bridging the gaps for improved CSO treatment

The primary treatment of PI during wet weather flow is inevitable at least for solids removal.

Yet its operation necessitates optimizing the process parameters to case-specific applications which also means upscaling bench scale experiments and testing various chemicals.

There are several concerns associated with the ozonation of PI during wet weather and wastewater in general. While the ultimate role of O_3 is the oxidation of specific target compounds, the complexity of a wastewater matrix entails the presence of O_3 sinks such as nitrites and nitrates. It is estimated that the stoichiometry of their reaction with O_3 is 1:1 in molar concentration (Lee

et al., 2013). Due to the presence of such compounds, ozonation is often applied after biological treatment to optimize its efficiency. Thus, these factors are to be accounted for when adjusting the O₃ dosage in complex matrices especially in CSO treatment.

Also, ozonation is known to affect the chemical structure and composition of wastewater (Jin et al., 2016). However, it has been claimed that although there is a variety of ozonation byproducts (bromate, aldehydes, ketones, peroxide), only very low concentrations of those compounds have been identified in the effluent and receiving waters (Wojtenko et al., 2001b). Therefore, the best approach to study the impact of the treatment on the overall toxicity of the effluent is under real environmental conditions and in real wastewater. Analyzing samples from the receiving water might also be needed.

2.1.5. Conclusions

Given the uniqueness of the wastewater during wet weather flow in terms of variability in its quality and frequency of occurrence, there is a need for an appropriate and convenient treatment that can be relatively fast and economic and can secure enhanced contaminants elimination. Achieving high removal of suspended solids and organic matter, pathogens and MPs are the top priority for a full comprehensive treatment. Due to the time limitations, an optimum treatment must be limited to few steps only. Therefore, primary treatment through coagulation, flocculation and settling combined with or followed by another treatment step that can guarantee disinfection and removal of MPs are needed. Among the several options, ozonation appears to be a viable treatment. A careful optimization of all the operation conditions is needed in order to achieve a comprehensive treatment and secure the best possible effluent quality before discharge.

2.2. Micropollutants degradation by PAC-catalyzed ozonation

2.2.1. Introduction

Ozone (O_3) is a strong selective oxidant and disinfectant. It is applied in water and wastewater treatment to oxidize, disinfect or assist other processes. For instance, during preozonation, O_3 is applied prior to coagulation to destabilize particles and enhance their removal process. In aqueous systems, O₃ undergoes direct or indirect reactions (Beltran, 2004). In direct O_3 reactions, molecular O_3 reacts with other chemical species through oxidation-reduction, substitution (i.e. nucleophilic and electrophilic reactions) and cycloaddition. These reactions are governed by kinetics and produce by-products such as aldehydes, ketones, carboxylic acids and polymers. Alternatively, in indirect O₃ reactions, targeted species do not react with molecular O₃ but rather with hydroxyl radicals ('OH) that are a product of O_3 decomposition in water either through reaction with hydroxyl ions (OH⁻) or through direct interactions with other species (Beltran, 2004). 'OH can initiate a series of chain reactions that either consume the 'OH or form radical intermediates producing another 'OH. Depending on the net production of 'OH, a compound can be classified as an initiator (reacting with O₃ to form the 'OH), promoter (reacting with the 'OH to from another 'OH via chain reactions) or inhibitor (reacting with and consuming the 'OH) in the ozonation process (Staehelin & Hoigne, 1985; Westerhoff et al., 1999). Also, some species in the aqueous system can react easily and preferentially with 'OH, and are thus known as 'OH scavengers or O₃ decomposition inhibitors. Such species include: carbonates and bicarbonates, tertiary-butanol (TBA), p-chlorobenzoate (pCBA), and excess natural organic matter (NOM) (Beltran, 2004; Stucki, 1988).

Single ozonation is one common technique that has been tested to oxidize a variety of micropollutants (MPs). It was examined for the removal of several MPs as a standalone treatment for model compounds (Seredyńska-Sobecka et al., 2005), or in combination with other processes

such as following secondary treatment of real wastewater (Huber et al., 2005; Lajeunesse et al., 2013; Lee et al., 2013; Nothe et al., 2009). Nevertheless, studies have identified several MPs with limited reactivity with ozone such as: phthalate (Beltran et al., 2002; Jin et al., 2012; Ning et al., 2007), bezafibrate, iomeprol (Altmann et al., 2014), benzotriazole (Altmann et al., 2014; Hollender et al., 2009), atenolol (Hollender et al., 2009), MPs with amine groups (Hubner et al., 2015), pesticides (Broseus et al., 2009), and saturated aliphatic compounds (Jin et al., 2012).

Recent trends in O₃-based advanced oxidation processes (AOPs) suggest the application of catalysts to improve the generation of radicals and thus the removal of compounds that are less reactive with molecular O₃. In order to improve the overall performance of ozonation, and thus the removal of organics and mineralization, a catalyst can be applied to accelerate the rate of O₃ reactions without undergoing any significant chemical reaction. Two types of catalytic ozonation processes can be discerned: homogeneous catalytic ozonation (HoCO) with water soluble catalyst, and heterogeneous catalytic ozonation (HeCO) with an insoluble catalyst. Some of those catalysts are listed in Table 2.3. In environmental applications, the latter is generally preferred over HoCO for being more cost-efficient, easily separable from the reaction media, and safer to the environment and public health, especially since the catalyst in HoCO remains in solution raising concerns about added pollutants (Beltran, 2004). The application of catalytic ozonation for the removal of MPs in aqueous systems is still a work in progress given the wide variety of available catalysts along with the huge list of MPs that exhibit different chemical structures and characteristics and the complexity in discerning the exact mechanisms involved in the process. So far, most of the work in this field has been performed on model compounds or simulated media with the exception of only few studies that tackled real wastewater samples (Kolosov et al., 2018; Mecha et al., 2016; Melero et al., 2009; Sui et al., 2012).

Catalytic ozonation process	Catalyst	Reference
НоСО	Transition metals: Mn ²⁺ , Co ²⁺ ,	(Guo et al., 2018)
	$Fe^{2+}, Fe^{3+}, Zn^{2+}, Cu^{2+}, Ni^{2+}, Ti^{2+}$	
	Ultraviolet light	(Mecha et al., 2016)
	Solar light	(Agustina et al., 2005)
HeCO	Iron oxides	(Wang & Bai, 2017)
	Titanium oxide	(Rodriguez et al., 2012)
	Aluminum Oxide	(Nawrocki & Fijołek, 2013)
	Carbon xerogel	(Alvarez et al., 2008; Orge et al., 2012)
	Multiwalled carbon nanotubes	(Gonçalves et al., 2013b; Oulton et al., 2015)
	Granular Activated Carbon	(Gümüs & Akbal, 2017)
	Powdered Activated Carbon	(Rozas et al., 2017)

Table 2.3 Soluble and insoluble catalysts applied in HoCO and HeCO and available in literature

Carbon-based catalysts are common in literature and have been applied in different forms, alone and in combination with other catalysts. The interest in carbon material is mainly associated with its practicality in terms of cost and ease of application, in addition to its potential dual role as catalyst and adsorbent. As such, multi-walled carbon nanotubes (Fan et al., 2014; Gonçalves et al., 2010; Gonçalves et al., 2013b; Liu et al., 2011; Oulton et al., 2015; Restivo et al., 2012), carbon nanofibers (Restivo et al., 2013), and activated carbon (AC) in powder or granular forms (Faria et al., 2008b; Goncalves et al., 2012; Gonçalves et al., 2013a; Gonçalves et al., 2014; Merle et al., 2010; Rozas et al., 2017; Sanchez-Polo et al., 2005) have been examined as single catalysts. Also, multi-walled carbon nanotubes with manganese oxide (Jalali, 2016; Sui et al., 2012), AC with cerium oxide (Faria et al., 2009), and granular activated carbon (GAC) with transition metals (Hammad Khan & Jung, 2008) have been investigated as combinations of catalysts. Studies on HeCO with PAC only are generally limited and focus on one (Faria et al., 2008a; Faria et al., 2008b; Faria et al., 2009; Goncalves et al., 2012; Gonçalves et al., 2013a; Gonçalves et al., 2014; Guzman-Perez et al., 2011) or a small mixture of four MPs only (Rozas et al., 2017). Those studies were performed using high concentration of target compounds in clean water and they seldom addressed the matrix effect.

The objective of this section in Chapter 2 is to provide an overview of the catalytic ozonation studies performed using PAC as well as the role of the catalyst in the mechanism of O_3 decomposition and MPs degradation. This section also addresses some limitations and discrepancies in the approach to PAC-catalyzed ozonation and means to mitigate them.

2.2.2. Fundamentals of the catalytic mechanisms in PAC-catalyzed ozonation: approaches and main findings

Faria et al. (2008a) studied the removal of oxamic and oxalic acids at a concentration of 1 mM each (equivalent to 90 mg/L) in a semi-batch system with 500 mg/L granular activate carbon (GAC) and ozone supplied at 150 cm³/ min over 2 to 3 hours. The study was performed at natural acidic pH and at pH 7 with phosphate buffer. The latter was found to slow down the removal of target compounds. This effect was associated with the possibility of phosphate buffer adsorbing onto the surface of GAC, limiting its interaction with O₃. At neutral and basic pH, the oxidation of oxamic acid by single or catalytic ozonation was reported very unlikely to happen unless the concentration of hydroxyl radicals is significantly increased. That is because under these conditions, oxamic acid is present in a stable hydrophilic form, and any removal would be due to adsorption. In contrast, catalytic ozonation demonstrated positive impact on the removal of oxalic acid. The addition of TBA inhibited its removal by catalytic ozonation at neutral pH. The removal pattern differed at acidic pH because of the impact of the solution pH on the surface functional groups of GAC (pH_{pzc}), on the dissociation of the target compound (pka) and on the O₃ decomposition pathway. At acidic pH, the addition of TBA did not impact the removal of oxalic acid. This observation was linked to the fact that lower solution pH promotes surface interactions between GAC and oxalic acid and demotes the interactions with radical species in the bulk. This conclusion was contrasted with findings from another study (Beltran et al., 2002). Two different mechanisms were proposed for the removal of oxalic acid during catalytic ozonation. The first mechanism suggests surface reactions by adsorption-surface reaction-desorption and the second suggests interactions in the bulk between oxalic acid and the active radical species, products of O_3 decomposition. The difference between the removal of oxalic and oxamic acids was associated with their chemical structure.

Faria et al. (2008b) researched the removal of sulfanilic acid and benzenesulfonic acid in a similar experimental setup, using the same doses of ozone (150 cm³/min), PAC (500 mg/L) and target compounds (1 mM equivalent to 173.2 mg/L and 158.18 mg/L respectively) at natural acidic pH and neutral pH (in the presence of phosphate buffer). At acidic pH complete removal was achieved for benzenesulfonic acid within 80 min, and the removal rate improved with catalytic ozonation. A major improvement was observed in the reduction of total organic carbon (TOC), since the oxidation of benzenesulfonic acid produced several intermediates, better mineralization was attained in catalytic ozonation systems. With catalytic ozonation at neutral pH, a small improvement in the removal of benzenesulfonic acid was observed while the most obvious impact was on TOC reduction. Maleic and oxalic acids were identified as oxidation intermediates for benzenesulfonic acid. The former was produced at the beginning of the reaction and oxidized by molecular O_3 while the latter formed later during the reaction and persisted and was more abundant at neutral pH. In fact, oxalic acid tends to react better with hydroxyl radicals than with molecular O_3 and its presence indicates the conversion of the parent compound. The addition of TBA prevented the removal of benzenesulfonic acid which proved that its degradation mainly occurs through the radical pathway. With sulfanilic acids, there were no major improvements in the compound's removal when catalytic ozonation was performed, yet a significant increase in TOC removal was noted. The addition of TBA had no impact on the removals. Intermediates that formed during its oxidation process were oxalic and oxamic acids. The fact that oxamic acid is refractory to oxidation justified the lower TOC removal compared to benzenesulfonic acid. The concentrations of ammonium and nitrate were also monitored throughout the process. Nitrate was more abundant at neutral pH while ammonia dominated at acidic pH. Monitoring these products was deemed important to indicate the possible mechanisms of reactions.

Faria et al. (2009) studied the removal of three different dyes in a catalytic ozonation system with the same O_3 and GAC doses as above. The authors also tested cerium oxide as an alternative catalyst at the same dose as GAC. The study was performed in simulated and real matrices of textile effluent after sludge treatment with target compounds at concentrations between 30 and 50 mg/L at natural pH (pH between 5 and 6). Experiments showed that adsorption had a minimal impact on the removal of dyes compared to single ozonation which also allowed an 88% reduction of TOC within 2 hours. In catalytic ozonation with GAC the removal of dyes did not significantly improve, yet the removal of TOC was accelerated. Cerium oxide was found superior to GAC as it showed better mineralization over a long reaction time. The initial concentration of the target compounds and the presence of carbonates and bicarbonates in the matrix were also examined. Carbonates addition did not show any significant impact, while increasing the concentration of dyes lowered the mineralization level. The applicability of single and catalytic ozonation in the treatment of treated (with activated sludge) and untreated effluent wastewater from textile plants was assessed. In treated wastewater, the level of mineralization achieved by single ozonation was lower than by catalytic ozonation with 30% and 57% TOC removal, respectively, after 30 min. This level was significantly lower when the treatments were applied to raw wastewater for the same duration with 6% TOC removal by single ozonation and around 18% by catalytic ozonation. The study claimed that the mechanisms of catalytic ozonation with AC for the removal of those dyes relied on interactions with the surface functional groups of the catalyst where 'OH did not

play any major role. There was no attempt to prove this claim by investigating the impact of radical scavengers on the removal of target compounds.

Merle et al. (2010) examined the removal of 2,4 dichlorophenol and nitrobenzene in a semibatch reactor using 250 mg/L of each of the target compounds with 30 L hr⁻¹ (equivalent to 500 cm³/ min) of O₃ and 1000 mg/L GAC at a natural neutral pH. The study reported that the rate constant of O_3 decomposition was proportional to the catalyst dose. It was also reported that while catalytic ozonation had no impact on the removal of 2,4 dichlorophenol which could be completely removed by single ozonation, it only improved the removal rate of nitrobenzene. Four by-products of the single ozonation of nitrobenzene were then identified and quantified: ortho-, meta- and paranitrophenol and acetic acid. Their concentrations varied throughout the reaction with acetic acid being generated at significantly higher concentrations than the other products. During catalytic ozonation, the same by-products were identified and their removal rates were relatively enhanced. A mass balance for the chemical oxygen demand (COD) of nitrobenzene was performed where the theoretical COD was calculated, the final COD was measured, and the actual COD of the byproducts was then calculated by subtracting COD final from COD theoretical. The theoretical COD of the by-products was computed upon their quantification and then compared to the actual COD of the by-products and was found lower. This meant that there were probably some other by-products that formed but were not quantified. It was also noticeable that the final COD obtained after 4 hours of single ozonation of nitrobenzene was more than 4 times higher than the value obtained after catalytic ozonation for the same period. In contrast, results of 2,4 dichlorophenol oxidation revealed the formation of eight by-products that were monitored over time. COD generation and evolvement with time did not differ between single and catalytic ozonation. The reaction pathways for the oxidation of both compounds were determined accordingly.

Guzman-Perez et al. (2011) studied in a semi-batch system, the removal of 10 mg/L atrazine from a clean water matrix with pH adjustment, using 5 mg/L O₃ and 250-1000 mg/L PAC. The study found that catalytic ozonation increased the removal rate of atrazine in comparison with adsorption and single ozonation (with removals of 80%, 20% and 55% respectively). It also accelerated the rate of O₃ decomposition compared to single ozonation at acidic pH. It was noticed that different types of AC catalysts with similar surface area and pHpzc exhibited the same catalytic activity. The application of 'OH scavenger TBA revealed a decrease in the rate of atrazine removal. TBA adsorbed to a limited extent onto PAC, it slowed down the removal of atrazine, confirming that removals occurred in the bulk solution rather than on the surface of the catalyst. The study showed that after complete O₃ depletion (30 min), at acidic pH 3, the role of adsorption in atrazine removal became more pronounced, as more sites were available on the surface of AC for adsorption. At acidic pH, O₃ adsorbed better onto the surface of the catalyst that had a high pH_{pzc} value of 9.82, while at neutral and basic pH, the hydroxyl surface groups as well as the abundant hydroxyl ions in solution were the major initiators of O₃ decomposition. The kinetics of atrazine removal by catalytic ozonation were performed and the ratio of 'OH to O₃ concentration, known as R_{ct} was computed for the different pH conditions studied. The rate constants for O₃ decomposition and atrazine removal, as well as R_{ct} were higher at neutral pH than those at basic pH where more hydroxyl ions OH^{-} are available that promote the decomposition of O_{3} . Those values also increased linearly as the catalyst dose increased; yet there was no obvious relationship between those constants and pH. As such, it was concluded that pH plays a major role in defining the type of interaction between O₃ and the surface of AC. It was claimed that at lower pH, O₃ adsorbs on the surface of PAC where OH⁻ is scarce, while at neutral and basic pH it reacts with the hydroxyl groups that adsorb onto PAC. This led to the conclusion that there are two different

mechanisms involved during catalytic ozonation and that the pH of the solution is the factor that determines which mechanism can take place.

In another semi-batch system, Goncalves et al. (2012) studied the removal of 50 mg/L sulfamethoxazole in water using 150 cm³/ min O₃ and 140 mg/L AC or multi-walled carbon nanotubes (MWCNT) at natural pH of 4.8. After characterization, the surface area of GAC was reportedly larger and consisted mainly of micropores with basic surface groups and pH_{pzc} of 8.5. Complete removal of sulfamethoxazole was achieved within 30 min of either single or catalytic ozonation. In contrast, adsorption with GAC removed 60% of sulfamethoxazole after 3 hours and more TOC compared to single ozonation. Catalytic ozonation also allowed better TOC removal than single ozonation with 45% reduction in TOC achieved in 3 hours. Scavenging 'OH by addition of TBA during single and catalytic ozonation led to faster removal of sulfamethoxazole. As such, it was concluded that the mechanism of sulfamethoxazole removal was based on the reaction with molecular O₃ and that the purpose of catalyst addition was the improvement of mineralization. The study further investigated the formation of oxidation intermediates and byproducts and tracked their evolvement with time. Two intermediates: 3-amino-5-methylisoxazole, p-benzoquinone, and four by-products: oxamic acid, oxalic acid, pyruvic acid ad maleic acid were identified. Sulfate, nitrite, nitrate and ammonium were also monitored during the reaction considering that the parent compound sulfamethoxazole contains sulfur and nitrogen. While it was reported that all the sulfur content could be transformed to sulfate in single or catalytic ozonation, nitrogen content of sulfamethoxazole was partially converted into ammonium at the beginning of the reaction, in addition to nitrate and other unidentified nitrogenated species. The products were identified in both single and catalytic ozonation but in different proportions. As a result, two pathways for the degradation of sulfamethoxazole during single and catalytic ozonation were suggested depending on the catalyst in use. Microtox bioassays with *Vibrio fischeri* were performed before treatment and after 30 min and 3 hours of single and catalytic ozonation. It was revealed that with both treatments, the toxicity increased after 30 min reaction time and then decreased again after 3 hours while remaining higher than the toxicity of the untreated sulfamethoxazole solution. It was noticeable that the acute toxicity with single ozonation was always higher than that with catalytic ozonation. The intermediates and by-products that were obtained during single or catalytic ozonation were deemed more toxic than the parent compound.

The removal of bezafibrate in a similar reaction set up was studied by Gonçalves et al. (2013a). 20 mg/L of the target compound were subjected to 150 cm³/ min of O₃ and 140 mg/L carbon-based catalysts: GAC or MWCNT at natural pH of 4.4. According to the catalysts characterization, GAC had larger surface area and was more basic (pHpzc value of 8.5) than MWCNT. Also, oxygenated surface groups on both materials were minimal and represented only around 1% by weight. The results showed that bezafibrate removal by adsorption with GAC was 95% and 70 % with MWCNT within 5 hours. Both single and catalytic ozonation achieved full removal of bezafibrate in 20 min, yet catalytic ozonation led to higher level of mineralization with the removal of 60 to 73% of TOC (with MWCT and GAC respectively) after 5 hours. A faster removal of bezafibrate was noted when 'OH scavenger TBA was added, implying that the mechanism of bezafibrate removal relied on molecular O3. The authors recognized that the presence of 0.5 mM TBA in those experiments interfered significantly with TOC results and opted not to track the effect of TBA addition in the system on TOC removal. The potential for catalyst reuse was evaluated in terms of the impact on bezafibrate and TOC removal: while reusing the catalyst had no impact on the degradation of bezafibrate, it was obvious that it limited the level of mineralization. Further investigations using temperature programmed desorption (TPD) spectra revealed that ozonation promoted the formation of oxygenated groups on the surface of the catalyst which tend to reduce the electron density around the catalyst, translating into lower catalytic activity. Ozonation also led to an increase in the overall surface area of the catalyst MWCNT. A closer look at the intermediates and by-products formed during single and catalytic ozonation of bezafibrate resulted in the identification of six different compounds that were monitored over 5hours reaction time and linked to TOC results. It was claimed that within the first few minutes, unsaturated organic compounds were formed and then transformed by reaction with molecular O_3 into lower molecular weight organics such as carboxylic acids (mainly oxamic acid and oxalic acid) that accumulated throughout the 5 hours. Inorganic ions were also produced and monitored, as such it was believed that all chlorine content of bezafibrate was transformed into chlorine ion Cl⁻ within 30 min, 60 min and 2 hours of catalytic ozonation with MWCNT, catalytic ozonation with GAC, and single ozonation, respectively. Nitrogen content was also converted to nitrite, nitrate and ammonia, it entered in the composition of unsaturated organic by-products and finally formed nitrate and some unidentified nitrogenated compounds. The study also noted that the simultaneous use of O₃ and catalyst improved the removal of bezafibrate oxidation by-products by means of adsorption onto the catalyst. It suggested possible bezafibrate degradation pathways during single and catalytic ozonation that culminate in the formation of oxamic, oxalic and pyruvic acids in addition to inorganic ions: nitrite, nitrate, ammonium and chloride. Microtox[®] bioassays with Vibrio fisheri showed a more pronounced increase in toxicity in the first 3 hours of single ozonation compared to catalytic ozonation. This was justified by the adsorption capacity of the catalyst that targeted both bezafibrate and its oxidation intermediates and by-products.

Carbon-based catalysts were also used to study the removal of erythromycin in catalytic ozonation (Gonçalves et al., 2014). 140 mg/L GAC or MWCNT before and after impregnation with cerium oxide were used along with 150 cm³/ min O₃ during the catalytic ozonation of 50

mg/L erythromycin in a semi-batch system at a natural pH of 6.5. The carbon-based catalysts in their different forms were characterized in terms of surface area following Brunauer-Emmett-Teller (BET) theory, surface chemistry with X-ray photoelectron spectroscopy (XPS), structure with X-ray diffraction (XRD) and morphology with Transmission Electron Microscopy (TEM). AC with and without CeO₂ revealed a higher surface area when compared to MWCNT. Upon impregnation with CeO_2 , Ce^{4+}/Ce^{3+} were identified on the surface of the carbon material. Results revealed that more TOC was removed by adsorption with MWCNT than with AC (46% and 25% removal respectively after 5 hours of contact time) which was linked to the surface and textural properties of both materials. The addition of catalyst enhanced the reduction of TOC compared to single ozonation, and the highest level of mineralization was achieved during catalytic ozonation with CeO₂/MWCNT. Three experiments were conducted with MWCNT as catalyst to assess its reutilization. The largest decline in the catalyst activity was observed between the first two cycles where the TOC removal decreased. Following the results of TPD analysis, it was noted that ozonation oxidized the surface of MWCNT producing more oxygenated surface groups that eventually became more abundant every time ozonation took place. The more oxygenated surface groups were present on the catalyst's surface, the less the electron density and thus the lower the catalytic effect in O₃ decomposition. Also, BET results showed that ozonation had little impact on the MWCNT surface texture translated in a small increase in the surface area from one run to the next. When investigating the by-products of catalytic ozonation, six compounds were recognized including oxamic, oxalic and pyruvic acids. Those compounds were tracked during the 5 hours reaction time, where it was noticed that they formed at different times and in varying concentrations depending on the catalyst used. Similarly, because of the nitrogen content of erythromycin, nitrate, nitrite and ammonium ions were monitored during the experiments. Nitrate was the most abundant and it was detected in catalytic ozonation experiments at higher

concentrations than in single ozonation experiments, which was justified by the better removal of erythromycin in the presence of catalyst. The study also indicated the formation of unidentified nitrogenated by-products that contributed to the total TOC in addition to nitrogenated gases that satisfied the nitrogen mass balance. Two pathways were suggested for the removal of erythromycin by catalytic ozonation. In the first pathway, the catalyst initiated O_3 decomposition into radicals, and reactions occurred in the bulk solution between those radicals and the organic pollutants. In the second pathway, the interaction between molecular O_3 and the catalyst's surface generated surface oxygenated radicals that reacted with the organic pollutant, intermediates and by-products on the surface by chemisorption. Those adsorbed compounds could still react with molecular O_3 and O_4 in the bulk becoming more hydrophilic, which caused them to desorb back into the bulk solution resulting in a TOC increase. Microtox® bioassays on Vibrio fischeri unveiled that single ozonation increased the acute toxicity by 42%, and the application of AC and CeO_2 in catalytic ozonation increased it further to 43% and 54%, respectively, despite the higher TOC removal. The study noted that the toxicity level decreased with longer reaction time, and that AC caused lower acute toxicity than MWCNT and its impregnation with ceria decreased the acute toxicity.

So far, all the studies presented above showcased the removal of one compound at a time and at significantly high concentrations compared to the real environmental concentrations. The first approach towards investigating the impact of catalytic ozonation on a mixture of MPs, at relatively low levels, was undertaken by Rozas et al. (2017) who performed the study on a mixture of 4 compounds: atrazine, carbamazepine, diclofenac and triclosan at concentrations 2.8, 2.8, 2.9 and 2.3 mg/L, respectively. Experiments were performed in a semi-batch system, with 2.4 L/ hr O₃ (40 cm³/ min) and 20 mg/L PAC in ultrapure water with pH adjusted to 7.5 using HNO₃ and NaOH, and in river water at natural neutral pH. The compounds were initially individually studied then studied in a mixture. When studied individually with single ozonation, atrazine was removed by 98% within 30 min while the other 3 compounds were completely removed within the first 4 min. Those results were comparable to findings from other studies. The reactivity of O_3 with these 4 compounds varied depending on their chemical structures and compositions. In contrast, in a mixture of compounds in ultrapure water, 90% of atrazine was removed in 30 min of single ozonation, and the application of PAC-catalyzed ozonation improved the removal in 20 min. In order to explain this improvement, adsorption experiments with higher PAC doses were conducted and they showed less than 35% removal of all compounds except triclosan that was removed by 75% in 30 min contact time. The addition of TBA slowed the removal of atrazine which confirmed that the mechanism of its oxidation involved 'OH, but did not impact the removal of the 3 other compounds verifying that molecular O_3 is the main driver for their oxidation. In general, the results were similar in pure and natural (river) water despite the fact that in ultrapure water pH dropped to 4.5 upon addition of O_3 while it remained neutral in river water due to the buffering effect of the low DOC level. Moreover, FTIR results showed that ozonation of PAC significantly affected the alcohol, carbonyl, aromatic and hydroxyl surface functional groups which reflected on the ability of PAC to adsorb organic compounds or generate hydroxyl radicals during catalytic ozonation. A closer look at the intermediates and by-products of the mixture's ozonation led to the identification of 5 different carboxylic acids: oxalic, oxamic, maleic, fumaric and acetic acids that were constantly present in the system. As such, since those carboxylic acids are refractory to O_3 , they can justify why TOC removal reached a plateau at 40% after 15 min of catalytic ozonation in ultrapure water. Toxicity tests were performed using Daphnia magna and they confirmed that toxicity of the initial mixture in ultrapure water, decreased with both single and catalytic ozonation and that the latter was more efficient in reducing the toxicity: 45% and 80% toxicity reduction by

single and catalytic ozonation, respectively, after 15 min.

In summary, the key findings pertaining to the catalytic mechanisms in PAC-catalysed

ozonation as reported in literature are available in Table 2.4.

Table 2.4 Summary of the key findings in literature pertaining to ozonation catalyzed by acti	ivated
carbon	

Effect of pH

Solution pH affects:

- The O₃ decomposition pathway;
- The surface functional groups of the catalyst;
- The dissociation of target compounds
- Abundance of hydroxyl surface groups on PAC and abundance of hydroxyl ions in the bulk solution promote the decomposition of O₃ (at neutral and basic pH);
- Buffer can adsorb onto the catalyst and limit its interaction with O₃.

Effect of O₃ on the catalyst's surface

- Oxidation of the catalyst's surface and formation of oxygenated surface groups;
- Reduction of electron density;
- Increased surface area of the catalyst;
- Reduced catalytic and adsorption capacities to be considered for catalyst's reuse.

Effect of radical scavenger TBA

- Faster degradation of target compounds proves that the degradation occurs mainly via direct pathway by reaction with molecular O₃;
- Decreased or inhibited removal of target compounds proves that the degradation occurs via indirect radical pathway in the bulk solution.

Removal Mechanism in catalyzed ozonation

- Reaction between target compounds adsorbed on the surface of the catalyst and O₃ or radicals in the bulk solution can cause them to desorb and increase TOC;
- Decomposition of O₃ into radicals in the bulk solution that react with target compounds;
- Reaction between target compounds and surface oxygenated radicals formed upon decomposition of O₃ on the catalyst's surface;
- Removal of target compounds by adsorption on the catalyst.

Impact of catalytic ozonation

- While single and catalytic ozonation can have a comparable effect on the removal of target compounds, catalytic ozonation can enhance TOC removal promoting mineralization;
- Catalytic ozonation reduces the acute toxicity (Microtox[®]) more than single ozonation.

2.2.3. Bridging the gaps for better understanding and application of PAC-catalyzed ozonation

2.2.3.1. Catalyst characterization

Detailed characterization of the catalyst is common in most studies on AC-catalyzed ozonation mainly with the purpose of proving or justifying the speculated catalytic mechanism. It includes the measurement of pH_{pzc}, and surface area with N₂ adsorption isotherm, in addition to the characterization of the surface structure and chemistry with X-ray diffraction (XRD) and X-ray photoelectron spectroscopy (XPS) and the quantification of adsorbed material with Fourier-transform infrared spectroscopy (FTIR). There is no doubt that these techniques help demonstrate the changes on the surface of PAC during the catalytic ozonation process, the doubt arises upon the interpretation of the causes of those changes should they exist.

Moreover, the exact composition of AC has an important role. For instance according to Sanchez-Polo et al. (2005), determining the ash content of AC can affect the interpretation of the catalytic mechanism since some metals on the surface of AC can act as catalysts. Also, it is claimed that those metals in addition to the basic surface functional groups induce the decomposition of O_3 in catalytic ozonation. For this reason, pre-treating AC with O_3 prior to AC-catalyzed ozonation can reduce the basic surface functional groups and the surface area. And while this does not affect the pores, it still can reduce the efficiency of AC in the treatment and therefore longer exposure time will be required. This study concluded that AC is an initiator or promoter of O_3 decomposition rather than a catalyst. In a later study (Sánchez-Polo et al., 2007), it was concluded that the role of AC when coupled with O_3 in drinking water treatment is to catalyze the degradation of O_3 and formation of •OH along with the adsorption of organic matter and alkalinity reduction. Alvarez et al. (2008) agreed with the previous study stating that O_3 chemisorbs onto the surface of AC by oxidizing the surface basic groups and that the presence of ash catalyzes its decomposition. The

authors confirmed that the reuse of AC as catalyst is inefficient unless it is preceded by a reactivation process to reproduce basic functional groups. Jaramillo et al. (2010) further confirmed that O_3 reacts with the basic surface groups generating acidic oxygenated groups. Dehouli et al. (2010) supported the concept of O_3 sorption onto AC but refuted that it reacts with basic groups. Instead, the authors suggested that O_3 and OH^- from the bulk adsorb on the AC surface and react together. The researchers argued that the presence of basic negatively charged groups on AC results in the repulsion of OH^- , thus preventing the interaction with O_3 . Dehouli et al. thus concluded that the presence of acidic groups affects the interaction between O_3 and AC.

2.2.3.2. Dosage

The common ranges of applied O_3 doses for MPs' removal are reported per mg DOC as specific O_3 doses: 0.21 - 1.53 mg O_3 /mg DOC (Song et al., 2017; Zimmermann et al., 2011). Several studies exceeded those values disregarding the importance of oxidant dosage in real applications in WWTPs in terms of increased cost, by-products formations and effluent toxicity.

Literature unveils a wide range of PAC doses (10 to 1000 mg/L) and particle sizes applied in catalytic ozonation in a clean water matrix. These significantly wide ranges were often coupled with high doses of target compounds in the mg/L range which clearly exceeded the real environmental concentrations and only satisfied research purposes. It was reported that 2 - 3 g PAC/g DOC is the common specific PAC dose applied in secondary treatment (Rizzo et al., 2019). Still, the majority of the research on this topic showed no attempt to justify the doses of O₃ or catalyst used. The effect of the catalyst dose was rarely presented with any data showing the least catalyst dose required to achieve catalytic activity under certain conditions. In practicality, less is more which means that maintaining the doses of O₃ and the catalyst at a minimum to achieve the maximum removal of contaminants is the most efficient option.

2.2.3.3. Catalytic mechanisms and pathways – Concept validation

Studying the mechanisms involved in the degradation of a specific compound or a mixture of compounds by catalytic ozonation is reported in literature through an evaluation of the effect of radical scavenger addition on the removals. TBA is the most used 'OH scavenger in literature on catalytic ozonation and its applied doses have not been justified. In AC-catalyzed ozonation, it was never clear if the dose of the radical scavenger added is related to the dose of the oxidant or that of the target compound, and if the radical scavenger can interfere in the catalytic ozonation system for instance by adsorbing onto the AC and affecting its interaction with O₃ and target compounds. Faria et al. (2008a) reported two mechanisms of AC catalytic ozonation. In the first mechanism, AC acts as an initiator for O_3 decomposition into 'OH in solution which in turn oxidizes the adsorbed MPs. In the second mechanism, molecular O_3 adsorbs and reacts with the AC surface generating surface radicals that attack the adsorbed MPs so that degradation occurs on the surface of the catalyst. These two proposed mechanisms assume that all degradation reactions occur on the surface and do not consider reactions in the bulk especially since not all types of MPs have high affinity for adsorption. Moreover, the organic carbon that is introduced in the system along with TBA contributes to the TOC measured and thus affects the interpretation of the level of mineralization that could be achieved.

It is remarkable that all studies on PAC-catalyzed ozonation resorted to TBA to scavenge 'OH as one approach towards proving the catalytic mechanism. There was no attempt to evaluate the formation and role of other types of radicals presumably because it is common to assess catalysis by the generation of 'OH, yet the mechanism can be more complex and other strong oxidizing agents or radicals can undertake major roles in the degradation of target compounds. Current trends in studying catalytic ozonation mechanisms resorted to investigating the formation of radials through electron paramagnetic resonance (EPR) spectroscopy as an alternative to radical probing (Song et al., 2019; Wang et al., 2019b). EPR is however not new, it has been applied for decades to identify free radicals: it involves using a spin trap to capture the free radical and form the adducts that are detected by EPR spectroscopy (Buettner, 1987). To the best of our knowledge this technology has not been used in PAC-catalyzed ozonation yet. It is crucial to consider however the strong adsorbing capacity of PAC and the impact that entails on the spin traps used in EPR systems and therefore the accuracy of the results.

2.2.3.4. Intermediates, by-products and toxicity tests

It is possible that the identification of intermediates and by-products during catalytic ozonation is one of the most meticulous tasks in the study. That is because they can form in very small concentrations posing challenges for quantification. They can also help justify the evolution of TOC through the experiment, and exacerbate the pollution and toxicity problems since some have been recognised as more toxic than the parent compound. Actually, toxicity tests have been performed in clean matrices with one or a few MPs using different types of Microtox[®] assays with *vibrio fischeri* (Goncalves et al., 2012; Gonçalves et al., 2014) and *daphnia magna* (Rozas et al., 2017). This approach might not be fully relevant as it does not portray realistic concentrations occurring in wastewater effluents and does not consider the synergistic effect of the mixture of contaminants, so it cannot reflect the real environmental hazard imposed on the receiving waters.

2.2.3.5. Matrix effect

In real water or wastewater applications, the complexity of the matrix has a huge implication on the reaction pathways and eventually on removals. Many factors can interfere or compete for the reaction with molecular O₃ and the radicals produced in the system which can hinder the removal of targeted compounds in wastewater. As such, there are many considerations for upscaling a bench scale study to a full-scale application for the degradation of specific pollutants.

One important parameter that cannot be overlooked when studying the efficiency of catalytic ozonation in real matrices is dissolved organic carbon (DOC) which is known to promote O_3 decomposition (at low concentrations), scavenge 'OH radicals (at high concentrations) and possibly reduce the catalyst's activity by adsorbing to its surface. Sanchez-Polo et al. (2005) proved that the effect of DOC on the catalyst's activity is not very significant; the authors ascertained that the presence of catalyst (AC) boosted the AOP with DOC adsorption. Kovalova et al. (2013) compared DOC removal by adsorption onto PAC to that by oxidation with O_3 and attested that the level of mineralization by O_3 was minimal compared to that by PAC.

Alkalinity and nitrites are other parameters that can weigh in the ozonation process and they can readily react with O₃ (Lee et al., 2013). They are rarely taken into account when real matrix is studied while they can impact the O₃ dosage. Similarly, the microbiological quality of the treated matrix affects O₃ consumption during treatment. Thus, considering O₃ for both disinfection and oxidation must entail satisfying the initial O₃ demand of the wastewater and securing enough residual to react with the target components.

2.2.3.6. Experimental limitations

The effect of pH in these oxidation experiments is of utmost importance, and is often neglected in the analysis of results. The high concentration of target compounds is reflected in the initial pH of the solution which has been mostly leaning towards the acidic side. Those studies tend to adjust the pH instead of using buffer. The importance of the solution's pH lies in the huge impact pH has on the interactions between the oxidant, the catalyst and the target compound. On one hand, it is not new that O₃ decomposition pathway is driven by the solution pH, whereby in acidic environment, the direct pathway is favored, i.e. the reaction between molecular O₃ and the target compound; yet in basic environment the indirect pathway is favoured, and that is the molecular O₃ decomposition into 'OH and the subsequent reaction between those radicals and the target compound. On the other hand, the solution pH affects the surface characteristics of PAC when assessed against the pH_{pzc}. As such, when the solution pH exceeds the pH_{pzc} of AC, more basic surface functional groups are expected, while when pH is lower than pH_{pzc} more acidic surface functional groups are expected. The surface chemistry of AC plays an important role in the interaction with O₃ and the target compounds. Finally, the solution pH also affects the chemistry of the target compound: if the solution pH is higher than pKa, of the compound, the latter is expected to be present in the system mainly in the deprotonated form, and if the pH is lower than pKa, the protonated forms are expected to be more abundant. The prevalence of the acidic or basic form of the target compound can be connected to the possibility of its adsorption onto the surface of AC.

At the same time, attempting to maintain the pH during the experiment at a fixed preferred value through the addition of buffer risks the introduction of species in the system that can interfere with the interaction between O₃/AC/target compound through either influencing the O₃ decomposition rate or adsorbing onto the surface of the catalyst or preferentially affecting the abundance of either the protonated or deprotonated form of the target compound. In contrast, a simple pH adjustment is performed to ensure specific initial conditions but is not enough to mitigate for the pH changes incurred by the reactions and impacting their pathways and thus cannot

reasonably simulate real applications in natural waters and wastewaters. Accordingly, discussing the role of the pH in the reaction mechanism is of high importance.

In a catalytic oxidation system using activated carbon, a very renowned adsorbent, adsorption becomes a crucial means of removal not only for the target compound, but also for any other species in the system. These include the buffer where there might be some interaction between the buffer and the surface of AC hindering or affecting the interaction with O₃ and generation of radicals. Another limitation comes with the addition of organic 'OH scavengers, such as TBA, which add up to the total organic carbon and interfere with reporting the total level of mineralization achieved in catalytic ozonation.

2.2.3.7. Fate of the catalyst: recovery and reuse or disposal

When PAC in particular is used to catalyze the decomposition of O_3 , important issues can arise pertaining to its application in WWTPs. First, there is the issue of its fate: recuperation and regeneration or recuperation and safe disposal or discharge with the effluent without risks to the receiving environment. This requires a meticulous understanding of the PAC size and the difficulty of its recuperation after suspension. If the recuperated PAC is destined for reuse, then there is a need for its reactivation especially after the surface changes caused by exposure to O_3 and adsorption of organic matter. If not, the recuperated PAC should be destined for final disposal with the other sludge generated at the WWTP. If those cannot be applicable, then disposal with the effluent is the least cumbersome yet a detailed assessment regarding its effect on the aquatic life and sediments is needed. Second, there is the issue of the additional cost it can entail for purchase, storage and disposal if any specific disposal measures are required. This can only be addressed through a case-specific cost analysis.

2.2.4 Conclusions

The use of O₃ for the oxidation of MPs has been tested and recommended by many researchers. Yet the fact that many MPs are not reactive with molecular O₃ prompted the addition of a catalyst such as PAC to enhance the formation of radicals. Those are capable of oxidizing more compounds by virtue of their non-selectivity and strong oxidizing potential. The exact mechanism of formation of those radicals along with the mechanism of oxidation of MPs appear to be case-specific. Significant efforts have been made to assess the efficiency of catalytic ozonation at bench-scale level. However, scale-up, cost analysis, and full-scale applications need further investigation.

References

- Agustina, T. E., Ang, H. M., & Vareek, V. K. (2005). A review of synergistic effect of photocatalysis and ozonation on wastewater treatment. Journal of Photochemistry and Photobiology C: Photochemistry Reviews, 6(4), 264-273. doi: 10.1016/j.jphotochemrev.2005.12.003
- Altmann, J., Ruhl, A. S., Zietzschmann, F., & Jekel, M. (2014). Direct comparison of ozonation and adsorption onto powdered activated carbon for micropollutant removal in advanced wastewater treatment. Water Res., 55, 185-193. doi: 10.1016/j.watres.2014.02.025
- Alvarez, P. M., Masa, J., Jaramillo, J., Beltran, F. J., & Gomez-Serrano, V. (2008). Kinetics of Ozone Decomposition by Granular Activated Carbon. Ind. Eng. Chem. Res., 47, 2545-2553.
- Arlos, M. J., Braggb, L. M., Servos, M. R., & Parker, W. J. (2014). Simulation of the fate of selected pharmaceuticals and personal care products in a highly impacted reach of a

Canadian watershed. Sci. of the Total Environ., 193-204. doi: http://dx.doi.org/10.1016/j.scitotenv.2014.03.092

- Baghvand, A., Zand, A. D., Mehrdadi, N., & Karbassi, A. (2010). Optimizing coagulation process for low to high turbidity waters using aluminum and iron salts. American J. of Environ. Sci., 6(5), 442-448. doi: 10.3844/ajessp.2010.442.448
- Beltran, F. J. (2004). Ozone reaction kinetics for water and wastewater systems United States of America: Lewis Publishers
- Beltran, F. J., Rivas, F. J., Fernandez, L. A., Alvarez, P. M., & Montero-de-Espinosa, R. (2002). Kinetics of Catalytic Ozonation of Oxalic Acid in Water with Activated Carbon. Ind. Eng. Chem. Res. 2, 41, 6510-6517. doi: https://doi.org/10.1021/ie020311d
- Broseus, R., Vincent, S., Aboulfadl, K., Daneshvar, A., Sauve, S., Barbeau, B., & Prevost, M. (2009). Ozone oxidation of pharmaceuticals, endocrine disruptors and pesticides during drinking water treatment. Water Res., 43(18), 4707-4717. doi: 10.1016/j.watres.2009.07.031
- Buettner, G. R. (1987). Spin trapping: parameters of spin adducts*. Free Radic. Biol. & Med., 3, 259-303.
- Can, Z. S., & Çakır, E. (2010). Treatability of Organic Constituents in the Paşaköy Wastewater Treatment Plant Effluent by O₃ and O₃/H₂O₂. Ozone: Sci. & Eng., 32(3), 209-214. doi: 10.1080/01919511003796087
- Chhetri, R. K., Bonnerup, A., & Andersen, H. R. (2016). Combined Sewer Overflow pretreatment with chemical coagulation and a particle settler for improved peracetic acid disinfection. J. of Ind. & Eng. Chem., 37, 372-379. doi: 10.1016/j.jiec.2016.03.049
- Chhetri, R. K., Flagstad, R., Munch, E. S., Hørning, C., Berner, J., Kolte-Olsen, A., . . . Andersen, H. R. (2015). Full scale evaluation of combined sewer overflows disinfection using

performic acid in a sea-outfall pipe. Chem. Eng. J., 270, 133-139. doi: 10.1016/j.cej.2015.01.136

- Chhetri, R. K., Thornberg, D., Berner, J., Gramstad, R., Ojstedt, U., Sharma, A. K., & Andersen,
 H. R. (2014). Chemical disinfection of combined sewer overflow waters using performic acid or peracetic acids. Sci. Total Environ., 490, 1065-1072. doi: 10.1016/j.scitotenv.2014.05.079
- Coyle, E. E., Ormsbee, L. E., & Brion, G. M. (2014). Peracetic Acid as an Alternative Disinfection
 Technology for Wet Weather Flows. Water Environ. Res., 86(8), 687-697. doi:
 10.2175/106143014x13975035525663
- Das, S., Ray, N. M., Wan, J., Khan, A., Chakraborty, T., & Ray, M. B. (2017). Micropollutants in Wastewater: Fate and Removal Processes. doi: 10.5772/65644
- Dehouli, H., Chedeville, O., Cagnon, B., Caqueret, V., & Porte, C. (2010). Influences of pH, temperature and activated carbon properties on the interaction ozone/activated carbon for a wastewater treatment process. Desalination, 254, 12-16. doi: doi:10.1016/j.desal.2009.12.021
- Diaz-Fierros, F. T., Puerta, J., Suarez, J., & Diaz-Fierros, F. V. (2002). Contaminant loads of CSOs at the wastewater treatment plant of a city in NW Spain Urban Water, 4, 291-299.
- El Samrani, A. G., Lartiges, B. S., & Villieras, F. (2008a). Chemical coagulation of combined sewer overflow: heavy metal removal and treatment optimization. Water Res., 42(4-5), 951-960. doi: 10.1016/j.watres.2007.09.009
- Evans, W. A., Davies, P. J., & McRae, C. (2016). The occurrence of methyl, ethyl, propyl, and butyl parabens in the urban rivers and stormwaters of Sydney Australia. Environ. Sci. Water Res. Technol. , 2, 733-742. doi: 10.1039/C5EW00240K
- Fan, X., Restivo, J., Órfão, J. J. M., Pereira, M. F. R., & Lapkin, A. A. (2014). The role of multiwalled carbon nanotubes (MWCNTs) in the catalytic ozonation of atrazine. Chem. Eng. J., 241, 66-76. doi: 10.1016/j.cej.2013.12.023
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2008a). Activated carbon catalytic ozonation of oxamic and oxalic acids. Appl. Catal. B: Environ., 79(3), 237-243. doi: 10.1016/j.apcatb.2007.10.021
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2008b). Catalytic ozonation of sulfonated aromatic compounds in the presence of activated carbon. Appl. Catal. B: Environ., 83(1-2), 150-159. doi: 10.1016/j.apcatb.2008.02.010
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2009). Activated carbon and ceria catalysts applied to the catalytic ozonation of dyes and textile effluents. Appl. Catal. B: Environ., 88(3-4), 341-350. doi: 10.1016/j.apcatb.2008.11.002
- Gasperi, J., Garnaud, S., Rocher, V., & Moilleron, R. (2008). Priority pollutants in wastewater and combined sewer overflow. Sci. Total Environ., 407(1), 263-272. doi: 10.1016/j.scitotenv.2008.08.015
- Gasperi, J., Zgheib, S., Cladiere, M., Rocher, V., Moilleron, R., & Chebbo, G. (2012). Priority pollutants in urban stormwater: part 2 - case of combined sewers. Water Res., 46(20), 6693-6703. doi: 10.1016/j.watres.2011.09.041
- Gibson, J., Farnood, R., & Seto, P. (2016). Chemical pretreatment of combined sewer overflows for improved UV disinfection. Water Sci. Technol., 73(2), 375-381. doi: 10.2166/wst.2015.447
- Gonçalves, A. G., Figueiredo, J. L., Órfão, J. J. M., & Pereira, M. F. R. (2010). Influence of the surface chemistry of multi-walled carbon nanotubes on their activity as ozonation catalysts. Carbon, 48(15), 4369-4381. doi: 10.1016/j.carbon.2010.07.051

- Goncalves, A. G., Orfao, J. J., & Pereira, M. F. (2012). Catalytic ozonation of sulphamethoxazole in the presence of carbon materials: catalytic performance and reaction pathways. J. Hazard. Mater., 239-240, 167-174. doi: 10.1016/j.jhazmat.2012.08.057
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2013a). Ozonation of bezafibrate promoted
 by carbon materials. Appl. Catal. B: Environ., 140-141, 82-91. doi: 10.1016/j.apcatb.2013.03.034
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2013b). Ozonation of sulfamethoxazole promoted by MWCNT. Catal. Commun., 35, 82-87. doi: 10.1016/j.catcom.2013.02.012
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2014). Ozonation of erythromycin over carbon materials and ceria dispersed on carbon materials. Chem. Eng. J., 250, 366-376. doi: 10.1016/j.cej.2014.04.012
- Green, M. B., Martin, J. R., & Griffin, P. (1999). Treatment of combined sewer overflows at small wastewater treatment works by constructed reed beds Wat. Sci. Tech, 40(3), 357-364.
- Guida, M., Mattei, M., Della Rocca, C., Melluso, G., & Meric, S. (2007). Optimization of alumcoagulation/flocculation for COD and TSS removal from five municipal wastewater. Desalination, 211(1-3), 113-127. doi: 10.1016/j.desal.2006.02.086
- Gümüs, D., & Akbal, F. (2017). A comparative study of ozonation, iron coated zeolite catalyzed ozonation and granular activated carbon catalyzed ozonation of humic acid. Chemosphere, 174, 218-231. doi: http://dx.doi.org/10.1016/j.chemosphere.2017.01.106
- Guo, Y., Wang, H., Wang, B., Deng, S., Huang, J., Yu, G., & Wang, Y. (2018). Prediction of micropollutant abatement during homogeneous catalytic ozonation by a chemical kinetic model. Water Res., 142(383-385). doi: https://doi.org/10.1016/j.watres.2018.06.019

- Guzman-Perez, C. A., Soltan, J., & Robertson, J. (2011). Kinetics of catalytic ozonation of atrazine in the presence of activated carbon. Sep. & Purif. Technol., 79, 8-14. doi: 10.1016/j.seppur.2011.02.035
- Hammad Khan, M., & Jung, J. Y. (2008). Ozonation catalyzed by homogeneous and heterogeneous catalysts for degradation of DEHP in aqueous phase. Chemosphere, 72(4), 690-696. doi: 10.1016/j.chemosphere.2008.02.037
- Haydar, S., & Aziz, J. A. (2009). Coagulation-flocculation studies of tannery wastewater using combination of alum with cationic and anionic polymers. J. Hazard. Mater, 168(2-3), 1035-1040. doi: 10.1016/j.jhazmat.2009.02.140
- Hollender, J., Zimmermann, S. G., Koepke, S., Krauss, M., Mcardell, C. S., Ort, C., . . . Siegrist,
 H. (2009). Elimination of organic micropollutants in a municipal wastewater treatment plant
 upgraded with a full-scale post-ozonation followed by sand filtration Environ. Sci. & Technol., 43(20), 7862–7869.
- Huber, M. M., Gobel, A., Joss, A., Hermann, N., Loffler, D., Mcardell, C. S., . . . von Gunten, U. (2005). Oxidation of Pharmaceuticals during Ozonation of Municipal Wastewater Effluents: A Pilot Study. Environ. Sci. Technol, 39(11), 4290-4299.
- Hubner, U., von Gunten, U., & Jekel, M. (2015). Evaluation of the persistence of transformation products from ozonation of trace organic compounds - a critical review. Water Res, 68, 150-170. doi: 10.1016/j.watres.2014.09.051
- Ikehata, K., Gamal El-Din, M., & Snyder, S. A. (2008). Ozonation and Advanced Oxidation Treatment of Emerging Organic Pollutants in Water and Wastewater. Ozone: Sci. & Eng., 30(1), 21-26. doi: 10.1080/01919510701728970

- Irvine, K. N., Perrelli, M. F., McCorkhill, G., & Caruse, J. (2005). Sampling and Modeling Approaches to Assess Water Quality Impacts of Combined Sewer Overflows - The Importance of a Watershed Perspective J. Great Lakes Res., 31, 105-115.
- Jalali, H. M. (2016). Kinetic study of antibiotic ciprofloxacin ozonation by MWCNT/MnO2 using Monte Carlo simulation. Mater. Sci. Eng. C. Mater. Biol. Appl., 59, 924-929. doi: 10.1016/j.msec.2015.10.085
- Jalliffier-Verne, I., Heniche, M., Madoux-Humery, A. S., Galarneau, M., Servais, P., Prevost, M.,
 & Dorner, S. (2016). Cumulative effects of fecal contamination from combined sewer overflows: Management for source water protection. J. Environ. Manage., 174, 62-70. doi: 10.1016/j.jenvman.2016.03.002
- Jaramillo, J., Alvarez, P. M., & Gomez-Serrano, V. (2010). Preparation and ozone-surface modification of activated carbon. Thermal stability of oxygen surface groups. Appl. Surf. Sci., 256, 5232-5236. doi: doi:10.1016/j.apsusc.2009.12.109
- Jin, P., Jin, X., Bjerkelund, V. A., Østerhus, S. W., Wang, X. C., & Yang, L. (2016). A study on the reactivity characteristics of dissolved effluent organic matter (EfOM) from municipal wastewater treatment plant during ozonation. Water Res., 88, 643-652. doi: http://dx.doi.org/10.1016/j.watres.2015.10.0600043-1354/©2015
- Jin, X., Peldszus, S., & Huck, P. M. (2012). Reaction kinetics of selected micropollutants in ozonation and advanced oxidation processes. Water Res., 46(19), 6519-6530. doi: 10.1016/j.watres.2012.09.026
- Johnson, A. C., & Williams, R. J. (2004). A Model To Estimate Influent and Effluent Concentrations of Estradiol, Estrone, and Ethinylestradiol at Sewage Treatment Works. Environ. Sci. & Technol., 38, 3649-3658. doi: 10.1021/es035342u CCC: \$27.50

- Jung, C., Oh, J., & Yoon, Y. (2015). Removal of acetaminophen and naproxen by combined coagulation and adsorption using biochar: influence of combined sewer overflow components. Environ. Sci. Pollut. Res. Int., 22(13), 10058-10069. doi: 10.1007/s11356-015-4191-6
- Kay, P., Hughes, S. R., Ault, J. R., Ashcroft, A. E., & Brown, L. E. (2016). Widespread, routine occurrence of pharmaceuticals in sewage effluent, combined sewer overflows and receiving waters. Environ. Pollut., 220, 1447e1455. doi: http://dx.doi.org/10.1016/j.envpol.2016.10.087
- Kolosov, P., Peyot, M.-L., & Yargeau, V. (2018). Novel materials for catalytic ozonation of wastewater for disinfection and removal of micropollutants. Sci. of the Total Environ, 1207-1218. doi: https://doi.org/10.1016/j.scitotenv.2018.07.022
- Kovalova, L., Siegrist, H., von Gunten, U., Eugster, J., Hagenbuch, M., Wittmer, A., ... McArdell,
 C. S. (2013). Elimination of micropollutants during post-treatment of hospital wastewater
 with powdered activated carbon, ozone, and UV. Environ Sci and Technol, 47(14), 78997908. doi: 10.1021/es400708w
- Kwon, K., Lee, S., Jung, Y., & Min, K. S. (2014). Contaminant loads of CSOs at the urban area in Korea. Desalination and Water Treat., 53(11), 3048-3053. doi: 10.1080/19443994.2014.922306
- Lajeunesse, A., Blais, M., Barbeau, B., Sauvé, S., & Gagnon, a. C. (2013). Ozone oxidation of antidepressants in wastewater-treatment evaluation and characterization of new by-products by LC-QToFMS. Chem. Cent. J., 7(15), 11.
- Launay, M. A., Dittmer, U., & Steinmetz, H. (2016). Organic micropollutants discharged by combined sewer overflows Characterisation of pollutant sources and stormwater-related processes. Water Res., 104, 82-92. doi: http://dx.doi.org/10.1016/j.watres.2016.07.068

- Lee, C. S., Robinson, J., & Chong, M. F. (2014a). A review on application of flocculants in wastewater treatment. Process Safety and Environ. Prot., 92(6), 489-508. doi: 10.1016/j.psep.2014.04.010
- Lee, I. S., Sim, W. J., Kim, C. W., Chang, Y. S., & Oh, J. E. (2011). Characteristic occurrence patterns of micropollutants and their removal efficiencies in industrial wastewater treatment plants. J. Environ. Monit., 13(2), 391-397. doi: 10.1039/c0em00130a
- Lee, Y., Gerrity, D., Lee, M., Bogeat, A. E., Salhi, E., Gamage, S., . . . Gunten, U. v. (2013). Prediction of Micropollutant Elimination during Ozonation of Municipal Wastewater Effluents: Use of Kinetic and Water Specific Information. Environ. Sci. & Technol., 47, 5872-5881. doi: dx.doi.org/10.1021/es400781r
- Lee, Y., Kovalova, L., McArdell, C. S., & von Gunten, U. (2014b). Prediction of micropollutant elimination during ozonation of a hospital wastewater effluent. Water Res, 64, 134-148. doi: 10.1016/j.watres.2014.06.027
- Liu, Z. Q., Ma, J., Cui, Y. H., Zhao, L., & Zhang, B. P. (2011). Factors affecting the catalytic activity of multi-walled carbon nanotube for ozonation of oxalic acid. Sep. & Purif. Technol., 78(2), 147-153. doi: 10.1016/j.seppur.2011.01.034
- Madoux-Humery, A. S., Dorner, S., Sauve, S., Aboulfadl, K., Galarneau, M., Servais, P., & Prevost, M. (2016). The effects of combined sewer overflow events on riverine sources of drinking water. Water Res, 92, 218-227. doi: 10.1016/j.watres.2015.12.033
- Margot, J., Rossi, L., Barry, D. A., & Holliger, C. (2015). A review of the fate of micropollutants in wastewater treatment plants. Wiley Interdisciplinary Reviews: Water, 2(5), 457-487. doi: 10.1002/wat2.1090

- Mecha, A. C., Onyango, M. S., Ochieng, A., Fourie, C. J. S., & Momba, M. N. B. (2016). Synergistic effect of UV–vis and solar photocatalytic ozonation on the degradation of phenol in municipal wastewater: A comparative study. J. of Catal., 341, 116-125. doi: 10.1016/j.jcat.2016.06.015
- Melero, J. A., Martinez, F., Botas, J. A., Molina, R., & Pariente, M. I. (2009). Heterogeneous catalytic wet peroxide oxidation systems for the treatment of an industrial pharmaceutical wastewater. Water Res, 43(16), 4010-4018. doi: 10.1016/j.watres.2009.04.012
- Merle, T., Pic, J. S., Manero, M. H., Mathé, S., & Debellefontaine, H. (2010). Influence of activated carbons on the kinetics and mechanisms of aromatic molecules ozonation. Catal. Today, 151(1-2), 166-172. doi: 10.1016/j.cattod.2010.03.044
- Metcalf, & Eddy. (2014). Wastewater Engineering: Treatment and Reuse (Fifth ed.). United States of America: Mc Graw Hill Education.
- Miege, C., Choubert, J. M., Ribeiro, L., Eusebe, M., & M.Coquery. (2009). Fate of pharmaceuticals and personal care products in wastewater treatment plants – Conception of a database and first results. Environ. Pollut., 157, 1721-1726. doi: doi:10.1016/j.envpol.2008.11.045
- Mutzner, L., Staufer, P., & Ort, C. (2016). Model-based screening for critical wet-weather discharges related to micropollutants from urban areas. Water Res., 104. doi: http://dx.doi.org/10.1016/j.watres.2016.08.003
- Nawrocki, J., & Fijołek, L. (2013). Effect of aluminium oxide contaminants on the process of ozone decomposition in water. Appl. Catal. B: Environ., 142, 533-537. doi: http://dx.doi.org/10.1016/j.apcatb.2013.05.069

- Ning, B., Graham, N., Zhang, Y., Nakonechny, M., & Gamal El-Din, M. (2007). Degradation of Endocrine Disrupting Chemicals by Ozone/AOPs. Ozone: Sci. & Eng., 29(3), 153-176. doi: 10.1080/01919510701200012
- Nothe, T., Fahlenkamp, H., & von Sonntag, C. (2009). Ozonation of wastewater: Rate of ozone consumption and hydroxyl radical yield. Environ. Sci. Technol., 43(15), 5990–5995.
- Orge, C. A., Órfão, J. J. M., & Pereira, M. F. R. (2012). Carbon xerogels and ceria–carbon xerogel materials as catalysts in the ozonation of organic pollutants. Appl. Catal. B: Environ, 126, 22-28. doi: 10.1016/j.apcatb.2012.06.029
- Oulton, R., Haase, J. P., Kaalberg, S., Redmond, C. T., Nalbandian, M. J., & Cwiertny, D. M. (2015). Hydroxyl radical formation during ozonation of multiwalled carbon nanotubes: performance optimization and demonstration of a reactive CNT filter. Environ Sci Technol, 49(6), 3687-3697. doi: 10.1021/es505430v
- Pongmala, K., Autixier, L., Madoux-Humery, A. S., Fuamba, M., Galarneau, M., Sauvé, S., . . . Dorner, S. (2015). Modelling total suspended solids, E. coli and carbamazepine, a tracer of wastewater contamination from combined sewer overflows. J. of Hydrol., 531, 830-839. doi: 10.1016/j.jhydrol.2015.10.042
- Qi, W., Singer, H., Berg, M., Muller, B., Pernet-Coudrier, B., Liu, H., & Qu, J. (2015). Elimination of polar micropollutants and anthropogenic markers by wastewater treatment in Beijing, China. Chemosphere, 119, 1054-1061. doi: 10.1016/j.chemosphere.2014.09.027
- Ratnaweera, H., & Fettig, J. (2015). State of the Art of Online Monitoring and Control of the Coagulation Process. [Review]. Water, 7(11), 6574-6597. doi: 10.3390/w7116574
- Restivo, J., Orfao, J. J., Armenise, S., Garcia-Bordeje, E., & Pereira, M. F. (2012). Catalytic ozonation of metolachlor under continuous operation using nanocarbon materials grown on

- a ceramic monolith. J. Hazard. Mater., 239-240, 249-256. doi: 10.1016/j.jhazmat.2012.08.073
- Restivo, J., Órfão, J. J. M., Pereira, M. F. R., Garcia-Bordejé, E., Roche, P., Bourdin, D., . . . Derrouiche, S. (2013). Catalytic ozonation of organic micropollutants using carbon nanofibers supported on monoliths. Chem. Eng. J., 230, 115-123. doi: 10.1016/j.cej.2013.06.064
- Rizzo, L., Malato, S., Antakyali, D., Beretsou, V. G., Đolić, M. B., Wolfgang Gernjak, ... Fatta-Kassinos, D. (2019). Consolidated vs new advanced treatment methods for the removal of contaminants of emerging concern from urban wastewater. Sci. of the Total Environ., 655, 986-1008. doi: https://doi.org/10.1016/j.scitotenv.2018.11.265
- Rodriguez, E. M., Fernandez, G., Alvarez, P. M., & Beltran, F. J. (2012). TiO2 and Fe (III) photocatalytic ozonation processes of a mixture of emergent contaminants of water. Water Res., 46(1), 152-166. doi: 10.1016/j.watres.2011.10.038
- Rozas, O., Baeza, C., Nunez, K., Rossner, A., Urrutia, R., & Mansilla, H. D. (2017). Organic micropollutants (OMPs) oxidation by ozone: Effect of activated carbon on toxicity abatement. Sci. of the Total Environ. 590-591, 430-439. doi: 10.1016/j.scitotenv.2016.12.120
- Ryu, J., Oh, J., Snyder, S. A., & Yoon, Y. (2014). Determination of micropollutants in combined sewer overflows and their removal in a wastewater treatment plant (Seoul, South Korea).
 Environ Monit Assess, 186(5), 3239-3251. doi: 10.1007/s10661-013-3613-5
- Sánchez-Polo, M., Salhi, E., Rivera-Utrilla, J., & Gunten, U. v. (2007). Combination of Ozone with Activated Carbon as an Alternative to Conventional Advanced Oxidation Processes. Ozone: Sci. & Eng., 28. doi: https://doi.org/10.1080/01919510600714170

- Sanchez-Polo, M., von Gunten, U., & Rivera-Utrilla, J. (2005). Efficiency of activated carbon to transform ozone into OH radicals: influence of operational parameters. Water Res., 39(14), 3189-3198. doi: 10.1016/j.watres.2005.05.026
- Seredyńska-Sobecka, B., Tomaszewska, M., & W.Morawski, A. (2005). Removal of micropollutants from water by ozonation/biofiltration process. Desalination, 182(1-3), 151-157. doi: https://doi.org/10.1016/j.desal.2005.03.015
- Shu, Z., Singh, A., Klamerth, N., McPhedran, K., Bolton, J. R., Belosevic, M., & Gamal-El-Din, M. (2016). Pilot-Scale UV/H₂O₂ Advanced Oxidation Process for Municipal Reuse Water: Assessing Micropollutant Degradation and Estrogenic Impacts on Goldfish (*Carassius auratus L.*). Water Res, 101, 157-166. doi: http://dx.doi.org/10.1016/j.watres.2016.05.0790043-1354/©2016
- Song, Y., Breider, F., Ma, J., & von Gunten, U. (2017). Nitrate formation during ozonation as a surrogate parameter for abatement of micropollutants and the N-nitrosodimethylamine (NDMA) formation potential. Water Res., 122, 246-257. doi: 10.1016/j.watres.2017.05.074
- Song, Z., Zhang, Y., Liu, C., Xu, B., Qi, F., Yuan, D., & Pu, S. (2019). Insight into 'OH and O₂⁻⁻ formation in heterogeneous catalytic ozonation by delocalized electrons and surface oxygen-containing functional groups in layered-structure nanocarbons. Chem. Eng. J., 357, 655-666. doi: https://doi.org/10.1016/j.cej.2018.09.182
- Soonthornnonda, P., & Christensen, E. R. (2008). Source apportionment of pollutants and flows of combined sewer wastewater. Water Res, 42(8-9), 1989-1998. doi: 10.1016/j.watres.2007.11.034

- Staehelin, J., & Hoigne, J. (1985). Decomposition of ozone in water in the presence of organic solutes acting as promoters and inhibitors of radical chain reactions. Environ. Sci. & Technol., 19(12), 1206. doi: https://doi.org/10.1021/es00142a012
- Stamm, C., Joss, A., Ackermann, M., & R.I.L.Eggen. (2016). Unravelling the Impacts of Micropollutants in Aquatic Ecosystems: Interdisciplinary Studies at the Interface of Large-Scale Ecology. Adv. in Ecol. Res. (Vol. 55, pp. 183-223). doi: http://dx.doi.org/10.1016/bs.aecr.2016.07.002183
- Stucki, S. (1988). Process technologies for water treatment. New York Plenum Press.
- Sui, M., Xing, S., Sheng, L., Huang, S., & Guo, H. (2012). Heterogeneous catalytic ozonation of ciprofloxacin in water with carbon nanotube supported manganese oxides as catalyst. J Hazard. Mater, 227-228, 227-236. doi: 10.1016/j.jhazmat.2012.05.039
- Sui, Q., Huang, J., Deng, S., Chen, W., & Yu, G. (2011). Seasonal Variation in the Occurrence and Removal of Pharmaceuticals and Personal Care Products in Different Biological Wastewater Treatment Processes. Environ. Sci. & Technol., 45, 3341-3348. doi: dx.doi.org/10.1021/es200248d |
- Sui, Q., Huang, J., Deng, S., Yu, G., & Fan, Q. (2010). Occurrence and removal of pharmaceuticals, caffeine and DEET in wastewater treatment plants of Beijing, China. Water Res, 44(2), 417-426. doi: 10.1016/j.watres.2009.07.010
- Tik, S., & Vanrolleghem, P. A. (2017). Chemically enhancing primary clarifiers: model-based development of a dosing controller and full-scale implementation. Water Sci. & Technol., 75(5), 1185-1193. doi: 10.2166/wst.2016.600
- Tolouei, S., Jean-Baptiste Burneta, Autixiera, L., Taghipoura, M., Bonsteelb, J., Duy, S. V., . . . Dorner, S. (2019). Temporal variability of parasites, bacterial indicators, and wastewater

micropollutants in a water resource recovery facility under various weather conditions. Water Res., 148, 446-458. doi: https://doi.org/10.1016/j.watres.2018.10.068

- Tondera, K., Klaer, K., Gebhardt, J., Wingender, J., Koch, C., Horstkott, M., . . . Pinnekamp, J. (2015). Reducing pathogens in combined sewer overflows using ozonation or UV irradiation. Int. J. Hyg. Environ. Health, 218(8), 731-741. doi: 10.1016/j.ijheh.2015.09.002
- Tondera, K., Klaer, K., Koch, C., Hamza, I. A., & Pinnekamp, J. (2016). Reducing pathogens in combined sewer overflows using performic acid. Int. J. Hyg. Environ. Health, 219(7 Pt B), 700-708. doi: 10.1016/j.ijheh.2016.04.009
- USEPA. (2010). NPDES Permit Writers' Manual (EPA-833-K-10-001). Washington, DC: Office of Water Washington DC.
- USEPA. (2018). Combined Sewer Overflows (CSOs), 2018, from https://www.epa.gov/npdes/combined-sewer-overflows-csos. Accessed June 2019
- Verlicchi, P., Galletti, A., Petrovic, M., & Barceló, D. (2012). Micro-pollutants in Hospital Effluent: Their Fate, Risk and Treatment Options. In D. Barceló (Ed.), Emerging Organic Contaminants and Human Health. The Handbook of Environmental Chemistry (Vol. 20, pp. 139-171). Berlin, Heidelberg: Springer. doi: https://doi.org/10.1007/698 2011 134
- von Sonntag, C., & von Gunten, U. (2012). Chemistry of Ozone in Water and Wastewater Treatment: IWA Publishing
- Wang, J., & Bai, Z. (2017). Fe-based catalysts for heterogeneous catalytic ozonation of emerging contaminants in water and wastewater. Chem. Eng. J., 312, 79-98. doi: http://dx.doi.org/10.1016/j.cej.2016.11.118

- Wang, W., Xu, Z., Li, H., & Jin, W. (2015). Optimization of coagulation–flocculation process for combined sewer overflow wastewater treatment using response surface methodology. Desalination & Water Treat., 57(32), 14824-14832. doi: 10.1080/19443994.2015.1067832
- Wang, Y., Cao, H., Chen, C., Xie, Y., Sun, H., Duan, X., & Wang, S. (2019). Metal-free catalytic ozonation on surface-engineered graphene: Microwave reduction and heteroatom doping. Chem. Eng. J. 355, 118-129. doi: https://doi.org/10.1016/j.cej.2018.08.134
- Westerhoff, P., Aiken, G., Amy, G., & Debroux, J. (1999). Relationships between the structure of natural organic matter and its reactivity towards molecular ozone and hydroxyl radicals.
 Water Res., 33(10), 2265-2276. doi: https://doi.org/10.1016/S0043-1354(98)00447-3
- Wildhaber, S. Y., Mestankova, H., Scharer, M., Schirmer, K., Salhi, E., & von Gunten, U. (2015).
 Novel test procedure to evaluate the treatability of wastewater with ozone. Water Res, 75, 324-335. doi: 10.1016/j.watres.2015.02.030
- Wojtenko, I., Stinson, M. K., & Field, R. (2001a). Challenges of Combined Sewer Overflow Disinfection by Ultraviolet Light Irradiation. Critical Reviews in Environ. Sci. & Technol., 31(3), 223-239. doi: 10.1080/20016491089217
- Wojtenko, I., Stinson, M. K., & Field, R. (2001b). Performance of Ozone as a Disinfectant for Combined Sewer Overflow. Critical Reviews in Environ. Sci. & Technol., 31(4), 295-309. doi: 10.1080/20016491089235
- Xu, P., Janex, M.-L., Savoye, P., Cockx, A., & Lazarova, V. (2002). Wastewater disinfection by ozone: main parameters for process design. Water Res., 36, 1043-1055.
- Zhou, H., Liu, J., Xia, H., Zhang, Q., Ying, T., & Hu, T. (2015). Removal and reduction of selected organic micro-pollutants in effluent sewage by the ozone-based oxidation processes. Chem. Eng. J., 269, 245-254. doi: 10.1016/j.cej.2015.01.116

Zimmermann, S. G., Wittenwiler, M., Hollender, J., Krauss, M., Ort, C., Siegrist, H., & von Gunten, U. (2011). Kinetic assessment and modeling of an ozonation step for full-scale municipal wastewater treatment: micropollutant oxidation, by-product formation and disinfection. Water Res., 45(2), 605-617. doi: 10.1016/j.watres.2010.07.080

CHAPTER 3Bench to Full-scale Enhanced Primary Treatment of Municipal
Wastewater under Wet Weather Flow for Minimized Pollution Load:
Evaluation of Chemical Addition and Process Control Indicators 1

3.1. Introduction

Wastewater collection networks represent a fundamental component of a city's infrastructure and its wastewater treatment process. In areas where those networks are designed to accommodate storm water runoffs in addition to regular domestic and industrial wastewaters, they become known as combined sewers. Their operation thus varies seasonally, reaching its full capacity during events of heavy rain and intensive snow melts. In such cases, wastewater treatment plants (WWTPs) to which those networks are connected become unable to contain the excess wastewater during wet weather conditions. Consequently, untreated wastewater ends up overflowing to the closest water stream as per the combined sewer system design. According to the USEPA (2018) this is known as combined sewer overflow (CSO) and it includes regular untreated wastewater diluted by storm water. The volume of poorly treated or untreated CSO and the frequency of its discharge will essentially contribute to the concentration of pollutants (including pathogens, organic and inorganic chemicals) in the receiving water body, and consequently will affect its suitability for drinking or recreational purposes. Problems with the discharge of untreated CSO can transcend, impacting the aquatic ecosystem and public health with detrimental waterborne diseases (Jalliffier-Verne et al., 2015). It is reported that in Canada, 87% of the population is connected via a wastewater collection network to certain type of treatment,

¹ A version of this chapter was published in a peer reviewed journal as: Alameddine, M., Al Umairi, A. R., Shaikh, M. Z., & Gamal El-Din, M. (2020). Bench to Full-Scale Enhanced Primary Treatment of Municipal Wastewater under Wet Weather Flow for Minimized Pollution Load: Evaluation of Chemical Addition and Process Control Indicators. *Can. J. of Civ. Eng.* doi: https://doi.org/10.1139/cjce-2019-0515

and that only few treatment plants undertake the initiative to store and treat CSO. As such, direct discharges from CSO have not been routinely monitored (Canada, 2019; CWN, 2018).

CSO is commonly characterized in terms of its mass loading of macropollutants. Some of its major constituents include around 45-114 NTUs, 83-600 mg/L of total suspended solids (TSS), 30-180 mg/L of biochemical oxygen demand (BOD), 80-200 mg/L of chemical oxygen demand (COD), 80-190 mg/L of total phosphorus, and 30-50 mg/L of ortho-phosphate (ortho-P) (El Samrani et al., 2008a; Gasperi et al., 2008; Li et al., 2004). CSOs can also be a relevant source of micropollutants (MPs) (pharmaceuticals, personal care products, herbicides and pesticides) as WWTPs (Kay et al., 2016; Phillips et al., 2012) and thus imply similar adverse effects on aquatic and terrestrial fauna and flora (Shu et al., 2016). Ryu et al. (2014) estimated that physical, chemical and biological processes in a WWTP can decrease the MPs in the effluent by 28% as compared to 97% decrease in suspended solids and 86% decrease in COD. This relatively low removal sheds the light on the importance of screening and modeling approaches to estimate the overall load of MPs in the environment and mitigate their cost-prohibitive and impractical quantification (Arlos et al., 2014; Johnson & Williams, 2004; Launay et al., 2016; Mutzner et al., 2016; Tolouei et al., 2019).

During wet weather conditions El Samrani et al. (2008a) demonstrated that the aluminumbased coagulant PACl was more efficient than FeCl₃ for CSO treatment. It had a lower optimum dose and allowed a pronounced reduction of turbidity and associated metals such as copper, lead, zinc and chromium. Gibson et al. (2016) reported similar effect on turbidity removal with alum coupled with a cationic polymer; yet, no major impact on TSS removal or percent ultraviolet transmittance (%UVT) was observed.

As a matter of fact, the addition of a coagulant that comes as a natural or synthetic polymer or a metal salt, initiates the colloidal destabilization and aggregation resulting in the formation of larger particles able to settle out. Different mechanisms can be involved in the particles removals depending on the water quality and coagulant dose (Pourrezaei et al., 2011). Particularly, during coagulation with aluminum-based salts, the salt hydrolysis product is responsible for particles destabilization (Exall & Marsalek, 2013). Two important precipitation reactions take place when alum is added to wastewater. The first one occurs between alum and calcium bicarbonate and produces aluminum hydroxide precipitate that can enmesh and remove colloids (Reaction 3.1). The second one occurs between alum and phosphorus and produces insoluble aluminum phosphate (Reaction 3.2) (Cao et al., 2010).

 $\begin{array}{l} Al_{2}(SO_{4})_{3} \cdot 18 \ H_{2}O + 3 \ Ca(HCO_{3})_{2} \leftrightarrow \ 3 \ CaSO_{4} + 2 \ Al(OH)_{3} + 6 \ CO_{2} + 18 \ H_{2}O & Reaction \ 3.1 \\ \\ Al^{3+} + \ PO_{4}^{\ 3-} \leftrightarrow AlPO_{4} & Reaction \ 3.2 \end{array}$

Following coagulation, larger particles or flocs are formed during flocculation by colloidal aggregation. This process is induced by mixing and polymer addition to strengthen the formed flocs (Young et al., 2000). Settling entails the separation of solids from the liquid and is the last step in the process. In real applications, settling can take place in sedimentation tanks often enhanced by plate settlers that promote large surface area and short settling distance. While the role of plate settlers is to maximize the particles removal, their efficiency remains a function of their geometric design, the wastewater quality and flow conditions (Sarkar et al., 2007).

To the best of our knowledge no study on the treatment of wet weather primary influent (PI) or CSO has comprehensively addressed the upscaling of coagulation conditions and the evaluation of process control indicators. Most studies were limited to bench or pilot-scale levels (El Samrani et al., 2008b; Exall & Marsalek, 2013; Li et al., 2003; Zhu et al., 2007). Evidence of full-scale application of the optimum coagulant dose is also scarce in literature (Baghvand et al., 2010; El Samrani et al., 2008b; Guida et al., 2007; Wang et al., 2015b). Few studies only evaluated the performance of disinfectants at full-scale level using performic acid and peracetic acid (Chhetri et

al., 2016; Chhetri et al., 2015; Chhetri et al., 2014). In this study, we undertook a practical approach towards the EPT of CSO to maximize the output, i.e. the quality of the effluent, while minimizing the chemical input.

The main objective of this chapter was to evaluate the performance of different coagulants and coagulant aids for the enhanced primary treatment of municipal wastewater during wet weather conditions at bench-scale level. The application of those compounds at full-scale level was investigated in order to optimize the EPT at the WWTP, its online process control, and the overall discharge of suspended solids and ortho-phosphate into the receiving environment.

3.2. Material and methods

3.2.1. Wastewater

Samples of municipal wastewater were collected in 20 L plastic pails from the influent stream (i.e., primary influent (PI)) to the Enhanced Primary Treatment (EPT) unit of a WWTP in central Edmonton between November 2016 and March 2017 and between August and October 2017 (Figure 3.1). The collected grab samples of PI during wet weather flow were transported to the laboratory for a preliminary characterization for TSS. During dry weather, PI samples were diluted with deionized water to the desired TSS concentration to emulate the average TSS value during wet weather flow. The obtained wastewater was immediately used in the coagulation/ flocculation experiments.



Figure 3.1 Simplified representation of the EPT system

3.2.2. Coagulants and polymers

All coagulants and chemicals used in the laboratory analyses were purchased from Fisher Scientific, except PACl which was obtained from ClearTech in the form of a stock solution of 4.6 mol/L. Polymers were purchased from SNF Inc., BASF, Kemira and AS Paterson. Stock solutions of coagulants and polymers were prepared in the laboratory by dissolving the respective solutes in deionized water to the desired concentration; they were then stored at 4 °C for up to four days until used in the jar test.

3.2.3. Jar test

During the coagulant and polymer optimization studies, three B-KER² Jar Test apparatuses with 3-inch wide Phipps & Bird stirrers were used for each mixing condition to cater for duplicate test runs. Jars and stirrers were thoroughly cleaned with phosphate-free detergent and rinsed with deionized water before and after each use. At the beginning of each experiment, jars were filled with 2 L of homogenized wastewater, then the designated doses of coagulants and polymers were consecutively added to the jars before turning the impellers on. In all experiments, one jar remained free of chemical addition to represent the effect of mixing only. Rapid and slow mixing were performed sequentially before stopping the impellers and carefully removing the stirrers in each apparatus to avoid any media disturbance. Settling was then allowed for one hour after which the supernatant was collected with syringes immersed at 2 inches below the surface. Samples were stored in dark glass bottles at 4°C for analysis.

A 2³ factorial design was adopted to determine the impact of rapid and slow mixing conditions and coagulant doses on the coagulation/ flocculation process of the wastewater (Table 3.1). These factors were selected due to their influence on the destabilization and aggregation process as well as their easy manipulation at bench-scale level. Keeping in mind that these factors are among the most important ones impacting the coagulation / flocculation efficiency and that various other factors have also been studied such at pH and temperature (Afzal et al., 2011; Chelme-Ayala et al., 2012; Chelme-Ayala et al., 2011a; Pourrezaei et al., 2010).

Table 3.1 Details of the 2^3 factorial design used for the coagulants assessment.

Factor	High Level	Low Level
Rapid Mixing Gt	68400 (300 rpm – 3 min)	9300 (150 rpm – 1 min)
Slow Mixing Gt	27600 (30 rpm – 20 min)	6000 (15 rpm – 10 min)
Coagulant Dose	125 mg/L Alum (10.14 mg Al/L)	50 mg/L Alum (4.05 mg Al/L)
	$61 \text{ mg/L FeCl}_3 (20.95 \text{ mg Fe/L})$	$24 \text{ mg/L FeCl}_3 (8.38 \text{ mg Fe/L})$

In total, there were four replicates for each measurement performed in the factorial design. Three different coagulants were tested: alum, PACl and FeCl₃. Literature was screened to identify the most common applicable doses of alum in the treatment of wet weather PI. In general, doses ranged between 25 and 200 mg/L, with 100 mg/L alum being identified for maximum COD removal and %UVT (Gibson et al., 2016; Jung et al., 2015; Wang et al., 2015b). Accordingly, the alum doses decided upon in this study were 50 and 125 mg/L as they represent a reasonable range of values. Metal-equivalent doses were then calculated to identify the corresponding doses of PACl and FeCl₃ (Table 3.2). They were elaborated based on the molar equivalent for the consumption of the same amount of alkalinity (Reactions 3.1 and 3.3), according to which 2 moles

of $FeCl_3$ are equivalent to 1 mole of alum. 1.25 mg/L cationic polymer (Table 3.7) was also uniformly added in all jars along with the coagulant.

Coagulant	Coagulant Molecular Weight (MW) (a/mol)	Active Species: Molecular Weight (g/mol)	Coagulant dose (mg/L)		Metal- equivalent Dose (mg Al or Ee/L)
A 1			25	0.07	
Aluminum	666.42	Al: 2/g/mol	25	0.07	2.03
Sulfate: Alum			50	0.15	4.05
$(Al_2O_{12}S_3.18H_2O)$			75	0.23	6.08
			100	0.30	8.10
			125	0.37	10.14
			150	0.45	12.15
Ferric Chloride	162.2	Fe: 55.84	24.34	0.15	8.38
(FeCl ₃)		g/mol	60.85	0.37	20.95
	Density	Metal-	Volume of	Mass of PACl required	Coagulant
	(g/mL)	equivalent	PACI	in 2 L Jar (mg)	dose
Coagulant		Dose	required in		(mg/L of
		(mg Al/L)	2 L Jar		PAC1)
			(mL)		, , , , , , , , , , , , , , , , , , ,
Polyhydroxyl	1.37	4.05	0.061	83.57	41.78
Aluminum		10.14	0.154	210.98	105.49
Chloride **					
(PACl)					

Table 3.2 Calculations of metal-equivalent coagulant doses.

Notes:

* n= mol of Active species in 1 mol of coagulant.

** PACl (Clear PAC 180) with a Specific Gravity of 1.37 (according to the manufacturer) was obtained in a solution form and was characterized in the lab for Al content. Results indicated the Al content of PACl as 131589.16 mg Al/L and calculations were made accordingly.

Reference equations for Alum and FeCl₃ in wastewater:

 $\begin{array}{l} \mathrm{Al}_2(\mathrm{SO}_4)_3 \cdot 18\mathrm{H}_2\mathrm{O} + 3\mathrm{Ca}(\mathrm{HCO}_3)_2 \Leftrightarrow 3\mathrm{Ca}\mathrm{SO}_4 + 2\mathrm{Al}(\mathrm{OH})_3 + 6\mathrm{CO}_2 + 18\mathrm{H}_2\mathrm{O} \\ \mathrm{2FeCl}_3 + 6\mathrm{H}_2\mathrm{O} + 3\mathrm{Ca}(\mathrm{HCO}_3)_2 \Leftrightarrow 3\mathrm{Ca}\mathrm{Cl}_2 + 2\mathrm{Fe}(\mathrm{OH})_3 + 6\mathrm{CO}_2 + 12\mathrm{H}_2\mathrm{O} \end{array}$

Two levels of rapid mixing with Gt values 68400 and 9300 and two levels of slow mixing

with Gt values 27600 and 6000 were applied. Where "G" is the velocity gradient and "t" is the

mixing time. Conversions from rpm-min to Gt are presented in Table 3.3.

$$2FeCl_3 + 6H_2O + 3Ca(HCO_3)_2 \leftrightarrow 3CaCl_2 + 2Fe(OH)_3 + 6CO_2 + 12H_2O$$
 Reaction 3.3

Faator	Level	S	Speed	Tin	Ct	
ractor		rpm	G (s ⁻¹)	T (min)	t (s)	G
Rapid Mixing	High	300	380	3	180	68400
	Low	150	155	1	60	9300
Slow Mixing	High	30	23	20	1200	27600
	Low	15	10	10	600	6000

Table 3.3 Conversion of mixing conditions from rpm, min to Gt (Based on B-KER² – Phipps & Bird Specifications).

Eight anionic polymers (labeled A to H) and one cationic polymer were arbitrarily selected and their performances as coagulant aids were evaluated at a bench scale using a jar test with alum as the primary coagulant. Similar mixing conditions were applied with all polymers uniformly injected with alum. After one hour of settling time, samples were collected from the supernatant. They were screened for TSS, turbidity and ortho-P. The polymer that revealed highest removals was selected and investigated under a range of dose combinations to identify the optimum condition for full-scale operation. All polymers were prepared from stock solutions and applied with a ratio of 100:1 as coagulant: polymer.

3.2.4. Full-scale application of alum/polymer

Different dose combinations of alum and polymer were tested at the existing EPT plant at the WWTP. The coagulant was dosed with mixing in the channel connected to the primary clarifier and the flocculent was dosed downstream with mixing in the channel connected to the flocculation chambers. Settling took place in the primary clarifier equipped with plate settlers for a retention time of approximately one to two hours.

3.2.5. Analytical methods

Samples were analyzed according to the Standard Methods for the Examination of Water and Wastewater (2012) for TSS (APHA (2017): Method 2540 D, using 0.45 µm filter paper), and COD (Method 5220 D). HACH method was used to measure ortho-P in filtered samples (HACH 8114 based on Standard Method 4500-P C). pH and turbidity were determined using Accumet Research AR20 pH/conductivity meter (Fischer-Scientific) and T-100 handheld Oakton Turbidity meter, respectively. UV absorbance (UVA) was measured at 254 nm using Thermo Scientific GENESYS 10S UV-Vis Spectrophotometer, and the %UV Transmittance (%UVT) was calculated using Equation 3.1. Data was analyzed using Microsoft Excel[®] which was also used to run the Three Factor ANOVA statistical test.

 $UVT = 100 \times 10^{-UVA}$

Equation 3.1

3.3. Results and discussion

3.3.1. Wastewater quality

As part of this study, samples of PI during real wet weather conditions were collected at different rain occasions and analyzed for TSS. The average obtained TSS value was 150 mg/L. Accordingly; during dry weather, PI samples were diluted with deionized water to emulate real wastewater during wet weather flow. Also, during wet weather conditions and a flow rate of 1000 MLD, the recorded turbidity and ortho-phosphate values were 128 NTUs and 6.6 mg PO_4^{3-}/L respectively, while at lower flow rate of 276 MLD, those values increased to 212 NTUs and 12.7 mg PO_4^{3-}/L . The sampling period extended over one year with a total of 12 samples. Details on all parameters are included in Table 3.4.

		рН	Turbidity (NTU)	TSS (mg/L)	%UVT	COD (mg O ₂ /L)	Ortho-P (mg PO ₄ ³⁻ /L)
Bench-scale	Range	7.17-7.88	43-110	98-167	<i>N.M</i> .	160-361	2.10-6.35
	Average (±STDEV)	7.38 (±0.22)	88 (±22)	137 (±29)	<i>N.M</i> .	293 (±71)	4.48 (±1.88)
Full-scale	Range	6.71-7.29	87-192	138-364	33-57	<i>N.M</i> .	0.88-11.10
	Average (±STDEV)	7.13 (±0.16)	150 (±31)	224 (±59)	44 (±6.85)	N.M.	6.54 (±2.44)

Table 3.4 Water quality parameters of the untreated diluted primary influent (PI) used in the bench-scale experiment and the influent wastewater to the WWTP in the full-scale runs.

*N.M. = Not measured

3.3.2. Performance assessment of coagulants

In general, the capacity of each coagulant to eliminate pollutants varied with the dose and mixing conditions. Neither mixing conditions nor coagulant type and dose affected the final pH of the wastewater which dropped slightly upon coagulant addition. At lower coagulant doses, the pH ranged between 6.90-7.21 for alum, 7.16-7.34 for PACl and 7.11-7.29 for FeCl₃. At higher coagulant doses, the pH of treated samples ranged between 6.35-6.99 for alum, 6.85-7.21 for PACl and 6.74-7.10 for FeCl₃.

According to Figure 3.2, turbidity removal ranged widely between 102 and 600 NTUs/M of equivalent metal. The highest removal was obtained with low dose of alum and low level rapid and slow mixing. Alum and PACl revealed close performances with removals often more than double those with FeCl₃. It was also noticeable that at lower coagulant doses, higher removals were achieved throughout the four mixing conditions.

Results of TSS removal indicated that alum outperformed other coagulants. At low alum doses, the removal of TSS was the highest with high level rapid mixing (153 mg TSS removed by M equivalent of of Al added as alum). In contrast, at high alum doses, low level of rapid and slow mixing were optimum. This is in-line with the sweep coagulation mechanism that dominates at

high coagulant doses and the efficiency of which is maximized by lower mixing intensity (Wei et al., 2015). Mixing showed different impact on TSS removal with PACl mainly because the flocs formed with PACl are more sensitive to breakage, and their settling velocity changes inversely with Gt (Dempsey et al., 1985; Yu et al., 2009). This is largely due to the fact that PACl acts on particles removal by sweep flocculation, which entails their enmeshment during precipitation (Wang et al., 2015a). So, at low dose where charge neutralization takes place, intensive rapid mixing provoked their irreversible shear, and at high dose, where sweep flocculation is in effect, intensive slow mixing lead to the same result. It is worth noting that in all cases, the final recorded TSS value was less than the target value set by the WWTP (30 mg/L).





Figure 3.2 Mass of contaminant removed by 1 M equivalent of active metal at high and low coagulant doses. The effect of mixing and settling on the removal was excluded in the calculation. In untreated wastewater, the parameters ranged as follows: Turbidity 47.2-106.0 NTU; TSS 99-167 mg/L; COD 160-361 mg O_2/L ; and Ortho-P 2.10-6.35 mg P/L.

With COD removal, the tested coagulant doses and mixing conditions turned out to have different effects. The condition of low-level rapid mixing with high level slow mixing was consistently the least favourable for all three coagulants, regardless of the dose. In most cases, PACl revealed better COD removal reaching up to almost 590 mg COD removed by M equivalent of active Al added as PACI. In fact, PACI is expected to exceed alum as it is a pre-polymerized coagulant that brings pre-formed stable aluminum hydroxide polymeric species to the matrix upon its addition. These species are also formed when alum is used and they are the product of alum hydrolysis (Jiang & Graham, 2010). Yet, with PACI application, they are readily available for longer time which justifies their higher potential for charge neutralization. Studies have demonstrated that for these reasons, PACl has high capacity to adsorb and remove dissolved organic carbon which would also imply removal of natural organic matter and lower COD (Jiang & Graham, 2010; Smoczynski et al., 2014). Alternatively, studies suggested increasing the dose of alum to improve COD removal (Zhou et al., 2008). Both Al-based coagulants were generally more efficient than FeCl₃ with which removal ratios ranged between 91 and 509 mg COD removed by M equivalent of Fe added as FeCl₃ at low dose.

The removal of ortho-P was best achieved with alum under all mixing conditions and doses, with the highest removal being 21 mg ortho-P removed by M equivalent of Al added. Similar to the results above, low coagulant doses lead to higher removal ratios, and FeCl₃ remained the least efficient among the three coagulants. The mixing intensity did not show significant impact on the removal of ortho-P which was also reported elsewhere (Ebeling & Ogden, 2004). Such observation can be associated with the formation of phosphate complexes that are influenced by pH and coagulant dose. When Al-based coagulants are applied, aluminum monomer species that form by the coagulant hydrolysis have the dominant effect on the formation of phosphate complexes (Trinh & Kang, 2015). It has been reported that iron-based coagulants are more effective for the removal of phosphate, while aluminum-based coagulants are more effective for the removal of COD (Smoczynski et al., 2014). This is not supported by our results that showed that FeCl₃ was less effective for the removal of both ortho-P and COD when applied in iron-equivalent doses comparable to those of the aluminum-based coagulants. This disagreement regarding the efficiency of both metal-based coagulants can be due to the different experimental conditions applied in both studies. As such, Smoczynski et al. (2014) used synthetic wastewater along with higher doses of aluminum and iron-based coagulant doses: 235 mg Al/L and 452 mg Fe/L, while we applied lower doses of both coagulants (10.14 mg Al/L and 20.95 mg Fe/L, equivalent to 0.37 mM of metal) on real wastewater.

The two components of mixing addressed in this study: intensity and time are generally capable of mitigating the effect each other. On one hand, studies have demonstrated that intense mixing provokes the breakage of flocs which results in small hard to settle particles and thus high turbidity (Yukselen & Gregory, 2004). On the other hand, long mixing time could reverse flocs breakage, yet this is not always guaranteed and tends to be less significant in metal-based

coagulants (Yu et al., 2011; Yukselen & Gregory, 2004). For all those reasons added to the extra power requirements for mixing, high level rapid mixing conditions are generally not favoured.

Since, during the analysis of variance, the effects with values near-zero are generally attributed to experimental error or noise (Anderson, 2015), factors in the current ANOVA showing such effects were considered negligible regardless of their p-values. Overlooking all three coagulants, it could be noticed that while turbidity removal was unilaterally impacted by coagulant dose, there was no agreement on the factors, impacting the removal of TSS, COD and ortho-P across the studied coagulants (Table 3.5).

Table 3.5 Results of the Three-Factor ANOVA test showing the interactions between contaminants percent removals and the three factors: rapid mixing, slow mixing, and coagulant dose at 95% confidence level.

	Alum								
	Tur	bidity	TSS		С	COD		Ortho-P	
Factors	% Removal		al % Removal		% Removal		% Removal		
	p-	Effect	p-	Effect	p-	Effect	p-	Effect	
	value	(±1.38)	value	(±1.38)	value	(±3.63)	value	(±1.11)	
Rapid Mixing A	0.000	-5.71	0.000	-7.32	0.103	-6.15	0.051	-2.28	
Slow Mixing B	0.151	1.24	0.207	1.78	0.000	-19.30	0.878	-0.17	
Coagulant Dose C	0.000	5.53	0.017	3.52	0.278	-4.03	0.000	18.35	
AB	0.009	-2.37	0.000	-6.78	0.001	-13.79	0.159	-1.61	
AC	0.004	-2.63	0.019	-3.47	0.033	-8.20	0.829	0.24	
BC	0.026	-1.99	0.004	-4.35	0.000	-21.09	0.000	-17.02	
ABC	0.618	0.42	0.134	2.13	0.115	-5.94	0.051	2.28	
				P	ACI				
	Tur	bidity	TS	S %	CO	D %	Ortho-P %		
	% R	emoval	Rer	noval	Ren	noval	Rer	noval	
	p-	Effect	p-	Effect	p-	Effect	p-	Effect	
	value	(±1.38)	value	(±1.40)	value	(±3.20)	value	(±2.05)	
Rapid Mixing A	0.015	-1.24	0.001	-5.40	0.046	-6.75	0.924	-0.20	
Slow Mixing B	0.000	3.84	0.772	0.41	0.000	-16.60	0.007	-6.09	
Coagulant Dose C	0.000	11.32	0.002	4.89	0.003	10.61	0.000	22.29	
AB	0.005	-1.48	0.000	-10.35	0.216	-4.07	0.282	2.26	
AC	0.000	2.94	0.000	8.98	0.360	-2.98	0.027	4.83	
BC	0.000	-5.65	0.000	-5.95	0.004	-10.27	0.000	-27.74	
ABC	0.000	2.89	0.501	-0.96	0.080	5.85	0.005	6.39	
				Fe	eCl ₃				
	Tur	bidity	TS	S %	CO	COD % Or		rtho-P %	
	% R	emoval	Rer	noval	Ren	noval	Removal		
	p-	Effect	p-	Effect	p-	Effect	p-	Effect	
	value	(±1.38)	value	(±1.35)	value	(±3.55)	value	(±1.34)	
Rapid Mixing A	0.006	-2.14	0.265	-1.54	0.057	-7.10	0.000	12.11	
Slow Mixing B	0.001	2.61	0.000	-5.93	0.000	-18.81	0.000	-23.54	
Coagulant Dose C	0.000	9.30	0.484	0.96	0.987	0.06	0.000	16.27	
AB	0.479	-0.51	0.366	-1.24	0.000	-20.98	0.000	14.40	
AC	0.871	-0.12	0.000	9.48	0.852	0.67	0.000	14.13	
BC	0.741	0.24	0.000	-12.76	0.004	-11.48	0.000	-41.41	
ABC	0.358	-0.67	0.294	1.45	0.757	1.11	0.000	15.50	

As such, factors affecting TSS removal by alum and the extent of this effect differed from those with PACl and FeCl₃. In fact, all three studied factors (coagulant dose, mixing time and mixing speed) showed a minor effect on the removal of turbidity and TSS. Similar results were obtained by Zhu et al. (2010) during the treatment of liquid swine manure. COD removal by alum was mostly affected by slow mixing alone and slow mixing and coagulant dose combined, while with PACl and FeCl₃ slow mixing was the most impactful. Factors affecting the removal of ortho-

P by alum and PACl were the same (coagulant dose only and slow mixing and coagulant dose). It must be highlighted that slow mixing and/or coagulant dose combined unveiled the highest effect for the removal of ortho-P and COD across all three coagulants. So, in order to maximize the contaminants removal, these factors must be optimized.

The following ANOVA models (Table 3.6) were developed to predict the percent removal of the different contaminants with each factor at two levels.

Turbidity % Removal by:						
Alum	88.57 – 5.71 A + 5.53 C – 2.37 AB – 2.63 AC – 1.99 BC					
PACI	89.49 - 1.24 A + 6.84 B + 11.32 C - 1.48 AB + 2.94 AC - 5.65 BC + 2.89 ABC					
FeCl ₃	81.46 – 2.14A + 2.61 B + 9.30 C					
	TSS % Removal by:					
Alum	90.29 – 7.32 A + 3.52 C – 6.78 AB – 3.47 AC + 4.35 BC					
PACI	86.36 - 5.40 A + 4.89 C - 10.35 AB + 8.98 AC - 5.95 BC					
FeCl ₃	85.66 – 5.93 B + 9.48 AC – 12.76 BC					
COD % Removal by:						
Alum	58.33 – 19.30 B – 13.79 AB – 8.20 AC – 21.09 BC					
PACI	63.47 - 6.75 A - 16.60 B + 10.61 C - 10.27 BC					
FeCl ₃	57.52 – 18.81 B – 20.98 AB – 11.48 BC					
Ortho-P % Removal by:						
Alum	80.88 + 18.35 C – 17.02 BC					
PACI	68.36 – 6.09 B + 22.29 CB + 4.83 AC – 27.74 BC + 6.39 ABC					
FeCl ₃	57.17 + 12.11 A - 23.54 B + 16.27 C + 14.40 AB + 14.13 AC - 41.41 BC + 15.50 ABC					

Table 3.6 ANOVA models.

3.3.3. Optimization of the alum dose

The above analysis deemed alum and PACl comparable for the removal of contaminants, yet alum is more cost efficient than PACl (with approximately 300 USD /MT of alum compared to 600 - 715 USD /MT of PACl), so it can be considered the best coagulant for full-scale application. In order to further optimize its dosage, a jar test was performed with a broader range of alum doses (0 to 150 mg/L) that were evaluated for the removal of TSS (Figure 3.3). Results

demonstrated that with an alum dose of 75 mg/L, the final average TSS was 33 mg/L. Further increasing the alum dose decreased the TSS to below the target value of 30 mg/L to reach near a plateau. Therefore, we recommended 75 mg/L alum as the typical dose because higher doses would incur higher costs with minimal returns.



Figure 3.3 Effect of alum dosage on the removal of TSS from wastewater during wet weather conditions.

3.3.4. Effect of polymer type on the performance of alum in the treatment of wastewater during wet weather flow

As alum was deemed the best coagulant for wastewater treatment, an attempt was made to further improve its performance under the effect of different types of polymers. The WWTP is currently implementing anionic polymer A along with alum. Therefore, 7 other types of anionic polymers and 1 cationic polymer were put to test to assess the role of polymer addition. The chemical and physical properties of all polymers are available in Table 3.7.

Polymer	Chemical Composition	Concentration (%)	Туре	Density (g/cm ³)	рН
	Distllates (petroleum, hyrdotreated light	20 to 23			
	Alcohol, c12-14, ethoxylated	0 to 3			6 to 8
Α	Alcohol, c10-16, ethoxylated	0 to 3	Anionic	1	
	Alcohol, c12-16, ethoxylated	0 to 3			
	Alcohol, c13-15, ethoxylated, branched and linear	0 to 3			
R	Distillates (petroleum), hydrotreated naphthenic acid	15 to 40	Anionic	11	3 9-4 4
В	Alcohol, c12-15, ethoxylated, propoxylayted	1 to 5	Allollic	1.1	5.7-4.4
	Distillates petroleum, hyrdotreated light	22 to 25			
	Alcohol, c10-16, ethoxylated	0 to 3.6			6 to 9.5
C	Alcohol, c12-14, ethoxylated	0 to 3.6	Anionic	1.1	
C	Alcohol, c12-16, ethoxylated	0 to 3.6	Amonic		
	(Z)-octadec-9-enylyamine, ethoxylayted	1.2 to 1.6			
	Ammonium acetate	2 to 10			
	Distillates petroleum, hyrdotreated light	22 to 25	Anionic	1.1	5 to 8 at 5 g/L
	Alcohol, c12-16, ethoxylated	0 to 3.6			
р	Alcohol, c12-14, ethoxylated	0 to 3.6			
D	Alcohol, c10-16, ethoxylated	0 to 3.6	7 unonic		
	(Z)-octadec-9-enylyamine, ethoxylayted	1.2 to 1.6			
	Ammonium acetate	2 to 10			
F	Distillates (petroleum), hydrotreated naphthenic acid	20 to 45	Anionic	11	5 to 8
12	Poly(oxy-1,2-ethanediyl), a-tridecyl-w-hydroxy-,branched	<3	7 unonic	1.1	at 5 g/L
F	Poly(oxy-1,2-ethanediyl), a-tridecyl-w-hydroxy-,branched	<3	Anionic	11	5 to 8
ľ	Distillates (petroleum), hydrotreated naphthenic acid	20 to 45	7 unionic	1.1	at 5 g/L
	Distillates (petroleum) hydrotreated light	<30%			
G	Alcohol, C12-16, Ethoxylated	<5%	Anionic	1	6 to 8
	Alcohol, C12-14, Ethoxylated	<5%	7 unonic	1	0100
	Alcohol, C10-16, Ethoxylated	<5%			
н	Distillates (Petroleum) Hydrotreated Light	20 to 28 %			
	Alcohol, C10-16, Ethoxylated	<5%	Anionic	1	6 to 8
	Alcohol, C12-16, Ethoxylated	<5%			
Cationic	COCO DIETHANOLAMIDE	>=0.5 < 1 %	Cationic	0.62	4 at 10 g/L

Table 3.7 Chemical and physical properties of the polymers used in this study.

The dose of coagulant: polymer was maintained constant at 100:1 and the mixing conditions were fixed at 300 rpm - 1 min for rapid mixing and 30 rpm - 20 minutes for slow mixing. Following the previous set of experiments, it was decided to investigate the performance of 75 mg/L alum applied with 0.75 mg/L polymer in terms of turbidity, TSS, % UVT and ortho-P removal. The obtained results (summarized in Figure 3.4) showed that while all tested polymers exhibited comparable performances, Alum with the anionic polymers A and B showed the best removal for all three tested parameters. All treated samples met the TSS and %UVT design values for the EPT effluent (30 mg/L and 40% respectively).



Figure 3.4 Effect of polymer addition on turbidity, TSS, ortho-P and %UVT (applied alum dose is 75 mg/L and polymer dose is 0.75 mg/L).

The addition of anionic polymer was further evaluated and the experiments were limited to polymer A which the WWTP was using during its operations at the time of this study. Thus, alum was applied with and without polymers A, (in doses ranging between 25/0.25 and 150/1.5 mg/L as alum/polymer (mg/L)) during a jar test and samples were analyzed for turbidity, TSS and ortho-P removal (Figure 3.5). With mixing only and no chemical addition, turbidity and TSS were significantly reduced compared to ortho-P. Adding alum only without polymers led to better removals compared to adding alum with polymer which is in line with findings from another study (Aguilar et al., 2005). An average alum dose of 75 mg/L achieved the target EPT effluent TSS value set by the WWTP (30 mg/L). Increasing the doses of alum and polymer did not efficiently improve the removals as samples treated with 125:1.25 alum/polymer could barely reach a final TSS value close to 30 mg/L. This may be referred to a mechanism of re-stabilization of particles in suspension due to the tremendous increase in the positively charged aluminum species (Saritha et al., 2017).



Figure 3.5 Impact of alum dose applied with anionic polymer A in a fixed dose ratio of 100:1 (as alum: polymer in mg/L) on the removal of turbidity, TSS and Ortho-P.

The removal of ortho-P is usually influenced by the wastewater quality including: initial ortho-P concentration, TSS, and pH, in addition to the type and dose of chemical added, and mixing conditions (Hussain et al., 2011). The pH of all tested samples was between 7.13 and 7.22 without adjustment, which is the optimum pH for ortho-P removal by alum and iron-based coagulants. During coagulation process, different aluminium and phosphate complexes are formed under specific pH values. It is reported that phosphorus adsorbs onto the hydrolysis products of
the metal-based coagulants, this chemisorption results in precipitates that separate easily from the liquid. Changes in the pH can affect the favorable conditions for the formation of phosphatealuminum complexes and the adsorption and desorption mechanisms (Smoczynski et al., 2014). The solubility diagram of aluminum hydroxide shows that pH dictates the form of hydrolysis products which are positively charged at low pH and negatively charged at high pH. It was found that the favourable pH for ortho-P removal by precipitation of aluminum phosphate (AlPO₄) is between 6 and 8 (David Jenkins & Menar, 1970). In this study, the removal of ortho-P was noticed to improve when higher chemical doses were applied; however less improvement was observed when alum doses exceeded 125 mg/L (Figure 3.5). Moreover, the application of anionic polymer might be hindering the removal of ortho-P as alum alone was able to achieve better removal.

3.3.5. Effect of alum and polymer addition during full-scale application

A detailed description of water quality parameters in the influent and effluent of the WWTP is available in Table 3.8.

		Flow rate	нрт	Turbidity	TSS	Ortho-	
Sample ID	Date	(MLD)	(hr)	(NTI)	$(m\sigma/L)$	Р	%UVT
		(MLD)	(111)	(((10)	(1115/12)	(mg/L)	
PI*	18-Aug	131.5	2.92	126	208	11.25	24
PI	19-Aug	239	1.61	86.45	200	6.65	33
Alum (100 mg/L)	19-Aug			26	26	1	44
PI	24-Aug	250.7	1.53	164	240	9.9	30
EPT Effluent**	24-Aug			73.55	54	9	36
PI	13-Sep	259.7	1.48	165.5	188	7.75	36
EPT Effluent	13-Sep			72.1	32	8.88	48
PI	13-Sep	320	1.2	120.5	244	7.25	42
Alum + Polymer (A) $(75/1.45)$ mg/L	13-Sep			30.75	38	2	45
PI	13-Sep	263	1.46	113.5	138	7.125	46
Alum + Polymer (A) $(75/1.45)$ mg/L	13-Sep			23.35	26	0.88	51
PI	14-Sep	214	1.79	190	364	8.88	40
EPT Effluent	14-Sep			67.1	40	4.75	51
PI	14-Sep	269	1.43	149	200	6.75	44
Alum 75	14-Sep			42.9	44	1.38	52
PI	19-Sep	192	1	1.64.7	2 10		-
PI	19-Sep			164.5	218	5.06	50
Alum + Polymer (B) $(75/1.1)$ mg/L	19-Sep			21.2	28	0.88	53
PI	19-Sep	272	0.71	154	180	4	56
Alum + Polymer (B) $(75/1.1)$ mg/L	19-Sep			68.55	64	0.5	68
PI	19-Sep	232	0.83	177.5	176	6.12	46
EPT Effluent	19-Sep	-		82.75	60	7.88	48
PI	19-Sep	306	0.63	215	360	2.63	62
Alum + Polymer (B) $(75/1.1)$ mg/L	19-Sep			62.8	98	0.5	65
PI	20-Sep	144.7	1.33	190	248	9.38	38
EPT Effluent	20-Sep	1	1.00	197.5	150	8.5	41
PI	20-Sep	179.3	1.07	191.5	280	5.75	41
Alum + Polymer (B) $(75/0.75)$ mg/L	20-Sep	1,5.0	1107	57.3	56	2.5	45
PI	20-Sep	162.5	1 18	170.5	280	7 75	42
Alum + Polymer (B) $(75/0.75)$ mg/L	20-Sep	102.0	1110	35.15	40	1 38	45
PI	20 Sep 21-Sep	247 7	0.78	124 5	200	0.88	57
Alum + Polymer (B) $(75/0.75)$ mg/L	21-Sep	2.7.7	0.70	14 085	14	0.00	62
PI	01-Oct	226	0.85	221	296	9.1	33
EPT Effluent	01-Oct	220	5.05	124	76	99	41
PI	02-Oct	167	1 1 5	147	244	11.1	43
Alum + Polymer (A) $(75/0.75)$ mg/I	02 Oct	107	1.1.2	39 55	48	39	47
EPT Effluent PI EPT Effluent PI Alum + Polymer (A) $(75/1.45) \text{ mg/L}$ PI Alum + Polymer (A) $(75/1.45) \text{ mg/L}$ PI EPT Effluent PI Alum + Polymer (B) $(75/1.1) \text{ mg/L}$ PI Alum + Polymer (B) $(75/1.1) \text{ mg/L}$ PI EPT Effluent PI Alum + Polymer (B) $(75/1.1) \text{ mg/L}$ PI EPT Effluent PI Alum + Polymer (B) $(75/0.75) \text{ mg/L}$ PI Alum + Polymer (B) $(75/0.75) \text{ mg/L}$ PI Alum + Polymer (B) $(75/0.75) \text{ mg/L}$ PI EPT Effluent PI Alum + Polymer (B) $(75/0.75) \text{ mg/L}$ PI EPT Effluent PI Alum + Polymer (A) $(75/0.75) \text{ mg/L}$	24-Aug 13-Sep 13-Sep 13-Sep 13-Sep 13-Sep 13-Sep 14-Sep 14-Sep 14-Sep 14-Sep 19-Sep 19-Sep 19-Sep 19-Sep 19-Sep 19-Sep 19-Sep 20	259.7 320 263 214 269 192 272 232 306 144.7 179.3 162.5 247.7 226 167	1.48 1.2 1.46 1.79 1.43 1 0.71 0.83 0.63 1.33 1.07 1.18 0.78 0.85 1.15	$\begin{array}{c} 73.55\\ 165.5\\ 72.1\\ 120.5\\ 30.75\\ 113.5\\ 23.35\\ 190\\ 67.1\\ 149\\ 42.9\\ 164.5\\ 21.2\\ 154\\ 68.55\\ 177.5\\ 82.75\\ 215\\ 62.8\\ 190\\ 197.5\\ 191.5\\ 57.3\\ 170.5\\ 35.15\\ 124.5\\ 14.085\\ 221\\ 124\\ 147\\ 39.55\\ \end{array}$	54 188 32 244 38 138 26 364 40 200 44 218 28 180 64 176 60 360 98 248 150 280 56 280 56 280 40 200 14 296 76 244 48	9 7.75 8.88 7.25 2 7.125 0.88 8.88 4.75 6.75 1.38 5.06 0.88 4 0.5 6.12 7.88 2.63 0.5 9.38 8.5 5.75 2.5 7.75 1.38 0.88 0.25 9.1 9.9 11.1 3.9	$\begin{array}{c} 36\\ 36\\ 48\\ 42\\ 45\\ 46\\ 51\\ 40\\ 51\\ 44\\ 52\\ 50\\ 53\\ 56\\ 68\\ 46\\ 48\\ 62\\ 65\\ 38\\ 41\\ 41\\ 45\\ 42\\ 45\\ 57\\ 62\\ 33\\ 41\\ 43\\ 47\\ \end{array}$

Table 3.8 Water quality parameters of untreated and treated samples during full-scale application of Alum and Alum + Polymer A at the WWTP.

Notes:

* PI = Primary Influent to the WWTP.

** EPT Effluent = effluent from the primary clarifier without any chemical addition (i.e., after mixing and settling only).

The initial TSS values for all events from the full-scale application ranged between 168 to 296 mg/L. As shown in Figure 3.6, the results of coagulant addition were in line with the findings from the jar test confirming that 100 mg/L of alum was capable of achieving the target TSS design value (30 mg/L). In fact, this is expected since the primary clarifier is provided with lamella plate

settlers to maximize the removal of suspended solids. Moreover, better removal was achieved with the lower dose of alum: polymer A 75/0.75 mg/L when the initial TSS was less than 200 mg/L. Yet, when initial TSS exceeded 200 mg/L, the same alum dose of 75 mg/L was unable to achieve the required removal even when increasing the polymer dose to 1.1 mg/L. It is therefore preferable to adjust the alum dose when the TSS values in the incoming PI are higher than 200 mg/L. Pollutants removals induced by the application of chemicals were evaluated against removals under the effect of lamella plate settlers (indicated by "No chemicals" in Figure 3.6). Results showed that the chemical addition is crucial for ortho-P reduction whereas the effect of lamella settler alone was almost null because ortho-P removal is mainly associated with precipitation (Gasperi et al., 2010; Mohammed & Shanshool, 2009). However, there was a noticeable reduction in TSS and turbidity where the main mechanism of removal is settling. In some cases, the effects of lamella settler on TSS reduction were comparable with the values obtained when chemicals were added. These findings are comparable to those reported by the effect of mixing only in the jar tests.



Figure 3.6 Removal of turbidity, TSS and ortho-P during the full-scale application of alum and alum + polymer A.

There are multiple physical and chemical factors affecting the process of coagulation/ flocculation such as the dose and type of chemical and its application point, reactor type, background effects and mixing intensities and times. In fact, it is possible to compare bench -scale with full-scale experiments by using the mean velocity gradient G or Gt values assuming the mixing environments will be the same under constant G or Gt (Cornwell & Bishop, 1983). Yet, during the jar tests, both chemicals were simultaneously added at the beginning of rapid mixing with known mixing intensities while during full-scale operation, the coagulant is mixed with the inflow to the coagulation tank and the polymer is mixed later with the outflow into the flocculation chamber. In addition, on either scale the temperature was not controlled, it varied between 5 and 20°C at full scale while room temperature was maintained during jar tests. Indeed, the comparison between bench and full-scale was strictly based upon the chemical addition. So, although the jar test could not mimic the exact operation conditions, our results acquired from bench-scale experiments were comparable to the results of the full-scale operation in terms of contaminants removals on mass basis (Figure 3.7). Bench and full-scale data (Figures 3.5 and 3.6) showed that polymer addition had no major contribution to the removal of turbidity, TSS and Ortho-P. Hence the addition of polymer was not a sustainable option neither economically nor environmentally.



Figure 3.7 Comparison between bench-scale and full-scale applications of 75 mg/L alum only and 75 mg/L alum + 0.75 mg/L polymer for the removal of turbidity, TSS and ortho-P.

3.3.6. Effective parameters for online process control

Investigating the correlation between different wastewater parameters is technically and economically beneficial. According to Ratnaweera and Fettig (2015), pH, turbidity, TSS, UVA and ortho-P are relevant parameters to evaluate the coagulation/ flocculation/ sedimentation process, and they are typically monitored by online sensors. Ortho-P is occasionally measured to monitor the influent and effluent quality. For these reasons, correlations between the different

parameters were studied in order to investigate downsizing the analysis to the most effective parameter for online process control.

Results obtained from the full-scale analysis revealed no strong association between the influent flow rate and any of the measured parameters. So, flow rate cannot be a reliable indicator for process control. Also, besides the redundant correlation between TSS and turbidity, there was a strong correlation between %UVT and ortho-P (Figure 3.8). This signifies that measuring turbidity and UVA during operation can help predict rough estimates of TSS and ortho-P and adjust the required coagulant dose without the hassle of multiple sample analyses. This is known as feed-forward control scheme where the quality of the influent is monitored to adjust the chemical dosing. (Ratnaweera & Fettig, 2015). Mass removal ratios can be used here to approximate the required coagulant dose.



Figure 3.8 Correlations between contaminants measured at full-sale level.

At bench-scale level, our results showed a good correlation between turbidity and TSS, TSS and COD, and COD and ortho-P (Fig. 9). The strongest linear correlations were between turbidity and TSS at both full-scale and bench-scale levels, where the R^2 values were 0.80 and 0.85 respectively (Figure 3.9). These results comply with those obtained by Grayson et al. (1996) who studied the correlation between turbidity and TSS and turbidity and total phosphorus in samples collected from a river catchment area and identified strong linear correlations with R^2 values of

0.86 and 0.90, respectively. It is noteworthy that the strength of such correlations is site specific and changes according to the tested wastewater matrix (Hannouche et al., 2011). Nevertheless, more data is essential to reduce the uncertainty and validate those correlations.



Figure 3.9 Correlations between contaminants measured at bench-scale level.

3.3.7. Implications of these removals on the monitoring of micropollutants

As discussed previously, CSOs are a considerable source of micropollutants. So, in order to assess the overall efficiency of the EPT and the need for any additional treatment, it is important to monitor not only conventional pollutants, but also micropollutants of concern. Thus, in order

to estimate the elimination of MPs during EPT, the correlation between differential UVA₂₅₄ (ΔUVA_{254}) and removal of MPs was surveyed from different studies. In the present study we project that UVA can serve not only for process control, but also as a surrogate parameter to estimate the removal of micropollutants during the EPT of CSO. Micropollutants removal correlates with ΔUVA_{254} in advanced oxidation and adsorption processes where an increase in ΔUVA_{254} is associated with higher removal of trace organics reaching up to 95% depending on the chemical structure and properties of the compound (Altmann et al., 2016; Altmann et al., 2015; Ning et al., 2007; Sui et al., 2010; Zhang et al., 2015; Ziska et al., 2016). In coagulation/ flocculation/ sedimentation processes, UVA254 was found to be an accurate surrogate parameter for trihalomethane formation in alum-treated waters (Pifer & Fairey, 2014). In the current study, ΔUVA_{254} ranged between 15% and 40% with the use of 75 mg/L alum at bench and full-scale levels. This range can be associated with up to complete removal of some trace organics by ozonation (Chys et al., 2017). In this WWTP final effluent, Shu et al. (2013) and Kerr et al. (2008) reported on the occurrence of some micropollutants such as carbamazepine, diclofenac, ibuprofen, gemfibrozil, mecoprop, diazinon and others at varying concentrations. In contrast, CSO contributes up to 10% of the total discharged load of trace organic compounds with low removal efficiency in a WWTP and up to 90% of those with high removal efficiency (Phillips et al., 2012). As such, we anticipate that the EPT of CSO can still achieve some removal of those compounds that have low water solubility, which entails hydrophobic interactions and sorption onto particulate matter (Margot et al., 2015). Other hydrophilic compounds can also aggregate with metal ions during the treatment with alum. Alum treatment can as such remove up to 75% of some micropollutants (Kim & Zoh, 2016) which can translate in higher ΔUVA_{254} . Moreover, the removal of selected MPs by different oxidants will be investigated extensively in other phases of this study.

3.3.8. Recommended future interventions

The local municipality projects that by 2020, the WWTP would contribute around 10% of the total TSS loading to the adjacent river among which 10 % (2700 Kg/d) would be attributed to the EPT effluent (at a concentration of 30 mg/L) compared to 72% to the final effluent (at a concentration of 7 mg/L) and 18% to the combined bypasses. Thus, the EPT was put under the Combined Sewer Overflow Control Strategy set by the city to control the discharges into the river (City-of-Edmonton, 2009). Despite prioritizing TSS, there are several macro and micropollutants that can implicate detrimental effect to the environment receiving the EPT effluent. It is believed that the discharge of CSO cumulatively increases the level of *E. coli* in the aquatic environment by 0.5 to 2 log (Madoux-Humery et al., 2016). The disinfection of CSOs was effectively achieved by ultraviolet irradiation (Tondera et al., 2015; Wojtenko et al., 2001a), ozone (Tondera et al., 2015; Wojtenko et al., 2014; Tondera et al., 2016), and peracetic acid (Chhetri et al., 2016; Chhetri et al., 2014; Coyle et al., 2014). Therefore, further polishing steps will be required for disinfection and oxidation purposes to minimize the environmental impact of the EPT effluent.

3.4. Conclusions

This study provided a practical solution for the enhanced primary treatment of municipal primary influent under wet weather conditions and proposed the best parameters for online process control. It demonstrated that alum is the best coagulant for the EPT. In fact, alum and PACl were both capable of maximizing the removal of major macropollutants under different mixing scenarios. However, the former had the advantage of cost efficiency and was therefore considered for further optimization. A dose of 75 mg/L alum without any polymer was guaranteed to achieve the target levels of TSS and reduce turbidity by 87% and ortho-P by 71%. While the addition of

polymer as a coagulant aid is the current practise at the WWTP, our results showed that this is not sustainable neither economically nor environmentally, as it adds costs by increasing the demand for chemicals with insignificant contribution to the reduction of the load of pollutants into the receiving environment.

Overall, bench and full-scale applications of coagulant and coagulant aid, despite their dissimilarities, unveiled comparable results especially for turbidity and TSS removal. The role of temperature in the overall optimization of the treatment process can be further investigated particularly where major seasonal and diurnal fluctuations are forecasted. The removal of macropollutants presented in this study correlated with that of micropollutants can serve as a guide for additional treatment with small footprint and high efficiency in order to refine the EPT of CSO. This treatment should consider sustainable options for disinfection and oxidation.

Acknowledgments

The authors acknowledge the financial support provided by the research grants from Natural Sciences and Engineering Research Council of Canada (NSERC) Collaborative Research and Development (CRD) program. Mr. A. Al Umairi was also supported by a scholarship from the Government of the Sultanate of Oman. The authors also recognize the valuable input of Mr. Abdul Mohammed and Ms. Bing Lin from EPCOR Water Services, Gold Bar Wastewater Treatment Plant in Edmonton, Canada, and the technical support of Dr. Shu Zhu.

References

Afzal, A., Pourrezaei, P., Ding, N., Moustafa, A., Hwang, G., Drzewicz, P., Kim, E.S., Pérez-Estrada, L.A., Chelme-Ayala, P., Liu, Y. and Gamal El-Din, M. (2011) Physico-Chemical Processes. Water Environ. Res. 83(10), 994-1091.

- Aguilar, M.I., Saez, J., Llorens, M., Soler, A., Ortuno, J.F., Meseguer, V. and Fuentes, A. (2005) Improvement of coagulation-flocculation process using anionic polyacrylamide as coagulant aid. Chemosphere 58(1), 47-56.
- Altmann, J., Massa, L., Sperlich, A., Gnirss, R. and Jekel, M. (2016) UV254 absorbance as realtime monitoring and control parameter for micropollutant removal in advanced wastewater treatment with powdered activated carbon. Water Res 94, 240-245.
- Altmann, J., Zietzschmann, F., Geiling, E.-L., Ruhl, A.S., Sperlich, A. and Jekel, M. (2015) Impacts of coagulation on the adsorption of organic micropollutants onto powdered activated carbon in treated domestic wastewater. Chemosphere 125.
- Anderson, M.J. (2015) Design of Experiments Simplified, Boca Raton, FL: CRC Press is an imprint of the Taylor & Francis Group, an Informa business.
- APHA (2012) Standard Methods for the Examination of Water and Wastewater, APHA, AWAA, WEF, Washington, DC.
- Arlos, M.J., Braggb, L.M., Servos, M.R. and Parker, W.J. (2014) Simulation of the fate of selected pharmaceuticals and personal care products in a highly impacted reach of a Canadian watershed. Sci. of the Total Environ. 193-204.
- Baghvand, A., Zand, A.D., Mehrdadi, N. and Karbassi, A. (2010) Optimizing coagulation process for low to high turbidity waters using aluminum and iron salts. Am. J. of Environ. Sci. 6(5), 442-448.
- Canada, E. (2019) Municipal Wastewater Status in Canada.

- Cao, B., Gao, B., Xu, C., Fu, Y. and Liu, X. (2010) Effects of pH on coagulation behavior and floc properties in Yellow River water treatment using ferric based coagulants. Chin. Sci. Bull. 55(14), 1382-1387.
- Chelme-Ayala, P., Afzal, A., Ding, N., Moustafa, A.M.A., Pourrezaei, P., Alpatova, A., Drzewicz,P., Hwang, G., Kim, E.-S., Liu, Y. and Gamal-El-Din, M. (2012) Physico-ChemicalProcesses. Water Environ. Res. 84(10), 971-1028.
- Chelme-Ayala, P., Gamal-El-Din, M., R., R.S. and Leonard, C. (2011) Advanced treatment of liquid swine manure using physico-chemical treatment. J. Hazard. Mater. 186, 1632-1638.
- Chhetri, R.K., Bonnerup, A. and Andersen, H.R. (2016) Combined Sewer Overflow pretreatment with chemical coagulation and a particle settler for improved peracetic acid disinfection. J. of Ind. & Eng. Chem. 37, 372-379.
- Chhetri, R.K., Flagstad, R., Munch, E.S., Hørning, C., Berner, J., Kolte-Olsen, A., Thornberg, D. and Andersen, H.R. (2015) Full scale evaluation of combined sewer overflows disinfection using performic acid in a sea-outfall pipe. Chem. Eng. J.1 270, 133-139.
- Chhetri, R.K., Thornberg, D., Berner, J., Gramstad, R., Ojstedt, U., Sharma, A.K. and Andersen, H.R. (2014) Chemical disinfection of combined sewer overflow waters using performic acid or peracetic acids. Sci Total Environ 490, 1065-1072.
- Chys, M., Audenaert, W.T.M., Deniere, E., Mortier, S.v.T.r.s.F.C., Langenhove, H.V., Nopens, I., Demeestere, K. and Hulle, S.W.H.V. (2017) Surrogate-Based Correlation Models in View of Real-Time Control of Ozonation of Secondary Treated Municipal Wastewater - Model Development and Dynamic Validation. Environ. Sci. & Technol. 51, 14233-14243.

City-of-Edmonton (2009) City of Edmonton Total Loadings Plan City of Edmonton

- Cornwell, D.A. and Bishop, M.M. (1983) Determining velocity gradients in laboratory and fullscale systems. American Water Works Association Journal 75(9), 470-475.
- Coyle, E.E., Ormsbee, L.E. and Brion, G.M. (2014) Peracetic Acid as an Alternative Disinfection Technology for Wet Weather Flows. Water Environ. Res. 86(8), 687-697.
- CWN (2018) Canada's Challenges and Opportunities to Address Contaminants in Wastewater Supporting Document 2
- David Jenkins, J.F.F., And Arnold and Menar, B. (1970) Review Paper: Chemical Processes for Phosphate Removal, pp. 369-389., Water Res. Pergamon Press.
- Dempsey, B.A., Sheu, H., Tanzeer Ahmed, T.M.T. and Mentink, J. (1985) Polyaluminum Chloride and Alum Coagulation of Clay-Fulvic Aicd Suspensions. American Water Works Association 77(3), 74-80.
- Ebeling, J.M. and Ogden, S.R. (2004) Application of chemical coagulation aids for the removal of suspended solids (TSS) and phosphorus from the microscreen effluent discharge of an intensive recirculating aquaculture system. North Am. J. of Aquacult. 66(3), 198-207.
- El Samrani, A.G., Lartiges, B.S. and Villieras, F. (2008) Chemical coagulation of combined sewer overflow: heavy metal removal and treatment optimization. Water Res 42(4-5), 951-960.
- Exall, K. and Marsalek, J. (2013) A Coagulant Survey for Chemically Enhanced Primary Treatment of Synthetic CSOs. Water Air and Soil Pollut. 224(2), 9.
- Gasperi, J., Garnaud, S., Rocher, V. and Moilleron, R. (2008) Priority pollutants in wastewater and combined sewer overflow. Sci Total Environ 407(1), 263-272.

- Gasperi, J., Rocher, V., Gilbert, S., Azimi, S. and Chebbo, G. (2010) Occurrence and removal of priority pollutants by lamella clarification and biofiltration. Water Res. 44(10), 3065-3076.
- Gibson, J., Farnood, R. and Seto, P. (2016) Chemical pretreatment of combined sewer overflows for improved UV disinfection. Water Sci Technol 73(2), 375-381.
- Grayson, R.B., Finlayson, B.L., Gippel, C.J. and Hart, B.T. (1996) The potential of field turbidity measurements for the computation of total phosphorus and suspended solids loads. J. of Environ. Manage. 47(3), 257-267.
- Guida, M., Mattei, M., Della Rocca, C., Melluso, G. and Meric, S. (2007) Optimization of alumcoagulation/flocculation for COD and TSS removal from five municipal wastewater. Desalination 211(1-3), 113-127.
- Hannouche, A., Chebbo, G., Ruban, G., Tassin, B., Lemaire, B.J. and Joannis, C. (2011) Relationship between turbidity and total suspended solids concentration within a combined sewer system. Water Sci. and Technolo. 64(12), 2445-2452.
- Hussain, S., Aziz, H.A., Isa, M.H., Ahmad, A., Van Leeuwen, J., Zou, L., Beecham, S. and Umar,
 M. (2011) Orthophosphate removal from domestic wastewater using limestone and granular activated carbon. Desalination 271(1-3), 265-272.
- Jalliffier-Verne, I., Leconte, R., Huaringa-Alvarez, U., Madoux-Humery, A.S., Galarneau, M., Servais, P., Prevost, M. and Dorner, S. (2015) Impacts of global change on the concentrations and dilution of combined sewer overflows in a drinking water source. Sci. of the Total Environ. 508, 462-476.

- Jiang, J.K. and Graham, N.J.D. (2010) Enhanced Coagulation Using Al/Fe(III) Coagulants: Effect of Coagulant Chemistry on the Removal of Colour-Causing NOM. Environ. Technol. 17, 937-950.
- Johnson, A.C. and Williams, R.J. (2004) A Model To Estimate Influent and Effluent Concentrations of Estradiol, Estrone, and Ethinylestradiol at Sewage Treatment Works. Environ. Sci. and Technol. 38, 3649-3658.
- Jung, C., Oh, J. and Yoon, Y. (2015) Removal of acetaminophen and naproxen by combined coagulation and adsorption using biochar: influence of combined sewer overflow components. Environ Sci Pollut Res Int 22(13), 10058-10069.
- Kay, P., Hughes, S.R., Ault, J.R., Ashcroft, A.E. and Brown, L.E. (2016) Widespread, routine occurrence of pharmaceuticals in sewage effluent, combined sewer overflows and receiving waters. Environ. Pollut. 220, 1447e1455.
- Kerr, J.L., Guo, Z., Smith, D.W., Goss, G.G. and Belosevic, M. (2008) Use of goldfish to monitor wastewater and reuse water for xenobiotics. J. of Environ. Eng. & Sci. 7.
- Kim, M.-K. and Zoh, K.-D. (2016) Occurrence and removals of micropollutants in water environment. Environ. Eng. Res. 21(4), 319-332.
- Launay, M.A., Dittmer, U. and Steinmetz, H. (2016) Organic micropollutants discharged by combined sewer overflows Characterisation of pollutant sources and stormwater-related processes. Water Res 104, 82-92.
- Li, J.G., Dhanvantari, S., Averill, D. and Biswas, N. (2003) Windsor combined sewer overflow treatability study with chemical coagulation. Water Quality Res. J. of Can. 38(2), 317-334.

- Li, J.G., Horneck, H., Averill, D., McCorquodale, J.A. and Biswas, N. (2004) High-Rate Retention Treatment Basins for CSO Control in Windsor, Ontario. Water Quality Res. J. of Can. 39(4), 449–456.
- Madoux-Humery, A.S., Dorner, S., Sauve, S., Aboulfadl, K., Galarneau, M., Servais, P. and Prevost, M. (2016) The effects of combined sewer overflow events on riverine sources of drinking water. Water Res 92, 218-227.
- Margot, J., Rossi, L., Barry, D.A. and Holliger, C. (2015) A review of the fate of micropollutants in wastewater treatment plants. Wiley Interdisciplinary Reviews: Water 2(5), 457-487.
- Mohammed, S. and Shanshool, H.A. (2009) Phosphorus Removal from Water and Waste Water by Chemical Precipitation Using Alum and Calcium Chloride. Iraqi J. of Chem. and Petroleum Eng. 10.
- Mutzner, L., Staufer, P. and Ort, C. (2016) Model-based screening for critical wet-weather discharges related to micropollutants from urban areas. Water Res 104.
- Ning, B., Graham, N., Zhang, Y., Nakonechny, M. and Gamal El-Din, M. (2007) Degradation of Endocrine Disrupting Chemicals by Ozone/AOPs. Ozone: Sci. & Eng. 29(3), 153-176.
- Phillips, P.J., Chalmers, A.T., Gray, J.L., Kolpin, D.W., Foreman, W.T. and Wall, G.R. (2012) Combined sewer overflows: an environmental source of hormones and wastewater micropollutants. Environ Sci Technol 46(10), 5336-5343.
- Pifer, A.D. and Fairey, J.L. (2014) Suitability of Organic Matter Surrogates to Predict Trihalomethane Formation in Drinking Water Sources. Environ. Eng. Sci. 31.

- Pourrezaei, P., Afzal, A., Ding, N., Islam, M.S., Moustafa, A., Drzewicz, P., Chelme-Ayala, P. and Gamal-El-Din, M. (2010) Physico-Chemical Processes. Water Environ. Res. 82(10), 997-1072.
- Pourrezaei, P., Drzewicz, P., Wang, Y., Gamal-El-Din, M., Perez-Estrada, L.A., Martin, J.W., Anderson, J., Wiseman, S., Liber, K. and Giesy, J.P. (2011) The Impact of Metallic Coagulants on the Removal of Organic Compounds from Oil Sands Process-Affected Water. Environ. Sci. Technol. 45(19), 8452-8459.
- Ratnaweera, H. and Fettig, J. (2015) State of the Art of Online Monitoring and Control of the Coagulation Process. Water 7(11), 6574-6597.
- Ryu, J., Oh, J., Snyder, S.A. and Yoon, Y. (2014) Determination of micropollutants in combined sewer overflows and their removal in a wastewater treatment plant (Seoul, South Korea). Environ Monit Assess 186(5), 3239-3251.
- Saritha, V., Srinivas, N. and Vuppala, S. (2017) Analysis and optimization of coagulation and flocculation process. Appl. Water Sci. 7, 451-460.
- Sarkar, S., Kamilya, D. and Mal, B.C. (2007) Effect of geometric and process variables on the performance of inclined plate settlers in treating aquacultural waste. Water Res 41, 993-1000.
- Shu, Z., Bolton, J.R., Belosevic, M. and Gamal-El-Din, M. (2013) Photodegradation of emerging micropollutants using the medium-pressure UV/H₂O₂ Advanced Oxidation Process. Water Res 47, 2881-2889.
- Shu, Z., Singh, A., Klamerth, N., McPhedran, K., Bolton, J.R., Belosevic, M. and Gamal-El-Din,
 M. (2016) Pilot-Scale UV/H₂O₂ Advanced Oxidation Process for Municipal Reuse Water:

Assessing Micropollutant Degradation and Estrogenic Impacts on Goldfish (*Carassius auratus L.*). Water Res 101, 157-166.

- Smoczynski, L., Munska, K.T., Kosobucka, M. and Pierozynski, B. (2014) Phosphorus and COD Removal from Chemically and Electrochemically Coagulated Wastewater. Environ. Prot. Eng. 40(3), 63-73.
- Sui, Q., Huang, J., Deng, S., Yu, G. and Fan, Q. (2010) Occurrence and removal of pharmaceuticals, caffeine and DEET in wastewater treatment plants of Beijing, China. Water Res 44(2), 417-426.
- Tolouei, S., Jean-Baptiste Burneta, Autixiera, L., Taghipoura, M., Bonsteelb, J., Duy, S.V., Sauve, S., Prevost, M. and Dorner, S. (2019) Temporal variability of parasites, bacterial indicators, and wastewater micropollutants in a water resource recovery facility under various weather conditions. Water Res 148, 446-458.
- Tondera, K., Klaer, K., Gebhardt, J., Wingender, J., Koch, C., Horstkott, M., Strathmann, M., Jurzik, L., Hamza, I.A. and Pinnekamp, J. (2015) Reducing pathogens in combined sewer overflows using ozonation or UV irradiation. Int J Hyg Environ Health 218(8), 731-741.
- Tondera, K., Klaer, K., Koch, C., Hamza, I.A. and Pinnekamp, J. (2016) Reducing pathogens in combined sewer overflows using performic acid. Int J Hyg Environ Health 219(7 Pt B), 700-708.
- Trinh, T.K. and Kang, L.S. (2015) Coagulation of phosphorus: effects of Al(III) species (Al-a, Al-b, and Al-c). Desalination and Water Treat. 53(2), 485-492.
- USEPA (2018) Combined Sewer Overflows (CSOs). From https://www.epa.gov/npdes/combinedsewer-overflows-csos. Accessed June 2019

- Wang, C., Alpatova, A., McPhedran, K.N. and Gamal-El-Din, M. (2015a) Coagulation/flocculation process with polyaluminum chloride for the remediation of oil sands process-affected water: Performance and mechanism study. J. Environ. Manage. 160, 254-262.
- Wang, W., Xu, Z., Li, H. and Jin, W. (2015b) Optimization of coagulation–flocculation process for combined sewer overflow wastewater treatment using response surface methodology. Desalination and Water Treat. 57(32), 14824-14832.
- Wei, N., Zhang, Z.G., Liu, D., Wu, Y., Wang, J. and Wang, Q.H. (2015) Coagulation behavior of polyaluminum chloride: Effects of pH and coagulant dosage. Chinese Journal of Chem. Eng. 23(6), 1041-1046.
- Wojtenko, I., Stinson, M.K. and Field, R. (2001a) Challenges of Combined Sewer Overflow Disinfection by Ultraviolet Light Irradiation. Critical Reviews in Environ. Sci. and Technol. 31(3), 223-239.
- Wojtenko, I., Stinson, M.K. and Field, R. (2001b) Performance of Ozone as a Disinfectant for Combined Sewer Overflow. Critical Reviews in Environ. Sci. & Technol. 31(4), 295-309.
- Young, S., Stanley, S.J. and Smith, D.W. (2000) Effect of mixing on the kinetics of polymer-aided flocculation. Journal of Water Services Research and Technology-Aqua 49(1), 1-8.
- Yu, W.-z., Gregory, J., Campos, L. and Li, G. (2011) The role of mixing conditions on floc growth, breakage and re-growth. Chem. Eng. J. 171(2), 425-430.
- Yu, W., Li, G., Xu, Y. and Yang, X. (2009) Breakage and re-growth of flocs formed by alum and PACl. Powder Technology 189(3), 439-443.

- Yukselen, M.A. and Gregory, J. (2004) The effect of rapid mixing on the break-up and reformation of flocs. Journal of Chem. Technol. & Biotechnol. 79(7), 782-788.
- Zhang, K., Deletic, A., Page, D. and McCarthy, D.T. (2015) Surrogates for herbicide removal in stormwater biofilters. Water Res 81, 64-71.
- Zhou, Y., Liang, Z. and Wang, Y. (2008) Decolorization and COD removal of secondary yeast wastewater effluents by coagulation using aluminum sulfate. Desalination 225(1-3), 301-311.
- Zhu, K., Gama-El-Din, M., Moawad, A.K. and Bromley, D. (2010) Physical and Chemical Processes for Removing Suspended Solids and Phosphorus from Liquid Swine Manure. Environ. Technol. 25(10), 1177-1187.
- Zhu, W., Seth, R. and Lalman, J. (2007) Evaluation of a micro carrier weighted coagulation flocculation process for the treatment of combined sewer overflow. Environ. Technol. 28(7), 761-770.
- Ziska, A.D., Park, M., Anumol, T. and Snyder, S.A. (2016) Predicting trace organic compound attenuation with spectroscopic parameters in powdered activated carbon processes. Chemosphere 156, 163-171.

CHAPTER 4: Treatment of a Mixture of Pharmaceuticals, Herbicides and Perfluorinated Compounds by PAC-Catalyzed Ozonation: Insight into the Non-Free 'OH Contingent Mechanisms²

4.1 Introduction

With the relentless advancements in agricultural, industrial and health care sectors matching the continuous evolvement in human lifestyle, new environmental challenges are constantly surfacing. One common problem emerging from those sectors is the presence of micropollutants (MPs) in the environment. MPs such as pharmaceuticals, personal care products, herbicides, insecticides and industrial compounds are manmade synthetic chemicals or naturally available products that have been detected in the environment at very low concentrations; however, they are capable of exerting adverse ecological impacts (Stamm et al., 2016). They find their way from production, to utilization to point or non-point disposal into landfills, soils and wastewaters, and are eventually detected in surface, ground and drinking waters (Barbosa et al., 2016). Nowadays, there is ample literature on the occurrence, fate and treatment of MPs. Yet treatment options remain under the spotlight because MPs vary in chemical structure and reactivity and accordingly numerous techniques to remove them are continuously investigated for both efficiency and practicality. Advanced oxidation processes (AOPs) constitute a viable option for the abatement of MPs that are hardly biodegradable and tend to resist conventional treatments (Ikehata et al., 2008; Ikehata et al., 2006).

The exposure to ozone (O_3) in a single ozonation process can selectively transform MPs into other products via direct oxidation/reduction, cycloaddition, electrophilic substitution or nucleophilic reactions (Beltran, 2004). While many MPs demonstrated high removals by single

² A version of this chapter was submitted to Applied Catalysis B: Environmental

ozonation, literature identified some MPs that are insufficiently reactive with O₃ such as: phthalate (Jin et al., 2012; Ning et al., 2007), bezafibrate, iomeprol (Altmann et al., 2014), benzotriazole (Altmann et al., 2014; Hollender et al., 2009), atenolol (Hollender et al., 2009), MPs with amine groups (Hubner et al., 2015), pesticides (Broseus et al., 2009; Chelme-Ayala et al., 2011b), and saturated aliphatic compounds (Jin et al., 2012). In general, the common range of applied O₃ doses for the removal of MPs is reported per mg of dissolved organic carbon (DOC): 0.2 - 1.5 mg O₃/mg DOC (Song et al., 2017; Zimmermann et al., 2011).

In order to improve the performance of ozonation in the degradation of MPs, catalysts are added to accelerate the rate of O₃ reactions, boost the removal and mineralization and reduce toxicity, all without undergoing significant chemical changes (Goncalves et al., 2012; Gonçalves et al., 2013a; Gonçalves et al., 2014; Wang et al., 2019a). There has been a specific interest in carbon-based catalysts associated with its practicality in terms of cost and ease of application, in addition to its potential dual role as catalyst and adsorbent. As such, powdered activated carbon (PAC) is used to catalyze the ozonation process by virtue of its ability to act as adsorbent material, or reaction support, or radical initiator or promoter or all of them combined (Rozas et al., 2017). The main factors reported to impact the catalytic activity of PAC are associated with its physical and chemical properties (including surface area, porosity, functional groups and pH_{pzc} (Pereira et al., 2014)), its concentration as well as its exposure time to O₃ (Alvarez et al., 2008; Sanchez-Polo et al., 2005). The activated carbon's (AC) surface area and its basic surface functional groups, in addition to the higher solution pH were found to enhance the rate of O₃ decomposition (Faria et al., 2006). The stability of PAC as a catalyst and its reusability have also been arguable (Feng et al., 2019). It is noticeable that many studies on catalytic ozonation with PAC in clean water matrix were limited to one MP or a small mixture of MPs at concentrations significantly higher than their expected environmental occurrence, i.e. in parts per million (Faria et al., 2008a; Faria et al., 2008b;

Goncalves et al., 2012; Gonçalves et al., 2013a; Gonçalves et al., 2014; Guzman-Perez et al., 2011; Merle et al., 2010; Rozas et al., 2017). Faster degradation was noted when MPs were treated individually compared to when they were treated in a mixture because of the competitive effect (Rozas et al., 2017). Literature unveils a wide range of AC dosage (10 to 850 mg/L) and particle sizes (75 to 300 µm) that can be applied in catalytic ozonation (Sanchez-Polo et al., 2005).

The mechanisms involved in catalytic ozonation with AC are numerous, generally complex and not completely understood. AC can act as an initiator for O_3 decomposition into 'OH in solution which in turn oxidizes the adsorbed MPs. Also, molecular O_3 can adsorb onto and react with the AC surface generating surface radicals that attack the adsorbed MPs (Faria et al., 2008b). A series of surface reactions have been presented and summarized to elucidate the role of AC in catalyzing the ozonation of organic compounds (Xing et al., 2014). The process was claimed to be relying mainly on 'OH. The decomposition of O_3 on the surface of AC was proposed to produce ozonide ion radicals (O_3^{-*}) , perhydroxyl radicals (HO_2^{-*}) and superoxide anion radicals (O_2^{-*}) as intermediates in a chain reaction. Hydrogen peroxide (H₂O₂) was reported as another intermediate product that eventually reacts with AC and forms 'OH. It was also recognized that 'OH can react with AC and form some other unidentified species. And, even though those compounds can diffuse in the solution where they can promote the decomposition of O_3 into 'OH to undertake the oxidation of target compounds, this diffusion is likely inefficient. The formation of those compounds was confirmed in a later study (Biernacki et al., 2015), where it was noted that the adsorbed 'OH can further react with the AC and transfer the free electron from the oxygen atom to a carbon atom resulting in AC⁻OH a compound that is stabilized by resonance.

The objective of the current study was to examine the efficiency of PAC-catalyzed ozonation for the removal of MPs in a diverse mixture of 17 MPs. The mixture included 8 pharmaceuticals and personal care products, 3 herbicides and 2 surfactants (perfluorinated compounds) that were considered for the first time in PAC-catalyzed ozonation at near-environmental concentrations. The study also aimed to provide a better understanding of the catalytic mechanisms in an O_3 +PAC system by identifying and monitoring the formed reactive species. This was also meticulously tackled through the inhibition effect of three different radical scavengers added to the O_3 +PAC system and through a detailed Electron Paramagnetic Resonance (EPR) spectroscopic analysis.

4.2 Material and methods

4.2.1. Chemicals and standards

All used chemicals: potassium indigo trisulfonate, phosphoric acid, sulfuric acid, sodium tetraborate, tertiary butanol, sodium thiosulfate, *para*-benzoquinone, sodium azide, were purchased from Sigma Aldrich, Canada. Commercial PAC WPX with a particle size of 45 µm was acquired from Calgon Carbon.

In this study, seventeen MPs were selected to represent three different categories: pharmaceuticals and personal care products, herbicides and industrial products. The list of all MPs along with their respective classes, structures, and chemical and physical characteristics are provided in Table 4.1.

#	Micropollutant: Molecular Formula	Abbreviation	Structure	Class	MW (g/mol)	рКа	log kow	Water Solubility	
	Pharmaceuticals and Personal Care Products (PPCPs)								
1	Carbamazepine C ₁₅ H ₁₂ N ₂ O	CBZ		Anticonvulsant	236.27	13.9	2.45	18 mg/L at 25 °C	
			H						
2	Carbendazim C₀H₀N₃O₂	CDM	H H	Antifungal	191.19	4.29	1.52	8 mg/L at 24 °C	
3	Clindamycin	CLN		Antibacterial	424 18	7 60	2 16	30.61 mg/L	
5	C ₁₈ H ₃₃ CN ₂ O ₅ S	CLIV	- The second sec	, inductorial	12 11 10	7.00	2.10	at 25 °C	
4	Fluconazole	FCZ		Antifungal	306.10	2.27 0.25	0.25	4363 mg/L at 25 °C	
	$C_{13}H_{12}F_2N_6O$								

Table 4.1 List of Micropollutants included in this study.

5	Gemfirbozil	GFL	0 0	Lipid-regulating agent	250.34	4.42 4.77	4.77	11 mg/L at 25 °C
	C ₁₅ H ₂₂ O ₃		°	C				
6	Ibuprofen	IBU	<mark>9</mark> ^. ^н	Anti- inflammatory	206.29	5.30	3.97	21 mg/L at 25 °C
	C ₁₃ H ₈ O ₂		C C C C C C C C C C C C C C C C C C C	initaliiniator y		T .21		at 25°C
7	Sulfamethoxazole	SMX	0	Antibacterial	253.05	1.6-5.7	0.79	610 mg/L at 37 °C
	$C_{10}H_{11}N_3O_3S$							
8	Naproxen C ₁₄ H ₁₄ O ₃	NPX	H.O	Anti- inflammatory, Anti-rheumatic	230.09	4.15	3.18	15.9 mg/L at 25 °C
9	Trimethoprim	ТМР	l .	Antibacterial	290.32	7.12	0.91	400 mg/L at
	$C_{14}H_{18}N_4O_3$					6.3		25 °C
			H_N_N_H H					

10	Ethinylestradiol C ₂₀ H ₂₄ O ₂	EE2		Semisynthetic Estrogen	296.41	10.40	3.67	11.3 mg/L at 27 °C
11	Triclosan C ₁₂ H ₇ Cl ₃ O ₂	TCN		Antiseptic	289.54	7.90	4.76	10 mg/L at 20 °C
			Herbicides					
12	Atrazine	ATZ	Н	Herbicide	215.69	1.60	2.61	34.7 mg/L
	$C_8H_{14}ClN_5$							at 26 °C
13	2,4-Dichlorophenoxya-cetic Acid	2,4-D	Ċı O O	Herbicide	221.03	2.73	2.81	677 mg/L at 25 °C
	C ₈ H ₆ Cl ₂ O ₃							
14	Mecoprop	МСР	0, H	Herbicide	214.65	3.78	3.13	880 mg/L
	C ₁₀ H ₁₁ ClO ₃		H					at 25°C

15	Diazinon	DZN		Insecticide	304.35	2.60	3.81	60 mg/L
	$C_{12}H_{21}N_2O_3PS$							at 20 °C
			Industrial Compo	unds				
16	Perfluorooctanoic Acid	PFOA	E F	Surfactant	414.07	1.30	4.81	2290 mg/L
	CE.(CE.).COOU		F F O H					at 24 °C
	CF3(CF2)6COOH							
17	Heptadecafluoro-	PFOS	F 0	Surfactant	538.22	1.0	4.49	3200 mg/L
	octanesulfonic acid		F F S K+					at 25 °C
	CF ₃ (CF ₂) ₇ SO K		F F F F					
			F F F					
			F					

Note: References:

[1] R. Broseus, S. Vincent, K. Aboulfadl, A. Daneshvar, S. Sauve, B. Barbeau, M. Prevost, Water Res. 43 (2009) 4707-4717.

[2] M. Patel, R. Kumar, K. Kishor, T. Mlsna, J. Charles U. Pittman, D. Mohan, Chem. Rev. 119 (2016) 3510-3673.

[3] P. Westerhoff, Y. Yoon, S. Snyder, E. Wert, Envrion. Sci. & Technol. 39 (2005) 6649-6663.

Some of these MPs have been found in ng/L to µg/L levels in the primary influent to a wastewater treatment plant (CBZ, FCZ, GFL, IBU, SMX, TMP, EE2, TCN), others are regulated by the Canadian Guidelines for Drinking Water Quality (ATZ, 2,4-D, DZN, PFOA, PFOS) (Health Canada, 2019). Out of the 17 MPs in the mixture, 13 MPs were studied for the first time in the O₃+PAC system: CDM, CLN, FCZ, GFL, IBU, NPX, TMP, EE2, 2,4-D, MCP, DZN, PFOA, PFOS. MPs were supplied by Sigma Aldrich, Canada and Supelco, Canada. A stock solution of borate buffer was prepared with a 1:1 mixture of sodium tetraborate and sulfuric acid and its pH was adjusted to 7 using sulfuric acid and sodium hydroxide. Stock solutions of the radical scavengers *tert*-butanol (TBA), *para*-benzoquinone (*p*-BZQ), sodium azide (NaN₃) and the O₃ quencher sodium thiosulfate were all prepared in MilliQ water and stored at 4 °C until used. All glassware used in the experiments were ozonated for 30 minutes in a washtub to ensure they were O₃ demand-free prior to each experiment.

4.2.2. Catalyst characterization

The surface area of PAC including the volume of micropores and mesopores were identified using the N₂ adsorption-desorption isotherm. The surface functional groups were identified by X-ray photoelectron spectroscopy (XPS; Kratos AXIS Ultra spectrometer, Shimadzu, Columbia, USA, equipped with monochromatic Al K α source) and the data were analyzed using XPS Vision-2 instrument software. Fourier-transform infrared spectra (FT-IR) were acquired from Thermo Scientific NICOLET 8700 FT-IR and used to assess the adsorption of materials onto the surface of PAC throughout the treatments. The point of zero charge pH_{pzc} of the catalyst was measured following the solid addition method detailed elsewhere (Pourrezaei et al., 2014).

4.2.3. Experimental procedures

All experiments were performed in duplicate 100 mL round bottom flask glass reactors with no free headspace, wrapped in aluminum foil at room temperature and atmospheric pressure. Seventeen individual stock solutions of MPs were prepared in MilliQ water at concentrations ranging between 5 and 100 ppm in conformity with their solubility limit at 20°C. Each solution was sonicated for 20 min to ensure complete dissolution and stored at 4°C in the dark until used. O₃ was generated from dry oxygen using the O₃ Generator Model GSO-30 and was bubbled into O₃ demand-free water (ODFW) in an ice bath to obtain a saturated stock solution. Experiments were performed with continuous stirring. Samples were withdrawn at fixed time intervals and analyzed for residual O₃ via the Indigo Method (Bader & Hoigne, 1981) with UV absorbance at 600 nm measured using UV spectrophotometer. Samples collected for MPs analysis were filtered with 0.22 µm nylon syringe filters into 1.5 mL amber polypropylene vials where 0.64 mM sodium thiosulfate was added to quench the residual O₃. All samples were stored at 4 °C until analysis.

4.2.4. Analytical methods

The analysis of MPs was performed using ultra-performance liquid chromatography – quadrupole time of flight mass spectrometry UPLC-qTOF-MS (Xevo G2-S, Waters). ACQUITY UPLC BEH C18, 50 x 2.1 mm column, at 50°C was utilized with an injection volume of 10 μ L. MPs were analyzed using either positive or negative ionization modes. In positive ionization mode, the mobile phase consisted of water with 0.1 % formic acid (solvent A) and acetonitrile with 0.1 % formic acid (solvent B). In negative ionization mode, 0.02% ammonium hydroxide in water (solvent A) and 0.02% ammonium hydroxide in acetonitrile (solvent B) were used. Details of the monitoring mode and mass are available in Table 4.2.

	МР	Monitored mass (<i>m/z</i>)	Mode
1	CBZ	237.103	Positive
2	CDM	192.077	Positive
3	CLN	425.188	Positive
4	FCZ	307.112	Positive
5	GFL	249.149	Negative
6	IBU	205.123	Negative
7	SMX	254.060	Positive
8	NPX	231.102	Positive
9	TMP	291.146	Positive
10	EE2	295.170	Negative
11	TCN	286.962	Negative
12	ATZ	216.102	Positive
13	2,4-D	218.962	Negative
14	MCP	213.032	Negative
15	DZN	305.109	Positive
16	PFOA	412.966	Negative
17	PFOS	498.930	Negative

Table 4.2 Monitored mass (m/z) and detection mode for the 17 micropollutants.

Electron paramagnetic resonance (EPR) measurements were performed using BRUKER ELEXSYS EPR spectrometer operated at 0.0001 T modulation amplitude, 100000 Hz modulation frequency, 9.839293 x 10^9 Hz microwave frequency and 0.02 W microwave power. Signals were acquired at 3504.95 G center field. Spin traps 5,5-diemthyl-1-1 pyrroline-N-oxide (DMPO), 2,2,6,6-tetramethylpiperidine (TEMP), 2-methyl-2-nitrosopropane (MNP), N-*tert*-butyl- α phenylnitrone (PBN) were supplied by Dojindo Molecular Tech. Inc. All experiments were run *in-situ* in 1.5 mL reactors incubated at 20 °C with 1000 rpm shaking. Samples were collected after 10 minutes reaction time using three capillary tubes and measurements were subsequently performed.

4.3. Results and discussion

4.3.1. Catalyst characterization

PAC was manufactured from bituminous coal. Its total surface area measured 632 m² g⁻¹ with a total pore volume of 0.452 cm³ g⁻¹ and it consisted mainly of micropores ($S_{micro} = 544 \text{ m}^2$ g⁻¹, $V_{micro} = 0.282 \text{ m}^3 \text{ g}^{-1}$) and few mesopores ($S_{meso} = 88 \text{ m}^2 \text{ g}^{-1}$, $V_{meso} = 0.170 \text{ m}^3 \text{ g}^{-1}$). The deconvolution of XPS spectra indicated that the major constituents of PAC surface were carbon, hydrogen and oxygen. Peaks formed for C=C (284.6 eV), C-OH (285.2 eV) and C(O)O (289.6 eV) (Feng et al., 2019). The pH_{pzc} value for the catalyst was 10 as shown in Figure.4.1. Accordingly, the catalyst presented a basic character.



Figure 4.1 Assessed PAC characteristics: (A) pH_{pzc} by Solid Addition Method; (B) XPS spectra for untreated PAC.

4.3.2. Ozone decomposition in single and catalytic ozonation

An initial screening of the O_3 and PAC dose combinations was performed. Single ozonation experiments were conducted using a mixture of 17 MPs at 500 µg/L each with two doses of O_3 (5 mg/L and 10 mg/L). The results showed that doubling the O_3 dose (10 mg/L) improved the MPs removal by up to 45% after 10 minutes of exposure time (Figure 4.2).



Figure 4.2 Comparison between the MPs removal (Ln C_0/C) in a mixture with two O_3 doses: 5 mg/L and 10 mg/L after 10 minutes of single ozonation (in borate buffer pH 7; Mixture of 17 MPs with 500 ppb each).

The dose of O_3 was then fixed at 10 mg/L and the O_3 decomposition kinetics were evaluated

against PAC doses ranging between 0 and 500 mg/L for 10 minutes in buffer at pH 7 (Figure 4.3).



Figure 4.3 O₃ decomposition kinetics in clean matrix using 10 mg/L O₃ in borate buffer at pH 7 and different doses of PAC. (x axis represents time in min; y axis represents ln $[O_3]/[O_3]_0$ where $[O_3]_0$ is the residual O₃ at time = 12 seconds)

Figure 4.4 illustrates the effect of PAC dose on the initial O₃ demand IOD and first-order rate constant of O₃ decomposition. The relationship between PAC dose and IOD was linear, revealing a significantly high IOD in presence of higher PAC doses: 66% of the applied O₃ dose was utilised in the first 12 seconds in presence of 500 mg/L PAC. On the other hand, the relationship between PAC dose and the rate of O₃ decomposition was not exactly linear as shown elsewhere in semi-batch reactors at pH 2 and 7 (Biernacki et al., 2015; Merle et al., 2010). However, the calculated rate constants were comparable to those reported at pH 7. In this study, increasing the PAC dose from 0 to 75 mg/L resulted in a ratio $\Delta k/\Delta PAC$ equal to 0.00013. That ratio increased to 0.003 when the PAC dose increased from 75 to 100 mg/L and then decreased to 0.000025 between higher PAC doses 100-500 mg/L. It is therefore noticeable that at a PAC dose of 100 mg/L the rate of O_3 decomposition increased substantially. This observed trend could be explained by a minor catalytic effect that PAC at low doses and small particle size (um level) had on O₃: in this case PAC could mimic the impact of a substrate rather than a catalyst (Biernacki et al., 2015). It was therefore noticeable that the PAC dose was correlated with the demand for O₃, yet it did not reveal similar correlation with the rate of O_3 decomposition because of the high demand in the batch reactor.



Figure 4.4 Effect of PAC dose on (A) the initial O_3 demand (IOD) in the first 12 seconds; and (B) the first-order rate constant of O_3 decomposition in borate buffer at pH 7 (O_3 dose: 10 mg/L – PAC doses: 0; 25; 50; 75; 100; 250; 500 mg/L – Borate Buffer, pH 7).

To further investigate, 10 mg/L O₃ was applied with 25 mg/L PAC on a buffered mixture of 17 MPs and the removals were compared to those achieved with 10 mg/L O₃ and 100 mg/L PAC as shown in Figure 4.5. While the rate of O₃ decomposition was higher with 25 mg/L PAC (similar to the rates obtained in buffered water), significantly less removals were noticed for: FCZ, IBU, TCN, ATZ, 2,4-D, MCP, DZN, PFOA and PFOS. These results are comparable to literature (Xing et al., 2014). They also further ascertain that at low doses, PAC did not catalyze the decomposition of O₃ but rather acted as a target compound.



Figure 4.5 Changes in the percent removal of MPs after 1 min of catalytic ozonation using different PAC doses: 25 mg/L PAC vs 100 mg/L PAC. The rates of O₃ decomposition were 1.74 L mg⁻¹ min⁻¹ with 25 mg/L PAC and 0.14 L mg⁻¹ min⁻¹ with 100 mg/L and (O₃: 10 mg/L; PAC: 25 mg/L – 100 mg/L; borate buffer pH 7; Mixture of 17 MPs with 500 ppb each).

The impact of buffer type on the kinetics of O_3 decomposition in presence of PAC in clean water was also evaluated. Experiments were performed with 10 mg/L O_3 and 100 mg/L PAC in presence of equimolar borate buffer or phosphate buffer to achieve neutral pH. It was obvious that both buffers accelerated the decomposition of O_3 in comparison with a system at natural pH (pH 5.3). The O_3 decomposition rate constant in presence of borate buffer at pH 7 was 14.8 times higher than that in natural pH 5.3, while that in presence of phosphate buffer was and 3.7 times
higher. These results are generally expected because of the high dependence of O_3 chemistry in aqueous systems on the pH whereby at higher pH more hydroxyl ions are available to initiate the decomposition of O_3 (Beltran, 2004). However, the difference noted in the effect of each buffer can be attributed to their possible interference with the chemistry of O_3 decomposition through the interaction with the catalyst surface.

Experiments performed on the mixture of 17 MPs were conducted in near-environmental conditions with each of the MPs added at a low initial concentration (500 μ g/L) in presence of 4 mM borate buffer to maintain a neutral pH. The O₃ dose was fixed a 10 mg/L, with a theoretical specific dose equivalent to 2.2 g O₃/g DOC. The O₃ decomposition kinetics (Figures 4.9 and 4.10 below) confirmed that O₃ decomposed faster in the absence of MPs. Also, its decomposition rate almost doubled with the addition of PAC in clean buffered water, while it increased by 1.3 times in the presence of MPs (from 0.11 min⁻¹ to 0.14 min⁻¹). This discrepancy in the effect of PAC on the decomposition of O₃ in presence and absence of target compounds can be attributed to the dual role of PAC as catalyst and adsorbent. Thus, when some MPs adsorb onto the surface of PAC, they could reduce the number of active sites capable of initiating O₃ decomposition which resulted in a smaller effect on the rate constant compared to when there were no target compounds to occupy potentially active sites.

4.3.3. Removal of micropollutants

In the list of 17 MPs treated in this study, only 4 MPs: ATZ (Fan et al., 2014; Guzman-Perez et al., 2011; Restivo et al., 2013; Rozas et al., 2017), CBZ (Rozas et al., 2017), TCN (Rozas et al., 2017) SMX (Goncalves et al., 2012; Gonçalves et al., 2013b) were previously examined with carbon material-based catalytic ozonation in clean water matrix. In most of those studies, the compounds were studied at higher concentrations in mg/L, individually or within a small mixture.

In this study, nine MPs were completely removed within the first minute of oxidation by single or catalytic ozonation: CBZ, CDM, CLN, GFL, SMX, NPX, TMP, EE2 and TCN (Figure 4.6). Since, these MPs demonstrated high reactivity with O₃ at low concentrations, ozonation turns out to be sufficient for their degradation.



Figure 4.6 MPs that were completely removed within the first 1 minute of oxidation in a buffered mixture of 17 MPs: adsorption (blue circle), single ozonation (orange, square), catalytic ozonation (green, triangle). (MPs mixture: 500 ppb each; O₃: 10 mg/L; PAC: 100 mg/L).

The removal results of the other eight MPs: FCZ, IBU, ATZ, 2,4-D, MCP, DZN, PFOA and PFOS after 10 minutes of adsorption, single and catalytic ozonation are summarized in Figure 4.7-A. The reactions in single and catalytic ozonation were almost complete within the first 1 minute

of treatment since the largest drop in concentration occurred in the first minute after which there was no significant removal. It was expected that under these experimental conditions, with 2.2 g O_3/g DOC, the reaction would proceed in a fast manner due to the high oxidant to pollutant ratio in a batch system. In order to elucidate whether the observed removal during catalytic ozonation was facilitated by enhanced oxidation or by adsorption, the synergistic effect was calculated as the sum of the removals during adsorption and during single ozonation. It was then compared with the catalytic effect represented by the obtained removals during catalytic ozonation (Figure 4.7-B). FCZ was the main MP that demonstrated the catalytic effect, followed by 2,4-D. In contrast, the removal of MCP and PFOA during catalytic ozonation lingered on the border line which highlights the probable role of synergy, while the removal of the other 4 MPs remained lower than that calculated by the synergistic effect. These results ascertain that the observed removals during catalytic ozonation occur for two reasons: simultaneous oxidation and adsorption as noted for MCP and PFOA, and degradation favoured by the catalytic mechanism, as noted for the remaining MPs. Generally, catalytic ozonation led to faster apparent degradation rate for most of those MPs (with the exception of IBU and PFOS, Figure 4.7-C) in comparison with single ozonation and adsorption. Also, as shown in Figure 4.8, the specific degradation rates of most MPs were higher in catalytic ozonation than in single ozonation. In particular, those rates increased 4.4 times for FCZ and 1.1 times for IBU, while they increased around 2.3 times for herbicides, 1.2 times for PFOA and PFOS. The faster removal of PFOA was mostly due to its adsorption onto PAC in catalytic ozonation. Knowing that PFOS exhibits a very low reactivity with O_3 (Dai et al., 2019b); its remarkably high removal in 1 minute that decreased with time can be associated with the foaming phenomenon observed in the ozonation experiments. It is thus possible that PFOS attached to the foam that formed during mixing by virtue of its hydrophobicity as reported elsewhere (Dai et al., 2019a).



Figure 4.7 (A) Removal of MPs in a buffered mixture of 17 MPs by different treatments: Adsorption (blue circle), Single Ozonation (orange, square), Catalytic ozonation (green, triangle); (**B**) Comparison between the 1 minute percent removal by synergistic effect (Adsorption + Single Ozonation) versus catalytic effect (catalytic ozonation) where C_0 represents the initial concentration of the MPs and C represents their concentration at 1 minute; (**C**) Comparison between the apparent degradation rate (k in min⁻¹) calculated for the first minute of the treatment.



Figure 4.8 Specific degradation rate for each MP in the mixture as μ M of MP removed per μ M O₃ utilized in 1 minute (MPs mixture: 500 ppb each; O₃: 10 mg/L; PAC: 100 mg/L, borate buffer pH 7).

4.3.4. Formation of reactive oxygen species (ROS)

4.3.4.1. Catalyst surface changes

To study the impact of O_3 and MPs on the surface of PAC, experiments were performed in a batch reactor and all reagents were mixed for 30 minutes. PAC was then allowed to decant; the supernatant was removed and PAC was dried in the oven at 100 °C for 2 hours. The recovered PAC was analyzed by FT-IR spectroscopy. The obtained spectra indicated that the surface of PAC did not undergo any significant permanent changes upon a 30 minutes exposure to borate buffer at pH 7 only, or to O_3 (10 mg/L) at natural pH of 5.3, or to the mixture of MPs at natural pH of 5.9. In contrast, in the presence of borate buffer at pH 7, some surface changes were depicted when PAC was added to the mixture of MPs alone and together with O_3 (Figure 4.9). This can be explained by the mutual interactions between pH_{pzc}, pH of the solution, and pKa of the MPs (Feng et al., 2019). In a buffered solution at pH 7, PAC with a basic pH_{pzc} of 10 loses some of its surface electron density, and its surface becomes positively charged attracting more hydroxide ions (OH⁻). When O₃ was applied under these conditions, the nucleophilic sites on PAC surface were targeted by the electrophilic sites in the O_3 molecule. As such, better catalysis was expected for a basic catalyst's surface, exhibiting higher electron density and more nucleophilic oxygenated surface groups (Gonçalves et al., 2010). With PAC, it is possible that the interaction between O_3 and the catalyst's surface resulted in radicals that can adsorb or diffuse to the bulk. The formation of these short-lived radicals did not translate into major changes on the surface groups. Other studies reported changes on the surface of PAC upon exposure to O_3 . For instance, the one hour contact between PAC and O_3 through continuous O_3 bubbling resulted in changes in C-OH and C=C bonds that were associated with the capacity of the carbon material (multi-walled carbon nanotube) for adsorption and 'OH generation (Rozas et al., 2017). It was also claimed that, depending on the nature and properties of AC, O_3 oxidized its outermost surface and increased the amount of oxygenated groups without causing any significant textural changes (Faria et al., 2006). It is therefore probable that O_3 did not have any pronounced impact on the surface of PAC utilized in this study because of its type and the O_3 exposure time and method.

The new FT-IR bands that formed in the presence of MPs, whether in adsorption only or in catalytic ozonation (Figure 4.9), corresponded to the groups alkene C=C and hydroxyl -OH (forming at 3110-3200 cm⁻¹), carboxylic salt –COONa, carbonyl group C=O (forming at 1414 cm⁻¹), and to the cyano group –CN (forming at 1199 cm⁻¹) (Coates, 2006). Those bands were generally narrower and less intense in catalytic ozonation. Therefore, all those detected groups on the surface of PAC can be attributed to the interactions between: the surface of PAC, MPs, O₃, and buffer at pH 7.



Figure 4.9 FT-IR Spectra for PAC samples recovered after 30 minutes exposure to O₃ and/or the mixture of MPs (with 4 mM borate buffer pH 7, 10 mg/L O₃, 100 mg/L PAC, mixture of 17 MPs at 500 ppb each).

(**a**): -OH, C=C; (**b**): -OH, C(=O)(-O), COONa; (**c**): C-O, C-O-SO₃⁻, -CN; (**d**): C-N, -SO₄²⁻, C-F, C-O-C; (e): aromatic C-H; (**f**): C-Br, O-H

4.3.4.2. Competitive radical test

Quenching tests were performed in the presence of different radical scavengers to explore the indirect radical pathways for degradation involving ROS in single and catalytic ozonation. Three scavengers were tested at a molar concentration equivalent to ten times that of the principal oxidant O₃. As such, the following compounds were added: 2mM TBA to scavenge 'OH ($k_{TBA/*OH}$ = (3.8-7.6) x 10⁸ M⁻¹.s⁻¹ (Wang et al., 2019b)), 2 mM *p*-BZQ to scavenge superoxide radicals (O₂^{-•}) ($k_{p-BZQ/*O2^-} = 2 \times 10^9 M^{-1}.s^{-1}$ (Wang et al., 2019b)) and 2 mM NaN₃ to scavenge singlet oxygen (¹O₂) ($k_{NaN3'}$ ¹O₂ = 2 x 10⁹ M⁻¹.s⁻¹ (Wang et al., 2015c)). It is worth noting that while those scavengers target mainly one radical, they can still react with other compounds. For instance, *p*-BZQ and NaN₃ are also strong 'OH scavengers as they react with 'OH at half or more than half the rate of their reactions with their main target radicals ($k_{p-BZQ'*OH} = 1.2 \times 10^9 M^{-1}.s^{-1}$ (Schuchmann et al., 1998) and $k_{NaN3'*OH} = 1 \times 10^9 M^{-1}.s^{-1}$ (Wang et al., 2015c)). Each of those scavengers was tested individually then all three scavengers were added together in catalytic and non-catalytic ozonation experiments. While, the most pursued impact of scavengers' addition is attacking the radicals responsible for the oxidation of MPs that favour the indirect pathway, two other impacts can also emerge. The first is the termination of O₃ decomposition, and therefore allowing residual O₃ to persist longer in solution to oxidize the MPs that are preferentially removed through the direct pathway. The second is the opposite: the promotion of O₃ decomposition which eventually means depleting the available residual O₃, responsible for the direct pathway. In these experiments, the effect of scavenger addition on the decomposition of O₃ varied with the type of matrix and presence of catalyst. That effect was more pronounced in the non-catalytic than in the catalytic O₃ decomposition (Figures 4.10 and 4.11), which hints to the presence of free radicals in both systems due to the auto-decomposition of O₃ initiated by OH⁻ at neutral pH. NaN₃ promoted the non-catalytic O₃ decomposition in clean water and inhibited it in the mixture of MPs. It can be noticed that the inhibition / promotion effect of each scavenger changed between systems.



Figure 4.10 Measured residual O_3 in buffered clean water at pH 7, during single ozonation (O_3) and catalytic ozonation ($O_3 + PAC$) with and without radical scavengers. O_3 : 10 mg/L; PAC: 100 mg/L; TBA: 2 mM; p-BZQ: 2 mM; NaN₃: 2 mM.

A. Single Ozonation of buffered clean water

B. Catalytic Ozonation of buffered clean water



Figure 4.11 Measured residual O_3 in buffered mixture of 17 MPs pH 7, during single ozonation (O_3) and catalytic ozonation ($O_3 + PAC$) with and without radical scavengers. Mixture of 17 MPs: 500 ppb each; O_3 : 10 mg/L; PAC: 100 mg/L PAC; TBA: 2 mM; *p*-BZQ: 2 mM; NaN₃: 2 mM.

On the other hand, the detailed data for the effect of adding scavengers on the removal

efficiencies (by PAC, O₃ and O₃+PAC) of MPs is available in Figure 4.12.







Figure 4.12 Effect of radical scavenger addition on the removal of each MP in a mixture of 17 MPs in borate buffer, pH 7 by (A) Adsorption; (B) Single Ozonation; (C) Catalytic ozonation (Mixture of 17 MPs: 500 ppb each; borate buffer: 4 mM; O₃: 10 mg/L; PAC: 100 mg/L; TBA: 2 mM; *p*-BZQ: 2 mM; NaN₃: 2 mM).

Control experiments to explore the effect of radical scavenger addition on the removal by adsorption during 10 minutes were performed. Since TBA is reported to poorly adsorb onto the surface of PAC and thus is more likely to scavenge free radicals in the bulk (Biernacki et al., 2015; Xing et al., 2014), experiments to assess the effect of TBA on the removal of MPs by adsorption were not performed. Nevertheless, *p*-BZQ seemed to infer some interaction with the surface of PAC whereby it almost completely inhibited the removal of more than half of the studied MPs (CLN, FCZ, GFL, IBU, NPX, 2,4-D, MCP, PFOA, PFOS). The inhibition of MPs adsorption was

unlikely to be due to *p*-BZQ as adsorption experiment with 100 mg/L PAC in presence of 2 mM *p*-BZQ in buffer showed little adsorption of *p*-BZQ on PAC with 3% (±0.35%) dissolved organic carbon (DOC) removal at 10 minutes. The effect of NaN₃ was generally more moderate and differed between MPs. It was noted elsewhere that 0.01 %W/W of NaN₃ did not seem to affect the adsorption of 1 μ g/L CBZ and NPX, onto 5 mg/L of granular activated carbon over a long period of 21 days (Yu et al., 2005). This means that NaN₃ did not impact the adsorption of CBZ and NPX which matched our results. However, it was also reported that NaN₃ can affect the adsorption capacity of AC by implying changes in the ionic strength of the solution (Yu et al., 2005) or by affecting the AC's surface properties (Klimenko et al., 2010). These could possibly explain why the addition of all three scavengers at the same time led to a decrease in removal for most MPs.

In single ozonation (Figure 4.13), all three scavengers showed relatively comparable effect for the inhibition of the removal of MPs. This was especially remarkable in the degradation of herbicides and PFAS. Some compounds showing full removal with single ozonation such CBZ, CLN, GFL, NPX, TMP, TCN were almost completely stopped from degrading by the addition of one or more radical scavengers. This is likely due to the very fast O₃ decomposition in presence of the three scavengers as displayed in Figure 4.11.



Figure 4.13 Inhibition effect of TBA, *p*-BZQ and NaN₃ added individually and combined during Single ozonation, with a buffered mixture of 17 MPs after 10 minutes of treatment (O₃: 10 mg/L, MPs in mixture: 500 ppb each; borate buffer pH 7; TBA: 2 mM; *p*-BZQ: 2 mM; NaN₃: 2 mM; exposure time: 10 minutes).

Among the three scavengers, TBA was less impactful indicating that 'OH was the least dominant oxidant in the ozonation system (Table 4.3). Also, it seemed that the removal of EE2 was favoured when any or all of the radicals were scavenged. It was reported that EE2 is highly reactive with O₃, with almost complete (99.7%) removal using 10 mg/L O₃ and 10 μ g/L EE2 at neutral pH (Maniero et al., 2008). This suggested that reaction with EE2 was more favorable with molecular O₃ than the ROS formed. For most of the remaining MPs, the combination of all three scavengers revealed similar complete scavenging effect as NaN₃ only. In single ozonation, the removal of most MPs was completely stopped upon the addition of one or a combination of scavengers, except for ATZ, 2,4-D and MCP that showed a drop from 20%, 69% and 50% to 10%, 5% and 21% respectively upon addition of TBA. Also, some removal was noted for 2,4-D and MCP with 1% and 5%, respectively, in presence of p-BZQ and 9% and 7% in presence of NaN₃

(Table 4.3).

MP	No	TBA	p-BZQ	NaN3	ТВА+ <i>p</i> -			
	scavengers				BZQ+NaN ₃			
% Removal by Single Ozonation at 10 min								
FCZ	10	0	0	0	0			
IBU	50	0	0	0	0			
ATZ	20	10	0	0	0			
2,4-D	69	5	1	9	0			
MCP	50	21	5	7	0			
DZN	3	7	0	0	0			
PFOA	10	0	0	0	0			
PFOS	57	0	0	0	0			
% Removal by Catalytic Ozonation at 10 min								
FCZ	70	60	25	29	21			
IBU	68	56	11	17	9			
ATZ	81	70	36	41	35			
2,4-D	68	61	15	1	18			
MCP	75	69	16	0	20			
DZN	96	92	71	3	45			
PFOA	43	46	15	11	0			
PFOS	88	98	0	20	0			

Table 4.3 Percent removal in single and catalytic ozonation before and after the addition of radical scavengers

In PAC-catalyzed ozonation (Figure 4.14), the removal of MPs was reduced in presence of TBA, *p*-BZQ, NaN₃ or all three scavengers combined. The only cases of complete inhibition were for 2,4-D, MCP and DZN in presence of NaN₃ (1%, 0%, 3% respectively); and for PFOA and PFOS in presence of the three scavengers. Even though when radicals were scavenged in catalytic ozonation, many MPs could still be degraded. This might be due to an alternative removal pathway: either adsorption or oxidation by other strong oxidizing species. Such species were probably associated simultaneously with the mixture of MPs and catalyst. These results do not match what was reported in literature where the effect of scavenger was slightly more pronounced in catalytic than in non-catalytic ozonation, when the assessment was based on the removal of one compound only (Guzman-Perez et al., 2011).



Figure 4.14 Inhibition effect of TBA, *p*-BZQ and NaN₃ added individually and combined during Catalytic Ozonation, with a buffered mixture of 17 MPs after 10 minutes of treatment (O₃: 10 mg/L, PAC, 100 mg/L; MPs in mixture: 500 ppb each; borate buffer pH 7; TBA: 2 mM; *p*-BZQ: 2 mM; NaN₃: 2 mM; exposure time: 10 minutes).

All examined radicals: 'OH, O_2^{-} , 1O_2 were involved, to different extents, in the indirect pathway in both single and catalytic ozonation of the mixture of MPs, with 'OH being the least impactful. It is also possible that carbon-centered radicals formed in both systems either as intermediates in the MPs degradation or by virtue of the interaction between O₃ and PAC. This was because *p*-BZQ that scavenges O_2^{-} , can also react with alkyl radicals (Denisov, 2006), which meant that its inhibition effect could be associated with the scavenging of both radicals, despite its minimal impact on the decomposition of O₃. A complex system was thus formed in single and catalytic ozonation of the mixture of MPs. 1O_2 was likely to lead the oxidation mechanism as selectively removing it by adding NaN₃ showed the largest impact. 1O_2 can contribute in the pathway of O₃ decomposition into 'OH or work directly as an oxidant. When the short-lived 1O_2 is generated as an intermediate in the reaction between O_3 and OH^- in solution, then the resulting 'OH would be free in the bulk. As such, the addition of TBA should have resulted in the same inhibition effect as the addition of NaN₃. Yet, that was not the case, since scavenging 1O_2 was more impactful than scavenging •OH in the bulk by TBA. Instead, it is very plausible that most 1O_2 formed either in the bulk during the oxidation of organic compounds (which explains the results obtained with the addition of NaN₃ in single ozonation) or on the surface of PAC upon its reaction with O_3 where it can still diffuse to the bulk.

4.3.4.3. Identification of ROS by EPR spectroscopy

A thorough investigation of the formation of ROS during single and catalytic ozonation processes was conducted with EPR. It was first observed that a characteristic signal for DMPO-OH adduct of DMPO and free 'OH (quartet signal with hyperfine splitting couplings of $a_N = a_H =$ 14.9G with a peak intensity ratio of 1:2:2:1) was obtained in single ozonation at pH 7 (Figure 4.15-A). This confirmed the formation of free 'OH in the bulk solution at pH 7 due to the non-catalytic, auto-decomposition of O₃ by reaction with ⁻OH. The same characteristic signal was also obtained with phosphate buffer at neutral pH (Figure 4.16). In contrast, the decomposition of O₃ catalyzed by PAC generated strong oxidative species or radicals, other than 'OH, that formed a DMPO-X adduct (with 7 lines 1:2:1:2:1:2:1 and a hyperfine splitting constant $a_N = 7.32$) in clean water at pH 7 with borate buffer. This DMPO-X represents an oxidized form of DMPO. There was no O2trapped since there no signal for DMPO-OOH (sextet signal) was obtained. Similarly, a DMPO-X adduct with a different spectrum (triple triplets) was obtained using phosphate buffer which further confirmed the absence of free 'OH in the bulk solution upon PAC-catalyzed O₃ decomposition. Other spin traps were then tested in an attempt to capture any formed radicals. With TEMP (Figure 4.15-B) a triplet signal (3 lines at equal intensity) was obtained representing

the typical spectrum for TEMPO which indicates the presence of ${}^{1}O_{2}$. Yet this signal was also obtained with TEMP and buffer only which means it could be an artifact. With MNP (Figure 4.15-C), a signal was detected in catalytic ozonation only representing a radical dimerization product of MNP: MNP was oxidized before trapping any radical, so neither ¹O₂ nor carbon-based radicals C' were trapped. With PBN (Figure 4.15-D) a signal for PBN-OH (triplet of doublets) was detected in single ozonation affirming the presence of 'OH, while there was no spectrum obtained in catalytic ozonation. While such observations ascertained that free 'OH could not be captured in the bulk in PAC-catalyzed ozonation, there was still a possibility that 'OH was formed on the surface or in the interface between PAC and the bulk. This possibility was supported by the reduction in DMPO-X signal intensity upon addition of TBA in borate buffer and in phosphate buffer (Figure 16). That was also proven after sample (PAC+ aqueous O₃) centrifugation and PAC separation where only the bulk was analyzed and the spectrum for DMPO-X was also reduced. Actually, in a proposed mechanism for the decomposition of O₃ on AC (Biernacki et al., 2015), it was suggested that when 'OH form near the surface, they can adsorb onto the surface of PAC where a transfer of free electron is likely to occur and result in a carbon-based radial AC⁻OH. It is likely that species such as AC-'OH or AC'-OH were responsible for the oxidation of DMPO into DMPO-X and that in presence of PAC the formed 'OH preferentially remained in proximity of the surface rather than diffused to the bulk solution.



Figure 4.15 EPR spectra obtained in clean water with (A) 100 mM DMPO, featuring a spectrum for DMPO-OH (quartet signal) in single ozonation, and a signal with 7 lines representing DMPO-X in catalytic ozonation; (B) 5.9 M TEMP, featuring a signal with 3 lines at equal intensity in both single and catalytic ozonation; (C) 20 mM MNP, featuring no spectrum in single ozonation, and low-intensity signal in catalytic ozonation; (D) 2.5 M PBN, featuring a spectrum for PBN-OH (triplet of doublets) in single ozonation, and no signal in catalytic ozonation.

(Experimental conditions: Single Ozonation: 10 mg O_3/L in 4 mM borate buffer; Catalytic Ozonation: 10 mg $O_3/L+100$ mg PAC/L in 4 mM borate buffer)



Figure 4.16 EPR spectra obtained with 100 mM DMPO in clean water under different experimental conditions: Single ozonation with 10 mg O_3/L ; Catalytic Ozonation with 10 mg O_3/L ; and 100 mg PAC/L; Catalytic Ozonation in presence of TBA with 10 mg O_3/L and 100 mg PAC/L and 100 mg PAC/L and 100 mg PAC/L and 2 mM TBA; in (A) Borate buffer, and (B) Phosphate buffer.

On the other hand, there was no proof of the formation of ${}^{1}O_{2}$, O_{2}^{\cdot} or carbon-centered radicals during PAC catalyzed ozonation since the characteristic spectra for the respective adducts with DMPO, TEMP or MNP were not obtained. Despite the fact that no free reactive oxygen species (ROS) were detected in the bulk, it was hard to demonstrate their occurrence on the surface of PAC or in the interface. In a possible scenario, with the presence of ROS in the solution, the paramagnetic centers on PAC surface, i.e. the formed carbon-centered radicals $-C^{\bullet}$, could transform into $-C-O^{\bullet}$ as suggested in literature (Wieckowski et al., 2015). When the EPR analysis was performed in a mixture of MPs, an intense spectrum for DMPO-OH was obtained during single ozonation revealing the abundant formation of free •OH in the bulk in comparison with the free 'OH formed in clean water. Yet, in catalytic ozonation it seems that DMPO-OH spectrum decreased and a new spectrum emerged signaling the formation of other radicals or strong oxidative species (Figure 4.17). The addition of TBA to scavenge free 'OH did not reveal any significant impact on the intensity of that spectrum. These findings were in conformity with the radical scavenging tests using TBA presented in section 4.3.4.1 where the addition of TBA only slightly inhibited the removal of MPs. As such, it can be confirmed that there were no free 'OH in the bulk solution during catalytic ozonation with PAC, and thus free 'OH did not partake in the catalytic oxidation of MPs.



Figure 4.17 EPR spectra obtained with 100 mM DMPO in a mixture of 17 MPs at 250 ppb each in 4 mM borate buffer pH 7, under different experimental conditions: Single Ozonation: 10 mg O_3/L ; Catalytic Ozonation: 10 mg O_3/L +100 mg PAC/L; Catalytic Ozonation + TBA: 10 mg O_3/L +100 mg PAC/L+2 mM TBA.

4.3.5. Proposed catalytic ozonation mechanisms

Based on these radical scavenging and EPR experimental results, it can be concluded that 'OH was not the key species in the catalytic mechanism. Instead other strong oxidizing species were probably responsible for the removal of target MPs in single ozonation and in PAC-catalyzed ozonation besides adsorption. By employing PAC as a catalyst, it seemed that it maintained its adsorbing role in addition to its role as initiator/promoter for O_3 decomposition. The latter was clearly justified when the rate of O_3 decomposition increased upon addition of PAC in clean water. Figure 4.18 displays a proposed schematic for the MPs removal mechanism by O_3 and PAC. In this figure, it is suggested that O_3 decomposition in presence of PAC and MPs at pH 7 takes place at two different levels simultaneously. O_3 reacts with OH⁻ and MPs in the bulk solution and with the surface of PAC. In this case with basic surface, PAC tends to lose its electron density when in solution and gain more protons which will facilitate the adsorption of deprotonated MPs (with pKa higher than pH 7 of the solution) making the adsorption pathway favourable for some MPs. Also, the now acidic PAC surface attracts electrophilic species such as O_3 and radicals with free electrons like 'OH, O_2^{-} and 1O_2 driving most of the reaction to occur on the surface/interface.



Figure 4.18 Suggested removal mechanisms in PAC-catalyzed ozonation.

In a complex matrix with a diverse mixture of MPs, the application of ${}^{1}O_{2}$ scavenger NaN₃ showed a significantly more pronounced inhibition effect for the removal of MPs in single and catalytic ozonation than TBA. This implies that ${}^{1}O_{2}$ was formed in O₃ + MPs in the bulk solution as intermediate or as product of the reaction of O₃ with either OH⁻ or MPs in the mixture (Equations 4.2 - 4.3 and 4.5). In its turn ${}^{1}O_{2}$ could further react to form 'OH in the bulk. It can also directly oxidize MPs in the bulk (Equation 4.6) as discussed earlier. When in water, H₂O molecules can chemisorb onto the surface of PAC and result in adsorbed hydroxyl groups as PAC–OH and H⁺ (Equation 4.7). In O₃ + PAC + MPs an additional source of ${}^{1}O_{2}$ and potentially other radicals is possibly the reaction between O₃ and the surface of the hydrated PAC (PAC-OH) (Equation 4.8),

and thus there could be more ${}^{1}O_{2}$ generated in a catalytic than in non-catalytic system. The formed ${}^{1}O_{2}$ can either diffuse to the bulk or adsorb onto the surface of PAC forming an oxidizing species that targeted MPs and resulted in oxidized MPs removed by PAC (Equation 4.9). The 'OH that form due to the interaction between O₃ and PAC (Equation 4.8) was more likely to adsorb on the surface forming a carbon-based radical PAC•–HO that can undertake the oxidation of MPs, rather than diffuse to the bulk (Equation 4.10). This was also suggested elsewhere (Biernacki et al., 2015).

Reactions in the bulk, radicals' formation in the bulk:

O₃ decomposition in water

$O_3 + H_2O$	\rightarrow	$OH + O_2$	Equation 4.1
$O_3 + OH^{-1}$	\rightarrow	$O_2 + HO_2 + HO_3 + O_3$	Equation 4.2
$2^{\circ}O_2^{-} + 2H^+$	\rightarrow	$^{1}O_{2}+H_{2}O_{2}$	Equation 4.3
$O_3 + HO_2$.	\rightarrow	•OH + 2O ₂	Equation 4.4

Reactions with MPs in solution

$$O_3 + MP$$
 $\rightarrow {}^1O_2 + MPO$ Equation 4.5
 ${}^1O_2/OH + MP$ $\rightarrow MPO$ Equation 4.6

Reactions with the surface of PAC, radicals' formation on the surface/interface:

$$PAC + H_2O$$
 $\rightarrow PAC-OH + H^+$ Equation 4.7 $O_3 + PAC-OH$ $\rightarrow {}^1O_2; {}^{\circ}O_2^-; {}^{\circ}OH$ Equation 4.8 $[{}^1O_2 + PAC] + MPs$ $\rightarrow PAC-MPO$ Equation 4.9 $\circ OH + PAC$ $\rightarrow PAC^-HO + MPs$ $\rightarrow PAC-MPO$ Equation 4.10

Hydrogen peroxide (H_2O_2) can also build up in the homogeneous bulk phase through reactions between 'OH molecules (Equation 4.11) or between 'OH and HO₃' formed in Equation 4.2 (Equation 4.12). It is then terminated by auto-decomposition or reaction with O_3 , O_2 , HO_2 (Zhao et al., 2008) (Equations 4.13 - 4.14 and 4.15). Yet, in our catalytic system, H_2O_2 did not seem to play any significant role where it measured less than 1 mg/L throughout the PAC-catalyzed ozonation experiments.

•OH+•OH	\rightarrow H ₂ O ₂ + O ₂	Equation 4.11
•OH + HO ₃ •	\rightarrow H ₂ O ₂ + O ₂	Equation 4.12
$H_2O_2 + O_3$	\rightarrow 'OH + HO ₂ ' + O ₂	Equation 4.13
$H_2O_2 + HO_2$	\rightarrow 'OH + HO ₂ + O ₂	Equation 4.14
$H_2O_2 + O_2$	\rightarrow 'OH + OH + OH + O ₂	Equation 4.15

4.4. Conclusions

In conclusion, while PAC is recognized for its high adsorption capacity, using it solely for adsorption ends up as a relatively slow process. Yet combining PAC with O₃ has proven effective for the degradation of organic micropollutants from different classes, and affinities towards adsorption and catalytic ozonation. In fact, MPs with limited reactivity with O₃ were removed by at least 68% within 10 minutes only of PAC-catalyzed ozonation with 10 mg/L O₃ and 100 mg/L PAC. In most cases, the parent compound degraded within the first minute. The catalytic mechanism with PAC relied heavily on the catalyst and the mixture of target compounds that substantially participated in the formation of radicals in PAC also acted as a support for radicals and intermediates generated in the solution by MPs oxidation. Unlike what has been reported previously, free 'OH was not the key radical, instead, singlet oxygen ¹O₂ and superoxide radical 'O₂⁻ dictated the mechanism that was mainly shifted to the interface between PAC and the bulk

solution. These results provided a new insight on the catalytic mechanisms involved in the ozonation process and can set the ground for further applications in real environmental matrices.

Acknowledgments

The authors acknowledge the financial support provided by a research grant from Natural Sciences and Engineering Research Council of Canada (NSERC) Collaborative Research and Development (CRD) program. The financial support of EPCOR Water Services is likewise acknowledged. The authors also recognize the kind assistance of Dr. Lusine Tonoyan in operating the EPR instrument.

References

- Altmann, J., Ruhl, A. S., Zietzschmann, F., & Jekel, M. (2014). Direct comparison of ozonation and adsorption onto powdered activated carbon for micropollutant removal in advanced wastewater treatment. Water Res., 55, 185-193. doi: 10.1016/j.watres.2014.02.025
- Alvarez, P. M., Masa, J., Jaramillo, J., Beltran, F. J., & Gomez-Serrano, V. (2008). Kinetics of Ozone Decomposition by Granular Activated Carbon. Ind. Eng. Chem. Res., 47, 2545-2553.
- Bader, H., & Hoigne, J. (1981). Determination of ozone in water by the Indigo Method. Water Res., 12, 449-456.
- Barbosa, M. O., Moreira, N. F., Ribeiro, A. R., Pereira, M. F., & Silva, A. M. (2016). Occurrence and removal of organic micropollutants: An overview of the watch list of EU Decision 2015/495. Water Res., 94, 257-279. doi: 10.1016/j.watres.2016.02.047
- Beltran, F. J. (2004). Ozone Reaction Kinetics for Water and Wastewater systems Lewis Publishers

- Biernacki, W., L.Fijołek, & Nawrocki, J. (2015). Dissolved Ozone Decomposition in Presence of Activated Carbon at Low pH: How Experimental Parameters Affect Observed Kinetics of the Process. Ozone: Sci. & Eng., 41(4), 296-311. doi: https://doi.org/10.1080/01919512.2018.1535889
- Broseus, R., Vincent, S., Aboulfadl, K., Daneshvar, A., Sauve, S., Barbeau, B., & Prevost, M. (2009). Ozone oxidation of pharmaceuticals, endocrine disruptors and pesticides during drinking water treatment. Water Res., 43(18), 4707-4717. doi: 10.1016/j.watres.2009.07.031
- Chelme-Ayala, P., Gamal El-Din, M., Smith, D. W., & Adams, C. D. (2011). Oxidation kinetics of two pesticides in natural waters by ozonation and ozone combined with hydrogen peroxide. Water Res., 45(8), 2517-2526. doi: 10.1016/j.watres.2011.02.007
- Coates, J. (2006). Interpretation of Infrared Spectra, A Practical Approach. In R. A. Meyers (Ed.), Encyclopedia of Analytical Chemistry: Applications, Theory and Instrumentation. doi: https://doi.org/10.1002/9780470027318.a5606
- Dai, X., Xie, Z., Dorian, B., Gray, S., & Zhan, J. (2019a). Comparative study of PFAS treatment by UV, UV/ozone, and fractionations with air and ozonated air. Environ. Sci. Water Res. & Technol., 5, 1897-1907. doi: 10.1039/c9ew00701f
- Dai, X., Xie, Z., Dorian, B., Gray, S., & Zhang, J. (2019b). Comparative study of PFAS treatment by UV, UV/ozone, and fractionations with air and ozonated air. Environ. Sci.: Water Res. Technol., 5(11), 1897-1907. doi: https://doi.org/10.1039/C9EW00701F
- Denisov, E. T. (2006). Reactivity of Quinones As Alkyl Radical Acceptors. Kinet. & Catal., 47(5), 662-671.

- Fan, X., Restivo, J., Órfão, J. J. M., Pereira, M. F. R., & Lapkin, A. A. (2014). The role of multiwalled carbon nanotubes (MWCNTs) in the catalytic ozonation of atrazine. Chem. Eng. J., 241, 66-76. doi: 10.1016/j.cej.2013.12.023
- Faria, P. C. C., Orfao, J. J. M., & Pereira, M. F. R. (2006). Ozone Decomposition in Water Catalyzed by Activated Carbon: Influence of Chemical and Textural Properties. Ind. Eng. Chem. Res., 45(8), 2715-2721.
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2008a). Activated carbon catalytic ozonation of oxamic and oxalic acids. Appl. Catal. B: Environ., 79(3), 237-243. doi: 10.1016/j.apcatb.2007.10.021
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2008b). Catalytic ozonation of sulfonated aromatic compounds in the presence of activated carbon. Appl. Catal. B: Environ, 83(1-2), 150-159. doi: 10.1016/j.apcatb.2008.02.010
- Feng, J., Xing, B., & Chen, H. (2019). Catalytic ozonation of humic acid in water with modified activated carbon: Enhancement and restoration of the activity of an activated carbon catalyst.
 J. of Environ. Manage, 237, 114-118. doi: https://doi.org/10.1016/j.jenvman.2019.02.061
- Gonçalves, A. G., Figueiredo, J. L., Órfão, J. J. M., & Pereira, M. F. R. (2010). Influence of the surface chemistry of multi-walled carbon nanotubes on their activity as ozonation catalysts. Carbon, 48(15), 4369-4381. doi: 10.1016/j.carbon.2010.07.051
- Goncalves, A. G., Orfao, J. J., & Pereira, M. F. (2012). Catalytic ozonation of sulphamethoxazole in the presence of carbon materials: catalytic performance and reaction pathways. J. Hazard. Mater, 239-240, 167-174. doi: 10.1016/j.jhazmat.2012.08.057

- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2013a). Ozonation of bezafibrate promoted
 by carbon materials. Appl. Catal. B: Environ., 140-141, 82-91. doi: 10.1016/j.apcatb.2013.03.034
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2013b). Ozonation of sulfamethoxazole promoted by MWCNT. Catal. Commun., 35, 82-87. doi: 10.1016/j.catcom.2013.02.012
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2014). Ozonation of erythromycin over carbon materials and ceria dispersed on carbon materials. Chem. Eng. J., 250, 366-376. doi: 10.1016/j.cej.2014.04.012
- Guzman-Perez, C. A., Soltan, J., & Robertson, J. (2011). Kinetics of catalytic ozonation of atrazine in the presence of activated carbon. Sep. & Purif. Technol., 79, 8-14. doi: 10.1016/j.seppur.2011.02.035
- Health Canada. (2019). Guidelines for Canadian Drinking Water Quality Summary Table. In H.E. a. C. S. B. Water and Air Quality Bureau (Ed.). Ottawa, Ontario: Health Canada.
- Hollender, J., Zimmermann, S. G., Koepke, S., Krauss, M., Mcardell, C. S., Ort, C., . . . Siegrist,
 H. (2009). Elimination of organic micropollutants in a municipal wastewater treatment plant
 upgraded with a full-scale post-ozonation followed by sand filtration Environ. Sci. & Technol., 43(20), 7862–7869.
- Hubner, U., von Gunten, U., & Jekel, M. (2015). Evaluation of the persistence of transformation products from ozonation of trace organic compounds - a critical review. Water Res, 68, 150-170. doi: 10.1016/j.watres.2014.09.051

- Ikehata, K., Gamal El-Din, M., & Snyder, S. A. (2008). Ozonation and Advanced Oxidation Treatment of Emerging Organic Pollutants in Water and Wastewater. Ozone: Sci. & Eng., 30(1), 21-26. doi: 10.1080/01919510701728970
- Ikehata, K., Jodeiri Naghashkar, N., & Gamal El-Din, M. (2006). Degradation of Aqueous
 Pharmaceuticals by Ozonation and Advanced Oxidation Processes: A Review. Ozone: Sci.
 & Eng., 28(6), 353-414. doi: 10.1080/01919510600985937
- Jin, X., Peldszus, S., & Huck, P. M. (2012). Reaction kinetics of selected micropollutants in ozonation and advanced oxidation processes. Water Res., 46(19), 6519-6530. doi: 10.1016/j.watres.2012.09.026
- Klimenko, N. A., Polyakova, T. V., & Savchina, L. A. (2010). Influence of Sodium Azide on the Removal of Fulvic Acids by Activated Carbon. J. of Water Chem. and Technol., 32(6), 329-335. doi: 10.3103/S1063455X10060032
- Maniero, M. G., Bila, D. M., & Dezotti, M. (2008). Degradation and estrogenic activity removal of 17β-estradiol and 17α-ethinylestradiol by ozonation and O3/H2O2. Sci. of the Total Environ., 407(1), 105-115. doi: https://doi.org/10.1016/j.scitotenv.2008.08.011
- Merle, T., Pic, J. S., Manero, M. H., Mathé, S., & Debellefontaine, H. (2010). Influence of activated carbons on the kinetics and mechanisms of aromatic molecules ozonation. Catal. Today, 151(1-2), 166-172. doi: 10.1016/j.cattod.2010.03.044
- Ning, B., Graham, N., Zhang, Y., Nakonechny, M., & Gamal El-Din, M. (2007). Degradation of Endocrine Disrupting Chemicals by Ozone/AOPs. Ozone: Science & Engineering, 29(3), 153-176. doi: 10.1080/01919510701200012

- Patel, M., Kumar, R., Kishor, K., Mlsna, T., Charles U. Pittman, J., & Mohan, D. (2016). Pharmaceuticals of Emerging Concern in Aquatic Systems: Chemistry, Occurrence, Effects, and Removal Methods. Chem. Rev , 119, 3510-3673. doi: 10.1021/acs.chemrev.8b00299
- Pereira, M. F. R., Gonçalves, A. G., & Orfão, J. J. M. (2014). Carbon materials as catalysts for the ozonation of organic pollutants in water. Bol. del Grupo Esp. del Carbon, 31, 18-24.
- Pourrezaei, P., Alpatova, A., Chelme-Ayala, P., Perez-Estrada, L. A., Jensen-Fontaine, M., Le, X. C., & Gamal-El-Din, M. (2014). Impact of petroleum coke characteristics on the adsorption of the organic fractions from oil sands process-affected water. Int. J of Environ. Sci. & Technol., 11, 2037–2050. doi: 10.1007/s13762-013-0406-x
- Restivo, J., Órfão, J. J. M., Pereira, M. F. R., Garcia-Bordejé, E., Roche, P., Bourdin, D., . . . Derrouiche, S. (2013). Catalytic ozonation of organic micropollutants using carbon nanofibers supported on monoliths. Chem. Eng. J., 230, 115-123. doi: 10.1016/j.cej.2013.06.064
- Rozas, O., Baeza, C., Nunez, K., Rossner, A., Urrutia, R., & Mansilla, H. D. (2017). Organic micropollutants (OMPs) oxidation by ozone: Effect of activated carbon on toxicity abatement. Sci. of the Total Environ., 590-591, 430-439. doi: 10.1016/j.scitotenv.2016.12.120
- Sanchez-Polo, M., von Gunten, U., & Rivera-Utrilla, J. (2005). Efficiency of activated carbon to transform ozone into .OH radicals: influence of operational parameters. Water Res., 39(14), 3189-3198. doi: 10.1016/j.watres.2005.05.026

- Schuchmann, M. N., Bothe, E., von-Sonntag, J., & von-Sonntag, C. (1998). Reaction of OH radicals with benzoquinone in aqueous solutions. A pulse radiolysis study. J. Chem. Soc., Perkin Trans. 2, 791-796. doi: https://doi.org/10.1039/A708772A
- Song, Y., Breider, F., Ma, J., & von Gunten, U. (2017). Nitrate formation during ozonation as a surrogate parameter for abatement of micropollutants and the N-nitrosodimethylamine (NDMA) formation potential. Water Res., 122, 246-257. doi: 10.1016/j.watres.2017.05.074
- Stamm, C., Joss, A., Ackermann, M., & R.I.L Eggen. (2016). Unravelling the Impacts of Micropollutants in Aquatic Ecosystems: Interdisciplinary Studies at the Interface of Large-Scale EcologyAdv. in Ecol. Res. (Vol. 55, pp. 183-223). doi: http://dx.doi.org/10.1016/bs.aecr.2016.07.002183
- Wang, B., Zhang, H., Wang, F., Xiong, X., Tian, K., Sun, Y., & Yu, T. (2019a). Application of Heterogeneous Catalytic Ozonation for Refractory Organics in Wastewater. Catal., 9(3). doi: https://doi.org/10.3390/catal9030241
- Wang, Y., Cao, H., Chen, C., Xie, Y., Sun, H., Duan, X., & Wang, S. (2019b). Metal-free catalytic ozonation on surface-engineered graphene: Microwave reduction and heteroatom doping. Chem. Eng. J. 355, 118-129. doi: https://doi.org/10.1016/j.cej.2018.08.134
- Wang, Y., Xie, Y., Sun, H., Xiao, J., Cao, H., & Wang, S. (2015). Hierarchical shape-controlled mixed-valence calcium manganites for catalytic ozonation of aqueous phenolic compounds. Catalysis Science and Technology. doi: 10.1039/c5cy01967b
- Westerhoff, P., Yoon, Y., Snyder, S., & Wert, E. (2005). Fate of Endocrine-Disruptor, Pharmaceutical, and Personal Care Product Chemicals during Simulated Drinking Water Treatment Processes. Environ. Sci. & Technol. 39, 6649-6663. doi: 10.1021/es0484799

- Wieckowski, A. B., Najder-Kozdrowska, L., Rechnia, P., Malaika, A., Krzyzynska, B., & Kozłowski, M. (2015). EPR Characteristics of Activated Carbon for Hydrogen Production by the Thermo-Catalytic Decomposition of Methane. Acta Phys. Polonica A, 130(3). doi: 0.12693/APhysPolA.130.701
- Xing, L., Xie, Y., Minakata, D., Cao, H., Xiao, J., Zhang, Y., & Crittenden6, J. C. (2014). Activated carbon enhanced ozonation of oxalate attributed to HO. oxidation in bulk solution and surface oxidation: Effect of activated carbon dosage and pH. J. of Environ. Sci, 26, 2095-2105. doi: http://dx.doi.org/10.1016/j.jes.2014.08.009
- Yu, Z., Peldszus, S., Anderson, W. B., & Huck, P. M. (2005). Adsorption of Selected Pharmaceuticals and Endocrine Disrupting Substances by GAC at Low Concentration Levels. Paper presented at the AWWA Water Quality Technology Conference and Exposition (WQTC), Quebec City, Quebec.
- Zhao, L., Ma, J., Sun, Z.-z., & Zha, X.-d. (2008). Catalytic ozonation for the degradation of nitrobenzene in aqueous solution by ceramic honeycomb-supported manganese. Appli.
 Catal. B: Environ., 83, 256-264. doi: 10.1016/j.apcatb.2008.02.009
- Zimmermann, S. G., Wittenwiler, M., Hollender, J., Krauss, M., Ort, C., Siegrist, H., & von Gunten, U. (2011). Kinetic assessment and modeling of an ozonation step for full-scale municipal wastewater treatment: micropollutant oxidation, by-product formation and disinfection. Water Res., 45(2), 605-617. doi: 10.1016/j.watres.2010.07.080

CHAPTER 5: Assessment of the Effect of Ozone Specific Dose on the Removal of Micropollutants in a Mixture by Single and PAC-catalyzed Ozonation through Benchmarks and Kinetics³

5.1. Introduction

Trace organic compounds such as pharmaceuticals, personal-care products, endocrine disrupting compounds, industrial surfactants, and others are frequently detected in the aquatic environment (Alexander et al., 2012). Commonly referred to as micropollutants (MPs), they occur in environmental waters at very low concentrations and can provoke adverse impacts to the environmental and public health (Stamm et al., 2016). Research has proved that treatment methods based on oxidation processes such as ozonation, are effective for their degradation in natural and synthetic media (Kolosov et al., 2018; Mecha et al., 2016; Melero et al., 2009; Sui et al., 2012). It is the presence of compounds that are not sufficiently reactive with molecular ozone (O₃) that prompts the addition of a catalyst to the process (Beltran, 2004) to improve the overall performance of ozonation. Powdered activate carbon (PAC) is one of the carbon-based catalysts that were tested in catalytic ozonation. It has the advantage of a high adsorption capacity by virtue of its small particle size. The dose selection of both O₃ and PAC is very critical for the efficiency of PACcatalyzed ozonation. Generally, the tested PAC doses ranged between 20 mg/L (Rozas et al., 2017) and 500 g/L (Faria et al., 2008b), while O_3 doses were commonly reported as specific doses realtive to the dissolved organic carbon (DOC) ranging between 0.21 - 1.53 mg O₃/mg DOC (Song et al., 2017; Zimmermann et al., 2011) with seldom doses outside this range such as 4.8 mg O₃/mg DOC (Tondera et al., 2015). The efficiency of PAC-catalyzed ozonation for the removal of several MPs that are simultaneously present in the same medium was approached in a mixture of four MPs

³ A version of this chapter will be submitted jointly with a version of chapter 6 to Journal of Hazardous Materials.

only (Rozas et al., 2017). MPs were monitored individually and in a mixture where they were reported to degrade in a slower fashion.

Despite the significant advances in analytical chemistry, the quantification of MPs has been cost-prohibitive and impractical due to the large number of structurally diverse compounds and the increasing frequency of their detection (Oulton et al., 2010). Therefore, mathematical models and surrogate parameters have been developed to understand and predict the behavior of MPs (Arlos et al., 2014; Johnson & Williams, 2004). In practical environmental applications, the use of surrogate parameters to predict the behavior of MPs has been extensively studied for municipal wastewater treatment (Acero et al., 2000). Surrogates used for municipal wastewaters included total organic carbon (TOC), chemical oxygen demand (COD), ultraviolet absorbance at 254 nm (UVA₂₅₄), and UV fluorescence (Gerrity et al., 2012; Lee et al., 2013). It was alleged that the selection of a specific surrogate parameter depends on the water treatment step being monitored (Bourgeois et al., 2001). Accordingly, in O₃-based treatments, UVA₂₅₄, total fluorescence, and dissolved organic carbon (DOC) were counted as promising surrogate parameters (Chys et al., 2018; Gerrity et al., 2012). Nevertheless, relying on these surrogates to estimate the removal of MPs is rather a holistic approach that looks at the general load of MPs and overlooks specific categories of compounds according to their predicted response to the treatment.

For a better understanding of the removals and as an alternative for actively measuring the exact concentrations of each compound, benchmarking which is the concept of assessing how one compound performs in comparison with another compound in the same media, can be used (McLachlan et al., 2016). In environmental studies, it was applied to estimate the behaviour of one pollutant relative to another pollutant that has been well studied and understood (McLachlan et al., 2016). It was also utilized to evaluate the performance of different treatments such as adsorption
with different types of adsorbents for the degradation of pollutants under multiple conditions (Ling et al., 2017).

The objective of this study was to assess the effect of O₃ specific dose on the performance of PAC-catalyzed ozonation for the degradation of a mixture of MPs in a clean matrix at neutral pH. The removals by PAC-catalyzed ozonation were benchmarked with single ozonation and adsorption. Correlations were developed between MPs to identify potential surrogate MPs for compounds with known or unknown reactivity towards O₃. Removal kinetics were attempted in batch experiments and intermediates were screened for both treatments.

5.2. Material and methods

5.2.1. Micropollutants and reagents

All micropollutants selected in this study were supplied by Sigma Aldrich, Canada and Supelco, Canada. Individual stock solutions of each MP were prepared in MilliQ water at concentrations ranging between 5 and 100 mg/L in conformity with their solubility limit at 20 °C. They were then sonicated for 20 minutes to ensure complete dissolution and used to prepare a stock mixture of MPs that was also sonicated for 20 minutes. All solutions were stored at 4 °C in the dark. Chemicals such as potassium indigo trisulfonate, phosphoric acid, sulfuric acid, sodium tetraborate, and sodium thiosulfate were purchased from Sigma Aldrich, Canada. Commercial PAC WPX with a particle size of 45 μ m and a measured surface area of 632 m²/g was acquired from Calgon Carbon. Further details on the characteristics of PAC are available in Chapter 4, section 4.3.1. Borate buffer was prepared in a stock solution with a 1:1 mixture of sodium tetraborate and sulfuric acid. The O₃ quencher sodium thiosulfate was prepared in MilliQ water and stored at 4 °C until used. All glassware used in the experiments were ozonated for 30 minutes in a washtub to ensure they were O₃ demand-free prior to each experiment.

5.2.2. Target mixture

A mixture of seventeen MPs belonging to the three different categories: pharmaceuticals and personal care products, herbicides and industrial products or surfactants was prepared in buffered water at pH 7. The MPs are enlisted in Table 5.1 along with their concentrations in the influent wastewater to a wastewater treatment plant in Alberta. Only four MPs were previously studied in a catalytic ozonation system with PAC: CBZ (Rozas et al., 2017); SMX (Goncalves et al., 2012); TCN (Rozas et al., 2017) and ATZ (Guzman-Perez et al., 2011; Rozas et al., 2017). The other thirteen were studied for the first time in the O₃+PAC system: CDM, CLN, FCZ, GFL, IBU, NPX, TMP, EE2, 2,4-D, MCP, DZN, PFOA, PFOS. To date, five MPs are regulated in the Guidelines for Canadian Drinking Water Quality (ATZ: 0.0005 mg/L; DZN: 0.02 mg/L; 2,4-D: 0.1 mg/L; PFOS: 0.0006 mg/L; PFOA: 0.0002 mg/L) (Health Canada, 2019). Also five MPs are regulated under the Environmental Quality Guidelines for Alberta Surface Waters (ATZ: 1.8 µg/L; 2,4-D: 4 μg/L; DZN: 0.17 μg/L ; EE2: 0.5 ng/L; MCP: 13 μg/L) (Alberta Environment & Sustainable Resource Development (ESRD), 2014). Two MPs are listed under the Canadian Water Quality Guidelines for the protection of Aquatic Life (in fresh water: ATZ: 1.8 μ g/L; CBZ: 10 μ g/L) (Canadian Council of Ministers of the Environment (CCME), 2014). There are still no national guidelines for the permissible concentrations in wastewater effluents.

Name		G ()	I	Influent conc. (ng/L)**	
Formula	Abbreviation	Structure	Use	2011	2019
Ethinylestradiol C ₂₀ H ₂₄ O ₂	EE2	CH ₃ OH H	PPCP*: Semisynthetic estrogen	287-2,260	3,500
Triclosan C ₁₂ H ₇ Cl ₃ O ₂	TCN	CI OH	PPCP: Antiseptic	1,210-946	N/D
Sulfamethoxazole C ₁₀ H ₁₁ N ₃ O ₃ S	SMX		PPCP: Antibacterial	597-1,420	4,900
Gemfibrozil C ₁₅ H ₂₂ O ₃	GFL		PPCP: Lipid-regulating agent	94-185	600
Carbamazepine C ₁₅ H ₁₂ N ₂ O	CBZ		PPCP: Anticonvulsant	549-1,060	N/D
Trimethoprim C ₁₄ H ₁₈ N ₄ O ₃	TMP	NH2 H2N N L L O	PPCP: Antibacterial	265-366	5,200
Naproxen C ₁₄ H ₁₄ O ₃	NPX	Hoch CH	PPCP: Anti-inflammatory, Anti-rheumatic	3,280-8,090	N/D
Mecoprop C ₁₀ H ₁₁ ClO ₃	МСР		Herbicide	N/A	4,600
2,4-Dichlorophenoxya- cetic Acid C ₈ H ₆ Cl ₂ O ₃	2,4-D	CI CI CI	Herbicide	N/A	4,700
Ibuprofen C ₁₃ H ₈ O ₂	IBU	CH ₈ CH ₈ OH	PPCP: Anti-inflammatory	7,580- 13,100	2,600
Atrazine C ₈ H ₁₄ ClN ₅	ATZ		Herbicide	N/A	N/D
Fluconazole C ₁₃ H ₁₂ F ₂ N ₆ O	FCZ		PPCP: Antifungal	N/A	100
Diazinon C ₁₂ H ₂₁ N ₂ O ₃ PS	DZN	H ₃ C CH ₃ H ₃ C CH ₃ CH ₃	Herbicide	N/A	N/D
Carbendazim C ₉ H ₉ N ₃ O ₂	CDM		PPCP: Antifungal	N/A	N/D
Clindamycin C ₁₈ H ₃₃ ClN ₂ O ₅ S	CLN	Ho Ho Ho Ho Ho Ho Ho Ho Ho Ho Ho	PPCP: Antibacterial	N/A	N/D
Perfluorooctanoic Acid CF ₃ (CF ₂) ₆ COOH	PFOA		Surfactant	N/A	N/D
Heptadecafluoro- octanesulfonic Acid CF ₃ (CF ₂) ₇ SO ₃ H	PFOS		Surfactant	N/A	N/D

 Table 5.1 Characteristics and occurrence of the studied MPs.

Notes: *PPCP = Pharmaceutical and Personal care product; N/A: Not available; N/D: Not detected;

** Concentrations in the influent wastewater to a wastewater treatment plant in Alberta.

5.2.3. Experimental procedures

5.2.3.1. Ozonation experiments

All experiments were performed in duplicates using 100 mL round bottom flask glass reactors wrapped in aluminum foil at room temperature and atmospheric pressure. O₃ was generated from dry oxygen using the O₃ Generator Model GSO-30 and was bubbled into O₃ demand-free water (ODFW) in an ice bath to obtain a saturated O₃ stock solution. A measured volume of O₃ stock solution was transferred to each reactor to achieve the desired final concentration. Reactors were then immediately capped with no free headspace, and stirred with a magnetic stirrer over a multi-position stir plate. Samples were withdrawn at fixed time intervals and analyzed for residual O₃ via the Indigo Method (Bader & Hoigne, 1981) with UV absorbance at 600 nm measured using Ultrospec 2100 UV-Visible spectrophotometer. Samples collected for MPs analysis were filtered with 0.22 µm nylon syringe filters into 1.5 mL amber polypropylene vials containing sodium thiosulfate to quench the residual O₃. All samples were stored at 4 °C in the dark until analysis.

5.2.3.2. Kinetics study

In this study, the O_3 specific dose was manipulated by varying the mass concentration of the MPs in the mixture, while keeping the mass concentrations of O_3 and PAC fixed throughout the experiments at 10 mg/L and 100 mg/L respectively. The concentrations of O_3 and PAC were selected in accordance with the study presented in Chapter 4, Figures 4.3 and 4.4. By changing the concentration of MPs in the mixture, or in other terms the DOC, the O_3 demand changed and thus the concentration of O_3 available to carry on the reaction changed. This is further discussed in section 5.3.3 below. MPs' removal kinetics were therefore attempted under the different O_3 specific doses in a batch reactor, following the same experimental procedure presented above.

5.2.4. Analytical methods

The analysis of MPs was performed using ultra-performance liquid chromatography – quadrupole time-of-flight mass spectrometry UPLC-qTOF-MS (Xevo G2-S, Waters). ACQUITY UPLC BEH C18, 50 x 2.1 mm column, at 50 °C was utilized with an injection volume of 10 μ L. MPs were analyzed using positive or negative ionization modes. In positive ionization mode, the mobile phase consisted of water with 0.1 % formic acid (solvent A) and acetonitrile with 0.1 % formic acid (solvent B). In negative ionization mode, 0.02% ammonium hydroxide in water (solvent A) and 0.02% ammonium hydroxide in acetonitrile (solvent B) were used. Details of the detection mode and monitored masses are available in Table 4.2 in Chapter 4.

5.2.5. Detection of intermediates

Literature was screened for the common intermediates and by-products that are produced during the oxidation process of each MP. The same analytical methods detailed above were employed to detect the presence of those compounds in the collected samples.

5.3. Results and discussion

5.3.1. Effect of O_3 specific dose on the removal of MPs by single ozonation

Table 5.2 summarizes the expected removals of the MPs in the mixture by adsorption onto PAC and by ozonation as reported in literature, along with their reported rate constants. The MPs were classified in four groups following their reactivity with molecular O₃ and with hydroxyl radicals ('OH). The reported second order rate constants were obtained for each MP studied individually rather than in a mixture with other MPs. Neither adsorption onto PAC nor ozonation seemed able to provide an effective solution for the degradation of all MPs from the different categories. While ozonation was less efficient for the removal of MPs in groups II and III in

addition to POFA and PFOS, adsorption onto PAC did not reveal high removal either, except for FCZ, PFOA and PFOS. Those three MPs were reportedly better removed by adsorption than by single ozonation. It is worth noting that the removal levels reported in literature were possibly a function of specific experimental conditions. For instance, the removal of PFOA and PFOS by adsorption onto activated carbon was evaluated in presence and absence of effluent organic matter (Yu & Hu, 2011). It was determined that the adsorption decreased in the presence of effluent organic matter where higher adsorbent doses would be required for mitigation.

	Removal in Literature				Kinetics in Literature				
МР	By PAC	References	Ву Оз	References	ko3 (M ⁻¹ s ⁻¹)	к. _{ОН} (М ⁻¹ s ⁻¹)	References		
	Group I: k ₀₃ > 1x10 ⁴ M ⁻¹ s ⁻¹								
EE2	Intermediate/High	(Westerhoff et al., 2005)	High	(Broseus et al., 2009)	7×10 ⁹	9.8×10 ⁹	(Huber et al., 2003)		
TCN	Intermediate/High	(Rozas et al., 2017; Westerhoff et al., 2005)	High	(Rozas et al., 2017)	3.8×10 ⁷	9.6×10 ⁹	(Suarez et al., 2007; Wert et al., 2009)		
SMX	Intermediate/Low	(Kovalova et al., 2013; Margot et al., 2013; Rizzo et al., 2019; Westerhoff et al., 2005)	High	(Goncalves et al., 2012; Kovalova et al., 2013; Margot et al., 2013; Rizzo et al., 2019)	5.7×10 ⁵	5.5×10 ⁹	(Dodd et al., 2006)		
GFL	Intermediate/Low	(Westerhoff et al., 2005)	High	(Uslu et al., 2015)	4.9×10 ⁵	7.1×10 ⁹	(Jin et al., 2012; Razavi et al., 2009)		
CBZ	Low	(Margot et al., 2013; Rozas et al., 2017)	High	(Margot et al., 2013; Rozas et al., 2017)	3×10 ⁵	8.8×10 ⁹	(Huber et al., 2003; Wert et al., 2009)		
ТМР	High	(Margot et al., 2013; Westerhoff et al., 2005)	High	(Kovalova et al., 2013; Margot et al., 2013)	2.7×10 ⁵	6.9×10 ⁹	(Dodd et al., 2006; Wert et al., 2009)		
NPX	High Intermediate/Low	(Margot et al., 2013) (Westerhoff et al., 2005)	Intermediate/High	(Margot et al., 2013)	2×10 ⁵	9.6×10 ⁹	(Huber et al., 2005; Wert et al., 2009)		
			Group II: 10 ≤ k _{O3} <	$1 x 10^4 M^{-1} s^{-1}$					
МСР	Intermediate	(Margot et al., 2013)	Intermediate	(Margot et al., 2013)	101	1.9×10 ⁹	(Beltrán & Rey, 2018)		
2,4-D	Low	(Javier-Benitez et al., 2004)	Low	(Javier-Benitez et al., 2004)	29.1	N/A	(Xiong & Graham, 2008)		
	Group III: $k_{O3} < 10 \text{ M}^{-1}\text{s}^{-1}$ and $k_{\bullet OH} > 1 \times 10^9 \text{ M}^{-1}\text{s}^{-1}$								
IBU	Intermediate/Low	(Westerhoff et al., 2005)	Intermediate	(Betancur-Corredor et al., 2016)	9.6	7.4×10 ⁹	(Beltrán & Rey, 2018; Wert et al., 2009)		
ATZ	Low Intermediate	(Rozas et al., 2017) (Margot et al., 2013; Westerhoff et al., 2005)	Low	(Guzman-Perez et al., 2011; Margot et al., 2013; Rozas et al., 2017)	6	3×10 ⁹	(Acero et al., 2000; Beltrán & Rey, 2018)		
FCZ	Intermediate/High	(Kovalova et al., 2013; Margot et al., 2013)	Low	(Kovalova et al., 2013; Margot et al., 2013)	<1	N/A	(Lee et al., 2014b)		
DZN	Intermediate/Low	(Ku et al., 1998)	Intermediate/Low	(Ku et al., 1998)	1.7-3.2×10 ⁻³ s ⁻¹	8.4×10 ⁹	(Ku et al., 1998; Real et al., 2007)		
Group IV: Data Not Available									
CDM	High	(Margot et al., 2013)	High	(Margot et al., 2013)	N/A	N/A			
CLN	High	(Margot et al., 2013)	High	(Kovalova et al., 2013; Margot et al., 2013)	N/A	N/A			
PFOA PFOS	Intermediate/High Intermediate/High	(AWWA, 2019; Yu & Hu, 2011 (AWWA, 2019; Yu & Hu, 2011)Low – Intermediate/High (pH 11))Low – Intermediate/High (pH 11)	(Lin et al., 2012) (Lin et al., 2012)	N/A N/A	<3×10 ⁷ <3×10 ⁷	(Trojanowicza et al., 2018) (Trojanowicza et al., 2018)		

Table 5.2 Summary of the removal of MPs by adsorption and single ozonation in different media as reported in literature and their reported rate constants at pH 7 and 20 °C, following similar classification used elsewhere (Guo et al., 2018; Lee et al., 2013).

Notes:

N/A = Not available; N/D = Not detected; Low removal: <50%; Intermediate removal: 50-80%; High removal: >80

In order to investigate the efficiency of single ozonation for the degradation of MPs in a diverse mixture, four different O₃ specific doses were applied to the mixture of MPs. The percent removal of each MP was separately calculated, and divided by 17, the total number of MPs in the mixture. The data was plotted in Figure 5.1 to show the percentage contribution of each MP to the total MPs removed. As the O₃ specific dose increased, the cumulative percent removal of MPs increased. The smallest O_3 specific dose tested (0.27 g O_3/g DOC) was able to remove 47% of the MPs in the mixture. The effect of increasing this dose was mainly observed through the improved removal of the less reactive MPs which eventually increased the cumulative removal. It was also noticeable that the relationship between the O_3 specific dose and the cumulative % removal of MPs in the mixture was linear for the first 3 doses, which fall within an acceptable range of applied doses (Song et al., 2017; Zimmermann et al., 2011). At the highest O_3 specific dose, which represented almost a 10 times dose increase, the cumulative percent removal increased only from 73% at 2.17 g O₃/g DOC to 79% at 21.64 g O₃/g DOC revealing near a plateau. This is better represented by the ratio of the differential cumulative % removal over the differential O₃ specific dose. This ratio was equal to 14 when the O₃ specific dose increased from 0.27 to 2.17 g O_3/g DOC; and it was reduced to 0.3 when the dose increased from 2.17 to 21.64 g O_3/g DOC. A similar trend was observed elsewhere on the DOC removal that slowed down after certain O_3 specific dose (Molnar et al., 2012). This was expected as a portion of the O₃ specific dose was used to satisfy the demand of fast reacting compounds. The less reactive ones have a limited capacity to react with O₃ thus increasing the dose would not considerably impact their removals. As such, three doses were studied in PAC-catalyzed ozonation: the low and high ends of the common range of O₃ specific doses, i.e. 0.27 and 2.17 g O₃/g DOC which also showed a significant improvement in the degradation of MPs, as well as the extreme O3 specific dose 21.64 g O3/g DOC for comparison.



Figure 5.1 Percent contribution of each MP in the mixture to the total cumulative removal after 1 minute of single ozonation under different specific O_3 doses, and linear correlation between the O_3 specific dose and the total % cumulative removal.

5.3.2. Effect of O₃ specific dose on the decomposition of O₃ in single and catalytic ozonation

The O₃ decomposition and one-minute removal of MPs under the three different O₃ specific doses are illustrated in Figure 5.2. It was noticeable that the higher the O₃ specific dose, the higher the residual O₃ and accordingly the more the O₃ available for reaction. Moreover, O₃ seemed to be degrading in more than one kinetic regime, with the first regime being the fastest. Similar behaviour was reported elsewhere where it was suggested that this is caused by the reaction with compounds that are highly or moderately reactive with O₃ (Benitez et al., 2007; Buffle et al., 2006; Chelme-Ayala et al., 2011b; Elovitz & von Gunten, 2008; Guo et al., 2018). Also, there was clearly a faster O₃ decomposition in the presence of PAC. In parallel, the removal rate of MPs as mg removed in 1 minute/mg O₃ applied, generally decreased as the O₃ specific dose increased because of the lower mass concentration of MPs at high O₃ specific dose. Yet, the removal rate reached its maximum for more MPs in the latter case than in any other condition because of the higher residual O₃. Also, it was noticeable that MPs belonging to group I exerted higher removal rates compared

to other MPs under the different O_3 specific doses. Moreover, the removal trend of MPs could only be detected at very high O_3 specific doses probably because this was the only condition that secured enough O_3 to oxidize slow-reacting MPs towards reaction completion.

Fast-reacting MPs belonging to group I were identified at below detection limit in both single and PAC-catalyzed ozonation. They could be considered equivalent to natural organic matter that exerted instantaneous O₃ demand (IOD). Therefore, their removal kinetics in a mixture could not be monitored and they are better studied individually through competition kinetics (Jin et al., 2012). IBU was present in the mixture but could not be detected at 21.64 g O₃/g DOC due to analytical errors associated with the limit of detection of IBU. The major improvement in PACcatalyzed ozonation was observed for 2 MPs in group IV: POFA and PFOS that showed almost no removal by single ozonation which was aligned with their expected reactivity (Table 5.2).



Figure 5.2 Residual O₃ and MPs specific degradation rates (mg MP removed after 1 minute/mg O₃ applied) in single and PAC catalyzed ozonation with different O₃ specific doses: Mixture of MPs at 0.05 mg/L, Mixture of MPs at 0.5 mg/L, Mixture of MPs at 4 mg/L each in borate buffer; O₃ dose: 10 mg/L, PAC dose: 100 mg/L, specific O₃ doses: (**A**) 0.27g O₃/g DOC; (**B**) 2.17 g O₃/g DOC; (**C**) 21.64 g O₃/g DOC

5.3.3. Benchmarking the removal of MPs by single and PAC-catalyzed ozonation

Along with the single and PAC-catalyzed ozonation experiments that were performed, adsorption experiments using 100 mg/L of PAC were also carried out as a control with mixtures of MPs at different concentrations. Figure 5.3 presents a comparison between the three treatments for the removal of MPs in the mixture during 10 minutes reaction time with data on all removals plotted and compared. At low O₃ specific dose (0.27 g O₃/g DOC), removals were generally low and concentrated around 0 implying minimal removal. Still, most MPs showed higher affinity for removal by catalytic ozonation than by single ozonation or adsorption. At higher O₃ specific dose (2.17g O₃/g DOC), the removal of MPs in the mixture revealed a wider spread and was mostly favoured by catalytic ozonation rather than by single ozonation or adsorption. MPs from groups II and III still showed less removal compared to other groups. At the highest O₃ specific dose (21.64 g O₃/g DOC) most MPs revealed higher affinity for removal by catalytic ozonation than by single ozonation than by single ozonation than by single ozonation or adsorption. MPs from groups II and III still showed less removal compared to other groups. At the highest O₃ specific dose (21.64 g O₃/g DOC) most MPs revealed higher affinity for removal by catalytic ozonation than by single ozonation. It was also remarkable that MPs belonging to groups III and IV were comparably or slightly better removed by adsorption on PAC.



Figure 5.3 Comparison between the removal as $\text{Ln } \text{C}_0/\text{C}$ by single ozonation, catalytic ozonation and adsorption for the different categories of MPs at three O₃ specific doses: (A) 0.27 g O₃/g DOC; (B) 2.17 g O₃/g DOC, and (C) 21.64 g O₃/g DOC during a 10 minutes reaction time: blue: *Group I*, red: *Group II*, green: *Group III*, yellow: *Group IV*.

Table 5.3 provides a benchmark for the removal of MPs at different O₃ specific doses. Removals were classified in accordance with the classification in Table 5.2 in four categories: <50%; between 50-80%, between 80-100%; and 100%. Results showed that the removal categories for most MPs were similar at 1 minute and 10 minutes which meant that the reactions were almost complete within the first 1 minute of exposure to O₃ and PAC. Also, increasing the O₃ specific dose shifted the percent removal of most MPs to at least the next higher level. An almost ten times increase in O₃ specific dose guaranteed complete degradation of fast reacting MPs (group I) by single or PAC–catalyzed ozonation. Yet, for MPs with limited reactivity with O₃ (groups II and III) that increase was only efficient in PAC-catalyzed ozonation. So, in order to enhance the degradation of diverse MPs, an increase in the O₃ dose was not enough. It was the addition of PAC to catalyze the ozonation process that prompted significant improvements in the removal of recalcitrant organic compounds. Since IBU could not be detected at 21.64 g O_3/g DOC as mentioned in section 5.3.2, it was not included in this table; yet, it is worth mentioning that its removal remained less than 50% and was shifted to 50-80% with 1 minute PAC-catalyzed ozonation and 10 minutes single or PAC-catalyzed ozonation at lower dose. The removal of PFOS was remarkably high and disagreed with the expected low removal in Table 5.2. A possible explanation was provided by Dai et al. (2019a) who claimed that the hydrophobic PFOS tends to adsorb onto the foam that results from the mixing process. The foaming phenomenon was clearly noted in our experiments and could be the reason behind its overestimated removal.

Table 5.3 Heatmap of MPs' removal by single and PAC catalyzed ozonation at 1 minute and 10 minutes of treatment and different O₃ specific doses.



Classifying MPs according to their reactivity with O_3 established a general understanding of their response to single and PAC-catalyzed ozonation in a mixture. However, estimating their removal still requires the identification of appropriate surrogates to validate the treatment. Several studies attempted to correlate the removal of MPs with other parameters that can be easily monitored. For instance, the ultraviolet absorbance at 254 nm (UVA₂₅₄) was appraised in different studies for its ability to predict the removal of some MPs in real time. Differential total phosphorus was also correlated with MPs (Zhang et al., 2015). In this study, correlations between the removals of MPs were attempted to identify surrogate compounds on the basis of their degradation in single (Figure 5.4) and PAC-catalyzed ozonation (Figure 5.5). The illustrated correlations correspond to the O_3 specific dose 2.17 g O_3/g DOC. Actually, there was a clear discrepancy between the different O_3 specific doses analyzed in most of the correlations where the data for all O_3 specific doses did not follow the same trend and deviated remarkably between low and high doses. This is possibly explained by the high level of DOC that occurred at the low O_3 specific dose and implied high IOD and not enough residual O₃ to react with the MPs. This also explained the low removals and the clear deviation from the data obtained at higher O₃ specific dose. This deviation was actually eased in PAC-catalyzed ozonation with some exceptions. It was thus determined that CBZ could serve as a surrogate for MPs that are highly reactive with O₃ while ATZ could serve as a surrogate for MPs that are less reactive with O_3 or with unknown reactivity.



Figure 5.4 Correlation between the removal (Ln C/C₀) of CBZ and the removal of MPs in group I, the removal of ATZ and the removal of MPs in groups II, III and IV under three different O_3 specific doses during single ozonation.



Figure 5.5 Correlation between the removal (Ln C/C₀) of CBZ and the removal of MPs in group I, the removal of ATZ and the removal of MPs in groups II, III and IV under three different O_3 specific doses during catalytic ozonation.

5.3.4. Kinetics of MPs removal in a mixture by single and catalytic ozonation

Under the different O₃ specific doses compared in this study during single and PACcatalyzed ozonation, the reaction seemed to be very fast in the first 1 minute where most of the degradation occurred, after which there was a minimal degradation if any. In order to confirm this observation, the predicted half-lives of MPs were calculated under the different O₃ specific doses based on the assumption that the overall reaction between the MP and O₃ or between MP and 'OH are bimolecular and follow second order rate kinetics Equation 5.1 (Metcalf & Eddy, 2014). The time required to degrade half the initial concentration of the MPs otherwise known as the half-life, was obtained from Equations 5.1 and 5.2 and calculated according to Equation 5.3.

$$-\frac{dC}{dt} = k_{[MP]}[MP][O_3] \qquad Equation 5.1$$

$$k' = k_{[MP]}[O_3] \times \frac{[MP]}{[MP]_t} \qquad Equation 5.2$$

$$ln2$$

$$t_{1/2} = \frac{ln2}{k'}$$
 Equation 5.3

Where $k_{[MP]}$ is the second order rate constant for the degradation of MP by O₃ in M⁻¹s⁻¹; [MP] is the concentration of the MP in M, [O₃] is the concentration of O₃ in M; k' is the first order rate constant in s⁻¹; t_{1/2} is the half-life of the MP, and $[MP]_t$ is the total molar concentration of all MPs in the mixture. k' was corrected to account for the relative concentration of the MP to all MPs in the mixture assuming that 1 M of O₃ would react with 1 M of MP and knowing that the MPs are present at different molar concentrations in the mixture. So, the overall reaction rate depends on the molar concentration of the oxidant as well as that of the target compound. On the other hand, the half-life depended on the oxidant's concentration provided that the oxidant is reacting strictly with the MP. Similarly, the half-lives for the reactions between MPs and 'OH only were calculated following the same assumptions made for O₃ and assuming that one molar of O₃ in water decomposes into one molar of 'OH. Details of the first order rate constants and half-lives are presented in Tables A.1 and A.2 in the Appendix.

It was remarkable that the predicted half-lives of MPs were longer when treated in a mixture compared to when treated individually and that the O_3 specific dose did not reveal any significant effect on the time predicted for the removal of MPs. Jin et al. (2012) tested and summarized three methods for measuring the rate constants for MPs with molecular O₃ depending on their reactivity. Those methods are: 1) Competition kinetics for fast reacting MPs, performed in the presence of the target compound and a reference compound (with known rate constant with O_3) at the same molar concentration, and 'OH scavenger (typically tertiary butanol TBA); 2) Compound monitoring for slow reacting MPs, performed in the presence of the target MP and O₃ at ten times the concentration of MP in addition to TBA; and 3) O₃ monitoring method for low solubility MPs, performed in the presence of the target MP at five times the concentration of O_3 and with TBA, the concentration of O_3 is monitored with time and used to calculate the rate constant. The rate constants for the MPs with O₃ reported in literature are thus obtained from the reaction between the target compound and molecular O₃ only generally in clean matrix and under known experimental conditions such as pH and temperature. It was also assumed that major radicals contributing to the reaction are scavenged. DOC is one main factor known to affect the reaction rate constant with O_3 in two possible manners: the reaction between DOC and molecular O_3 generates radicals that contribute to the oxidation of MPs; or the reaction between DOC and molecular O₃ consumes the oxidant and reduces its availability to degrade MPs (Uslu et al., 2015). In this study, since DOC was the only factor that changed, it can be presumed that the concentration of MPs expressed as DOC was controlling the removal kinetics as discussed below. Overall, only 3 MPs in the mixture presented removal trends as illustrated in Figures 5.6 used to compute the overall rate constants. 2,4-D (group II) was better removed than ATZ and FCZ (group III) in both single and catalytic ozonation.



Figure 5.6 Removal of 2,4-D, ATZ and FCZ by (A) Single ozonation and (B) Catalytic ozonation as a function of time.

Generally, the apparent rate constants were in agreement with the reactivity of those compounds with O₃ (Table 5.2). In single ozonation, the degradation of MPs could be divided into three zones. In zone 1 including the first 12 seconds, the removal of MPs was very fast with all three MPs dropping by more than 50%. The removal started slowing down in zone 2 that extends between 12 seconds and 5 minutes before starting in zone 3 that lasted until 60 minutes. In zone 1, the measured first order rate constants in single ozonation (Figure 5.7) were comparable to those in catalytic ozonation (Figure 5.8). However, in zone 2, the reactions were faster in catalytic ozonation than in single ozonation whereby the rate constants in the former case were 6.7×10^{-2} s⁻¹, 5.7×10^{-3} s⁻¹, 6.7×10^{-3} s⁻¹ for 2,4-D, ATZ and FCZ respectively compared to 3.7×10^{-3} s⁻¹, 1.6×10^{-2} s⁻¹, $3 \times 1, 2.4 \times 10^{-3} \times 1^{-1}$ in single ozonation. Zone 3 in single ozonation was obviously the slowest. It is also worth noting that the obtained rate constants are higher than those estimated for the reaction with molecular O₃ in Table 5.4 and significantly lower than those for the reaction with 'OH predicted in Table 5.5. Therefore, the overall kinetics in single and PAC-catalyzed ozonation were

closer to the degradation kinetics under the effect of molecular O₃ only which ties with the competition between the different MPs in the mixture and their preferential degradation pathways. Also, when measured in real matrix, those rates are expected to decrease (Rozas et al., 2017; Shu et al., 2016). Guo et al. (2018) raised the issue of incomplete mixing that can occur in batch systems resulting in an overestimation of the O₃ degradation and accordingly the degradation of MPs with limited reactivity with O₃. Thus, the use of stopped-flow systems was recommended to better estimate the O₃ degradation and MPs' removal.



Figure 5.7 Degradation kinetics of MPs by single ozonation (Mixture of MPs at 0.05 mg/L each, borate buffer: 4 mM, O₃ dose: 10 mg/L).



Figure 5.8 Degradation kinetics of MPs by catalytic ozonation divided into three different zones (Mixture of MPs at 0.05 mg/L each, borate buffer: 4 mM, O₃ dose: 10 mg/L, PAC: 100 mg/L).

5.3.5. Formation of intermediates in single and PAC-catalyzed ozonation.

Literature was surveyed for the common intermediates that form during the oxidation of each of the 17 MPs present in the mixture (Table A.3 in the Appendix). To the best of our knowledge, the oxidation mechanisms of seven MPs in the mixture GFL, MCP, FCZ, DZN, CLN, PFOA and PFOS have not been studied yet. The thirteen intermediate compounds provided in Table 5.4, were detected after 10 minutes of single ozonation using 0.27 g O₃/g DOC, among which 11 compounds were also detected in catalytic ozonation.

Glutaric acid was identified during single ozonation only and it is reportedly an intermediate in the ozonation of EE2 that is generated at later stages before complete mineralization into carbon dioxide (Zhang et al., 2006). *p*-Benzoquinone was likewise formed in single and catalytic ozonation of SMX as an intermediate. In fact, oxamic, oxalic, maleic and pyruvic acids have been reported in literature as the last ozonation products that form prior to the complete mineralization and formation of carbon dioxide and inorganic ions (Goncalves et al., 2012; Zhang et al., 2006). Multiple intermediates have been identified for CBZ with C₁₃H₉NO detected in single ozonation only. The mechanism of CBZ ozonation suggested by Hubner et al. (2014) comprises C₁₃H₉NO that is formed via both direct pathway, by reaction with molecular O_3 , and indirect pathway by reaction with hydroxyl radicals ('OH) and it does not undergo further transformations. $C_{15}H_{10}N_2O_3$ could also be generated through both direct and indirect pathways and transformed by reaction with O_3 into $C_{15}H_{10}N_2O_4$. Similarly, the study implied that $C_{15}H_{12}N_2O_3$ is formed through the indirect pathway and doesn't proceed through further reactions. C₁₀H₁₀O₃ and C₉H₁₀O₂ identified in this study have also been detected as intermediates in the 60 minutes single and catalytic ozonation of NPX in presence of nickel oxide (Aguilar et al., 2019). It was suggested that direct pathway is responsible for the degradation of NPX and indirect pathway might contribute to the removal of the intermediates. C₈H₁₅N₅O was reported as an intermediate in the single ozonation of ATZ where it is formed though the substitution of Cl by OH (Li & Zhou, 2019). It was claimed that it can undergo other transformations in ozonation catalyzed by zero valent iron. C₇H₆N₂ and $C_6H_8N_2$ that were identified here can be potentially associated with the ozonation of CDM since they were also reported as by-products of the photodegradation of CDM by UV irradiation (Kiss & Virág, 2009b).

It is therefore possible that the different oxidation mechanisms occurring in single and PACcatalyzed discussed in Chapter 4, in addition to the possible adsorption onto PAC resulted in the formation of more intermediates in single ozonation. **Table 5.4** Oxidation intermediates and by-products identified in the mixture with $0.27 \text{ g O}_3/\text{g DOC}$ at 10 minutes by single and catalytic ozonation: compounds in bold were identified in single ozonation only

МР	Oxidation intermediates/by-products		Mass (m/z)	Reference
EE2	C5H8O4	Glutaric acid	133.0501	(Zhang et al., 2006)
SMX	$C_6H_4O_2$	p-Benzoquinone	109.029	(Goncalves et al., 2012)
CBZ	$C_8H_6N_2O_2$	1-(2-benzoic acid)- (1H,3H)-quinazoline-2,4-dione	163.0508	(McDowell et al., 2005)
	C14H9NO2	2-Nitroanthracene or 9-Nitroanthracene	224.0710	(Hubner et al., 2014)
	$C_{15}H_{10}N_2O_4$	Phencomycin	283.0725	
	$C_{15}H_{12}N_2O_3$	Hydrofuramide	269.0933	
	$C_{15}H_{10}N_2O_3$	Carbmazepine-o-quinone	267.0770	
	C13H9NO	Acridone	196.0767	
NPX	$C_{10}H_{10}O_3$	2-(4-(2-Hydroxyvinyl) phenyl) acetic acid	177.0554	(Aguilar et al., 2019)
	$C_9H_{10}O_2$	2-(<i>p</i> -Tolyl) acetic acid	149.0592	
ATZ	C8H15N5O	2-Hydroxy-4-ethylamino-6-isopropylamino-1,3,5-triazine	198.1355	(Li & Zhou, 2019)
CDM	C7H6N2	Benzimidazole	119.0609	(Kiss & Virág, 2009b)
	$C_6H_8N_2$	1,2-diaminobenzene	109.0766	

For a better understanding of the differences between intermediate compounds formed in single ozonation and those formed in catalytic ozonation, the corresponding peaks were integrated and their areas were reported and plotted in Figure 5.9. Assuming proportionality between the peak area and the compound's concentration, it can be noted that most of the intermediate compounds were more abundantly generated in single ozonation than in catalytic ozonation. Similar results were obtained in other studies (Goncalves et al., 2012; Gonçalves et al., 2013a; Rozas et al., 2017). Such findings were associated with the greater ability of catalytic ozonation for the mineralization of target compound(s). In this case, where ozonation was catalyzed by carbon material, the enhanced mineralization could be due to either the catalytic effect or the adsorptive capacity of the catalyst or both combined.



Figure 5.9 Comparison between the peak areas corresponding to the detected intermediates in single ozonation and in PAC-catalyzed ozonation at 10 minutes reaction time.

5.4. Conclusions

In this study, the effect of single and PAC-catalyzed ozonation were compared under different O₃ specific doses using a diverse mixture of MPs from different categories and different classes of reactivity. Overall, most MPs showed higher affinity for the removal by PAC-catalyzed ozonation which was mainly observed at higher O₃ specific dose. Under this condition more than 50% removal could be achieved for the less reactive MPs. CBZ and ATZ were determined good surrogate MPs for fast and slow reacting MPs, respectively. The removal kinetics were affected by the level of DOC and its capacity to consume the applied O₃ dose and exert an instantaneous demand. Still faster kinetics were obtained in PAC-catalyzed ozonation leaning more towards the removal kinetics with molecular O₃. Also, in this case, less intermediate compounds were formed possibly because the removal occurs through different pathways in single and PAC-catalyzed ozonation. Further studies are required for a better understanding of the degradation mechanisms of MPs in individual treatments in comparison with treatments in a mixture.

Acknowledgments

The authors acknowledge the financial support provided by the research grants from Natural Sciences and Engineering Research Council of Canada (NSERC) Collaborative Research and Development (CRD) program. The financial support of EPCOR Water Services is also acknowledged.

References

- Acero, J. L., Stemmler, K., & von Gunten, U. (2000). Degradation Kinetics of Atrazine and Its Degradation Products with Ozone and OH Radicals: A Predictive Tool for Drinking Water Treatment. Environ. Sci. & Technol., 34(4), 591-597. doi: https://doi.org/10.1021/es990724e
- Aguilar, C. M., Chairez, I., Rodríguez, J. L., Tiznado, H., Santillán, R., Arrieta, D., & Poznyak, T.
 (2019). Inhibition effect of ethanol in naproxen degradation by catalytic ozonation with NiO.
 RSC Advances 9. doi: 10.1039/c9ra02133g
- Alberta Environment & Sustainable Resource Development (ESRD). (2014). Environmental Quality Guidelines for Alberta Surface Waters. Edmonton, Alberta.
- Alexander, J. T., Hai, F. I., & Al-aboud, T. M. (2012). Chemical coagulation-based processes for trace organic contaminant removal: Current state and future potential. J. of Environ. Manag., 111, 195-207. doi: https://doi.org/10.1016/j.jenvman.2012.07.023
- Altmann, J., Massa, L., Sperlich, A., Gnirss, R., & Jekel, M. (2016). UV254 absorbance as realtime monitoring and control parameter for micropollutant removal in advanced wastewater

treatment with powdered activated carbon. Water Res., 94, 240-245. doi: http://dx.doi.org/10.1016/j.watres.2016.03.001

Arlos, M. J., Braggb, L. M., Servos, M. R., & Parker, W. J. (2014). Simulation of the fate of selected pharmaceuticals and personal care products in a highly impacted reach of a Canadian watershed. Sci. of the Total Environ., 193-204. doi: http://dx.doi.org/10.1016/j.scitotenv.2014.03.092

AWWA. (2019). Per- and Polyfluoroalkyl Substances (PFAS) Treatment.

- Bader, H., & Hoigne, J. (1981). Determination of ozone in water by the Indigo Method. Water Res., 12, 449-456.
- Beltrán, F. J., & Rey, A. (2018). Free Radical and Direct Ozone Reaction Competition to Remove Priority and Pharmaceutical Water Contaminants with Single and Hydrogen Peroxide Ozonation Systems. Ozone: Sci. & Eng., 40(4), 251-265. doi: 10.1080/01919512.2018.1431521
- Benitez, F. J., Real, F. J., Acero, J. L., & Garcia, C. (2007). Kinetics of the transformation of phenyl-urea herbicides during ozonation of natural waters: Rate constants and model predictions. Water Res., 41(18), 4073-4084. doi: https://doi.org/10.1016/j.watres.2007.05.041
- Betancur-Corredor, B., Soltan, J., & Peñuela, G. A. (2016). Mineralization of Ibuprofen and Humic Acid through Catalytic Ozonation. Ozone: Sci. & Eng., 38, 203-210. doi: 10.1080/01919512.2015.1114880

- Bourgeois, W., Burgess, J. E., & Stuetz, R. M. (2001). On-line monitoring of wastewater quality:
 a review. J. of Chem. Technol. & Biotechnol., 76, 337-348. doi: https://doi.org/10.1002/jctb.393
- Broseus, R., Vincent, S., Aboulfadl, K., Daneshvar, A., Sauve, S., Barbeau, B., & Prevost, M. (2009). Ozone oxidation of pharmaceuticals, endocrine disruptors and pesticides during drinking water treatment. Water Res., 43(18), 4707-4717. doi: 10.1016/j.watres.2009.07.031
- Buffle, M. O., Schumacher, J., Salhi, E., Jekel, M., & von Gunten, U. (2006). Measurement of the Initial Phase of Ozone Decomposition in Water and Wastewater by Means of a Continuous Quench-Flow System: Application to Disinfection and Pharmaceutical Oxidation. Water Res., 40(9), 1884-1894. doi: 10.1016/j.watres.2006.02.026
- Canadian Council of Ministers of the Environment (CCME). (2014). Canadian Water Quality Guidelines for the protection of Aquatic Life Canadian Environmental Quality Guidelines, from http://ceqg-rcqe.ccme.ca/en/index.html#void
- Chelme-Ayala, P., Gamal El-Din, M., Smith, D. W., & Adams, C. D. (2011). Oxidation kinetics of two pesticides in natural waters by ozonation and ozone combined with hydrogen peroxide. Water Res., 45(8), 2517-2526. doi: 10.1016/j.watres.2011.02.007
- Chys, M., Audenaert, W. T. M., Stapel, H., Ried, A., Wieland, A., Weemaes, M., . . . Hulle, S. W.
 H. V. (2018). Techno-economic assessment of surrogate-based real-time control and monitoring of secondary effluent ozonation at pilot scale. Chem. Eng. J., 352, 431-440. doi: https://doi.org/10.1016/j.cej.2018.07.041

- Dai, X., Xie, Z., Dorian, B., Gray, S., & Zhan, J. (2019). Comparative study of PFAS treatment by UV, UV/ozone, and fractionations with air and ozonated air. Environ. Sci. Water Res. & Technol., 5, 1897-1907. doi: 10.1039/c9ew00701f
- Dodd, M. C., Buffle, M.-O., & von Gunten, U. (2006). Oxidation of Antibacterial Molecules by Aqueous Ozone: Moiety-Specific Reaction Kinetics and Application to Ozone-Based Wastewater Treatment. Environ. Sci. & Technol., 60(6), 1969-1977. doi: https://doi.org/10.1021/es051369x
- Elovitz, M. S., & von Gunten, U. (2008). Hydroxyl Radical/Ozone Ratios During Ozonation Processes. I. The Rct Concept. Ozone Sci. & Eng., 21(3), 239-260. doi: https://doi.org/10.1080/01919519908547239
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2008). Catalytic ozonation of sulfonated aromatic compounds in the presence of activated carbon. Appl. Catal. B: Environ, 83(1-2), 150-159. doi: 10.1016/j.apcatb.2008.02.010
- Gerrity, D., Gamage, S., Jones, D., Korshin, G. V., Lee, Y., Pisarenko, A., ... Snyder, S. A. (2012).
 Development of surrogate correlation models to predict trace organic contaminant oxidation and microbial inactivation during ozonation. Water Res, 46, 6257-6272. doi: http://dx.doi.org/10.1016/j.watres.2012.08.037
- Goncalves, A. G., Orfao, J. J., & Pereira, M. F. (2012). Catalytic ozonation of sulphamethoxazole in the presence of carbon materials: catalytic performance and reaction pathways. J. Hazard. Mater, 239-240, 167-174. doi: 10.1016/j.jhazmat.2012.08.057

- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2013a). Ozonation of bezafibrate promoted
 by carbon materials. Appl. Catal. B: Environ., 140-141, 82-91. doi: 10.1016/j.apcatb.2013.03.034
- Guo, Y., Wang, H., Wang, B., Deng, S., Huang, J., Yu, G., & Wang, Y. (2018). Prediction of micropollutant abatement during homogeneous catalytic ozonation by a chemical kinetic model. Water Res., 142(383-385). doi: https://doi.org/10.1016/j.watres.2018.06.019
- Guzman-Perez, C. A., Soltan, J., & Robertson, J. (2011). Kinetics of catalytic ozonation of atrazine in the presence of activated carbon. Sep. & Purif. Technol., 79, 8-14. doi: 10.1016/j.seppur.2011.02.035
- Health Canada. (2019). Guidelines for Canadian Drinking Water Quality Summary Table. In H.E. a. C. S. B. Water and Air Quality Bureau (Ed.). Ottawa, Ontario: Health Canada.
- Huber, M. M., Canonica, S., Park, G.-Y., & vonGunten, U. (2003). Oxidation of Pharmaceuticals during Ozonation and Advanced Oxidation Processes. Environ. Sci. and Technol., 37(5). doi: https://doi.org/10.1021/es025896h
- Huber, M. M., Gobel, A., Joss, A., Hermann, N., Loffler, D., Mcardell, C. S., . . . von Gunten, U.
 (2005). Oxidation of Pharmaceuticals during Ozonation of Municipal Wastewater Effluents:
 A Pilot Study. Environ. Sci. & Technol., 39(11), 4290-4299.
- Hubner, U., Seiwert, B., Reemtsma, T., & Jekel, M. (2014). Ozonation products of carbamazepine and their removal from secondary effluents by soil aquifer treatment--indications from column experiments. Water Res., 49, 34-43. doi: 10.1016/j.watres.2013.11.016

- Javier-Benitez, F., Acero, J. L., Real, F. J., & Roman, S. (2004). Oxidation of MCPA and 2,4-d by UV Radiation, Ozone, and the Combinations UV/H2O2 and O3/H2O2. J. of Environ. Sci. & Health, Part B, B39, 393-409. doi: 10.1081/ PFC-120035925
- Jin, X., Peldszus, S., & Huck, P. M. (2012). Reaction kinetics of selected micropollutants in ozonation and advanced oxidation processes. Water Res., 46(19), 6519-6530. doi: 10.1016/j.watres.2012.09.026
- Johnson, A. C., & Williams, R. J. (2004). A Model To Estimate Influent and Effluent Concentrations of Estradiol, Estrone, and Ethinylestradiol at Sewage Treatment Works. Environ. Sci. & Technol., 38, 3649-3658. doi: 10.1021/es035342u CCC: \$27.50
- Kiss, A., & Virág, D. (2009). Photostability and photodegradation pathways of distinctive pesticides. J. Environ. Qual., 13(38), 157-163. doi: 10.2134/jeq2007.0504.
- Kolosov, P., Peyot, M.-L., & Yargeau, V. (2018). Novel materials for catalytic ozonation of wastewater for disinfection and removal of micropollutants. Sci. of the Total Environ., 1207-1218. doi: https://doi.org/10.1016/j.scitotenv.2018.07.022
- Kovalova, L., Siegrist, H., von Gunten, U., Eugster, J., Hagenbuch, M., Wittmer, A., ... McArdell,
 C. S. (2013). Elimination of micropollutants during post-treatment of hospital wastewater
 with powdered activated carbon, ozone, and UV. Environ. Sci. and Technol, 47(14), 78997908. doi: 10.1021/es400708w
- Ku, Y., Chang, J.-L., Shen, Y.-S., & Lin, S.-Y. (1998). Decomposition of diazinon in aqueous solution by ozonation. Water Res., 32(6), 1957-1963. doi: https://doi.org/10.1016/S0043-1354(97)00353-9

- Lee, Y., Gerrity, D., Lee, M., Bogeat, A. E., Salhi, E., Sujanie Gamage, S., . . . von Gunten, U. (2013). Prediction of Micropollutant Elimination during Ozonation of Municipal Wastewater Effluents: Use of Kinetic and Water Specific Information. Environ. Sci. & Technol., 47, 5872-5881. doi: dx.doi.org/10.1021/es400781r
- Lee, Y., Kovalova, L., McArdell, C. S., & von Gunten, U. (2014). Prediction of micropollutant elimination during ozonation of a hospital wastewater effluent. Water Res, 64, 134-148. doi: 10.1016/j.watres.2014.06.027
- Li, H., & Zhou, B. (2019). Degradation of atrazine by catalytic ozonation in the presence of iron scraps: performance, transformation pathway, and acute toxicity. J. of Environ. Sci. & Health, Part B, 54, 432-440. doi: https://doi.org/10.1080/03601234.2019.1574175
- Lin, A. Y.-C., S.C.Panchangam, Chang, C. Y., Hong, P. K. A., & Hsueh, H. F. (2012). Removal of perfluorooctanoic acid and perfluorooctane sulfonate via ozonation under alkaline condition. J. of Hazard. Mater., 243, 272-277. doi: http://dx.doi.org/10.1016/j.jhazmat.2012.10.029
- Ling, Y., Klemes, M. J., Xiao, L., Alsbaiee, A., Dichtel, W. R., & Helbling, D. E. (2017). Benchmarking Micropollutant Removal by Activated Carbon and Porous β-Cyclodextrin Polymers under Environmentally Relevant Scenarios. Environ. Sci. & Technol., 51, 7590-7598. doi: 10.1021/acs.est.7b00906
- Margot, J., Kienle, C., Magnet, A., Weil, M., Rossi, L., de Alencastro, L. F., . . . Barry, D. A. (2013). Treatment of micropollutants in municipal wastewater: ozone or powdered activated carbon? Sci Total Environ, 461-462, 480-498. doi: 10.1016/j.scitotenv.2013.05.034

- McDowell, D. C., Huber, M. M., Wagner, M., Gunten, U. V., & Ternes, T. A. (2005). Ozonation of Carbamazepine in Drinking Water: Identification and Kinetic Study of Major Oxidation Products. Environ. Sci. Technol., 39, 8014-8022.
- McLachlan, M. S., Zou, H., & Gouin, T. (2016). Using Benchmarking To Strengthen the Assessment of Persistence Environ. Sci. & Technol., 51(1), 4-11. doi: 10.1021/acs.est.6b03786
- Mecha, A. C., Onyango, M. S., Ochieng, A., Fourie, C. J. S., & Momba, M. N. B. (2016).
 Synergistic effect of UV-vis and solar photocatalytic ozonation on the degradation of phenol in municipal wastewater: A comparative study. J. of Catal., 341, 116-125. doi: 10.1016/j.jcat.2016.06.015
- Melero, J. A., Martinez, F., Botas, J. A., Molina, R., & Pariente, M. I. (2009). Heterogeneous catalytic wet peroxide oxidation systems for the treatment of an industrial pharmaceutical wastewater. Water Res, 43(16), 4010-4018. doi: 10.1016/j.watres.2009.04.012
- Metcalf, & Eddy. (2014). Wastewater Engineering: Treatment and Reuse (Fifth ed.). United States of America: Mc Graw Hill Education.
- Molnar, J., Agbaba, J., Dalmacija, B., Klašnja, M., Watson, M., & Kragulj, M. (2012). Effects of Ozonation and Catalytic Ozonation on the Removal of Natural Organic Matter from Groundwater. J. of Environ. Eng., 138(7), 804-808.
- Oulton, R. L., Kohn, T., & Cwiertny, D. M. (2010). Pharmaceuticals and personal care products in effluent matrices: A survey of transformation and removal during wastewater treatment and implications for wastewater management. J. of Environ. Monit. (11), 1956-1978. doi: 10.1039/c0em00068j.

- Razavi, B., Song, W., Cooper, W. J., Greaves, J., & Jeong, J. (2009). Free-Radical-Induced Oxidative and Reductive Degradation of Fibrate Pharmaceuticals: Kinetic Studies and Degradation Mechanisms. J. of Phys. Chem., 113, 1287-1294. doi: https://doi.org/10.1021/jp808057c
- Real, F. J., Benitez, F. J., Acero, J. L., & Gonzalez, M. (2007). Removal of diazinon by various advanced oxidation processes. J. of Chem. Technol. & Biotechnol., 82, 566-574. doi: https://doi.org/10.1002/jctb.1702
- Rizzo, L., Malato, S., Antakyali, D., Beretsou, V. G., Đolić, M. B., Wolfgang Gernjak, ... Fatta-Kassinos, D. (2019). Consolidated vs new advanced treatment methods for the removal of contaminants of emerging concern from urban wastewater. Sci. of the Total Environ, 655, 986-1008. doi: https://doi.org/10.1016/j.scitotenv.2018.11.265
- Rozas, O., Baeza, C., Nunez, K., Rossner, A., Urrutia, R., & Mansilla, H. D. (2017). Organic micropollutants (OMPs) oxidation by ozone: Effect of activated carbon on toxicity abatement. Sci. of the Total Environ. 590-591, 430-439. doi: 10.1016/j.scitotenv.2016.12.120
- Shu, Z., Singh, A., Klamerth, N., McPhedran, K., Bolton, J. R., Belosevic, M., & Gamal-El-Din,
 M. (2016). Pilot-Scale UV/H₂O₂ Advanced Oxidation Process for Municipal Reuse Water:
 Assessing Micropollutant Degradation and Estrogenic Impacts on Goldfish (*Carassius auratus L.*). Water Res, 101, 157-166. doi:
 http://dx.doi.org/10.1016/j.watres.2016.05.0790043-1354/©2016

- Song, Y., Breider, F., Ma, J., & von Gunten, U. (2017). Nitrate formation during ozonation as a surrogate parameter for abatement of micropollutants and the N-nitrosodimethylamine (NDMA) formation potential. Water Res., 122, 246-257. doi: 10.1016/j.watres.2017.05.074
- Stamm, C., Joss, A., Ackermann, M., & R.I.L.Eggen. (2016). Unravelling the Impacts of Micropollutants in Aquatic Ecosystems: Interdisciplinary Studies at the Interface of Large-Scale Ecology Adv. in Ecol. Res. (Vol. 55, pp. 183-223). doi: http://dx.doi.org/10.1016/bs.aecr.2016.07.002183
- Suarez, S., Dodd, M. C., Omil, F., & vonGunten, U. (2007). Kinetics of triclosan oxidation by aqueous ozone and consequent loss of antibacterial activity: Relevance to municipal wastewater ozonation. Water Res., 41, 2481-2490. doi: https://doi.org/10.1016/j.watres.2007.02.049
- Sui, M., Xing, S., Sheng, L., Huang, S., & Guo, H. (2012). Heterogeneous catalytic ozonation of ciprofloxacin in water with carbon nanotube supported manganese oxides as catalyst. J. Hazard. Mater., 227-228, 227-236. doi: 10.1016/j.jhazmat.2012.05.039
- Sui, Q., Huang, J., Deng, S., Yu, G., & Fan, Q. (2010). Occurrence and removal of pharmaceuticals, caffeine and DEET in wastewater treatment plants of Beijing, China. Water Res., 44(2), 417-426. doi: 10.1016/j.watres.2009.07.010
- Tondera, K., Klaer, K., Gebhardt, J., Wingender, J., Koch, C., Horstkott, M., . . . Pinnekamp, J. (2015). Reducing pathogens in combined sewer overflows using ozonation or UV irradiation. Int. J. Hyg. Environ. Health., 218(8), 731-741. doi: 10.1016/j.ijheh.2015.09.002
- Trojanowicza, M., Bojanowska-Czajkaa, A., Bartosiewicza, I., & Kulisaa, K. (2018). Advanced Oxidation/Reduction Processes treatment for aqueous perfluorooctanoate (PFOA) and

perfluorooctanesulfonate (PFOS) – A review of recent advances. Chem. Eng. J., 336, 170-199. doi: https://doi.org/10.1016/j.cej.2017.10.153

- Uslu, M., Seth, R., Jasim, S., Tabe, S., & Biswas, N. (2015). Reaction Kinetics of Ozone with Selected Pharmaceuticals and Their Removal Potential from a Secondary Treated Municipal Wastewater Effluent in the Great Lakes Basin. Ozone: Sci. & Eng., 37, 36-44. doi: 10.1080/01919512.2014.929520
- Wert, E. C., L.Rosario-Ortiz, F., & Snyder, S. A. (2009). Effect of ozone exposure on the oxidation of trace organic contaminants in wastewater. Water Res., 43(4), 1005-1014. doi: 10.1016/j.watres.2008.11.050
- Westerhoff, P., Yoon, Y., Snyder, S., & Wert, E. (2005). Fate of Endocrine-Disruptor, Pharmaceutical, and Personal Care Product Chemicals during Simulated Drinking Water Treatment Processes. Envrion. Sci. & Technol, 39, 6649-6663. doi: 10.1021/es0484799
- Xiong, F., & Graham, N. J. D. (2008). Rate Constants for Herbicide Degradation by Ozone. Ozone
 Sci. & Eng., 14(4), 283-301. doi: 10.1080/01919519208552274
- Yu, J., & Hu, J. (2011). Adsorption of Perfluorinated Compounds onto Activated Carbon and Activated Sludge. J.of Environ. Eng., 137(10), 945-951. doi: 10.1061/(asce)ee.1943-7870.0000402
- Zhang, K., Deletic, A., Page, D., & McCarthy, D. T. (2015). Surrogates for herbicide removal in stormwater biofilters. Water Res., 81, 64-71. doi: http://dx.doi.org/10.1016/j.watres.2015.05.043
- Zhang, X., Chen, P., Wu, F., Deng, N., Liu, J., & Fang, T. (2006). Degradation of 17 aethinylestradiol in aqueous solution by ozonation. J. Hazard. Mater. B, 291-298.
Zimmermann, S. G., Wittenwiler, M., Hollender, J., Krauss, M., Ort, C., Siegrist, H., & von Gunten, U. (2011). Kinetic assessment and modeling of an ozonation step for full-scale municipal wastewater treatment: micropollutant oxidation, by-product formation and disinfection. Water Res., 45(2), 605-617. doi: 10.1016/j.watres.2010.07.080

CHAPTER 6: Advancing the Treatment of Primary Influent and Effluent Wastewater during Wet Weather Flow by Single versus Powdered Activated Carbon-catalyzed Ozonation for Micropollutants removal, Disinfection and Toxicity Abatement⁴

6.1. Introduction

Combined sewer overflows (CSO) contain storm water runoffs along with domestic and industrial wastewater and toxic materials and debris (USEPA, 2018). Their direct release to the environment promotes a cumulative increase in *Escherichia coli* (E.coli) in the aquatic system by up to 2 log (Madoux-Humery et al., 2016). There have also been several studies on the contribution of CSO to the release of micropollutants (MPs) into receiving waters (Evans et al., 2016; Gasperi et al., 2008; Madoux-Humery et al., 2016; Phillips et al., 2012). While these problems persist, several treatment options surfaced in order to prevent the discharge of untreated wastewater during wet weather conditions. In a first step, the enhanced primary treatment consisting of coagulation/ flocculation/ sedimentation has been investigated (Alameddine et al., 2020). Then, different solutions for disinfection have been explored including: ultraviolet irradiation (UV) (Tondera et al., 2015; Wojtenko et al., 2001a), performic acid (Chhetri et al., 2015; Chhetri et al., 2014; Tondera et al., 2016), peracetic acid (Chhetri et al., 2016; Chhetri et al., 2014; Coyle et al., 2014) and ozone (O₃) (Tondera et al., 2015; Wojtenko et al., 2001b). In comparison, studies on the removal of MPs from the primary influent (PI) during wet weather have been relatively scarce (Jung et al., 2015; Pongmala et al., 2015). Therefore, there is a need for an efficient treatment that can address both disinfection and MPs removal and partake in the enhanced primary treatment of PI during wet weather conditions.

⁴ A version of this chapter will be submitted jointly with a version of chapter 5 to Journal of Hazardous Materials.

Ozonation is one widely recognized technique for its capacity to both disinfect and oxidize (Altmann et al., 2014). The application of ozone (O_3) is often favored because O_3 has strong and fast disinfection abilities that cover a wide range of microorganisms, and tends to form less regulated disinfection by-products (DBPs) (Cho et al., 2002; Rodayan et al., 2014). In the literature, ozonation was widely applied following secondary treatment, and was deemed efficient for the removal of a variety of MPs such as triclosan, gemfibrozil, atenolol, sulfamethoxazole, diclofenac, carbamazepine, among others (Altmann et al., 2014; Can & Cakır, 2010; Huber et al., 2005; Ikehata et al., 2008; Jin et al., 2016; Lee et al., 2013; Lee et al., 2014b; Nothe et al., 2009; Wildhaber et al., 2015; Zhou et al., 2015). Ozonation was also previously tested on CSO samples after settling where it achieved good pathogens removal and significant reductions in quality parameters such as turbidity, dissolved organic carbon (DOC), total organic carbon (TOC) and total suspended solids (TSS), but there was no data reported on its effect on MPs removal from CSO (Tondera et al., 2015). The common ranges of applied O₃ doses for the treatment of secondary effluent were reported as O₃ concentrations: 5 - 15 mg O₃/L (Nothe et al., 2009) or O₃ specific doses: 0.21 - 1.53 mg O₃/mg DOC (Song et al., 2017; Zimmermann et al., 2011). Doses outside that range such as 2.24 mg O₃/mg DOC were also studied (Jin et al., 2016). It was alleged that ozonation exhibits better energy efficiency and higher capacity to target even the most recalcitrant compounds compared to other techniques such as UV or UV/H₂O₂ (Kovalova et al., 2013; von Sonntag & von Gunten, 2012). Nevertheless, there are several concerns commonly brought forward when discussing O₃ for the treatment of wastewater in general and CSO in particular. In real life applications, ozonation is associated with several limitations pertaining to its low stability and the need to manufacture it *in-situ* and therefore safety concerns and maintenance costs that can be entailed (Wojtenko et al., 2001a). Moreover, the complexity of a wastewater matrix entails the presence of O_3 sinks such as: nitrites, nitrates, sulphites, iron and manganese that can hinder the O₃ available for disinfection and oxidation (Lee et al., 2013). Similarly, the higher the content of CSO in TSS, COD and TOC, the higher the O₃ demand. Natural organic matter (NOM), pH and temperature can also affect the O₃ decomposition rate (Wojtenko et al., 2001b). These factors are thus to be accounted for when adjusting the O₃ dosage. Likewise, ozonation affects the chemical structure and composition of wastewater, and can promote the formation of by-products and intermediates (carboxylic acids, aldehydes, bromates, ketone, peroxides) that might be more toxic than the parent compound (Jin et al., 2016; Wojtenko et al., 2001a).

In order to mitigate these limitations, recent trends in O₃-based advanced oxidation processes (AOPs) suggest the application of catalysts to improve the generation of radicals and thus the removal of compounds that are less reactive with molecular O₃. Single ozonation and other AOPs are likely applied as pre-treatments to partially oxidize organic compounds and hence improve their biodegradation, and most commonly as tertiary treatments on low organics matrices (Ikehata et al., 2006). Among the various types of catalysts, carbon-based catalysts can be the most practical in wastewater treatment because they can be used for adsorption, they are relatively affordable, and do not leave metal residues in the treated media. There is a variety of carbon-based catalysts including multi-walled carbon nanotubes (Fan et al., 2014; Gonçalves et al., 2010; Gonçalves et al., 2013b; Liu et al., 2011; Oulton et al., 2015; Restivo et al., 2012), carbon nanofibers (Restivo et al., 2013), and activated carbon (AC). The latter whether in powdered or granular form (as powdered activated carbon PAC or granular activated carbon GAC), has the advantage of very high adsorption capacity due to its small particle size. Therefore, applying it as a catalyst in the wastewater treatment industry can secure dual roles: catalyzing the oxidation of target compounds and removing them and/or their by-products by adsorption. Since the oxidation process entails the chemical transformation of one compound into another, while adsorption involves their physical removal, the synergy between the two processes can be promising for enhanced removal and

lowered toxicity. However, the role of PAC in catalytic ozonation can be deterred by the dissolved organic matter (DOM) which compete with target pollutants for the adsorption site on PAC affecting the reaction with molecular O_3 and the radicals, which results in the need for increasing its concentration (Margot et al., 2013). In secondary treatment, the common specific PAC dose range is 2 – 3 g PAC/g DOC (Rizzo et al., 2019). To the best of our knowledge, PAC-catalyzed ozonation has not been studied yet for the treatment of PI during wet weather conditions.

This study aimed primarily to test the applicability of single and PAC-catalyzed ozonation to the treatment PI during wet weather through two possible treatment trains. A dose screening was initially performed followed by an evaluation of the impact of both treatment trains on conventional contaminants, micropollutants, and disinfection. The safety of the discharge of the effluent was appraised through acute toxicity, genotoxicity and Yeast Estrogen Screen (YES) bioassays. Finally a detailed cost analysis was performed to predict the applicability of the proposed ozonation processes.

6.2. Material and methods

6.2.1. Sample collection and preparation

During the first stage of this study, samples of PI were collected from a wastewater treatment plant in Alberta over a six months period, extending between April and November 2018. Those samples were acquired before the primary clarifier, after grit removal and were used for screening the effect of adsorption and single ozonation on the removal of major contaminants. During the second stage, another sampling campaign was conducted in March 2020 where three types of wastewater samples were acquired: PI, secondary effluent before UV treatment and final effluent after UV treatment. They were used to compare the effect of PAC-catalyzed ozonation for the treatment of PI during wet weather. All samples were collected in plastic pails that were soaked in 20% bleach overnight and then thoroughly rinsed with deionized water. Samples were transferred to the laboratory and characterized for pH, *E. coli* and DOC on the same day and PI samples were immediately used in the ozonation experiments. It is worth noting that in order to emulate wet weather conditions, the acquired PI samples were uniformly diluted to 50% either with deionized (DI) water (to represent untreated PI during wet weather) or directly in the reactor during treatment (with DI water and O_3 stock solution) as described later in the experimental procedure. All samples were stored at 4 °C in the dark until analysis completion.

6.2.2. Experimental procedures

Initially, a set of ozonation experiments was performed on samples of diluted PI at different O₃ specific doses to determine the optimum dose to be applied in the proposed treatment trains. Another set of adsorption experiments with different PAC doses was also conducted as a control to screen the effect of PAC on the removal of DOC and COD.

During the second stage of the study, two types of experiments were performed at bench scale level to simulate two possible options for applying ozonation during the enhanced treatment of the PI during wet weather conditions (Figure 6.1). The first option, hereafter referred to as simultaneous treatment (ST), consisted of applying single ozonation (SO) or catalytic ozonation (CO) during the coagulation/ flocculation/ sedimentation (CFS) process that takes place in the primary clarifier. In order to address this treatment option, jar tests were carried in a B-KER² Jar Test apparatus with 3-inch wide Phipps & Bird stirrers. In each jar, 75 mg/L of the coagulant aluminum sulfate was simultaneously added with O₃ stock solution with or without PAC such as the volume of O₃ stock solution added did not exceed 30% of the total volume. One minute of rapid mixing was initiated at 150 rotations per minute (rpm) (equivalent to a Gt value of 9300)

while all jars were covered with aluminum foil to minimize the volatilization of O_3 . Slow mixing followed for 10 minutes at 15 rpm (equivalent to a Gt value of 6000) after which the stirrers were carefully removed to avoid any media disturbance. Settling was then allowed for 1 hour after which the supernatant was carefully collected with a syringe immersed at 2 inches below the surface. The acquired samples had no residual O_3 and were immediately tested for *E. coli* and then stored at 4 °C in the dark for further analysis. All experiments were performed in duplicates at room temperature (20 °C).



Figure 6.1 Schematic representation of the performed single and catalytic ozonation experiments as simultaneous and post treatments.

The second option entailed applying single or catalytic ozonation on the CFS effluent before discharge into the river. Although such treatment is conventionally performed in a contact chamber, having this addition to a wastewater treatment plant might not be always feasible. This is not only because of space and cost limitations, but also because it will result in an increase in the overall retention time which does not coincide with the high influent flow rates, a major characteristic of the CSO. Instead, ozonation is suggested as a treatment in the pipe for the CFS effluent to improve its quality before discharge. This treatment will be referred to as post treatment (PT). At bench-scale level, PT was performed following a full jar test using 75 mg/L of aluminum sulfate with rapid mixing (1 minute at 150 rpm), slow mixing (10 minute at 15 rpm) and settling (1 hour). The supernatant was then collected and the required volume was transferred to a 250 mL round bottom flask reactor. O_3 and PAC were added from their respective saturated stock solutions and the reactor was immediately capped with no free headspace. The reactor was stirred with a magnetic stirrer over a multi-position stir plate. After 1 minute, residual O₃ was measured and sodium thiosulfate (2mM) was added to the reactor to stop any reaction with residual O₃. All experiments were performed in duplicates, at room temperature (20 °C). A short exposure time to the oxidant and catalyst was selected in order to emulate an in-pipe treatment whereby the travel time between the primary clarifier and the point of discharge was estimated between 0.6 and 3 minutes for a flow rate ranging between 600 mega liters per day (MLD) and 100 MLD.

6.2.3. Wastewater spiking with MPs

A similar set of simultaneous and post experiments was conducted after spiking the PI that was diluted to emulate wet weather conditions with carbamazepine (CBZ) and atrazine (ATZ) to achieve a final concentration of 2 mg/L of each MP. These experiments were performed to validate the efficiency of single and PAC-catalyzed ozonation for the degradation of different types of MPs when applied for the chemical treatment of PI during wet weather. CBZ and ATZ were selected to represent organic MPs that are highly reactive with O₃ and those with low reactivity: CBZ is rapidly degraded by molecular O₃ ($k_{O3/CBZ}$ = 3×10⁵ M⁻¹s⁻¹, (Huber et al., 2003)) while ATZ is not $(k_{O3/ATZ} = 6 \text{ M}^{-1}\text{s}^{-1})$, (Acero et al., 2000)). Also, those MPs were determined good surrogates for other MPs as discussed in section 5.3.2, Chapter 5. Both MPs were weighed and dissolved directly in a small volume of PI before transferring them to the dedicated PI sample.

6.2.4. Chemicals

All the used chemicals, including sodium indigo trisulfonate, sodium thiosulfate, CBZ and ATZ were acquired from Sigma Aldrich, Canada. O₃ was generated from dry oxygen using the O₃ Generator Model GSO-30. It was bubbled into O₃ demand-free water (ODFW) in an ice bath to obtain a saturated stock solution. Commercial PAC WPX with a particle size of 45 µm was acquired from Calgon Carbon. It has a measured surface area of 632 m²g⁻¹and its full characterization is available in section 4.3.1, Chapter 4. PAC was added from a stock solution of 6 g/L in ODFW. Microtox[®] reagents were purchased from MODERNWATER. SOS ChromoTest test kit for genotoxicity was obtained from Environmental Bio-detection Products Inc. Canada. XenoScreen XL YES test kit for YES bioassays was procured by XENOMETRIX, Switzerland.

6.2.5. Analytical methods

pH and turbidity were determined using Accumet Research AR20 pH/conductivity meter (Fischer-Scientific) and T-100 handheld Oakton Turbidity meter, respectively. UV absorbance (UVA) was measured at 254 nm using Thermo Scientific GENESYS 10S UV-Vis Spectrophotometer in 1-cm path length quartz cells using samples that were filtered with 0.45 μm nylon syringe filters. The %UV Transmittance (%UVT) was calculated using Equation 6.1. DOC was measured using a high temperature combustion TOC analyzer (TOC-L analyzers, Shimadzu), as per Standard Method 5310-B (APHA, 2017). In the first stage of the study, the total COD was measured for unfiltered samples following the Standard Method 5220D; and *E. coli* was analyzed using the Membrane Filter Technique, Standard Method 9222 and reported as CFUs/100 mL (colony forming units). During the second stage, COD was measured in filtered samples with 0.45 µm nylon filtered and analyzed as COD dissolved COD_d using HACH method 8000. *E. coli* was quantified based on the most probable number as MPN/100 mL using Colilert-18 method from IDEXX to better accommodate for a prompt analysis of the different samples of treated and untreated wastewater. It should be noted that 1 MPN/100 mL is equivalent to 1 CFU/100 mL according to IDEXX official website (Leonard, 2020). Phosphate anions were quantified using Dionex ICS Ion Chromatography.

MPs were analyzed using ultra-performance liquid chromatography – quadrupole time-offlight mass spectrometry (Xevo G2-S, Waters), operated in positive mode. Chromatographic separation was achieved using ACQUITY UPLC BEH C18, 50×2.1 mm column, at 50 °C with an injection volume of 10 µL. The mobile phase consisted of water with 0.1 % formic acid (solvent A) and acetonitrile with 0.1 % formic acid (solvent B).

The instantaneous O₃ demand (IOD) for untreated and coagulated PI during wet weather was measured using Standard Method 2350 D (APHA, 2017). The residual O₃ in all single and catalytic ozonation experiments was measured following the Indigo method (Bader & Hoigne, 1981).

6.2.6. Toxicity tests

The acute toxicity of untreated and treated wastewater samples was assessed via standard Microtox[®] assays with the bacterial reagent *Vibrio fischeri*. Bioluminescence inhibition tests were undertook in duplicates in 96-well plates. Samples were transferred to the wells and an adequate volume of osmotic adjustment solution was added to adjust their salinity. In another 96-well plate equal amounts of bacterial solutions were transferred into each well and the luminescence was measured using Synergy Microplate reader. Samples were added to the bacteria-containing 96-well plate, which was then incubated at 15 °C for 5 minutes after which the luminescence was measured again. The inhibition of light emission was then calculated.

The potential carcinogenic effect of wastewater samples was evaluated through SOS-ChromoTest, a colorimetric method used to measure the relative strength of a potential genotoxic compound. Upon exposure to a genotoxic agent, *E. coli* PQ37, a modified strain of *E. coli* uses SOS gene repair complex to repair the DNA damage and synthesizes β -galactosidase, an enzyme that reacts with blue chromogen. In 96-well plates, different serial dilutions were prepared for each tested sample with dimethyl sulfoxide (DMSO). *E. coli* PQ37 was then added to each well and the plates were incubated at 37 °C for 2 hours after which blue chromogen was added. The absorbance was measured at 420 nm and 600 nm using Synergy Microplate reader. The plates were incubated again at 37 °C for 90 minutes and the final absorbance was measured at 420 nm and 600 nm. The cytotoxic effect of the samples was calculated through the bacteria survival rate in Equation 6.2. The genotoxic effect was calculated using the growth factor G in Equation 6.3, the β -galactosidase activity β -gal in Equation 6.4, and the induction ratio IF in Equation 6.5 (SOS-ChromoTest, 2019). IF ≥ 1.2 was the criteria used to declare a sample genotoxic (Kocak, 2015).

% Survival rate =
$$100 \times \frac{OD_{420}}{OD_{420 c}}$$

 $G = \frac{A_{420} - A_{420 b}}{A_{420 c} - A_{420 b}}$
 $\beta - gal = \frac{A_{600} - A_{600 b}}{A_{600 c} - A_{600 b}}$
 $IF = \frac{\beta - gal}{G}$
 $Equation 6.5$

Where: OD_{420} and $OD_{420 c}$ represent the optical density at 420 nm of the sample and negative control respectively; A_{420} , $A_{420 c}$, $A_{420 c}$ represent the absorbance at 420 nm of the sample, blank and negative control, respectively; A_{600} , $A_{600 c}$ represent the absorbance at 600 nm of the sample, blank and negative control, respectively.

YES bioassays were also conducted on all samples to measure their hormonal activity on estrogen receptors. The Yeast cell contains an estrogen receptor gene that binds to estrogen and gets activated. Then it binds to an estrogen responsive element that contains a reporter gene responsible for the synthesis of β -galactosidase. The latter enzyme turns the yellow chlorophenol red β -D-galactopyranoside into purple or red. A 96 well assay plate was prepared with samples and controls added and serial dilutions performed using DMSO, Yeast culture was prepared and diluted according to its measured OD₆₉₀ and then distributed to the sample plate and incubated at 31 °C and 100 rpm for 18 hours after which the OD₆₉₀ was measured again. A known volume was then transferred to a fresh plate containing lysis buffer. The fresh plate was then incubated at 31 °C and 100 rpm for up to 1 hour after which OD₆₉₀ and OD₅₇₀ were measured and used to calculate the % induction and growth factor. 10% Induction was the criteria used to declare the samples as exhibiting an estrogenic toxicity.

6.3. Results and discussion

6.3.1. Wastewater characterization and IOD

Table 6.1 summarizes the characteristics of the PI samples used to screen the effect of O_3 and PAC specific doses on the removal of conventional contaminants. IOD was measured for the PI before and after coagulation with 30 seconds exposure time in order to assess the minimum required O_3 dose for the treatment. With 15 mg/L O_3 and 14 mg/L O_3 as applied O_3 doses, the IOD was 9.69 ±1.19 mg O_3/L for raw PI during wet weather and it dropped by around 36% to 6.21 ±0.68 mg O_3/L after CFS. It is crucial to note that this IOD is dose specific, may not be extrapolated to other doses, and is expected to vary as the quality of PI varies (APHA, 2017). However, it can still provide an estimate for the O_3 doses required in ST and PT.

Table 6.1 Average characteristics of primary influent samples used in the first stage of the study, after dilution with deionized water (50:50) to simulate primary influent during wet weather conditions.

Measured Parameter	PI during wet weather flow* (± STDEV)		
рН	7.3	(±0.1)	
Turbidity (NTU)	66.87	(±3.62)	
%UVT	52	(±9)	
COD total (mg O/L)	219.3	(±37.4)	
DOC (mg C/L)	25.10	(±5.91)	
<i>E. coli</i> (CFUs/100 mL)	2.16×10^{6}	$(\pm 0.47 \times 10^6)$	

*Average of five sampling campaigns

Table 6.2 describes the characteristics of the PI acquired for the second set of experiments and used in the performance assessment of single and catalytic ozonation in ST and PT. It also shows the effect of spiking the PI during wet weather with CBZ and ATZ in terms of increase in COD_d and DOC. Table 6.2 also compares PI during wet weather to secondary and final effluents in a conventional treatment train. Secondary and tertiary treatments removed 94% of COD_t and 80% and 82% of the DOC, respectively.

Table 6.2 Characteristics of the primary influent (PI) used in the performance assessment of single and PAC-catalyzed ozonation as simultaneous or post treatments in comparison with the characteristics of secondary and final effluents.

Measured Parameter	PI during w	vet weather flow	Secondary Effluent	Final Effluent
	Before spiking	After spiking with CBZ and ATZ	Before UV treatment	After UV treatment
рН	7.1 (±0.1)	7.2 (±0.0)	7.1	7.1
Turbidity (NTU)	88.75 (±2.85)	93.20 (±5.80)	0.60	1.44
%UVT	63 (±0.5)	52 (±0.3)	64	64
COD dissolved (mg O/L)	56 (±2)	76 (±0)	N.M.*	N.M.
COD total (mg O/L)	219	N.M.	27 (±4)	29 (±1)
DOC (mg C/L)	24.52	27.63	10.09	8.87
Phosphate (mg PO ₄ ^{3-/} L)	6.37 (±0.01)	6.37 (±0.01)	< 0.05	< 0.05
<i>E.coli</i> (MPN/100 mL)	87×10^{4}	87×10^4	24.2×10^{4}	0

* N.M = Not measured; CBZ: carbamazepine; ATZ: atrazine.

6.3.2. Dose screening for O_3 and PAC

An initial screening of the effect of O_3 and PAC doses on the removal of conventional macropollutants was conducted on samples of diluted PI to simulate PI during wet weather, during CFS (i.e. ST) and after CFS (i.e. PT). Only low O_3 specific doses were tested in order to explore the effect of minimizing the chemical addition. In simultaneous treatment (Figure 6.2A), the small increment in the O_3 specific dose did not show any significant effect on the removal of turbidity and total COD and on the %UVT beyond the effect of CFS only. However, a negative effect was observed on DOC removal as 0.5 mg O_3 /mg DOC increased the final DOC compared to CFS. Similar results were found elsewhere (Tondera et al., 2015) where increasing the O_3 dose to 10 mg/L increased the DOC concentration by 27% and a higher O_3 dose of 15 mg/L increased it further by 41% compared to the influent untreated wastewater (7.8 ±5.4 mg/L COD). In fact, Chandrakanth and Amy (1996) alleged that it is possible that O_3 oxidized NOM that is adsorbed onto the suspended colloids, increasing the acidic functional groups and rendering them more hydrophilic. As a result, NOM desorbed into the aqueous phase and was measured as DOC. It is

also possible that O₃ oxidized NOM in the aqueous phase into a lower molecular weight matter resulting in an increased affinity for adsorption onto the colloids. The authors observed that at lower O₃ doses, the adsorption of aqueous NOM onto the colloid either balanced or exceeded its desorption; while at higher O₃ doses, which was the case observed in Figure 6.2A desorption exceeded adsorption. In PT (Figure 6.2B), the O₃ specific dose showed a similar effect on turbidity and total COD; however, there was a substantial improvement in the removal of DOC and an increase in the %UVT. As such 0.5 mg O₃/mg DOC achieved 72% removal of DOC when applied after CFS, which was almost 60% improvement from CFS only. Accordingly, the effect of the O₃ specific dose on DOC removal in PT opposed that in ST because of the dissimilarities between the two targeted media. In PT, O₃ was applied after the removal of suspended matter which means that the applied O_3 was not utilized to breakdown large particles instead it targeted the oxidation of dissolved matter. This enhancement in DOC removal was aligned with an increase in %UVT since the latter is an indicator of the matrix content in organic matter and more specifically in aromatic carbon C=C. Those compounds absorb UV light at 254 nm in a proportional manner to their concentration (APHA, 2017; Chow et al., 2008). An O₃ specific dose of 0.5 g O₃/g DOC was thus selected for further investigation.



Figure 6.2 Effect of O_3 dose on the removal of turbidity, COD and DOC after 30 minutes of single ozonation as **(A)** Simultaneous Treatment, and **(B)** Post Treatment (a dose of 0 g O_3 /g DOC corresponds to CFS only).

Similarly, PAC doses up to 500 mg/L were tested in simultaneous and post-treatments for the removal of COD and DOC where they showed no significant effect, except some improvement in DOC removal at higher PAC doses (Figure 6.3) after 30 minutes. Actually, a typical contact time for PAC application ranges between 18 minutes and 3 hours (Rizzo et al., 2019). It is possible that when PAC is added at low concentrations in ST, it aggregates with alum and thus loses its efficiency, while at higher doses it acted as coagulant aid (Szlachta & Adamsk, 2009; USEPA, 2020). A dose of 100 mg/L was chosen to catalyze the ozonation in the next set of experiments because this dose did not have the effect of coagulant aid and was proven to catalyze the degradation of O₃ and improve removals when applied with O₃ specific doses between 0.2 and 2.2 g O₃/g DOC (Chapter 5). Also, this dose matches the common specific PAC dose range, which is 2 - 3 g PAC/g DOC as reported in literature for application in secondary treatment (Rizzo et al., 2019), and it did not increase the final TSS to more than 30 mg/L, the desired value for primary effluent discharge.



Figure 6.3 Effect of PAC dose on the removal of COD and DOC after 30 minutes of adsorption as a control in (A) Simultaneous Treatment and (B) Post Treatment (a dose of 0 mg/L PAC corresponds to CFS only).

6.3.3. Simultaneous versus post treatment of PI during wet weather by single and PAC-catalyzed ozonation

A fixed O₃ specific dose of 0.5 mg O₃/mg DOC was selected to assess the performance of single and catalytic ozonation with 100 mg/L PAC. 0.5 mg O₃/mg DOC corresponded to 14 mg O₃/L in ST and 5.5 mg O₃/L in PT. In the former case, there was no residual O₃ at the time of sample collection (i.e. after 1 hour and 11 minutes). However, in the latter case, where the contact time was set to 1 minute only to simulate an in-pipe treatment, the residual O₃ was 2.98 ± 0.12 mg/L O₃ that corresponded to almost 45% of the applied O₃ dose.

Figure 6.4 summarizes the effect of each treatment on the removal of key wastewater quality monitoring parameters. In ST, single ozonation unveiled a negative impact on the removal of turbidity, COD and DOC while catalytic ozonation did not entail any significant effect past the removal by CFS. This result can be associated with the mechanism of oxidation of the organic matter in wastewater by O_3 . Whereby O_3 targets specific sites in the hydrophobic fraction of the wastewater through substitution or oxidation-reduction, leading to their cleavage into less hydrophobic compounds (Jin et al., 2016) thus an increase in turbidity and in the soluble portion of the organic material measured by DOC. These observations were slightly attenuated in the presence of PAC by virtue of its role as adsorbent and catalyst. Our results showed that even a PAC dose as high as 250 mg/L did not increase the TSS past the desirable limit in the primary effluent (30 mg/L), probably because PAC was expected to settle during the 1 hour settling time. It is possible that the observed improvement in %UVT upon treatment was due to the mechanism of oxidation through cycloaddition where the electrophilic site in the O₃ molecule reacted with the unsaturated target molecules and broke the double bond C=C, thus decreasing the UV absorbance at 254 nm (Beltran, 2004; Chow et al., 2008). Nevertheless, despite the effect on COD_d and DOC,



the main benefit from applying single or catalytic ozonation in ST resides in the improved disinfection to almost 3 log removal of *E. coli*.

Figure 6.4 Comparison between Simultaneous and Post treatments for the removal of turbidity, COD_d , DOC, *E. coli* and enhancement of %UVT during the treatment of PI during wet weather flow (O₃ specific dose: 0.5 g O₃/g DOC; PAC dose: 100 mg/L). Results correspond to samples collected after 1 hour settling in ST and 1 minute treatment in PT.

In PT, PAC-catalyzed ozonation increased the turbidity significantly which was expected as one disadvantage for suspending PAC in the CFS effluent. It was, however, more efficient for the removal of COD_d and DOC compared to single ozonation which indicated that PAC-catalyzed ozonation enhanced the degree of mineralization. The role of PAC as a strong adsorbent could also be contributing to these results. As such, single ozonation removed 22% and 11% of COD_d and DOC, respectively, while catalytic ozonation removed 43% of COD_d and 27% of DOC. These results are better than those reported by Lei et al. (2007) who combined O₃ and granular activated carbon (GAC) for the treatment of industrial wastewater from a pharmacy plant. O₃ was continuously bubbled in a column reactor to which GAC (60 g/L) and wastewater with adjusted pH were added. Under basic conditions (pH 10), after 1 minute of treatment, there were almost 31% and 0% COD removal by O₃+GAC and O₃ only, respectively; while after 1 hour treatment, the COD removal attained 91% with O₃+GAC and 29% O₃ only. Moreover, there was at least 99% removal of phosphate by CFS only, which eliminates the risk of eutrophication in the receiving natural water (Yamashita & Yamamoto-Ikemoto, 2014).

In PT, both single and PAC-catalyzed ozonation were capable of enhancing the disinfection realizing up to 4 log removal. The initial E. coli concentration was in agreement with ranges reported in the literature: $1 \times 10^5 - 1 \times 10^7$ (Tondera et al., 2015), $1 \times 10^6 - 8 \times 10^6$ MPN/100 mL (Tondera et al., 2013), $2 \times 10^3 - 1 \times 10^7$ MPN/100 mL (Christoffels et al., 2014). However, it is interesting that within 1 minute only of PT, single and catalytic ozonation achieved 4 log removal of E. coli. In fact, knowing that within this contact time 55% of the applied O₃ dose was utilized, it is anticipated that a longer exposure time could achieve even better disinfection. In comparison, Tondera et al. (2015) achieved 3.4 log removal with O₃ doses between 8.4-12.1 mg/L O₃ (equivalent to 1.08 -1.55 g O₃/g DOC) and 15 minutes exposure time. Also, some suggested that a 2 minutes O₃ exposure time is enough to achieve efficient removal of fecal coliform from wastewater (Xu et al., 2002). It was reported that due to the specificity of this wastewater in terms of its high volume and flow rate, an improved disinfection can be achieved using higher disinfectant dose and/or mixing intensity (Wojtenko et al., 2001b). Overall, the quality of the PT effluent was better than that of the secondary effluent for E.coli and comparable for DOC and phosphate.

The removal of selected MPs by single and catalytic ozonation in both treatment options is displayed in Figure 6.5. The CFS removed 5% and 10% of CBZ and ATZ, respectively. This low removal by CFS can be anticipated since both MPs exhibit some hydrophobic character, (logKow = 2.45 and 2.61, respectively) and thus are expected to persist after CFS (Reungoat et al., 2010). The O₃ specific dose of 0.5 mg O₃/mg DOC was sufficient to completely remove CBZ by single or catalytic ozonation in any treatment form. Single ozonation only was not efficient for the removal of ATZ (11% in ST and 15% in PT) which was expected as ATZ has low reactivity with O₃. However, the removal of ATZ significantly improved when catalytic ozonation was applied reaching 97% in ST after the complete one hour treatment and 75% in PT after 1 minute treatment. Our results are comparable to those reported in the literature for CBZ and slightly lower for ATZ. Kovalova et al. (2013), attained complete removal of CBZ from secondary effluent using an O_3 specific dose of 0.64 mg O₃/mg DOC or by adsorption only with 23 mg/L PAC. Margot et al. (2013) reported more than 90% removal of CBZ using 0.8 mg O₃/mg DOC or 12 mg/L PAC. Altmann et al. (2014) achieved comparable results with 0.5 mg O₃/mg DOC or 20 mg/L PAC. Also, Lee et al. (2013) tested different O₃ specific doses ranging between 0.25 and 1.5 g O_3/g DOC and reported roughly between 20 and 50% removal of ATZ from spiked secondary effluent originating from different WWTPs. Parallel to the removal of MPs, the removal of DOC in spiked PI hinted to the attained degree of mineralization by each of the applied treatments. As such, SO did not affect DOC in ST and decreased it by 11% in PT. In contrast, CO decreased DOC by 5% and 27% in ST and PT, respectively. Thus, the highest degree of mineralization was achieved in PT-CO despite the fact that less ATZ was removed in this case. It is possible that since the primary effluent carried a lower load of particles, PT targeted via oxidation and adsorption the dissolved organics that could have been otherwise hindered by the presence of larger particles in ST.



Figure 6.5 Effect of the treatment on the removal of CBZ, ATZ and DOC in PI during wet weather that was spiked with 2 mg/L CBZ and 2 mg/L ATZ. Results correspond to samples collected after 1 hour settling in ST and 1 minute treatment in PT.

6.3.4. Toxicity assays and fate of PAC

Figure 6.6A depicts the different treatments' effects on the toxicity of PI during wet weather assessed using *V. fischeri*. All treatments reduced the acute toxicity of the PI to different extents. In general, ST increased the wastewater's acute toxicity in comparison with CFS alone. Particularly single ozonation turned out more toxic than catalytic ozonation which goes along with the increase in DOC illustrated in Figure 6.5. As discussed in section 6.3.3, ozonation contributes to the breaking down of non-polar compounds into smaller polar compounds. Reungoat et al. (2010), claimed that toxicity is related to the level of hydrophobicity: more hydrophobic compounds are more toxic. It is then possible that the cleavage of the parent compound in single ozonation generated several less hydrophobic products that collectively maintained a more toxic effect. This was less pronounced in PAC-catalyzed ozonation due to either removal of those hydrophobic compounds by adsorption, or to an enhanced oxidation process that shifted the reaction further towards the generation of more hydrophilic products. In PT, single and catalytic ozonation maintained the same acute toxicity level as CFS, and did not impose any negative effect.

Results of the genotoxicity test are summarized in Figure 6.6B. The survival rate in all samples exceeded 80% which, according to the test's supplier, validates the test results proving that the samples were not cytotoxic to the test bacteria. Also, all the tested samples resulted in IF values lower than the threshold IF value of 1.2. This signifies that neither the untreated nor the treated samples were genotoxic, or contained gentoxic substances. Yet, it was remarkable that the IF was significantly lower in samples from the PT process train (both SO and CO). Therefore, wastewater after CFS only or ST showed the same genetic toxicity as the PI, while PT further reduced the genetic toxicity of the effluent. Similarly, YES bioassays (Figure 6.6C) conducted on all samples revealed that none of them showed an estrogenic toxicity since the % induction was below 10%. However, it was observed that the estrogenic activity was reduced in wastewater samples pertaining to the PT train.



Figure 6.6 Effect of the treatment on the acute toxicity and genotoxicity of the primary effluent: (A) Results of the Microtox[®] bioassay after 5 minutes exposure time; (B) Results of the SOS Chromotest for genotoxicity (Induction Factor $\ge 1.2 =$ genotoxic); (C) Results of the YES assay for the estrogenic activity (% Induction > 10% = estrogenic toxicity).

While sludge analysis was out of this study's scope, it was noticeable that applying PAC during ST would entail disposal of the used PAC with the primary sludge according to the conventional operations of the WWTP. However, applying PAC during PT could result in its build up in the pipe, before flushing it through the outfall to the receiving water. As such, PAC can end up as sediments in the river and its actual impact was out of the scope of this study.

6.3.5. Cost Analysis

While it is universally approved that O₃ is highly competent for pathogens reduction, and despite all the knowledge regarding its high capacity for oxidizing organic matter; it is still argued that applying it in CSO treatment in particular, might not be a cost-effective option. As a matter of fact, as the CSO quality is prone to changes due to weather conditions, higher O₃ doses can be required for mitigation and hence the expanding costs of treatment (Wojtenko et al., 2001b). Therefore, and due to concerns with its high capital cost, the USEPA (2018) did not endorse its application for CSO disinfection. Wojtenko et al. (2001b) reported that in a disinfection project in Philadelphia, USA, the ozonation capital cost for a 2 minutes contact time and oxygen-fed generator, was 8.6 times and 7.7 times higher than other disinfection processes (hypochlorite and conventional chlorination, respectively).

In this study, cost calculations were made based on the assumptions in Table 6.3. It was estimated that the expected range of flow rates during wet weather conditions 100-600 MLD corresponded to a retention time ranging between 52-310 minutes in ST, and 0.56 to 3.34 minutes in PT. The costs were projected for the different treatment options based on assumptions made for service life (15 years) and interest rate (6%), and O₃ dose (approximate doses based on a specific O₃ dose of 0.5 g O₃/g DOC). The cost of PAC was estimated at 2.53\$/kg (Yang et al., 2010). Moreover, as the suggested treatment options are applied seasonally for the treatment of PI during wet weather, the smallest possible footprint is usually favoured. In this study, none of the suggested treatments required the addition of a contact chamber which resulted in waiving the corresponding costs. The operation was also presumed at a maximum of 100 days per year according to which the operation costs were corrected. It should be noted that the operation and maintenance costs associated with PAC application in PAC-catalyzed ozonation consist of PAC supply as well as disposal. In ST-CO those costs are waived since PAC is removed with the sludge. Yet in PT-CO,

they are to be accounted for especially if it is deemed that the river sediments require remediation, which was not considered in this research and was excluded from the calculations.

Parameter	Value
Flow rate during wet weather conditions, MLD	100 - 600
Residence time in the clarifier, min	52 - 310
Residence time in effluent channel before discharge, min	0.56 - 3.34
Service life, yr	15
Days of operation per year	100
O ₃ dose in simultaneous treatment, mg/L	14
O ₃ dose in post treatment, mg/L	5.5
PAC dose, mg/L	100
Price of PAC, \$/kg	2.53
Interest rate	6%

 Table 6.3 Assumptions for the cost estimates.

The estimated costs for all four proposed treatments are detailed in Table 6.4. In general, the cost of ST was 15 to 30 times higher than that of PT. This renders the former a cost-prohibitive alternative. Moreover, in ST, the addition of PAC did not result in considerable cost increase as the cost of SO and CO were relatively close. However, in PT, PAC addition led to at least doubling the estimated cost for high flow rate (0.32 and 0.63 m^3 in ST and PT, respectively). Reungoat et al. (2010) reported a comparable ozonation cost of 0.16 e/m^3 equivalent to 0.21 m^3 (based on 2010 exchange rate of 1.33) for an average O₃ dose of 5.7 mg O₃/L applied to a secondary effluent.

	Simultaneous Treatment		Post Treatment		
	SO °	CO d	SO °	CO d	
Estimated CapEx, \$	94,368,072	94,956,893	3,004,090	3.592,911	
Estimated OpEx, \$/yr	1.220,459 - 3,878,902	10,454,959 - 59,285,902	529,883 - 2,476,115	9,764,383 -57,886,115	
Corrected OpEx, \$/yr	334,372 - 1,062,713	2,864,372 -16,242,713	145,173 - 678,388	2,675,173 - 15,858,388	
Total annual cost, \$/yr	94,702,444 - 95,430,785	97,821,265 - 111,199,606	3,149,263 - 3,682,478	6,268,084 - 19,451,299	
Estimated cost per volume					
of PI during wet weather,	1.59 - 9.47	1.85 - 9.78	0.06 - 0.32	0.32 - 0.63	
\$/m ³					

Table 6.4 CapEx^a and OpEx^b analyses for the different treatment options with single and PAC-catalyzed ozonation, assuming the total period of operation is 15 years.

^a CapEx: capital expenditures.

^b OpEx: operational and maintenance expenditures. Both CapEx and OpEx values are present worth (PW) in the year 2016, adjusted with Engineering News Record (ENR) index.

^c Estimated with cost equations for ozonation systems by Sharma et al. (2013). Reference: Sharma, J.R., Najafi M., and Qasim S.R., Preliminary cost estimation models for construction, operation, and maintenance of water treatment plants. Journal of Infrastructure Systems, 2013. 19(4): p. 451-464.

^dEstimated with CapEx for a PAC dry feed system obtained from McGivney and Kawamura, 2008, while the OpEx cost was estimated from the cost of chemical addition obtained from (Yang et al., 2010), added to the OpEx of single ozonation. References: McGivney W.T. and Kawamura S. (2008) Cost estimating manual for water treatment facilities, John Wiley & Sons, Inc.; Yang, W., Paetkau, M., & Cicek, N. (2010). Improving the performance of membrane bioreactors by powdered activated carbon dosing with cost considerations *Water Science and Technology*, *62*(1), 172-179. doi: https://doi.org/10.2166/wst.2010.276

6.3.6. Overall performance assessment

The overall performance of each suggested CSO treatment option was summarized in Table

6.5. Each treatment was rated based on its effect on wastewater quality parameters and toxicity, as well as operational limitations and costs. As such, a ST whether through single or PAC-catalyzed ozonation was unfavourable since it had a negative effect on the wastewater content of organic matter and its toxicity. It also implied higher additional cost compared to PT. The latter was only favoured with single ozonation because PAC-catalyzed ozonation incurred major financial and environmental concerns associated with the disposal of PAC. Consequently, single ozonation applied in the pipe after a complete EPT is a promising treatment to improve the quality of the primary effluent before final disposal.

Table 6.5 Color-coded performance assessment of the different treatment options in comparison with primary treatment (CFS) only.

	Simultaneous Treatment		Post Treatment	
	O 3	O3+PAC	O3	O3+PAC
pH				
Turbidity removal				
COD removal				
DOC removal				
<i>E.coli</i> removal				
Removal of highly reactive MPs				
Removal of less reactive MPs				
Toxicity to V. fisheri				
Genetic and Estrogenic toxicity				
Problem with PAC disposal				
Cost				
Significant Improvement / No issue or concern				
Improvement	Improvement			
No Improvement	No Improvement			
Decline / Potential issue or	Decline / Potential issue or concern			

6.4. Conclusions

This study showed that a post treatment approach consisting of coagulation/ flocculation/ sedimentation followed by an ozonation process is a more viable option for the treatment of primary influent during wet weather conditions. In fact, under this treatment train, and 1 minute exposure time, both single and PAC-catalyzed ozonation were satisfactory for disinfection, achieving 4 log removal of *E. coli*. Complete removal of MPs that are highly reactive with O₃ was also attained, and none of the treatment approaches left any concerns about the effluent's acute, genetic or estrogenic toxicity. However, PAC-catalyzed ozonation proved more effective for the elimination of MPs that are less reactive with O₃ and subsequently DOC and COD removal. Therefore, if disinfection is the only concern, single ozonation in the pipe is a reasonable solution at an average cost 0.19 \$/m³. Yet, if disinfection along with the removal of a wider range of MPs are to be attained, PAC-catalyzed ozonation can be a promising solution with a higher average cost of 0.48 \$/m³. Further studies exploring options for PAC removal and disposal as well as cost optimization are needed.

Acknowledgement

The authors acknowledge the financial supports provided by the research grants from Natural Sciences and Engineering Research Council of Canada (NSERC) Collaborative Research and Development (CRD) program and NSERC Discovery grant. The financial supports of EPCOR Water Services and the Helmholtz-Alberta Initiative (HAI) through the Alberta Environment and Parks' EcoTrust Program are likewise acknowledged.

References

- Acero, J. L., Stemmler, K., & von Gunten, U. (2000). Degradation Kinetics of Atrazine and Its Degradation Products with Ozone and OH Radicals: A Predictive Tool for Drinking Water Treatment. Envrion. Sci. & Technol., 34(4), 591-597. doi: https://doi.org/10.1021/es990724e
- Alameddine, M., Al Umairi, A. R., Shaikh, M. Z., & Gamal El-Din, M. (2020). Bench to Full-Scale Enhanced Primary Treatment of Municipal Wastewater under Wet Weather Flow for Minimized Pollution Load: Evaluation of Chemical Addition and Process Control Indicators. Can. J. of Civ. Eng. doi: https://doi.org/10.1139/cjce-2019-0515
- Altmann, J., Ruhl, A. S., Zietzschmann, F., & Jekel, M. (2014). Direct comparison of ozonation and adsorption onto powdered activated carbon for micropollutant removal in advanced wastewater treatment. Water Res., 55, 185-193. doi: 10.1016/j.watres.2014.02.025
- APHA. (2017). Standard Methods for the Examination of Water and Wastewater (22nd ed.). Washington, DC: APHA, AWAA, WEF.

- Bader, H., & Hoigne, J. (1981). Determination of ozone in water by the Indigo Method. Water Res., 12, 449-456.
- Beltran, F. J. (2004). Ozone Reaction Kinetics for Water and Wastewater systems Lewis Publishers
- Can, Z. S., & Çakır, E. (2010). Treatability of Organic Constituents in the Paşaköy Wastewater Treatment Plant Effluent by O₃ and O₃/H₂O₂. Ozone: Sci. & Eng., 32(3), 209-214. doi: 10.1080/01919511003796087
- Chandrakanth, M. S., & Amy, G. L. (1996). Effects of Ozone on the Colloidal Stability and Aggregation of Particles Coated with Natural Organic Matter. Environ. Sci. and Technol., 30(2), 431-443. doi: https://doi.org/10.1021/es9500567
- Chhetri, R. K., Bonnerup, A., & Andersen, H. R. (2016). Combined Sewer Overflow pretreatment with chemical coagulation and a particle settler for improved peracetic acid disinfection. J. of Ind. & Eng. Chem., 37, 372-379. doi: 10.1016/j.jiec.2016.03.049
- Chhetri, R. K., Flagstad, R., Munch, E. S., Hørning, C., Berner, J., Kolte-Olsen, A., ... Andersen,
 H. R. (2015). Full scale evaluation of combined sewer overflows disinfection using performic acid in a sea-outfall pipe. Chem. Engi. J., 270, 133-139. doi: 10.1016/j.cej.2015.01.136
- Chhetri, R. K., Thornberg, D., Berner, J., Gramstad, R., Ojstedt, U., Sharma, A. K., & Andersen,
 H. R. (2014). Chemical disinfection of combined sewer overflow waters using performic acid or peracetic acids. Sci. Total Environ., 490, 1065-1072. doi: 10.1016/j.scitotenv.2014.05.079

- Cho, M., Chung, H., & Yoon, J. (2002). Effect of pH and Importance of Ozone initiated Radical Reactions In Inactivating Bacillus subtilis Spore. Ozone Sci. and Eng. 24, 145-150.
- Chow, A. T., Dahlgren, R. A., Zhang, Q., & Wong, P. K. (2008). Relationships between specific ultraviolet absorbance and trihalomethane precursors of different carbon sources. J. of Water Supply: Res. and Technol., 57(7), 471-480. doi: 10.2166/aqua.2008.064
- Christoffels, E., Mertens, F. M., Kistemann, T., & Schreiber, C. (2014). Retention of pharmaceutical residues and microorganisms at the Altendorf retention soil filter. Water Sci. and Technol., 70(9), 1503-1509. doi: https://doi.org/10.2166/wst.2014.404
- Coyle, E. E., Ormsbee, L. E., & Brion, G. M. (2014). Peracetic Acid as an Alternative Disinfection
 Technology for Wet Weather Flows. Water Environ. Res., 86(8), 687-697. doi:
 10.2175/106143014x13975035525663
- Evans, W. A., Davies, P. J., & McRae, C. (2016). The occurence of methyl,ethyl, propyl, and butyl parabens in the urban rivers and stormwaters of Sydneyn Australua. Environ. Sci. Water Res. Technol., 2, 733-742. doi: 10.1039/C5EW00240K
- Fan, X., Restivo, J., Órfão, J. J. M., Pereira, M. F. R., & Lapkin, A. A. (2014). The role of multiwalled carbon nanotubes (MWCNTs) in the catalytic ozonation of atrazine. Chem. Eng. J., 241, 66-76. doi: 10.1016/j.cej.2013.12.023
- Gasperi, J., Garnaud, S., Rocher, V., & Moilleron, R. (2008). Priority pollutants in wastewater and combined sewer overflow. Sci. Total Environ., 407(1), 263-272. doi: 10.1016/j.scitotenv.2008.08.015

- Gonçalves, A. G., Figueiredo, J. L., Órfão, J. J. M., & Pereira, M. F. R. (2010). Influence of the surface chemistry of multi-walled carbon nanotubes on their activity as ozonation catalysts. Carbon, 48(15), 4369-4381. doi: 10.1016/j.carbon.2010.07.051
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2013). Ozonation of sulfamethoxazole promoted by MWCNT. Catal. Commun., 35, 82-87. doi: 10.1016/j.catcom.2013.02.012
- Huber, M. M., Canonica, S., Park, G.-Y., & vonGunten, U. (2003). Oxidation of Pharmaceuticals during Ozonation and Advanced Oxidation Processes. Environ. Sci. and Technol., 37(5). doi: https://doi.org/10.1021/es025896h
- Huber, M. M., Gobel, A., Joss, A., Hermann, N., Loffler, D., Mcardell, C. S., . . . von Gunten, U.
 (2005). Oxidation of Pharmaceuticals during Ozonation of Municipal Wastewater Effluents:
 A Pilot Study. Environ. Sci. & Technol., 39(11), 4290-4299.
- Ikehata, K., Gamal El-Din, M., & Snyder, S. A. (2008). Ozonation and Advanced Oxidation Treatment of Emerging Organic Pollutants in Water and Wastewater. Ozone: Sci. & Eng., 30(1), 21-26. doi: 10.1080/01919510701728970
- Ikehata, K., Jodeiri Naghashkar, N., & Gamal El-Din, M. (2006). Degradation of Aqueous
 Pharmaceuticals by Ozonation and Advanced Oxidation Processes: A Review. Ozone: Sci.
 & Eng., 28(6), 353-414. doi: 10.1080/01919510600985937
- Jin, P., Jin, X., Bjerkelund, V. A., Østerhus, S. W., Wang, X. C., & Yang, L. (2016). A study on the reactivity characteristics of dissolved effluent organic matter (EfOM) from municipal wastewater treatment plant during ozonation. Water Res., 88, 643-652. doi: http://dx.doi.org/10.1016/j.watres.2015.10.0600043-1354/©2015

- Jung, C., Oh, J., & Yoon, Y. (2015). Removal of acetaminophen and naproxen by combined coagulation and adsorption using biochar: influence of combined sewer overflow components. Environ. Sci. Pollut. Res. Int., 22(13), 10058-10069. doi: 10.1007/s11356-015-4191-6
- Kocak, E. (2015). Investigation of potential genotoxic activity using the SOS Chromotest for real paracetamol wastewater and the wastewater treated by the Fenton process. J. of Environ. Health Sci. & Eng., 13(66). doi: 10.1186/s40201-015-0220-0
- Kovalova, L., Siegrist, H., von Gunten, U., Eugster, J., Hagenbuch, M., Wittmer, A., ... McArdell,
 C. S. (2013). Elimination of micropollutants during post-treatment of hospital wastewater
 with powdered activated carbon, ozone, and UV. Environ Sci Technol, 47(14), 7899-7908.
 doi: 10.1021/es400708w
- Lee, Y., Gerrity, D., Lee, M., Bogeat, A. E., Salhi, E., Sujanie Gamage, S., . . . von Gunten, U. (2013). Prediction of Micropollutant Elimination during Ozonation of Municipal Wastewater Effluents: Use of Kinetic and Water Specific Information. Environ. Sci. & Technol., 47, 5872-5881. doi: dx.doi.org/10.1021/es400781r
- Lee, Y., Kovalova, L., McArdell, C. S., & von Gunten, U. (2014). Prediction of micropollutant elimination during ozonation of a hospital wastewater effluent. Water Res, 64, 134-148. doi: 10.1016/j.watres.2014.06.027
- Lei, L., Gu, L., Zhang, X., & Su, Y. (2007). Catalytic oxidation of highly concentrated real industrialwastewater by integrated ozone and activated carbon. Appl. Catal. A: Gen., 327, 287–294. doi: 10.1016/j.apcata.2007.05.027

- Leonard, N. (2020). IDEXX water testing solution blog, Accessed April 2020 from https://www.idexx.com/en/blogs/idexx-water-testing-solutions/what-s-the-difference-between-mpn-and-cfu/
- Liu, Z. Q., Ma, J., Cui, Y. H., Zhao, L., & Zhang, B. P. (2011). Factors affecting the catalytic activity of multi-walled carbon nanotube for ozonation of oxalic acid. Sep. & Purif. Technol., 78(2), 147-153. doi: 10.1016/j.seppur.2011.01.034
- Madoux-Humery, A. S., Dorner, S., Sauve, S., Aboulfadl, K., Galarneau, M., Servais, P., & Prevost, M. (2016). The effects of combined sewer overflow events on riverine sources of drinking water. Water Res, 92, 218-227. doi: 10.1016/j.watres.2015.12.033
- Margot, J., Kienle, C., Magnet, A., Weil, M., Rossi, L., de Alencastro, L. F., . . . Barry, D. A. (2013). Treatment of micropollutants in municipal wastewater: ozone or powdered activated carbon? Sci Total Environ, 461-462, 480-498. doi: 10.1016/j.scitotenv.2013.05.034
- Nothe, T., Fahlenkamp, H., & von Sonntag, C. (2009). Ozonation of wastewater: Rate of ozone consumption and hydroxyl radical yield. Environ. Sci. Technol., 43(15), 5990–5995.
- Oulton, R., Haase, J. P., Kaalberg, S., Redmond, C. T., Nalbandian, M. J., & Cwiertny, D. M. (2015). Hydroxyl radical formation during ozonation of multiwalled carbon nanotubes: performance optimization and demonstration of a reactive CNT filter. Environ Sci Technol, 49(6), 3687-3697. doi: 10.1021/es505430v
- Phillips, P. J., Chalmers, A. T., Gray, J. L., Kolpin, D. W., Foreman, W. T., & Wall, G. R. (2012). Combined sewer overflows: an environmental source of hormones and wastewater micropollutants. Environ Sci Technol, 46(10), 5336-5343. doi: 10.1021/es3001294

- Pongmala, K., Autixier, L., Madoux-Humery, A. S., Fuamba, M., Galarneau, M., Sauvé, S., . . . Dorner, S. (2015). Modelling total suspended solids, E. coli and carbamazepine, a tracer of wastewater contamination from combined sewer overflows. J. of Hydrol., 531, 830-839. doi: 10.1016/j.jhydrol.2015.10.042
- Restivo, J., Orfao, J. J., Armenise, S., Garcia-Bordeje, E., & Pereira, M. F. (2012). Catalytic ozonation of metolachlor under continuous operation using nanocarbon materials grown on a ceramic monolith. J. Hazard. Mater., 239-240, 249-256. doi: 10.1016/j.jhazmat.2012.08.073
- Restivo, J., Órfão, J. J. M., Pereira, M. F. R., Garcia-Bordejé, E., Roche, P., Bourdin, D., . . . Derrouiche, S. (2013). Catalytic ozonation of organic micropollutants using carbon nanofibers supported on monoliths. Chem. Eng. J., 230, 115-123. doi: 10.1016/j.cej.2013.06.064
- Reungoat, J., Macova, M., Escher, B. I., Carswell, S., Mueller, J. F., & Keller, J. (2010). Removal of micropollutants and reduction of biological activity in a full scale reclamation plant using ozonation and activated carbon filtration. Water Res, 44(2), 625-637. doi: 10.1016/j.watres.2009.09.048
- Rizzo, L., Malato, S., Antakyali, D., Beretsou, V. G., Đolić, M. B., Wolfgang Gernjak, ... Fatta-Kassinos, D. (2019). Consolidated vs new advanced treatment methods for the removal of contaminants of emerging concern from urban wastewater. Sci. of the Total Environ., 655, 986-1008. doi: https://doi.org/10.1016/j.scitotenv.2018.11.265

- Rodayan, A., Segura, P. A., & Yargeau, V. (2014). Ozonation of wastewater: Removal and transformation products of drugs of abuse. Sci. of the Total Environ., 487, 763-770. doi: https://doi.org/10.1016/j.scitotenv.2013.11.023
- Song, Y., Breider, F., Ma, J., & von Gunten, U. (2017). Nitrate formation during ozonation as a surrogate parameter for abatement of micropollutants and the N-nitrosodimethylamine (NDMA) formation potential. Water Res., 122, 246-257. doi: 10.1016/j.watres.2017.05.074
- SOS-ChromoTest. (2019). SOS Chromotest Procedure S9 Activation and Express Strains Version6.6. In E. B.-D. P. Inc. (Ed.). Ontario, Canada: ebpi Inc.
- Szlachta, M., & Adamsk, W. (2009). Effects of natural organic matter removal by integrated processes: alum coagulation and PAC-adsorption. Water Sci. and Technol., 59(10), 1951-1957. doi: 10.2166/wst.2009.217
- Tondera, K., Klaer, K., Gebhardt, J., Wingender, J., Koch, C., Horstkott, M., . . . Pinnekamp, J. (2015). Reducing pathogens in combined sewer overflows using ozonation or UV irradiation. Int. J. Hyg. Environ. Health., 218(8), 731-741. doi: 10.1016/j.ijheh.2015.09.002
- Tondera, K., Klaer, K., Koch, C., Hamza, I. A., & Pinnekamp, J. (2016). Reducing pathogens in combined sewer overflows using performic acid. Int. J. Hyg. Environ. Health, 219(7 Pt B), 700-708. doi: 10.1016/j.ijheh.2016.04.009
- Tondera, K., Koenen, S., & Pinnekamp, J. (2013). Survey monitoring results on the reduction of micropollutants, bacteria, bacteriophages and TSS in retention soil filters. Water Sci. & Technol., 68(5), 1004-1012. doi: https://doi.org/10.2166/wst.2013.340
- USEPA. (2018). Combined Sewer Overflows (CSOs), 2018, from https://www.epa.gov/npdes/combined-sewer-overflows-csos. Accessed June 2019

- USEPA. (2020). Drinking water treatability database Powdered Activated Carbon Accesse from https://iaspub.epa.gov/tdb/pages/treatment/treatmentOverview.do?treatmentProcessId=207 4826383. Accessed April 2020
- von Sonntag, C., & von Gunten, U. (2012). Chemistry of Ozone in Water and Wastewater Treatment: IWA Publishing.
- Wildhaber, S. Y., Mestankova, H., Scharer, M., Schirmer, K., Salhi, E., & von Gunten, U. (2015).
 Novel test procedure to evaluate the treatability of wastewater with ozone. Water Res, 75, 324-335. doi: 10.1016/j.watres.2015.02.030
- Wojtenko, I., Stinson, M. K., & Field, R. (2001a). Challenges of Combined Sewer Overflow Disinfection by Ultraviolet Light Irradiation. Critical Reviews in Environ. Sci. & Technol., 31(3), 223-239. doi: 10.1080/20016491089217
- Wojtenko, I., Stinson, M. K., & Field, R. (2001b). Performance of Ozone as a Disinectant for Combined Sewer Overflow. Critical Reviews in Environ. Sci. & Technol., 31(4), 295-309. doi: 10.1080/20016491089235
- Xu, P., Janex, M.-L., Savoye, P., Cockx, A., & Lazarova, V. (2002). Wastewater disinfection by ozone: main parameters for process design. Water Res., 36, 1043-1055.
- Yamashita, T., & Yamamoto-Ikemoto, R. (2014). Nitrogen and Phosphorus Removal from Wastewater Treatment Plant Effluent via Bacterial Sulfate Reduction in an Anoxic Bioreactor Packed with Wood and Iron Int. J. Environ. Res. Public Health 2, 11, 9835-9853.
- Yang, W., Paetkau, M., & Cicek, N. (2010). Improving the performance of membrane bioreactors by powdered activated carbon dosing with cost considerations Water Sci. and Technol., 62(1), 172-179. doi: https://doi.org/10.2166/wst.2010.276
- Zhou, H., Liu, J., Xia, H., Zhang, Q., Ying, T., & Hu, T. (2015). Removal and reduction of selected organic micro-pollutants in effluent sewage by the ozone-based oxidation processes. Chem. Eng. J., 269, 245-254. doi: 10.1016/j.cej.2015.01.116
- Zimmermann, S. G., Wittenwiler, M., Hollender, J., Krauss, M., Ort, C., Siegrist, H., & von Gunten, U. (2011). Kinetic assessment and modeling of an ozonation step for full-scale municipal wastewater treatment: micropollutant oxidation, by-product formation and disinfection. Water Res., 45(2), 605-617. doi: 10.1016/j.watres.2010.07.080

CHAPTER 7: Conclusions and Recommendations

7.1. Thesis overview

Combined sewer overflows (CSOs) are characterized by their high level of suspended solids, pathogens and orthophosphate. It is undeniable that the release of raw untreated CSO into environmental waters distresses the environmental and public health. However, its treatment remains very challenging. In practicality, CSO is equivalent to municipal wastewater that was diluted with storm water and snow melts. It occurs seasonally and at high flow rates which triggers a prompt and effective treatment with minimal cost and footprint taking into consideration the complex matrix composition. A primary treatment consisting of coagulation/ flocculation and sedimentation is inevitable to remove conventional contaminants, yet the choice of chemicals remains critical for the treatment success. Those chemicals usually added to aggregate and remove suspended matter, cannot compete with oxidation techniques aimed at disinfection and removal of organic compounds. Ozonation is one such technique that has been studied to disinfect and remove organic compounds and has been recently upgraded with the use of a catalyst to further enhance its efficiency. Catalytic ozonation has been therefore competing with other processes although its fundamental catalytic mechanisms are arguable and not very well understood. In real life application, carbon-based catalysts such as PAC can be used to catalyze the ozonation of wastewater for enhanced disinfection and removal of micropollutants. However, little is known about the catalytic mechanisms involved, the efficiency and overall performance of PAC catalyzed ozonation, especially with the high adsorption capacity of PAC.

To shed the light on a basic treatment method that mitigates for the direct discharge of CSO, enhanced primary treatment was investigated through a comprehensive approach from bench to full scale presented in Chapter 3. The study tested and compared three metal-based coagulants and investigated the additional effect of polymer to aid the flocculation and improve the removal of colloidal matters. Chapter 4 delivered a detailed examination of the fundamentals of the catalytic mechanisms involved in PAC-catalyzed ozonation. It paved the road for a better understanding of the role of PAC-catalyzed ozonation in the treatment of different types of MPs. The study was performed using a mixture of different MPs including pharmaceuticals, herbicides and perfluorinated compounds known for their diverse affinities towards ozone. Several of those MPs were studied for the first time in a PAC and ozone system at near environmental concentrations. Chapter 5 proceeded with a comparison between the O_3 specific doses and their effect on the removal of MPs by single and PAC-catalyzed ozonation. It suggested that some MPs can serve as surrogates to help predict the removal of other MPs. In this chapter there was an attempt to study the overall removal kinetics of MPs in a mixture treated by O_3 only and by O_3 and PAC, as well as an overview of the intermediate products that form in each process. Chapter 6 resumed with the application of PAC-catalyzed ozonation for the treatment of primary influent (PI) during wet weather conditions. It evaluated several options for an all-inclusive treatment by considering coagulation/ flocculation/ sedimentation for the removal of suspended solids, as well as single and catalytic ozonation for the removal of MPs, disinfection and toxicity reduction.

7.2. Conclusions

To sum up, the research presented in this thesis focused on two solutions for the treatment of the PI during wet weather conditions in response to the issue of CSO discharge. The thesis answered questions related to practical, fundamental and theoretical aspects of the PI treatment in a cohesive manner aspiring to optimize the quality of the effluent and ultimately preserve the environment and public health. There was a special emphasis on PAC-catalyzed ozonation and its fundamentals being an emerging technique with excessive potentials. The key conclusions are recapped below.

Chapter 3: EPT of PI during wet weather conditions

- Alum outperformed other metal-based coagulants in the primary treatment of PI during wet weather flow: 1 mg of Al added as alum with low mixing conditions was able to remove 22 NTUs, 19 mg chemical oxygen demand (COD) and 0.8 mg orthophosphate (ortho-P).
- Three-factor ANOVA indicated that TSS removal depended mostly on rapid mixing while COD and ortho-P removals depended on slow mixing and coagulant dose.
- The addition of polymer did not lead to any pronounced improvements in the quality of the treated wastewater neither in bench-scale nor in full scale operations.
- Turbidity and percentage UV transmission showed good correlation with total suspended solid and ortho-P. Therefore, they can be used as surrogates for estimating micropollutants' (MPs) removal and online process control.

Chapter 4: Fundamentals of the catalytic mechanisms in PAC-catalyzed ozonation

- PAC-catalyzed ozonation improved the removal of MPs that are resistant to ozonation by up to 27% and promoted faster specific degradation rates.
- Radical probing experiments showed that scavenging hydroxyl radical ('OH) did not have any significant impact on the removal of MPs, while scavenging other reactive oxygen species, namely singlet oxygen ¹O₂, was more influential.
- Electron Paramagnetic Resonance spectroscopy ascertained that the decomposition of O₃ in presence of PAC at neutral pH did not boost the generation of free 'OH.
- Free 'OH did not play any significant role in PAC-catalyzed ozonation.

- It is likely that the catalytic effect was expressed at the surface of the catalyst and in the interface between PAC and the bulk solution.
- It is likely that the catalytic decomposition of O₃ by PAC produced adsorbed 'OH as PAC'–
 HO and other strong oxidizing species like [¹O₂+PAC].

Chapter 5: Effect of O_3 specific dose on MPs' removal by single and PAC-catalyzed ozonation

- In a diverse mixture of MPs, most compounds showed higher affinity for the removal by PAC-catalyzed ozonation which was mainly observed at higher O₃ specific dose. Under this condition more than 50% removal could be achieved for the less reactive MPs.
- Carbamazepine and atrazine can serve as surrogate MPs for fast and slow reacting compounds, respectively.
- The removal kinetics were affected by the level of dissolved organic carbon (DOC) available to consume O₃.
- Faster kinetics were obtained in PAC-catalyzed ozonation leaning more towards the removal kinetics with molecular O₃.
- Less intermediate compounds were formed in PAC-catalyzed ozonation possibly because the removal occurs through different pathways in single and PAC-catalyzed ozonation, in addition to the combined role of catalysis and adsorption.

Chapter 6: Applicability of PAC-catalyzed ozonation for the treatment of PI during wet weather

- Post treatment approach consisting of coagulation/ flocculation/ sedimentation followed by ozonation processes is a viable option for the treatment of PI during wet weather conditions.
- Single and PAC-catalyzed ozonation applied as post-treatment were satisfactory for disinfection and removal of MPs that are highly reactive with O₃ without any concerns about the effluent's acute, genetic or estrogenic toxicity.

- PAC-catalyzed ozonation proved more effective for the removal of MPs that are less reactive with O₃ and subsequently DOC and COD.
- If disinfection is the only concern, single ozonation in the pipe is a reasonable solution at an average cost 0.19 \$/m³.
- If removal of a wider range of MPs are to be attained, PAC-catalyzed ozonation is a promising solution at an average cost of 0.48 \$/m³.

7.3. Recommendations

The above-mentioned conclusions can lay the ground for further investigations. We therefore propose the recommendations summarized below to be addressed in future works.

- Bench-scale experiments performed for the optimization of coagulant dose suggested the use of turbidity and UVA₂₄₅ for online process control. This requires validation through bench-scale coagulation and ozonation experiments that account for the varying PI quality through different dilutions and different temperatures. Moreover, there is a need to confirm their use as surrogates for MPs removal after coagulation and single or catalytic ozonation processes with and without spiking.
- Since testing the addition of polymer during the primary treatment of PI revealed some improvements only at full scale operation, further experiments that investigate the effect of slow mixing and retention time during bench scale are recommended.
- The confusion between the role of PAC as catalyst or initiator/ promoter of O₃ decomposition and its role as adsorbent has been behind the reluctance in its application in catalytic ozonation. With all the difficulties that arise when demonstrating the exact removal mechanism during PAC-catalyzed ozonation, mainly because of its strong adsorbing nature, there is still no clear evidence of the exact species that are involved. Also, knowing that there

are different types of activated carbon, any suggested mechanism for one carbon-based catalyst might not essentially apply to all other carbon-based catalysts. This eventually blurs the fundamentals of ozonation catalyzed by carbon materials despite carbon catalysts being more preferred in environmental applications. The challenge remains in the identification of reliable and targeted radical probes and scavengers that do not adsorb onto PAC and can provide strong exact answers with high certainty pertaining to the species involved in the mechanisms.

- The attempt to study the degradation kinetics of the mixture of MPs by single and PACcatalyzed ozonation did not prove very successful in a batch system since only three of the less O₃-reactive MPs could be monitored. Semi-batch or stopped flow systems deserve a try where continuous supply of O₃ is required in order to mitigate for the fast O₃ decomposition in batch systems and thus securing enough oxidant to carry on the reaction for a longer time and allowing for a better representation of the compounds decay over time.
- Out of the seventeen studied MPs, the pathways of the oxidation of five MPs have not been investigated before. Detailed studies investigating the oxidation mechanisms of fluconazole, diazinon, clindamycin, mecoprop and gemfibrozil individually are needed. Moreover, a comparison between the intermediates formed when MPs are studied individually and those formed in a mixture can give a better insight into the matrix effect on the catalytic and non-catalytic mechanisms.
- Bench-scale experiments performed in the ozonation study for PI mainly portrayed batch reactors that do not represent the suggested full-scale applications in the pipe. Therefore, another experimental setup that can model in-pipe treatment with plug flow reactors and direct O₃ bubbling is recommended to better elucidate and validate the removal of organics and *E.coli*. A fluidized bed reactor could represent another possible configuration. The effect

of other environmental conditions, namely temperature, is also worth the investigation knowing that O₃ is more stable at lower temperatures and that the studied PI occurs in spring and summer seasons due to snow melts and rains.

- Upgrading the treatment of PI during wet weather conditions through the addition of PACcatalyzed ozonation after primary treatment delivered several advantages regarding the quality of the effluent. A major drawback of the process is associated with the disposal of utilized PAC when the treatment is suggested in the pipe. Further studies that explore options for PAC removal, such as end of pipe settling tanks, and disposal as well as cost optimization are needed.
- The O₃ and PAC doses that were tested in this research for the treatment of PI during wet weather flow were justified. Yet, further optimization of the doses and exposure time is beneficial to account for the fluctuating quality of the PI during wet weather conditions.
- Studying the effect of single and PAC-catalyzed ozonation on the removal of pre-existing MPs in the PI would provide data pertaining to a broad range of MPs. Solid phase extraction (SPE) could be applied to wastewater samples before and after the suggested treatments to acquire the exact MPs' degradation information. This can confirm the validity of the proposed surrogates and process control parameters as well as the treatment's compliance with target values or projected guidelines whenever they are implemented.
- Although PAC has numerous advantages, being an affordable and excellent adsorbent that is commonly used in the water and wastewater treatment industries, it is worth exploring other types of carbon-based catalysts. For instance, adsorbents that can be prepared from the sludge generated in primary or secondary clarifiers or anaerobic digesters, also referred to as biochar, can also be evaluated for their capacity to catalyze the ozonation process.

239

- In this study we evaluated the treatment of PI during wet weather flow with two approaches coagulation/ flocculation/ sedimentation and oxidation. Since, the former is compulsory for the removal of suspended solids, the latter is an additional polishing step that showed great potential for disinfection and removal of micropollutants. Still, other processes that can compete with single or catalytic ozonation can be tested and compared in terms of effluent quality, feasibility and cost assessment. UV disinfection, ferrate addition, chlorination and membrane filtration could possibly be good candidates.
- While the work presented in this study targeted the treatment of PI during wet weather flow, it could lay the ground for engineered green approaches for CSO treatment. As such, in areas with split sewers, the installation of porous pavement and manholes through which storm water and snowmelts can infiltrate, can provide a route for the removal of trace organics through adsorption and oxidation in presence of dissolved oxygen.

BIBLIOGRAPHY

- Acero, J. L., Stemmler, K., & von Gunten, U. (2000). Degradation Kinetics of Atrazine and Its Degradation Products with Ozone and OH Radicals: A Predictive Tool for Drinking Water Treatment. *Envrion. Sci. & Technol.*, 34(4), 591-597. doi: https://doi.org/10.1021/es990724e
- Afzal, A., Pourrezaei, P., Ding, N., Moustafa, A., Hwang, G., Drzewicz, P., . . . Gamal-El-Din, M. (2011). Physico-Chemical Processes. *Water Environ. Res.*, 83(10), 994-1091.
- Aguilar, C. M., Chairez, I., Rodríguez, J. L., Tiznado, H., Santillán, R., Arrieta, D., & Poznyak, T.
 (2019). Inhibition effect of ethanol in naproxen degradation by catalytic ozonation with NiO.
 RSC Advances 9. doi: 10.1039/c9ra02133g
- Aguilar, M. I., Saez, J., Llorens, M., Soler, A., Ortuno, J. F., Meseguer, V., & Fuentes, A. (2005). Improvement of coagulation-flocculation process using anionic polyacrylamide as coagulant aid. *Chemosphere*, 58(1), 47-56. doi: 10.1016/j.chemosphere.2004.09.008
- Agustina, T. E., Ang, H. M., & Vareek, V. K. (2005). A review of synergistic effect of photocatalysis and ozonation on wastewater treatment. *Journal of Photochemistry and Photobiology C: Photochemistry Reviews*, 6(4), 264-273. doi: 10.1016/j.jphotochemrev.2005.12.003
- Alameddine, M., Al Umairi, A. R., Shaikh, M. Z., & Gamal El-Din, M. (2020). Bench to Full-Scale Enhanced Primary Treatment of Municipal Wastewater under Wet Weather Flow for Minimized Pollution Load: Evaluation of Chemical Addition and Process Control Indicators. *Can. J. of Civ. Eng.* doi: https://doi.org/10.1139/cjce-2019-0515
- Alberta Environment & Sustainable Resource Development (ESRD). (2014). Environmental Quality Guidelines for Alberta Surface Waters. Edmonton, Alberta.

- Alexander, J. T., Hai, F. I., & Al-aboud, T. M. (2012). Chemical coagulation-based processes for trace organic contaminant removal: Current state and future potential. *J. of Environ. Manag.*, *111*, 195-207. doi: https://doi.org/10.1016/j.jenvman.2012.07.023
- Altmann, J., Massa, L., Sperlich, A., Gnirss, R., & Jekel, M. (2016). UV₂₅₄ absorbance as realtime monitoring and control parameter for micropollutant removal in advanced wastewater treatment with powdered activated carbon. *Water Res.*, 94, 240-245. doi: <u>http://dx.doi.org/10.1016/j.watres.2016.03.001</u>
- Altmann, J., Ruhl, A. S., Zietzschmann, F., & Jekel, M. (2014). Direct comparison of ozonation and adsorption onto powdered activated carbon for micropollutant removal in advanced wastewater treatment. *Water Res.*, 55, 185-193. doi: 10.1016/j.watres.2014.02.025
- Altmann, J., Zietzschmann, F., Geiling, E.-L., Ruhl, A. S., Sperlich, A., & Jekel, M. (2015). Impacts of coagulation on the adsorption of organic micropollutants onto powdered activated carbon in treated domestic wastewater. *Chemosphere*, 125. doi: <u>http://dx.doi.org/10.1016/j.chemosphere.2014.12.061</u>
- Alvarez, P. M., Masa, J., Jaramillo, J., Beltran, F. J., & Gomez-Serrano, V. (2008). Kinetics of Ozone Decomposition by Granular Activated Carbon. *Ind. Eng. Chem. Res.*, 47, 2545-2553.
- Anderson, M. J. (2015). *Design of Experiments Simplified* (Third ed.): Boca Raton, FL: CRC Press is an imprint of the Taylor & Francis Group, an Informa business.
- APHA. (2017). Standard Methods for the Examination of Water and Wastewater (22nd ed.). Washington, DC: APHA, AWAA, WEF.
- Arlos, M. J., Braggb, L. M., Servos, M. R., & Parker, W. J. (2014). Simulation of the fate of selected pharmaceuticals and personal care products in a highly impacted reach of a Canadian watershed. *Sci. of the Total Environ.*, 193-204. doi: <u>http://dx.doi.org/10.1016/j.scitotenv.2014.03.092</u>

AWWA. (2019). Per- and Polyfluoroalkyl Substances (PFAS) Treatment.

- Bader, H., & Hoigne, J. (1981). Determination of ozone in water by the Indigo Method. *Water Res.*, 12, 449-456.
- Baghvand, A., Zand, A. D., Mehrdadi, N., & Karbassi, A. (2010). Optimizing coagulation process for low to high turbidity waters using aluminum and iron salts. [article]. *American J. of Environ. Sci.*, 6(5), 442-448. doi: 10.3844/ajessp.2010.442.448
- Barbosa, M. O., Moreira, N. F., Ribeiro, A. R., Pereira, M. F., & Silva, A. M. (2016). Occurrence and removal of organic micropollutants: An overview of the watch list of EU Decision 2015/495. *Water Res.*, 94, 257-279. doi: 10.1016/j.watres.2016.02.047
- Beltran, F. J. (2004). Ozone reaction kinetics for water and wastewater systems United States of America: Lewis Publishers
- Beltrán, F. J., & Rey, A. (2018). Free Radical and Direct Ozone Reaction Competition to Remove
 Priority and Pharmaceutical Water Contaminants with Single and Hydrogen Peroxide
 Ozonation Systems. *Ozone: Sci. & Eng., 40*(4), 251-265. doi: 10.1080/01919512.2018.1431521
- Beltran, F. J., Rivas, F. J., Fernandez, L. A., Alvarez, P. M., & Montero-de-Espinosa, R. (2002).
 Kinetics of Catalytic Ozonation of Oxalic Acid in Water with Activated Carbon. *Ind. Eng. Chem. Res. 2, 41*, 6510-6517. doi: https://doi.org/10.1021/ie020311d
- Benitez, F. J., Real, F. J., Acero, J. L., & Garcia, C. (2007). Kinetics of the transformation of phenyl-urea herbicides during ozonation of natural waters: Rate constants and model predictions. *Water Res.*, 41(18), 4073-4084. doi: https://doi.org/10.1016/j.watres.2007.05.041

- Betancur-Corredor, B., Soltan, J., & Peñuela, G. A. (2016). Mineralization of Ibuprofen and Humic Acid through Catalytic Ozonation. *Ozone: Sci. & Eng.*, 38, 203-210. doi: 10.1080/01919512.2015.1114880
- Biernacki, W., L.Fijołek, & Nawrocki, J. (2015). Dissolved Ozone Decomposition in Presence of Activated Carbon at Low pH: How Experimental Parameters Affect Observed Kinetics of the Process. *Ozone: Sci. & Eng., 41*(4), 296-311. doi: https://doi.org/10.1080/01919512.2018.1535889
- Bourgeois, W., Burgess, J. E., & Stuetz, R. M. (2001). On-line monitoring of wastewater quality:
 a review. J. of Chem. Technol. & Biotechnol., 76, 337-348. doi: https://doi.org/10.1002/jctb.393
- Bozoglu, F. (2011). Impact of Pesticides as Organic Micro-Pollutants on the Environment and Risks for Mankind. In H. Alpas, S. M. Berkowics & I. Ermakova (Eds.), NATO Science for Peace and Security Series -C: Environmental Security - Environmental Security and Ecoterrorism (pp. 73-82): Springer, Dordrecht. doi: https://doi.org/10.1007/978-94-007-1235-5_6
- Broseus, R., Vincent, S., Aboulfadl, K., Daneshvar, A., Sauve, S., Barbeau, B., & Prevost, M. (2009). Ozone oxidation of pharmaceuticals, endocrine disruptors and pesticides during drinking water treatment. *Water Res.*, 43(18), 4707-4717. doi: 10.1016/j.watres.2009.07.031
- Buettner, G. R. (1987). Spin trapping: parameters of spin adducts*. *Free Radic. Biol. & Med., 3*, 259-303.
- Buffle, M. O., Schumacher, J., Salhi, E., Jekel, M., & von Gunten, U. (2006). Measurement of the Initial Phase of Ozone Decomposition in Water and Wastewater by Means of a Continuous Quench-Flow System: Application to Disinfection and Pharmaceutical Oxidation. *Water Res.*, 40(9), 1884-1894. doi: 10.1016/j.watres.2006.02.026

- Can, Z. S., & Çakır, E. (2010). Treatability of Organic Constituents in the Paşaköy Wastewater Treatment Plant Effluent by O₃ and O₃/H₂O₂. *Ozone: Sci. & Eng., 32*(3), 209-214. doi: 10.1080/01919511003796087
- Canada, E. (2019). Municipal Wastewater Status in Canada, from <u>http://ec.gc.ca/eu-ww/default.asp?lang=En&n=6E4ACEEE-21A9-4A1A-89F1</u>
- Canadian Council of Ministers of the Environment (CCME). (2014). Canadian Water Quality Guidelines for the protection of Aquatic Life *Canadian Environmental Quality Guidelines*, Accessed June 2020 from <u>http://ceqg-rcqe.ccme.ca/en/index.html#void</u>
- Cao, B., Gao, B., Xu, C., Fu, Y., & Liu, X. (2010). Effects of pH on coagulation behavior and floc properties in Yellow River water treatment using ferric based coagulants. *Chinese Science Bulletin*, 55(14), 1382-1387. doi: 10.1007/s11434-010-0087-5
- Chandrakanth, M. S., & Amy, G. L. (1996). Effects of Ozone on the Colloidal Stability and Aggregation of Particles Coated with Natural Organic Matter. *Environ. Sci. and Technol.*, 30(2), 431-443. doi: https://doi.org/10.1021/es9500567
- Chelme-Ayala, P., Afzal, A., Ding, N., Moustafa, A. M. A., Pourrezaei, P., Alpatova, A., . . . Gamal-El-Din, M. (2012). Phys.-Chem. Process. *Water Environ. Res.*, *84*(10), 971-1028.
- Chelme-Ayala, P., Gamal-El-Din, M., R., R. S., & Leonard, C. (2011a). Advanced treatment of liquid swine manure using physico-chemical treatment. *J. Hazard. Mater.*, 186, 1632-1638. doi: 10.1016/j.jhazmat.2010.12.047
- Chelme-Ayala, P., Gamal El-Din, M., Smith, D. W., & Adams, C. D. (2011b). Oxidation kinetics of two pesticides in natural waters by ozonation and ozone combined with hydrogen peroxide. *Water Res.*, *45*(8), 2517-2526. doi: 10.1016/j.watres.2011.02.007

- Chen, X., Richard, J., Liu, Y., Dopp, E., Tuerk, J., & Beste, K. (2012). Ozonation products of triclosan in advanced wastewatertreatment. *Water Research Pergamon Press*, 46. doi: 10.1016/j.watres.2012.01.039
- Chhetri, R. K., Bonnerup, A., & Andersen, H. R. (2016). Combined Sewer Overflow pretreatment with chemical coagulation and a particle settler for improved peracetic acid disinfection. J. of Ind. & Eng. Chem., 37, 372-379. doi: 10.1016/j.jiec.2016.03.049
- Chhetri, R. K., Flagstad, R., Munch, E. S., Hørning, C., Berner, J., Kolte-Olsen, A., . . . Andersen,
 H. R. (2015). Full scale evaluation of combined sewer overflows disinfection using performic acid in a sea-outfall pipe. *Chem. Engi. J.*, 270, 133-139. doi: 10.1016/j.cej.2015.01.136
- Chhetri, R. K., Thornberg, D., Berner, J., Gramstad, R., Ojstedt, U., Sharma, A. K., & Andersen,
 H. R. (2014). Chemical disinfection of combined sewer overflow waters using performic acid or peracetic acids. *Sci. Total Environ.*, 490, 1065-1072. doi: 10.1016/j.scitotenv.2014.05.079
- Cho, M., Chung, H., & Yoon, J. (2002). Effect of pH and Importance of Ozone initiated Radical Reactions In Inactivating Bacillus subtilis Spore. Ozone Sci. and Eng., 24, 145-150.
- Chow, A. T., Dahlgren, R. A., Zhang, Q., & Wong, P. K. (2008). Relationships between specific ultraviolet absorbance and trihalomethane precursors of different carbon sources. *J. of Water Supply: Res. and Technol.*, 57(7), 471-480. doi: 10.2166/aqua.2008.064
- Christoffels, E., Mertens, F. M., Kistemann, T., & Schreiber, C. (2014). Retention of pharmaceutical residues and microorganisms at the Altendorf retention soil filter. *Water Sci.* and Technol., 70(9), 1503-1509. doi: https://doi.org/10.2166/wst.2014.404
- Chys, M., Audenaert, W. T. M., Deniere, E., Mortier, S. v. T. r. s. F. C., Langenhove, H. V., Nopens, I., . . . Hulle, S. W. H. V. (2017). Surrogate-Based Correlation Models in View of

Real-Time Control of Ozonation of Secondary Treated Municipal Wastewater - Model Development and Dynamic Validation. *Environ. Sci. & Technol., 51*, 14233-14243. doi: DOI: 10.1021/acs.est.7b04905

Chys, M., Audenaert, W. T. M., Stapel, H., Ried, A., Wieland, A., Weemaes, M., . . . Hulle, S. W.
H. V. (2018). Techno-economic assessment of surrogate-based real-time control and monitoring of secondary effluent ozonation at pilot scale. *Chem. Eng. J.*, 352, 431-440. doi: https://doi.org/10.1016/j.cej.2018.07.041

City of Edmonton Total Loadings Plan (2009).

- Coates, J. (2006). Interpretation of Infrared Spectra, A Practical Approach. In R. A. Meyers (Ed.), Encyclopedia of Analytical Chemistry: Applications, Theory and Instrumentation. doi: https://doi.org/10.1002/9780470027318.a5606
- Cornwell, D. A., & Bishop, M. M. (1983). Determining velocity gradients in laboratory and fullscale systems. *American Water Works Assoc. J.*, 75(9), 470-475.
- Coyle, E. E., Ormsbee, L. E., & Brion, G. M. (2014). Peracetic Acid as an Alternative Disinfection
 Technology for Wet Weather Flows. *Water Environ. Res.*, 86(8), 687-697. doi: 10.2175/106143014x13975035525663
- CWN. (2018). Canada's Challenges and Opportunities to Address Contaminants in Wastewater Supporting Document 2
- Wastewater Treatment Practice and Regulations in Canada and Other Jurisdictions: Environment and Climate Change Canada.
- Dai, X., Xie, Z., Dorian, B., Gray, S., & Zhan, J. (2019a). Comparative study of PFAS treatment by UV, UV/ozone, and fractionations with air and ozonated air. *Environ. Sci. Water Res. & Technol.*, 5, 1897-1907. doi: 10.1039/c9ew00701f

- Dai, X., Xie, Z., Dorian, B., Gray, S., & Zhang, J. (2019b). Comparative study of PFAS treatment by UV, UV/ozone, and fractionations with air and ozonated air. *Envrion. Sci.: Water Res.Technol.*, 5(11), 1897-1907. doi: https://doi.org/10.1039/C9EW00701F
- Das, S., Ray, N. M., Wan, J., Khan, A., Chakraborty, T., & Ray, M. B. (2017). Micropollutants in Wastewater: Fate and Removal Processes. doi: 10.5772/65644
- David Jenkins, J. F. F., And Arnold, & Menar, B. (1970). Review Paper:Chemical Processes for Phosphate Removal (Vol. 5, pp. 369-389.): Water Research Pergamon Press.
- Dehouli, H., Chedeville, O., Cagnon, B., Caqueret, V., & Porte, C. (2010). Influences of pH, temperature and activated carbon properties on the interaction ozone/activated carbon for a wastewater treatment process. *Desalination*, 254, 12-16. doi: doi:10.1016/j.desal.2009.12.021
- Dempsey, B. A., Sheu, H., Tanzeer Ahmed, T. M. T., & Mentink, J. (1985). Polyaluminum Chloride and Alum Coagulation of Clay-Fulvic Aicd Suspensions. *American Water Works Association*, 77(3), 74-80.
- Denisov, E. T. (2006). Reactivity of Quinones As Alkyl Radical Acceptors. *Kinet. & Catal., 47*(5), 662-671.
- Diaz-Fierros, F. T., Puerta, J., Suarez, J., & Diaz-Fierros, F. V. (2002). Contaminant loads of CSOs at the wastewater treatment plant of a city in NW Spain *Urban Water*, *4*, 291-299.
- Dodd, M. C., Buffle, M.-O., & vonGunten, U. (2006). Oxidation of Antibacterial Molecules by Aqueous Ozone: Moiety-Specific Reaction Kinetics and Application to Ozone-Based Wastewater Treatment. *Environ. Sci. & Technol., 60*(6), 1969-1977. doi: https://doi.org/10.1021/es051369x
- Ebeling, J. M., & Ogden, S. R. (2004). Application of chemical coagulation aids for the removal of suspended solids (TSS) and phosphorus from the microscreen effluent discharge of an

intensive recirculating aquaculture system. [Article]. North American J. of Aquaculture, 66(3), 198-207. doi: 10.1577/a03-056.1

- Eggen, R. I. L., Hollender, J., Joss, A., Scharer, M., & Stamm[†], C. (2014). Reducing the Discharge of Micropollutants in the Aquatic Environment: The Benefits of Upgrading Wastewater Treatment Plants. *Environ. Sci. & Technol., 48*, 7683-7689. doi: dx.doi.org/10.1021/es500907n
- El Samrani, A. G., Lartiges, B. S., & Villieras, F. (2008a). Chemical coagulation of combined sewer overflow: heavy metal removal and treatment optimization. *Water Res.*, 42(4-5), 951-960. doi: 10.1016/j.watres.2007.09.009
- El Samrani, A. G., Lartiges, B. S., & Villieras, F. (2008b). Chemical coagulation of combined sewer overflow: Heavy metal removal and treatment optimization. *Water Res.*, 42(4-5), 951-960. doi: 10.1016/j.watres.2007.09.009
- Elovitz, M. S., & von Gunten, U. (2008). Hydroxyl Radical/Ozone Ratios During Ozonation Processes. I. The Rct Concept. Ozone Sci. & Eng., 21(3), 239-260. doi: https://doi.org/10.1080/01919519908547239
- Evans, W. A., Davies, P. J., & McRae, C. (2016). The occurrence of methyl, ethyl, propyl, and butyl parabens in the urban rivers and stormwaters of Sydneyn Australua. *Environ. Sci. Water Res. Technol.*, 2, 733-742. doi: 10.1039/C5EW00240K
- Exall, K., & Marsalek, J. (2013). A Coagulant Survey for Chemically Enhanced Primary Treatment of Synthetic CSOs. *Water Air and Soil Pollut.*, 224(2), 9. doi: 10.1007/s11270-012-1414-z
- Fan, X., Restivo, J., Órfão, J. J. M., Pereira, M. F. R., & Lapkin, A. A. (2014). The role of multiwalled carbon nanotubes (MWCNTs) in the catalytic ozonation of atrazine. *Chem. Eng. J.*, 241, 66-76. doi: 10.1016/j.cej.2013.12.023

- Faria, P. C. C., Orfao, J. J. M., & Pereira, M. F. R. (2006). Ozone Decomposition in Water Catalyzed by Activated Carbon: Influence of Chemical and Textural Properties. *Ind. Eng. Chem. Res.*, 45(8), 2715-2721.
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2008a). Activated carbon catalytic ozonation of oxamic and oxalic acids. *Appl. Catal. B: Environ.*, 79(3), 237-243. doi: 10.1016/j.apcatb.2007.10.021
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2008b). Catalytic ozonation of sulfonated aromatic compounds in the presence of activated carbon. *Appl. Catal. B: Environ.*, 83(1-2), 150-159. doi: 10.1016/j.apcatb.2008.02.010
- Faria, P. C. C., Órfão, J. J. M., & Pereira, M. F. R. (2009). Activated carbon and ceria catalysts applied to the catalytic ozonation of dyes and textile effluents. *Appl. Catal. B: Environ.*, 88(3-4), 341-350. doi: 10.1016/j.apcatb.2008.11.002
- Feng, J., Xing, B., & Chen, H. (2019). Catalytic ozonation of humic acid in water with modified activated carbon: Enhancement and restoration of the activity of an activated carbon catalyst. *J. of Environ.l Manag.*, 237, 114-118. doi: https://doi.org/10.1016/j.jenvman.2019.02.061
- Gasperi, J., Garnaud, S., Rocher, V., & Moilleron, R. (2008). Priority pollutants in wastewater and combined sewer overflow. *Sci. Total Environ.*, 407(1), 263-272. doi: 10.1016/j.scitotenv.2008.08.015
- Gasperi, J., Rocher, V., Gilbert, S., Azimi, S., & Chebbo, G. (2010). Occurrence and removal of priority pollutants by lamella clarification and biofiltration. *Water Res.*, 44(10), 3065-3076. doi: 10.1016/j.watres.2010.02.035
- Gasperi, J., Zgheib, S., Cladiere, M., Rocher, V., Moilleron, R., & Chebbo, G. (2012). Priority pollutants in urban stormwater: part 2 - case of combined sewers. *Water Res.*, 46(20), 6693-6703. doi: 10.1016/j.watres.2011.09.041

- Gerrity, D., Gamage, S., Jones, D., Korshin, G. V., Lee, Y., Pisarenko, A., ... Snyder, S. A. (2012).
 Development of surrogate correlation models to predict trace organic contaminant oxidation and microbial inactivation during ozonation. *Water Res.*, 46, 6257-6272. doi: http://dx.doi.org/10.1016/j.watres.2012.08.037
- Gibson, J., Farnood, R., & Seto, P. (2016). Chemical pretreatment of combined sewer overflows for improved UV disinfection. *Water Sci. Technol.*, 73(2), 375-381. doi: 10.2166/wst.2015.447
- Gonçalves, A. G., Figueiredo, J. L., Órfão, J. J. M., & Pereira, M. F. R. (2010). Influence of the surface chemistry of multi-walled carbon nanotubes on their activity as ozonation catalysts. *Carbon, 48*(15), 4369-4381. doi: 10.1016/j.carbon.2010.07.051
- Goncalves, A. G., Orfao, J. J., & Pereira, M. F. (2012). Catalytic ozonation of sulphamethoxazole in the presence of carbon materials: catalytic performance and reaction pathways. *J. Hazard. Mater.*, 239-240, 167-174. doi: 10.1016/j.jhazmat.2012.08.057
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2013a). Ozonation of bezafibrate promoted by carbon materials. *Appl. Catal. B: Environ.*, 140-141, 82-91. doi: 10.1016/j.apcatb.2013.03.034
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2013b). Ozonation of sulfamethoxazole promoted by MWCNT. *Catal. Commun.*, *35*, 82-87. doi: 10.1016/j.catcom.2013.02.012
- Gonçalves, A. G., Órfão, J. J. M., & Pereira, M. F. R. (2014). Ozonation of erythromycin over carbon materials and ceria dispersed on carbon materials. *Chem. Eng. J.*, 250, 366-376. doi: 10.1016/j.cej.2014.04.012
- Grayson, R. B., Finlayson, B. L., Gippel, C. J., & Hart, B. T. (1996). The potential of field turbidity measurements for the computation of total phosphorus and suspended solids loads. J. of Environ. Manag., 47(3), 257-267. doi: 10.1006/jema.1996.0051

- Green, M. B., Martin, J. R., & Griffin, P. (1999). Treatment of combined sewer overflows at small wastewater treatment works by constructed reed beds *Wat. Sci. Tech*, *40*(3), 357-364.
- Guida, M., Mattei, M., Della Rocca, C., Melluso, G., & Meric, S. (2007). Optimization of alumcoagulation/flocculation for COD and TSS removal from five municipal wastewater. *Desalination*, 211(1-3), 113-127. doi: 10.1016/j.desal.2006.02.086
- Gümüs, D., & Akbal, F. (2017). A comparative study of ozonation, iron coated zeolite catalyzed ozonation and granular activated carbon catalyzed ozonation of humic acid. *Chemosphere*, 174, 218-231. doi: <u>http://dx.doi.org/10.1016/j.chemosphere.2017.01.106</u>
- Guo, Y., Wang, H., Wang, B., Deng, S., Huang, J., Yu, G., & Wang, Y. (2018). Prediction of micropollutant abatement during homogeneous catalytic ozonation by a chemical kinetic model. *Water Res.*, 142(383-385). doi: https://doi.org/10.1016/j.watres.2018.06.019
- Guzman-Perez, C. A., Soltan, J., & Robertson, J. (2011). Kinetics of catalytic ozonation of atrazine in the presence of activated carbon. *Sep. & Purif. Technol.*, 79, 8-14. doi: 10.1016/j.seppur.2011.02.035
- Hammad Khan, M., & Jung, J. Y. (2008). Ozonation catalyzed by homogeneous and heterogeneous catalysts for degradation of DEHP in aqueous phase. *Chemosphere*, 72(4), 690-696. doi: 10.1016/j.chemosphere.2008.02.037
- Hannouche, A., Chebbo, G., Ruban, G., Tassin, B., Lemaire, B. J., & Joannis, C. (2011).
 Relationship between turbidity and total suspended solids concentration within a combined sewer system. *Water Sci. and Technol.*, 64(12), 2445-2452. doi: 10.2166/wst.2011.779
- Haydar, S., & Aziz, J. A. (2009). Coagulation-flocculation studies of tannery wastewater using combination of alum with cationic and anionic polymers. J. Hazard. Mater., 168(2-3), 1035-1040. doi: 10.1016/j.jhazmat.2009.02.140

- Health Canada. (2019). Guidelines for Canadian Drinking Water Quality Summary Table. In H.E. a. C. S. B. Water and Air Quality Bureau (Ed.). Ottawa, Ontario: Health Canada.
- Hollender, J., Zimmermann, S. G., Koepke, S., Krauss, M., Mcardell, C. S., Ort, C., . . . Siegrist,
 H. (2009). Elimination of organic micropollutants in a municipal wastewater treatment plant
 upgraded with a full-scale post-ozonation followed by sand filtration *Environ. Sci. & Technol.*, 43(20), 7862–7869.
- Huang, H., Liu, G., Lv, W., Yao, K., Kang, Y., Li, F., & Lin, L. (2015). Ozone-Oxidation Products of Ibuprofen and Toxicity Analysis in Simulated Drinking Water. *J Drug Metab Toxicol*, 6(3). doi: 0.4172/2157-7609.1000181
- Huber, M. M., Canonica, S., Park, G.-Y., & vonGunten, U. (2003). Oxidation of Pharmaceuticals during Ozonation and Advanced Oxidation Processes. *Environ. Sci. and Technol.*, 37(5). doi: https://doi.org/10.1021/es025896h
- Huber, M. M., Gobel, A., Joss, A., Hermann, N., Loffler, D., Mcardell, C. S., . . . von Gunten, U. (2005). Oxidation of Pharmaceuticals during Ozonation of Municipal Wastewater Effluents: A Pilot Study. *Environ. Sci. & Technol.*, *39*(11), 4290-4299.
- Huber, M. M., Ternes, T. A., & von-Gunten, U. (2004). Removal of Estrogenic Activity and Formation of Oxidation Products during Ozonation of 17a-Ethinylestradiol. *Environ. Sci. Technol, 38*, 5177-5186.
- Hubner, U., Seiwert, B., Reemtsma, T., & Jekel, M. (2014). Ozonation products of carbamazepine and their removal from secondary effluents by soil aquifer treatment--indications from column experiments. *Water Res.*, 49, 34-43. doi: 10.1016/j.watres.2013.11.016

- Hubner, U., von Gunten, U., & Jekel, M. (2015). Evaluation of the persistence of transformation products from ozonation of trace organic compounds - a critical review. *Water Res, 68*, 150-170. doi: 10.1016/j.watres.2014.09.051
- Hussain, S., Aziz, H. A., Isa, M. H., Ahmad, A., Van Leeuwen, J., Zou, L., . . . Umar, M. (2011).
 Orthophosphate removal from domestic wastewater using limestone and granular activated carbon. *Desalination*, 271(1-3), 265-272. doi: 10.1016/j.desal.2010.12.046
- Ikehata, K., Gamal El-Din, M., & Snyder, S. A. (2008). Ozonation and Advanced Oxidation Treatment of Emerging Organic Pollutants in Water and Wastewater. *Ozone: Sci. & Eng.*, 30(1), 21-26. doi: 10.1080/01919510701728970
- Ikehata, K., Jodeiri Naghashkar, N., & Gamal El-Din, M. (2006). Degradation of Aqueous
 Pharmaceuticals by Ozonation and Advanced Oxidation Processes: A Review. *Ozone: Sci.*& Eng., 28(6), 353-414. doi: 10.1080/01919510600985937
- Irvine, K. N., Perrelli, M. F., McCorkhill, G., & Caruse, J. (2005). Sampling and Modeling Approaches to Assess Water Quality Impacts of Combined Sewer Overflows - The Importance of a Watershed Perspective J. Great Lakes Res., 31, 105-115.
- Jalali, H. M. (2016). Kinetic study of antibiotic ciprofloxacin ozonation by MWCNT/MnO2 using Monte Carlo simulation. *Mater. Sci. Eng. C. Mater. Biol. Appl.*, 59, 924-929. doi: 10.1016/j.msec.2015.10.085
- Jalliffier-Verne, I., Heniche, M., Madoux-Humery, A. S., Galarneau, M., Servais, P., Prevost, M.,
 & Dorner, S. (2016). Cumulative effects of fecal contamination from combined sewer overflows: Management for source water protection. *J. Environ. Manage.*, 174, 62-70. doi: 10.1016/j.jenvman.2016.03.002
- Jalliffier-Verne, I., Leconte, R., Huaringa-Alvarez, U., Madoux-Humery, A. S., Galarneau, M., Servais, P., . . . Dorner, S. (2015). Impacts of global change on the concentrations and

dilution of combined sewer overflows in a drinking water source. *Sci. of the Total Environ.*, *508*, 462-476. doi: 10.1016/j.scitotenv.2014.11.059

- Jaramillo, J., Alvarez, P. M., & Gomez-Serrano, V. (2010). Preparation and ozone-surface modification of activated carbon. Thermal stabilityof oxygen surface groups. *Appl. Surf. Sci.*, 256, 5232-5236. doi: doi:10.1016/j.apsusc.2009.12.109
- Javier-Benitez, F., Acero, J. L., Real, F. J., & Roman, S. (2004). Oxidation of MCPA and 2,4-d by UV Radiation, Ozone, and the Combinations UV/H₂O₂ and O₃/H₂O₂. *J. of Environ. Sci. & Health, Part B, B39*, 393-409. doi: 10.1081/ PFC-120035925
- Jiang, J. K., & Graham, N. J. D. (2010). Enhanced Coagulation Using Al/Fe(III) Coagulants: Effect of Coagulant Chemistry on the Removal of Colour-Causing NOM. *Environ. Technol.* , 17, 937-950. doi: https://doi.org/10.1080/09593330.1996.9618422
- Jin, P., Jin, X., Bjerkelund, V. A., Østerhus, S. W., Wang, X. C., & Yang, L. (2016). A study on the reactivity characteristics of dissolved effluent organic matter (EfOM) from municipal wastewater treatment plant during ozonation. *Water Res.*, 88, 643-652. doi: http://dx.doi.org/10.1016/j.watres.2015.10.0600043-1354/©2015
- Jin, X., Peldszus, S., & Huck, P. M. (2012). Reaction kinetics of selected micropollutants in ozonation and advanced oxidation processes. *Water Res.*, 46(19), 6519-6530. doi: 10.1016/j.watres.2012.09.026
- Johnson, A. C., & Williams, R. J. (2004). A Model To Estimate Influent and Effluent Concentrations of Estradiol, Estrone, and Ethinylestradiol at Sewage Treatment Works. *Environ. Sci. & Technol., 38*, 3649-3658. doi: 10.1021/es035342u CCC: \$27.50
- Joss, A., Schärer, M., & Abegglen, C. (2016). *Micropollutants: The Swiss Strategy*. Paper presented at the Water 2020.

- Jung, C., Oh, J., & Yoon, Y. (2015). Removal of acetaminophen and naproxen by combined coagulation and adsorption using biochar: influence of combined sewer overflow components. *Environ. Sci. Pollut. Res. Int., 22*(13), 10058-10069. doi: 10.1007/s11356-015-4191-6
- Kay, P., Hughes, S. R., Ault, J. R., Ashcroft, A. E., & Brown, L. E. (2016). Widespread, routine occurrence of pharmaceuticals in sewage effluent, combined sewer overflows and receiving waters. *Environ. Pollut.*, 220, 1447e1455. doi: <u>http://dx.doi.org/10.1016/j.envpol.2016.10.087</u>
- Kerr, J. L., Guo, Z., Smith, D. W., Goss, G. G., & Belosevic, M. (2008). Use of goldfish to monitor wastewater and reuse water for xenobiotics. *J.of Environ. Eng. & Sci.*, 7. doi: the total discharged load of trace organic compounds with low removal efficiency in a WWTP
- Khalil, N., Chen, A., Lee, M., Czerwinski, S. A., Ebert, J. R., DeWitt, J. C., & Kannan, K. (2016).
 Association of Perfluoroalkyl Substances, Bone Mineral Density, and Osteoporosis in the
 U.S. Population in NHANES 2009-2010. *Environ. Health Perspectives 124*(1), 81-87. doi:
 doi: 10.1289/ehp.1307909
- Kim, M.-K., & Zoh, K.-D. (2016). Occurrence and removals of micropollutants in water environment. *Environ. Eng. Res.*, 21(4), 319-332. doi: 10.4491/eer.2016.115
- Kiss, A., & Virág, D. (2009a). Photostability and Photodegradation Pathways of Distinctive J. *Environ. Qua, 38*, 157-163.
- Kiss, A., & Virág, D. (2009b). Photostability and photodegradation pathways of distinctive pesticides. *J. Environ. Qual.*, *13*(38), 157-163. doi: 10.2134/jeq2007.0504.
- Klimenko, N. A., Polyakova, T. V., & Savchina, L. A. (2010). Influence of Sodium Azide on the Removal of Fulvic Acids by Activated Carbon. *J. of Water Chem. and Technol.*, 32(6), 329-335. doi: 10.3103/S1063455X10060032

- Kocak, E. (2015). Investigation of potential genotoxic activity using the SOS Chromotest for real paracetamol wastewater and the wastewater treated by the Fenton process. J. of Environ.l Health Sci. & Eng., 13(66). doi: 10.1186/s40201-015-0220-0
- Kolosov, P., Peyot, M.-L., & Yargeau, V. (2018). Novel materials for catalytic ozonation of wastewater for disinfection and removal of micropollutants. *Sci. of the Total Environ.*, 1207-1218. doi: https://doi.org/10.1016/j.scitotenv.2018.07.022
- Kovalova, L., Siegrist, H., von Gunten, U., Eugster, J., Hagenbuch, M., Wittmer, A., . . . McArdell,
 C. S. (2013). Elimination of micropollutants during post-treatment of hospital wastewater
 with powdered activated carbon, ozone, and UV. *Environ Sci Technol, 47*(14), 7899-7908.
 doi: 10.1021/es400708w
- Ku, Y., Chang, J.-L., Shen, Y.-S., & Lin, S.-Y. (1998). Decomposition of diazinon in aqueous solution by ozonation. *Water Res.*, 32(6), 1957-1963. doi: https://doi.org/10.1016/S0043-1354(97)00353-9
- Kuang, J., Huang, J., Wang, B., Cao, Q., Deng, S., & Yu, G. (2013). Ozonation of trimethoprim in aqueous solution: identification of reaction products and their toxicity. *Water Res*, 47(8), 2863-2872. doi: 10.1016/j.watres.2013.02.048
- Kwon, K., Lee, S., Jung, Y., & Min, K. S. (2014). Contaminant loads of CSOs at the urban area in Korea. *Desalination and Water Treat.*, 53(11), 3048-3053. doi: 10.1080/19443994.2014.922306
- Lajeunesse, A., Blais, M., Barbeau, B., Sauvé, S., & Gagnon, a. C. (2013). Ozone oxidation of antidepressants in wastewater-treatment evaluation and characterization of new by-products by LC-QToFMS. *Chem. Cent. J.*, 7(15), 11.

- Launay, M. A., Dittmer, U., & Steinmetz, H. (2016). Organic micropollutants discharged by combined sewer overflows Characterisation of pollutant sources and stormwater-related processes. *Water Res.*, 104, 82-92. doi: <u>http://dx.doi.org/10.1016/j.watres.2016.07.068</u>
- Lee, C. S., Robinson, J., & Chong, M. F. (2014a). A review on application of flocculants in wastewater treatment. *Process Safety and Environ. Prot.*, 92(6), 489-508. doi: 10.1016/j.psep.2014.04.010
- Lee, I. S., Sim, W. J., Kim, C. W., Chang, Y. S., & Oh, J. E. (2011). Characteristic occurrence patterns of micropollutants and their removal efficiencies in industrial wastewater treatment plants. J. Environ. Monit., 13(2), 391-397. doi: 10.1039/c0em00130a
- Lee, Y., Gerrity, D., Lee, M., Bogeat, A. E., Salhi, E., Sujanie Gamage, S., . . . von Gunten, U. (2013). Prediction of Micropollutant Elimination during Ozonation of Municipal Wastewater Effluents: Use of Kinetic and Water Specific Information. *Environ. Sci. & Technol.*, 47, 5872-5881. doi: dx.doi.org/10.1021/es400781r
- Lee, Y., Kovalova, L., McArdell, C. S., & von Gunten, U. (2014b). Prediction of micropollutant elimination during ozonation of a hospital wastewater effluent. *Water Res, 64*, 134-148. doi: 10.1016/j.watres.2014.06.027
- Lei, L., Gu, L., Zhang, X., & Su, Y. (2007). Catalytic oxidation of highly concentrated real industrialwastewater by integrated ozone and activated carbon. *Appl.Catal. A: Gen.*, 327, 287–294. doi: 10.1016/j.apcata.2007.05.027
- Leonard, N. (2020). IDEXX water testing solution blog, Accessed April 2020, from https://www.idexx.com/en/blogs/idexx-water-testing-solutions/what-s-the-differencebetween-mpn-and-cfu/

- Li, H., & Zhou, B. (2019). Degradation of atrazine by catalytic ozonation in the presence of iron scraps: performance, transformation pathway, and acute toxicity. *J. of Environ. Sci. & Health, Part B, 54*, 432-440. doi: https://doi.org/10.1080/03601234.2019.1574175
- Li, J. G., Dhanvantari, S., Averill, D., & Biswas, N. (2003). Windsor combined sewer overflow treatability study with chemical coagulation. *Water Qual. Res. J. of Can. 38*(2), 317-334.
- Li, J. G., Horneck, H., Averill, D., McCorquodale, J. A., & Biswas, N. (2004). High-Rate Retention Treatment Basins for CSO Control in Windsor, Ontario. *Water Qual. Res. J. of Can.*, 39(4), 449–456. doi: 0.2166/wqrj.2004.056
- Lin, A. Y.-C., S.C.Panchangam, Chang, C. Y., Hong, P. K. A., & Hsueh, H. F. (2012). Removal of perfluorooctanoic acid and perfluorooctane sulfonate via ozonation under alkaline condition. J. of Hazard. Mater., 243, 272-277. doi: http://dx.doi.org/10.1016/j.jhazmat.2012.10.029
- Ling, Y., Klemes, M. J., Xiao, L., Alsbaiee, A., Dichtel, W. R., & Helbling, D. E. (2017). Benchmarking Micropollutant Removal by Activated Carbon and Porous β-Cyclodextrin Polymers under Environmentally Relevant Scenarios. *Environ. Sci. & Technol., 51*, 7590-7598. doi: 10.1021/acs.est.7b00906
- Liu, Z. Q., Ma, J., Cui, Y. H., Zhao, L., & Zhang, B. P. (2011). Factors affecting the catalytic activity of multi-walled carbon nanotube for ozonation of oxalic acid. *Sep. & Purif. Technol.*, 78(2), 147-153. doi: 10.1016/j.seppur.2011.01.034
- Madoux-Humery, A. S., Dorner, S., Sauve, S., Aboulfadl, K., Galarneau, M., Servais, P., & Prevost, M. (2016). The effects of combined sewer overflow events on riverine sources of drinking water. *Water Res*, 92, 218-227. doi: 10.1016/j.watres.2015.12.033

- Maniero, M. G., Bila, D. M., & Dezotti, M. (2008). Degradation and estrogenic activity removal of 17β-estradiol and 17α-ethinylestradiol by ozonation and O₃/H₂O₂. *Sci. of the Total Environ.*, 407(1), 105-115. doi: https://doi.org/10.1016/j.scitotenv.2008.08.011
- Margot, J., Kienle, C., Magnet, A., Weil, M., Rossi, L., de Alencastro, L. F., . . . Barry, D. A. (2013). Treatment of micropollutants in municipal wastewater: ozone or powdered activated carbon? *Sci Total Environ*, 461-462, 480-498. doi: 10.1016/j.scitotenv.2013.05.034
- Margot, J., Rossi, L., Barry, D. A., & Holliger, C. (2015). A review of the fate of micropollutants in wastewater treatment plants. *Wiley Interdisciplinary Reviews: Water*, 2(5), 457-487. doi: 10.1002/wat2.1090
- McDowell, D. C., Huber, M. M., Wagner, M., Gunten, U. V., & Ternes, T. A. (2005). Ozonation of Carbamazepine in Drinking Water: Identification and Kinetic Study of Major Oxidation Products. *Environ. Sci. Technol.*, 39, 8014-8022.
- McLachlan, M. S., Zou, H., & Gouin, T. (2016). Using Benchmarking To Strengthen the Assessment of Persistence Environ. Sci. & Technol. , 51(1), 4-11. doi: 10.1021/acs.est.6b03786
- Mecha, A. C., Onyango, M. S., Ochieng, A., Fourie, C. J. S., & Momba, M. N. B. (2016). Synergistic effect of UV–vis and solar photocatalytic ozonation on the degradation of phenol in municipal wastewater: A comparative study. *J. of Catal., 341*, 116-125. doi: 10.1016/j.jcat.2016.06.015
- Melero, J. A., Martinez, F., Botas, J. A., Molina, R., & Pariente, M. I. (2009). Heterogeneous catalytic wet peroxide oxidation systems for the treatment of an industrial pharmaceutical wastewater. *Water Res, 43*(16), 4010-4018. doi: 10.1016/j.watres.2009.04.012

- Merle, T., Pic, J. S., Manero, M. H., Mathé, S., & Debellefontaine, H. (2010). Influence of activated carbons on the kinetics and mechanisms of aromatic molecules ozonation. *Catal. Today*, 151(1-2), 166-172. doi: 10.1016/j.cattod.2010.03.044
- Metcalf, & Eddy. (2014). *Wastewater Engineering: Treatment and Reuse* (Fifth ed.). United States of America: Mc Graw Hill Education.
- Miege, C., Choubert, J. M., Ribeiro, L., Eusebe, M., & M.Coquery. (2009). Fate of pharmaceuticals and personal care products in wastewater treatmentplants – Conception of a database and first results. *Environ. Pollut.*, 157, 1721-1726. doi: doi:10.1016/j.envpol.2008.11.045
- Mohammed, S., & Shanshool, H. A. (2009). Phosphorus Removal from Water and Waste Water by Chemical Precipitation Using Alum and Calcium Chloride. *Iraqi J. of Chem. and Petroleum Eng.*, 10.
- Molnar, J., Agbaba, J., Dalmacija, B., Klašnja, M., Watson, M., & Kragulj, M. (2012). Effects of Ozonation and Catalytic Ozonation on the Removal of Natural Organic Matter from Groundwater. J. of Environ. Eng., 138(7), 804-808.
- Mutzner, L., Staufer, P., & Ort, C. (2016). Model-based screening for critical wet-weather discharges related to micropollutants from urban areas. *Water Res.*, 104. doi: <u>http://dx.doi.org/10.1016/j.watres.2016.08.003</u>
- Nawrocki, J., & Fijołek, L. (2013). Effect of aluminium oxide contaminants on the process of ozonedecomposition in water. *Appl. Catal. B: Environ.*, 142, 533-537. doi: <u>http://dx.doi.org/10.1016/j.apcatb.2013.05.069</u>
- Ning, B., Graham, N., Zhang, Y., Nakonechny, M., & Gamal El-Din, M. (2007). Degradation of Endocrine Disrupting Chemicals by Ozone/AOPs. *Ozone: Sci. & Eng.*, 29(3), 153-176. doi: 10.1080/01919510701200012

- Nothe, T., Fahlenkamp, H., & von Sonntag, C. (2009). Ozonation of wastewater: Rate of ozone consumption and hydroxyl radical yield. *Environ. Sci. Technol.*, *43*(15), 5990–5995.
- Orge, C. A., Órfão, J. J. M., & Pereira, M. F. R. (2012). Carbon xerogels and ceria–carbon xerogel materials as catalysts in the ozonation of organic pollutants. *Appl.Catal. B: Environ.*, 126, 22-28. doi: 10.1016/j.apcatb.2012.06.029
- Oulton, R., Haase, J. P., Kaalberg, S., Redmond, C. T., Nalbandian, M. J., & Cwiertny, D. M. (2015). Hydroxyl radical formation during ozonation of multiwalled carbon nanotubes: performance optimization and demonstration of a reactive CNT filter. *Environ Sci Technol*, 49(6), 3687-3697. doi: 10.1021/es505430v
- Oulton, R. L., Kohn, T., & Cwiertny, D. M. (2010). Pharmaceuticals and personal care products in effluent matrices: A survey of transformation and removal during wastewater treatment and implications for wastewater management. *J. of Environ. Monit*.(11), 1956-1978. doi: 10.1039/c0em00068j.
- Patel, M., Kumar, R., Kishor, K., Mlsna, T., Charles U. Pittman, J., & Mohan, D. (2016). Pharmaceuticals of Emerging Concern in Aquatic Systems: Chemistry, Occurrence, Effects, and Removal Methods. *Chem. Rev.*, 119, 3510-3673. doi: 10.1021/acs.chemrev.8b00299
- Pelch, K. E., Reade, A., A.M.Wolffe, T., & F.Kwiatkowski, C. (2019). PFAS health effects database: Protocol for a systematic evidence map. *Environ. Int.*, 130. doi: https://doi.org/10.1016/j.envint.2019.05.045
- Pereira, M. F. R., Gonçalves, A. G., & Orfão, J. J. M. (2014). Carbon materials as catalysts for the ozonation of organic pollutants in water. *Bol. del Grupo Esp. del Carbon, 31*, 18-24.
- Phillips, P. J., Chalmers, A. T., Gray, J. L., Kolpin, D. W., Foreman, W. T., & Wall, G. R. (2012). Combined sewer overflows: an environmental source of hormones and wastewater micropollutants. *Environ Sci Technol*, 46(10), 5336-5343. doi: 10.1021/es3001294

- Pifer, A. D., & Fairey, J. L. (2014). Suitability of Organic Matter Surrogates to Predict Trihalomethane Formation in Drinking Water Sources. *Environ. Eng. Sci.*, 31. doi: 10.1089/ees.2013.0247
- Pongmala, K., Autixier, L., Madoux-Humery, A. S., Fuamba, M., Galarneau, M., Sauvé, S., . . . Dorner, S. (2015). Modelling total suspended solids, E. coli and carbamazepine, a tracer of wastewater contamination from combined sewer overflows. *J. of Hydrol.*, *531*, 830-839. doi: 10.1016/j.jhydrol.2015.10.042
- Pourrezaei, P., Afzal, A., Ding, N., Islam, M. S., Moustafa, A., Drzewicz, P., . . . Gamal-El-Din,
 M. (2010). Physico-Chemical Processes. *Water Environ. Res.*, 82(10), 997-1072.
- Pourrezaei, P., Alpatova, A., Chelme-Ayala, P., Perez-Estrada, L. A., Jensen-Fontaine, M., Le, X.
 C., & Gamal-El-Din, M. (2014). Impact of petroleum coke characteristics on the adsorption of the organic fractions from oil sands process-affected water. *Int. J of Environ. Sci. & Technol., 11*, 2037–2050. doi: 10.1007/s13762-013-0406-x
- Pourrezaei, P., Drzewicz, P., Wang, Y., Gamal-El-Din, M., Perez-Estrada, L. A., Martin, J. W., .
 . . Giesy, J. P. (2011). The Impact of Metallic Coagulants on the Removal of Organic Compounds from Oil Sands Process-Affected Water. *Environ. Sci. Technol.*, 45(19), 8452-8459. doi: dx.doi.org/10.1021/es201498v | Environ. Sci. Technol. 2011, 45, 8452–8459
- Qi, W., Singer, H., Berg, M., Muller, B., Pernet-Coudrier, B., Liu, H., & Qu, J. (2015). Elimination of polar micropollutants and anthropogenic markers by wastewater treatment in Beijing, China. *Chemosphere*, *119*, 1054-1061. doi: 10.1016/j.chemosphere.2014.09.027
- Ratnaweera, H., & Fettig, J. (2015). State of the Art of Online Monitoring and Control of the Coagulation Process. *Water*, 7(11), 6574-6597. doi: 10.3390/w7116574
- Razavi, B., Song, W., Cooper, W. J., Greaves, J., & Jeong[†], J. (2009). Free-Radical-Induced Oxidative and Reductive Degradation of Fibrate Pharmaceuticals: Kinetic Studies and

Degradation Mechanisms. J. of Phys. Chem., 113, 1287-1294. doi: https://doi.org/10.1021/jp808057c

- Real, F. J., Benitez, F. J., Acero, J. L., & Gonzalez, M. (2007). Removal of diazinon by various advanced oxidation processes. J. of Chem. Technol. & Biotechnol., 82, 566-574. doi: https://doi.org/10.1002/jctb.1702
- Restivo, J., Orfao, J. J., Armenise, S., Garcia-Bordeje, E., & Pereira, M. F. (2012). Catalytic ozonation of metolachlor under continuous operation using nanocarbon materials grown on a ceramic monolith. *J. Hazard. Mater.*, 239-240, 249-256. doi: 10.1016/j.jhazmat.2012.08.073
- Restivo, J., Órfão, J. J. M., Pereira, M. F. R., Garcia-Bordejé, E., Roche, P., Bourdin, D., . . . Derrouiche, S. (2013). Catalytic ozonation of organic micropollutants using carbon nanofibers supported on monoliths. *Chem. Eng. J., 230*, 115-123. doi: 10.1016/j.cej.2013.06.064
- Reungoat, J., Macova, M., Escher, B. I., Carswell, S., Mueller, J. F., & Keller, J. (2010). Removal of micropollutants and reduction of biological activity in a full scale reclamation plant using ozonation and activated carbon filtration. *Water Res.*, 44(2), 625-637. doi: 10.1016/j.watres.2009.09.048
- Rizzo, L., Malato, S., Antakyali, D., Beretsou, V. G., Đolić, M. B., Wolfgang Gernjak, . . . Fatta-Kassinos, D. (2019). Consolidated vs new advanced treatment methods for the removal of contaminants of emerging concern from urban wastewater. *Sci. of the Total Environ.*, 655, 986-1008. doi: https://doi.org/10.1016/j.scitotenv.2018.11.265
- Rodayan, A., Segura, P. A., & Yargeau, V. (2014). Ozonation of wastewater: Removal and transformation products of drugs of abuse. *Sci. of the Total Environ.*, 487, 763-770. doi: https://doi.org/10.1016/j.scitotenv.2013.11.023

- Rodriguez, E. M., Fernandez, G., Alvarez, P. M., & Beltran, F. J. (2012). TiO2 and Fe (III) photocatalytic ozonation processes of a mixture of emergent contaminants of water. *Water Res.*, 46(1), 152-166. doi: 10.1016/j.watres.2011.10.038
- Rozas, O., Baeza, C., Nunez, K., Rossner, A., Urrutia, R., & Mansilla, H. D. (2017). Organic micropollutants (OMPs) oxidation by ozone: Effect of activated carbon on toxicity abatement. *Sci. of the Total Environ.*, 590-591, 430-439. doi: 10.1016/j.scitotenv.2016.12.120
- Ryu, J., Oh, J., Snyder, S. A., & Yoon, Y. (2014). Determination of micropollutants in combined sewer overflows and their removal in a wastewater treatment plant (Seoul, South Korea). *Environ Monit Assess, 186*(5), 3239-3251. doi: 10.1007/s10661-013-3613-5
- Sánchez-Polo, M., Salhi, E., Rivera-Utrilla, J., & Gunten, U. v. (2007). Combination of Ozone with Activated Carbon as an Alternative to Conventional Advanced Oxidation Processes. *Ozone: Sci. & Eng., 28.* doi: https://doi.org/10.1080/01919510600714170
- Sanchez-Polo, M., von Gunten, U., & Rivera-Utrilla, J. (2005). Efficiency of activated carbon to transform ozone into OH radicals: influence of operational parameters. *Water Res.*, 39(14), 3189-3198. doi: 10.1016/j.watres.2005.05.026
- Saritha, V., Srinivas, N., & Vuppala, S. (2017). Analysis and optimization of coagulation and flocculation process. *Applied Water Science*, *7*, 451-460.
- Sarkar, S., Kamilya, D., & Mal, B. C. (2007). Effect of geometric and process variables on the performance of inclined plate settlers in treating aquacultural waste. *Water Res.*, 41, 993-1000. doi: 10.1016/j.watres.2006.12.015
- Schuchmann, M. N., Bothe, E., von-Sonntag, J., & von-Sonntag, C. (1998). Reaction of OH radicals with benzoquinone in aqueous solutions. A pulse radiolysis study. J. Chem. Soc., Perkin Trans. 2, 791-796. doi: https://doi.org/10.1039/A708772A

- Seredyńska-Sobecka, B., Tomaszewska, M., & W.Morawski, A. (2005). Removal of micropollutants from water by ozonation/biofiltration process. *Desalination*, 182(1-3), 151-157. doi: https://doi.org/10.1016/j.desal.2005.03.015
- Shu, Z., Bolton, J. R., Belosevic, M., & Gamal-El-Din, M. (2013). Photodegradation of emerging micropollutants using the medium-pressure UV/H₂O₂ Advanced Oxidation Process. *Water Res.*, 47, 2881-2889. doi: <u>http://dx.doi.org/10.1016/j.watres.2013.02.045</u>
- Shu, Z., Singh, A., Klamerth, N., McPhedran, K., Bolton, J. R., Belosevic, M., & Gamal-El-Din, M. (2016). Pilot-Scale UV/H₂O₂ Advanced Oxidation Process for Municipal Reuse Water: Assessing Micropollutant Degradation and Estrogenic Impacts on Goldfish (Carassius auratus L.). *Water Res, 101*, 157-166. doi: http://dx.doi.org/10.1016/j.watres.2016.05.0790043-1354/©2016
- Smoczynski, L., Munska, K. T., Kosobucka, M., & Pierozynski, B. (2014). Phosphorus and COD Removal from Chemically and Electrochemically Coagulated Wastewater. *Environ. Prot. Eng.*, 40(3), 63-73. doi: 10.5277/epe140305
- Song, Y., Breider, F., Ma, J., & von Gunten, U. (2017). Nitrate formation during ozonation as a surrogate parameter for abatement of micropollutants and the N-nitrosodimethylamine (NDMA) formation potential. *Water Res.*, 122, 246-257. doi: 10.1016/j.watres.2017.05.074
- Song, Z., Zhang, Y., Liu, C., Xu, B., Qi, F., Yuan, D., & Pu, S. (2019). Insight into OH and O2⁻⁷ formation in heterogeneous catalytic ozonation by delocalized electrons and surface oxygen-containing functional groups in layered-structure nanocarbons. *Chem. Eng. J.*, 357, 655-666. doi: https://doi.org/10.1016/j.cej.2018.09.182
- Soonthornnonda, P., & Christensen, E. R. (2008). Source apportionment of pollutants and flows of combined sewer wastewater. *Water Res, 42*(8-9), 1989-1998. doi: 10.1016/j.watres.2007.11.034

- SOS-ChromoTest. (2019). SOS Chromotest Procedure S9 Activation and Express Strains Version6.6. In E. B.-D. P. Inc. (Ed.). Ontario, Canada: ebpi Inc. .
- Staehelin, J., & Hoigne, J. (1985). Decomposition of ozone in water in the presence of organic solutes acting as promoters and inhibitors of radical chain reactions. *Environ. Sci. & Technol.*, 19(12), 1206. doi: https://doi.org/10.1021/es00142a012
- Stamm, C., Joss, A., Ackermann, M., & R.I.L.Eggen. (2016). Unravelling the Impacts of Micropollutants in Aquatic Ecosystems: Interdisciplinary Studies at the Interface of Large-Scale Ecology Adv. in Ecol. Res. (Vol. 55, pp. 183-223). doi: <u>http://dx.doi.org/10.1016/bs.aecr.2016.07.002183</u>

Stucki, S. (1988). Process technologies for water treatment. New York Plenum Press.

- Suarez, S., Dodd, M. C., Omil, F., & vonGunten, U. (2007). Kinetics of triclosan oxidation by aqueous ozone and consequent loss of antibacterial activity: Relevance to municipal wastewater ozonation. *Water Res.*, 41, 2481-2490. doi: https://doi.org/10.1016/j.watres.2007.02.049
- Sui, M., Xing, S., Sheng, L., Huang, S., & Guo, H. (2012). Heterogeneous catalytic ozonation of ciprofloxacin in water with carbon nanotube supported manganese oxides as catalyst. J. Hazard. Mater., 227-228, 227-236. doi: 10.1016/j.jhazmat.2012.05.039
- Sui, Q., Huang, J., Deng, S., Chen, W., & Yu, G. (2011). Seasonal Variation in the Occurrence and Removal of Pharmaceuticals and Personal Care Products in Different Biological Wastewater Treatment Processes. *Environ. Sci. & Technol., 45*, 3341-3348. doi: dx.doi.org/10.1021/es200248d |
- Sui, Q., Huang, J., Deng, S., Yu, G., & Fan, Q. (2010). Occurrence and removal of pharmaceuticals, caffeine and DEET in wastewater treatment plants of Beijing, China. *Water Res.*, 44(2), 417-426. doi: 10.1016/j.watres.2009.07.010
- Szlachta, M., & Adamsk, W. (2009). Effects of natural organic matter removal by integrated processes: alum coagulation and PAC-adsorption. *Water Sci. and Technol.*, 59(10), 1951-1957. doi: 10.2166/wst.2009.217
- Tik, S., & Vanrolleghem, P. A. (2017). Chemically enhancing primary clarifiers: model-based development of a dosing controller and full-scale implementation. *Water Science and Technology*, 75(5), 1185-1193. doi: 10.2166/wst.2016.600
- Tolouei, S., Jean-Baptiste Burneta, Autixiera, L., Taghipoura, M., Bonsteelb, J., Duy, S. V., ...
 Dorner, S. (2019). Temporal variability of parasites, bacterial indicators, and wastewater micropollutants in a water resource recovery facility under various weather conditions. *Water Res.*, 148, 446-458. doi: https://doi.org/10.1016/j.watres.2018.10.068
- Tondera, K., Klaer, K., Gebhardt, J., Wingender, J., Koch, C., Horstkott, M., . . . Pinnekamp, J. (2015). Reducing pathogens in combined sewer overflows using ozonation or UV irradiation. *Int. J. Hyg. Environ. Health.*, 218(8), 731-741. doi: 10.1016/j.ijheh.2015.09.002
- Tondera, K., Klaer, K., Koch, C., Hamza, I. A., & Pinnekamp, J. (2016). Reducing pathogens in combined sewer overflows using performic acid. *Int. J. Hyg. Environ. Health, 219*(7 Pt B), 700-708. doi: 10.1016/j.ijheh.2016.04.009
- Tondera, K., Koenen, S., & Pinnekamp, J. (2013). Survey monitoring results on the reduction of micropollutants, bacteria, bacteriophages and TSS in retention soil filters. *Water Sci. & Technol.*, 68(5), 1004-1012. doi: https://doi.org/10.2166/wst.2013.340
- Trinh, T. K., & Kang, L. S. (2015). Coagulation of phosphorus: effects of Al(III) species (Al-a, Al-b, and Al-c). *Desalination and Water Treatment*, 53(2), 485-492. doi: 10.1080/19443994.2013.841106
- Trojanowicza, M., Bojanowska-Czajkaa, A., Bartosiewicza, I., & Kulisaa, K. (2018). Advanced Oxidation/Reduction Processes treatment for aqueous perfluorooctanoate (PFOA) and

perfluorooctanesulfonate (PFOS) – A review of recent advances. *Chem. Eng. J.*, *336*, 170-199. doi: https://doi.org/10.1016/j.cej.2017.10.153

- USEPA. (2010). *NPDES Permit Writers' Manual* (EPA-833-K-10-001). Washington, DC: Office of Water Washington DC.
- USEPA. (2018). Combined Sewer Overflows (CSOs), 2018, from https://www.epa.gov/npdes/combined-sewer-overflows-csos. Accessed June 2019
- USEPA. (2020). Drinking water treatability database Powdered Activated Carbon. From https://iaspub.epa.gov/tdb/pages/treatment/treatmentOverview.do?treatmentProcessId=207 4826383. Accessed April 2020
- Uslu, M., Seth, R., Jasim, S., Tabe, S., & Biswas, N. (2015). Reaction Kinetics of Ozone with Selected Pharmaceuticals and Their Removal Potential from a Secondary Treated Municipal Wastewater Effluent in the Great Lakes Basin. *Ozone: Sci. & Eng.*, 37, 36-44. doi: 10.1080/01919512.2014.929520
- Verlicchi, P., Galletti, A., Petrovic, M., & Barceló, D. (2012). Micro-pollutants in Hospital Effluent: Their Fate, Risk and Treatment Options. . In D. Barceló (Ed.), Emerging Organic Contaminants and Human Health. The Handbook of Environmental Chemistry (Vol. 20, pp. 139-171). Berlin, Heidelberg: Springer. doi: https://doi.org/10.1007/698 2011 134
- von Sonntag, C., & von Gunten, U. (2012). *Chemistry of Ozone in Water and Wastewater Treatment*: IWA Publishing
- Wang, B., Zhang, H., Wang, F., Xiong, X., Tian, K., Sun, Y., & Yu, T. (2019a). Application of Heterogeneous Catalytic Ozonation for Refractory Organics in Wastewater. *Catal.*, 9(3). doi: https://doi.org/10.3390/catal9030241
- Wang, C., Alpatova, A., McPhedran, K. N., & Gamal El-Din, M. (2015a). Coagulation/flocculation process with polyaluminum chloride for theremediation of oil

sands process-affected water: Performance and mechanism study. *J. Environ. Manage.*, *160*, 254-262. doi: http://dx.doi.org/10.1016/j.jenvman.2015.06.0250301-4797/©2015

- Wang, J., & Bai, Z. (2017). Fe-based catalysts for heterogeneous catalytic ozonation of emerging contaminants in water and wastewater. *Chem. Eng. J.*, 312, 79-98. doi: http://dx.doi.org/10.1016/j.cej.2016.11.118
- Wang, W., Xu, Z., Li, H., & Jin, W. (2015b). Optimization of coagulation–flocculation process for combined sewer overflow wastewater treatment using response surface methodology. *Desalination & Water Treat.*, 57(32), 14824-14832. doi: 10.1080/19443994.2015.1067832
- Wang, Y., Cao, H., Chen, C., Xie, Y., Sun, H., Duan, X., & Wang, S. (2019b). Metal-free catalytic ozonation on surface-engineered graphene: Microwave reduction and heteroatom doping. *Chem. Eng. J.*, 355, 118-129. doi: https://doi.org/10.1016/j.cej.2018.08.134
- Wang, Y., Xie, Y., Sun, H., Xiao, J., Cao, H., & Wang, S. (2015c). Hierarchical shape-controlled mixed-valence calcium manganites for catalytic ozonation of aqueous phenolic compounds. *Catal. Sci. and Technol.* doi: 10.1039/c5cy01967b
- Wei, N., Zhang, Z. G., Liu, D., Wu, Y., Wang, J., & Wang, Q. H. (2015). Coagulation behavior of polyaluminum chloride: Effects of pH and coagulant dosage. *Chinese Journal of Chemical Engineering*, 23(6), 1041-1046. doi: 10.1016/j.cjche.2015.02.003
- Wert, E. C., L.Rosario-Ortiz, F., & Snyder, S. A. (2009). Effect of ozone exposure on the oxidation of trace organic contaminants in wastewater. *Water Res.*, 43(4), 1005-1014. doi: 10.1016/j.watres.2008.11.050
- Westerhoff, P., Aiken, G., Amy, G., & Debroux, J. (1999). Relationships between the structure of natural organic matter and its reactivity towards molecular ozone and hydroxyl radicals. *Water Res.*, 33(10), 2265-2276. doi: https://doi.org/10.1016/S0043-1354(98)00447-3

- Westerhoff, P., Yoon, Y., Snyder, S., & Wert, E. (2005). Fate of Endocrine-Disruptor, Pharmaceutical, and Personal Care Product Chemicals during Simulated Drinking Water Treatment Processes. *Envrion. Sci. & Technol.*, 39, 6649-6663. doi: 10.1021/es0484799
- Wieckowski, A. B., Najder-Kozdrowska, L., Rechnia, P., Malaika, A., Krzyzynska, B., & Kozłowski, M. (2015). EPR Characteristics of Activated Carbon for Hydrogen Production by the Thermo-Catalytic Decomposition of Methane. *ACTA PHYS. POLONICA A, 130*(3). doi: 0.12693/APhysPolA.130.701
- Wildhaber, S. Y., Mestankova, H., Scharer, M., Schirmer, K., Salhi, E., & von Gunten, U. (2015).
 Novel test procedure to evaluate the treatability of wastewater with ozone. *Water Res, 75*, 324-335. doi: 10.1016/j.watres.2015.02.030
- Wojtenko, I., Stinson, M. K., & Field, R. (2001a). Challenges of Combined Sewer Overflow Disinfection by Ultraviolet Light Irradiation. *Critical Reviews in Environ. Sci. & Technol.*, 31(3), 223-239. doi: 10.1080/20016491089217
- Wojtenko, I., Stinson, M. K., & Field, R. (2001b). Performance of Ozone as a Disinfectant for Combined Sewer Overflow. *Critical Reviews in Environ. Sci. & Technol.*, 31(4), 295-309. doi: 10.1080/20016491089235
- Waters Protection Ordinance AS 1998 2863 C.F.R. (2020).
- Xing, L., Xie, Y., Minakata, D., Cao, H., Xiao, J., Zhang, Y., & Crittenden6, J. C. (2014). Activated carbon enhanced ozonation of oxalate attributed to HO. oxidation in bulk solution and surface oxidation: Effect of activated carbon dosage and pH. *J. of Environ. Sci.*, 26, 2095-2105. doi: <u>http://dx.doi.org/10.1016/j.jes.2014.08.009</u>
- Xiong, F., & Graham, N. J. D. (2008). Rate Constants for Herbicide Degradation by Ozone. *Ozone Sci. & Eng., 14*(4), 283-301. doi: 10.1080/01919519208552274

- Xu, P., Janex, M.-L., Savoye, P., Cockx, A., & Lazarova, V. (2002). Wastewater disinfection by ozone: main parameters for process design. *Water Res.*, 36, 1043-1055.
- Yamashita, T., & Yamamoto-Ikemoto, R. (2014). Nitrogen and Phosphorus Removal from Wastewater Treatment Plant Effluent via Bacterial Sulfate Reduction in an Anoxic Bioreactor Packed with Wood and Iron *Int. J. Environ. Res. Public Health 2, 11*, 9835-9853.
- Yang, W., Paetkau, M., & Cicek, N. (2010). Improving the performance of membrane bioreactors by powdered activated carbon dosing with cost considerations *Water Sci. and Technol.*, 62(1), 172-179. doi: https://doi.org/10.2166/wst.2010.276
- Ying-hui, Y., Jun, M., & Yan-jun, H. (2006). Degradation of 2,4-dichlorophenoxyacetic acid in water by ozonehydrogen peroxide process. J. of Environ. Sci. , 18(6), 1043-1049.
- Young, S., Stanley, S. J., & Smith, D. W. (2000). Effect of mixing on the kinetics of polymeraided flocculation. [Article]. Journal of Water Services Research and Technology-Aqua, 49(1), 1-8.
- Yu, J., & Hu, J. (2011). Adsorption of Perfluorinated Compounds onto Activated Carbon and Activated Sludge. J.of Environ. Eng., 137(10), 945-951. doi: 10.1061/(asce)ee.1943-7870.0000402
- Yu, W.-z., Gregory, J., Campos, L., & Li, G. (2011). The role of mixing conditions on floc growth, breakage and re-growth. *Chem. Eng. J.*, 171(2), 425-430. doi: 10.1016/j.cej.2011.03.098
- Yu, W., Li, G., Xu, Y., & Yang, X. (2009). Breakage and re-growth of flocs formed by alum and PAC1. *Powder Technol.*, 189(3), 439-443. doi: 10.1016/j.powtec.2008.07.008
- Yu, Z., Peldszus, S., Anderson, W. B., & Huck, P. M. (2005). Adsorption of Selected Pharmaceuticals and Endocrine Disrupting Substances by GAC at Low Concentration Levels. Paper presented at the AWWA Water Quality Technology Conference and Exposition (WQTC), Quebec City, Quebec.

- Yukselen, M. A., & Gregory, J. (2004). The effect of rapid mixing on the break-up and reformation of flocs. J. of Chem. Technol. & Biotechnol., 79(7), 782-788. doi: 10.1002/jctb.1056
- Zhang, K., Deletic, A., Page, D., & McCarthy, D. T. (2015). Surrogates for herbicide removal in stormwater biofilters. *Water Res.*, 81, 64-71. doi: http://dx.doi.org/10.1016/j.watres.2015.05.043
- Zhang, X., Chen, P., Wu, F., Deng, N., Liu, J., & Fang, T. (2006). Degradation of 17 aethinylestradiol in aqueous solution by ozonation. *J. Hazard. Mater. B*, 291-298.
- Zhao, L., Ma, J., Sun, Z.-z., & Zha, X.-d. (2008). Catalytic ozonation for the degradation of nitrobenzene in aqueous solution by ceramic honeycomb-supported manganese. *Appli. Catal. B: Environ.*, 83, 256-264. doi: 10.1016/j.apcatb.2008.02.009
- Zhou, H., Liu, J., Xia, H., Zhang, Q., Ying, T., & Hu, T. (2015). Removal and reduction of selected organic micro-pollutants in effluent sewage by the ozone-based oxidation processes. *Chem. Eng. J.*, 269, 245-254. doi: 10.1016/j.cej.2015.01.116
- Zhou, Y., Liang, Z., & Wang, Y. (2008). Decolorization and COD removal of secondary yeast wastewater effluents by coagulation using aluminum sulfate. *Desalination*, 225(1-3), 301-311. doi: 10.1016/j.desal.2007.07.010
- Zhu, K., Gama-El-Din, M., Moawad, A. K., & Bromley, D. (2010). Physical and Chemical Processes for Removing Suspended Solids and Phosphorus from Liquid Swine Manure. *Environ. Technol.*, 25(10), 1177-1187.
- Zhu, W., Seth, R., & Lalman, J. (2007). Evaluation of a micro carrier weighted coagulation flocculation process for the treatment of combined sewer overflow. *Environmental Technology*, 28(7), 761-770. doi: 10.1080/09593332808618831

- Zimmermann, S. G., Wittenwiler, M., Hollender, J., Krauss, M., Ort, C., Siegrist, H., & von Gunten, U. (2011). Kinetic assessment and modeling of an ozonation step for full-scale municipal wastewater treatment: micropollutant oxidation, by-product formation and disinfection. *Water Res.*, 45(2), 605-617. doi: 10.1016/j.watres.2010.07.080
- Ziska, A. D., Park, M., Anumol, T., & Snyder, S. A. (2016). Predicting trace organic compound attenuation with spectroscopic parameters in powdered activated carbon processes. *Chemosphere 156*, 163-171. doi: <u>http://dx.doi.org/10.1016/j.chemosphere.2016.04.073</u>

APPENDIX

1. Supplementary Tables and Figures

Table A	.1 Predicted	half-lives of	of MPs in	a reaction w	ith molecula	r O3 onl	y under	different O	3 S	pecific doses.
---------	--------------	---------------	-----------	--------------	--------------	----------	---------	-------------	-----	----------------

Individually treated		In mixture 0.27 g O ₃ /g DOC		In mixture 2.17 g O ₃ /g DOC			In mixture 21.64 g O ₃ /g DOC					
	kos			Molar			Molar			Molar		
MP	(M ⁻¹ s ⁻¹)*	k' (s ⁻¹)	t1/2 (s)	conc.	k' (s ⁻¹)**	t1/2 (s)***	conc.	k' (s-1)	t1/2 (s)	conc.	k' (s-1)	t1/2 (s)
				(μΝΙ)			(μΝΙ)			(μΝΙ)		
EE2	7.00E+09	1.46E+06	4.75E-07	13.50	7.77E+04	8.92E-06	1.69	7.71E+04	8.99E-06	0.17	7.71E+04	8.99E-06
TCN	3.80E+07	7.92E+03	8.76E-05	13.82	4.32E+02	1.60E-03	1.73	4.29E+02	1.62E-03	0.17	4.29E+02	1.62E-03
SMX	5.70E+05	1.19E+02	5.84E-03	15.81	7.41E+00	9.35E-02	1.98	7.36E+00	9.42E-02	0.20	7.36E+00	9.42E-02
GFL	4.90E+05	1.02E+02	6.79E-03	15.98	6.44E+00	1.08E-01	2.00	6.39E+00	1.08E-01	0.20	6.39E+00	1.08E-01
CBZ	3.00E+05	6.25E+01	1.11E-02	16.93	4.18E+00	1.66E-01	2.12	4.15E+00	1.67E-01	0.21	4.15E+00	1.67E-01
ТМР	2.70E+05	5.63E+01	1.23E-02	13.78	3.06E+00	2.26E-01	1.72	3.03E+00	2.29E-01	0.17	3.03E+00	2.29E-01
NPX	2.00E+05	4.17E+01	1.66E-02	17.38	2.86E+00	2.42E-01	2.17	2.83E+00	2.45E-01	0.22	2.83E+00	2.45E-01
МСР	101	2.10E-02	3.29E+01	16.34	1.36E-03	5.11E+02	2.33	1.53E-03	4.52E+02	0.23	1.53E-03	4.52E+02
2,4-D	29.1	6.06E-03	1.14E+02	18.10	4.33E-04	1.60E+03	2.26	4.29E-04	1.62E+03	0.23	4.29E-04	1.62E+03
IBU	9.6	2.00E-03	3.47E+02	19.39	1.53E-04	4.53E+03	2.42	1.51E-04	4.58E+03	0.24	1.51E-04	4.58E+03
ATZ	6	1.25E-03	5.55E+02	18.56	9.16E-05	7.57E+03	2.32	9.08E-05	7.64E+03	0.23	9.08E-05	7.64E+03
FCZ	1	2.08E-04	3.33E+03	13.07	1.08E-05	6.45E+04	1.63	1.06E-05	6.52E+04	0.16	1.06E-05	6.52E+04
DZN	N/A	1.70E-03	4.08E+02	13.14	8.82E-05	7.86E+03	1.64	8.73E-05	7.94E+03	0.16	8.73E-05	7.94E+03
CDM	N/A			20.92			2.62			0.26		
CLN	N/A			9.43			1.18			0.12		
PFOA	N/A			9.66			1.21			0.12		
PFOS	N/A			7.43			0.93			0.09		
Total M	Iolar conc. (of MPs [MP]t	253.24			31.95			3.19		
Total E	quiv. DOC			36.93			4.62			0.46		

Notes:

N/A = Not available

*Obtained from literature

** $k^{1}(s^{-1}) = k_{O_{3}}(M^{-1}s^{-1}) \times [O_{3}](M) \times \frac{[MP]}{[MP]_{t}}$ *** $t_{1/2} = \frac{Ln2}{k'(s^{-1})}$

	Individually treated		In mixture 0.27 g O ₃ /g DOC		In mixture 2.17 g O ₃ /g DOC			In mixture 21.64 g O ₃ /g DOC				
МР	К•он (М ⁻¹ s ⁻¹)*	k' (s ⁻¹)	t1/2 (s)	Molar conc. (µM)	k' (s ⁻¹)**	t1/2 (s)***	Molar conc. (µM)	k' (s ⁻¹)	t1/2 (s)	Molar conc. (µM)	k' (s ⁻¹)	t1/2 (s)
EE2	9.80E+09	2.04E+06	3.40E-07	13.5	1.09E+05	6.37E-06	1.69	1.08E+05	6.42E-06	0.17	1.08E+05	6.42E-06
TCN	9.60E+09	2.00E+06	3.47E-07	13.82	1.09E+05	6.35E-06	1.73	1.08E+05	6.40E-06	0.17	1.08E+05	6.40E-06
SMX	5.50E+09	1.15E+06	6.05E-07	15.81	7.15E+04	9.69E-06	1.98	7.10E+04	9.76E-06	0.20	7.10E+04	9.76E-06
GFL	7.10E+09	1.48E+06	4.69E-07	15.98	9.33E+04	7.43E-06	2	9.26E+04	7.49E-06	0.20	9.26E+04	7.49E-06
CBZ	8.80E+09	1.83E+06	3.78E-07	16.93	1.23E+05	5.66E-06	2.12	1.22E+05	5.70E-06	0.21	1.22E+05	5.70E-06
ТМР	6.90E+09	1.44E+06	4.82E-07	13.78	7.82E+04	8.86E-06	1.72	7.74E+04	8.96E-06	0.17	7.74E+04	8.96E-06
NPX	9.60E+09	2.00E+06	3.47E-07	17.38	1.37E+05	5.05E-06	2.17	1.36E+05	5.10E-06	0.22	1.36E+05	5.10E-06
МСР	1.9E+09	3.96E+05	1.75E-06	16.34	2.55E+04	2.71E-05	2.33	2.89E+04	2.40E-05	0.23	2.89E+04	2.40E-05
2,4-D	N/A			18.1			2.26			0.23		
IBU	7.4E+09	1.54E+06	4.50E-07	19.39	1.18E+05	5.87E-06	2.42	1.17E+05	5.94E-06	0.24	1.17E+05	5.94E-06
ATZ	3.0E+09	6.25E+05	1.11E-06	18.56	4.58E+04	1.51E-05	2.32	4.54E+04	1.53E-05	0.23	4.54E+04	1.53E-05
FCZ	N/A			13.07	0.00E+00		1.63			0.16		
DZN	8.40E+09	1.75E+06	3.96E-07	13.14	9.08E+04	7.63E-06	1.64	8.98E+04	7.72E-06	0.16	8.98E+04	7.72E-06
CDM	N/A			20.92			2.62			0.26		
CLN	N/A			9.43			1.18			0.12		
PFOA	3.00E+07	6250	1.11E-04	9.66	2.38E+02	2.91E-03	1.21	2.37E+02	2.93E-03	0.12	2.37E+02	2.93E-03
PFOS	3.00E+07	6250	1.11E-04	7.43	1.83E+02	3.78E-03	0.93	1.82E+02	3.81E-03	0.09	1.82E+02	3.81E-03
Total M	olar conc. of	MPs [MP] _t		253.24			31.95			3.19		
Total E	quiv. DOC			36.93			4.62			0.46		

Table A.2 Predicted half-lives of MPs in a reaction with hydroxyl radicals 'OH only under different O₃ specific doses.

Notes:

N/A = Not available

*Obtained from literature

**
$$k^{1}(s^{-1}) = k \cdot_{OH}(M^{-1}s^{-1}) \times [\circ OH](M) \times \frac{[MP]}{[MP]_{t}}$$

*** $t_{1/2} = \frac{Ln2}{k'(s^{-1})}$

МР	Ovidation int	Mass	Reference	
NIF	Oxidation int	(m/z)		
EE2	C6H10O3		129	(Huber et
	C6H11O4		145	al., 2004)
	C14H20O5		267	
	C15H22O7		313	
	$C_{16}H_{21}O_8$		341	
	$C_{14}H_{20}O_4$		251	
	C16H22O7		325	
	$C_2H_2O_4$	Oxalic acid	91.0031	(Zhang et
	$C_3H_4O_4$	Malonic acid	105.0188	al., 2006)
	$C_4H_6O_4$	Succinic acid	119.0344	
	C6H10O4	Adipic acid	147.0657	
	C5H8O4	Glutaric acid	133.0501	
TCN	C ₆ H ₄ Cl ₂ O	2,4-Dichlorophenol	162.9717	(Chen et al.,
	C ₆ H ₅ ClO ₂	4-Chlorocatechol	143.9978	2012)
SMX	C4H6N2O	3-Amino-5-methylisoxazole	99.0558	(Goncalves
	$C_6H_4O_2$	p-Benzoquinone	109.029	et al., 2012)
	C ₂ H ₃ NO ₃	Oxamic acid	90.0191	
	C ₃ H ₄ O ₃	Pyruvic acid		
	$C_2H_2O_4$	Oxalic acid	91.0031	
	C4H4O4	Maleic acid	117.0188	
GFL	N/A			
CBZ	$C_{15}H_{10}N_2O_2$	1-(2-benzaldehyde)-4-hydro-(1H,3H)-quinazoline-2-one	251.082	(McDowell
	$C_{15}H_{10}N_2O_3$	1-(2-benzaldehyde)-(1H,3H)-quinazoline-2,4-dione	267.077	et al., 2005)
	$C_8H_6N_2O_2$	1-(2-benzoic acid)- (1H,3H)-quinazoline-2,4-dione	163.0508	
	C14H9NO2	2-Nitroanthracene or 9-Nitroanthracene	224.0710	(Hubner et
	$C_{15}H_{14}N_2O_3$		309.0645	al., 2014)
	$C_{15}H_{10}N_2O$		267.0775	
	$C_{15}H_{10}N_2O_4$	Phencomycin	283.0725	
	$C_{15}H_{12}N_2O_2$		253.0980	
	$C_{15}H_{12}N_2O_3$	Hydrofuramide	269.0933	
	$C_{15}H_{10}N_2O_2$		269.0933	
	$C_{15}H_{12}N_2O_2$		253.0980	
	$C_{15}H_{10}N_2O_3$	Carbmazepine-o-quinone	267.0770	
	C14H9NO2		224.0709	
	C13H9NO	Acridone	196.0767	
	$C_{14}H_{11}NO_4$		258.0761	
	$C_{14}H_{11}NO_4$		208.0760	
ТМР	$C_{14}H_{19}N_4O_3$		291.1441	(Kuang et
	C13H19N4O4		295.1391	al., 2013)
	C14H19N4O5		323.1335	
	$C_{14}H_{21}N_4O_5$		325.1503	
	$C_{14}H_{19}N_4O_6$		339.1309	
	$C_{14}H_{17}N_4O_3$		277.1216	
	$C_{13}H_{16}N_{3}O_{4}$		278.1156	
	$C_{12}H_{17}N_4O_4$		281.1220	
	$C_{14}H_{19}N_2O_6$		311.1195	
	C14H19N4O7		355.1224	
	$C_{14}H_{21}N_4O_7$		357.1369	
	$C_{11}H_{14}N_3O_3$		236.0939	
	C14H19N4O4		307.1197	
	C11H17N4O7		317.1183	
	C5H6N4O		139.062	

 Table A.3 Oxidation intermediates and by-products for the 17 studied MPs reported in literature.

МР	Oxidation inte	Mass	Reference		
1111	O Aldation inte	inculates by products	(m/z)		
	C5H8N4O		141.0776		
NPX	$C_{13}H_{18}O_4$	2-(4-(2-Mmethoxyethyl)-3-methylphenyl) propanoic acid	239.1283	(Aguilar et	
	$C_{13}H_{14}O_3$	1-(6-Methoxynaphthalene-2-yl) ethylhydroperoxide	219.1021	al., 2019)	
	$C_{12}H_{14}O_3$	2-(4-(2-Hydroxyvinyl)phenyl) acetic acid	207.1021		
	$C_{10}H_{10}O_3$	2-(4-(2-Hydroxyvinyl)phenyl) acetic acid	177.0554		
	$C_9H_{10}O_2$	2-(<i>p</i> -Tolyl)acetic acid	149.0592		
	$C_2H_2O_4$	Oxalic acid	91.0031		
	CH ₂ O ₂		47.0133		
MCP	N/A				
2,4-D	C ₆ H ₄ Cl ₂ O	2,4-Dichlorophenol	162.9717	(Ying-hui et	
	C ₆ H ₅ ClO ₂	Chlorohydroxyquinone	143.9978	al., 2006)	
	C ₆ H ₅ ClO ₂	4-chlorocatechol	143.9978		
	C ₆ H ₄ Cl ₂ O ₂	3,5-Dichlorocatechole	178.9667		
	$C_2H_2O_4$	Oxalic acid	91.0031		
	$C_2H_4O_2$	Acetic acid	61.029		
IBU	C9H9O		133.0653	(Huang et	
	C12H15O5		239.0919	al., 2015)	
	C12H17O		177.1279		
	C12H17O		176.1279		
	C12H15O		175.1123		
	$C_{13}H_{17}O_3$		221.1178		
	C26H33O8		473.2175		
	$C_{13}H_{17}O_2$		205.1229		
ATZ	C ₈ H ₁₃ ClN ₆ O	N-(4-chloro-6-(isopropylamino)-1,3,5-triazin-2-yl)		(Li & Zhou,	
	$C_8H_{15}N_5O$	2-Hydroxy-4-ethylamino-6-isopropylamino-1,3,5-triazine	198.1355	2019)	
	C ₈ H ₁₄ Cl ₁₅ NN ₄	2-Chloro-4-amino-6-isoproylamino-1,3,5-triazine			
	C ₃ H ₃ ClN ₅	2-Chloro-4,6-diamino-1,3,5-triazine	145.0155		
	C ₃ H ₅ N ₅ O	2-Hydroxyl-4,6-diamino-1,3,5-triazine			
FCZ	N/A				
DZN	N/A				
CDM	$C_8H_7N_3O_2$	benzimidazole-2-ylcarbamic-acide		(Kiss &	
	$C_7H_7N_3$	2-amino-benzimidazole	134.0718	Virág,	
	C7H6N2	Benzimidazole	119.0609	2009a)	
	$C_7H_{11}N_3$	2-methyl-amino-aniline	138.1031		
	C ₆ H ₈ N ₂	1,2-diaminobenzene	109.0766		
	C ₆ H ₅ OH	benzene			
	C ₆ H ₅ OH	Phenol			
	C ₆ H ₅ NH ₂	Aniline			
CLN	N/A				
PFOA	N/A				
PFOS	N/A				

Note: N/A = Not available

		Simultaneou	ıs Treatment	Post Tr		
		Max flow 600 MLD	Min Flow 100 MLD	Max flow 600 MLD	Min Flow 100 MLD	Unit
Flow rate	2	600,000	100,000	600,000	100,000	m ³ /d
Service li	fe	15	15	15	15	yr
Interest r	ate	6	6	6	6	%
Present w	vorth (PW) factor	9.712	9.712	9.712	9.712	-
ENR inde	ex (May 2011)	9,027.23	9,027.23	9,027.23	9,027.23	-
ENR inde	ex (March 2016)	10,279.94	10,279.94	10,279.94	10,279.94	-
Adjustme	ent factor	1.1	1.1	1.1	1.1	-
Ozone do	Ozone dose		14.0	5.5	5.5	mg/L
		0.014	0.014	0.006	0.006	kg/m ³
Ozone ge	neration capacity	8400.0	1400.0	3300.0	550.0	kg/d
		18480	3080	7260	1210	lb/d
Contact t	ime	52	310	0.56	3.34	min
Contact c	chamber volume	-	-	-	-	ft ³
CC*	Ozone generation	804,818,004.6	804,818,004.6	25,620,375.8	25,620,375.8	\$
00	Ozone contact chamber	0.0	0.0	0.0	0.0	\$
	Adjusted annual CC	94,368,071.9	94,368,071.9	3,004,089.7	3,004,089.7	\$
O&M	Ozone generation	3,406,220.2	1,071,734.2	2,174,376.4	465,311.9	\$
	Adjusted value	3,878,901.9	1,220,458.9	2,476,114.9	529,883.3	\$
Note:						

Table A.4 Cost analysis for single and PAC-catalyzed ozonation as ST and as PT

In CC:

Ozone generation = $0.0002x^3$ -1.5794x² + 4,424.4x + 214,180; where x = ozone generation capacity (lb/day) Adjusted annual CC = (Ozone generation + Ozone contact chamber)/FW factor) * Adjustment factor In O&M:

Ozone generation = $(-0.01 \times x^2) + 367.19 \times x + 35653$; where x = ozone generation capacity (lb/day) Adjusted value = Ozone generation * Adjustment factor



Figure A.1 Graphical summary of Chapter 3.



Figure A.2 Graphical summary of Chapter 4.



Figure A.3 Graphical summary of Chapters 5 and 6.

2. Canadian Science Publishing – License Terms and Conditions

/27/2020 https://marketplace.org	copyright.com/rs-ul-web/mp/license/2e0	345eb-bfe3-4e35-967e-5d4691709aae	/Daf06bde-1e25-415d-b0d2-0ffa17aefc
Copyright Clearance Market	place [™]		
Canadian Science Publis	hing - License Terms and C	onditions	
This is a License Agreement betwee ("CCC"). The license consists of you conditions.	en Mirna Alameddine ("You") and Canadia ir order details, the terms and conditions	in Science Publishing ("Publisher") provi provided by Canadian Science Publishin	ded by Copyright Clearance Center g, and the CCC terms and
All payments must be made in full	to CCC.		
Order Date Order license ID ISSN	24-jul-2020 1050626-1 1208-6029	Type of Use Publisher Portion	Republish in a thesis/dissertation Canadian Science Publishing Chapter/article
LICENSED CONTENT			
Publication Title	Canadian journal of civil engineering : Revue canadienne de génie civil	Country Rightsholder	Canada Canadian Science Publishing
Article Title	Bench to Full-Scale Enhanced Primary Treatment of Municipal Wastewater under Wet Weather Flow for Minimized Pollution Load: Evaluation of Chemical Addition and Process Control Indicators	Рибікаціон туре	egournai
Date	12/31/1995		
Language	English, French, French		
REQUEST DETAILS			
Portion Type	Chapter/article	Rights Requested	Main product and other
Page range(s)	All		compilations/derivative products
Total number of pages	39	Distribution	Worldwide
Format (select all that apply)	Electronic	Translation	Original language of publication
Who will republish the content?	Academic institution	Minor editing privileges?	Vec
Duration of Use	Life of current and all future editions	Incidental promotional use?	No
Lifetime Unit Quantity	Up to 499	Currency	CAD
NEW WORK DETAILS			
Title	Bench to Full-Scale Enhanced Primary Treatment of Municipal Wastewater under Wet Weather Flow for Minimized Pollution Load: Evaluation of Chemical Addition and Process Control Indicators	Institution name Expected presentation date	University of Alberta 2020-07-30
Instructor name	Mirna Alameddine		
ADDITIONAL DETAILS			
Order reference number	N/A	The requesting person / organization to appear on the license	Mirna Alameddine
REUSE CONTENT DETAIL	.S		
Title, description or numeric reference of the portion(s)	ench to Full-Scale Enhanced Primary Treatment of Municipal Wastewater under Wet Weather Flow for Minimized Pollution Load: Evaluation of Chemical Addition and Process Control Indicators	Title of the article/chapter the portion is from	Bench to Full-Scale Enhanced Primary Treatment of Municipal Wastewater under Wet Weather Flow for Minimized Pollution Load: Evaluation of Chemical Addition and Process Control Indicators

https://marketplace.copyright.com/rs-ui-web/mp/license/2e0845eb-bfe3-4e35-967e-5d4691709aae/0af06bde-1e25-415d-b0d2-0ffa17aefc94

1/3

7/27/2020 https://marketplace.copyright.com/rs-ul-web/mp/license/2e0845eb-bfe3-4e35-967e-5d4691709aae/0af06bde-1e25-415d-b0d2-0ffa17aefc...

Editor of portion(s)	Gamal El-Din, Mohamed; Shaikh, Mohammed Zakee; Al Umairi, Abdul Rahim; Alameddine, Mirna	Author of portion(s)	Gamal El-Din, Mohamed; Shaikh, Mohammed Zakee; Al Umairi, Abdul Rahim; Alameddine, Mirna	
Volume of serial or monograph	N/A	Issue, if republishing an article	N/A	
Page or page range of portion	All	from a serial		
		Publication date of portion	2020-04-14	

CCC Republication Terms and Conditions

- Description of Service; Defined Terms. This Republication License enables the User to obtain licenses for republication of one or more copyrighted works as described in detail on the relevant Order Confirmation (the "Work(s)"). Copyright Clearance Center, Inc. ("CCC") grants licenses through the Service on behalf of the rightsholder identified on the Order Confirmation (the "Rightsholder"). "Republication", as used herein, generally means the inclusion of a Work, in whole or in part, in a new work or works, also as described on the Order Confirmation. "User", as used herein, means the person or entity making such republication.
- 2. The terms set forth in the relevant Order Confirmation, and any terms set by the Rightsholder with respect to a particular Work, govern the terms of use of Works in connection with the Service. By using the Service, the person transacting for a republication license on behalf of the User represents and warrants that he/she/it (a) has been duly authorized by the User to accept, and hereby does accept, all such terms and conditions on behalf of User, and (b) shall inform User of all such terms and conditions. In the event such person is a "freelancer" or other third party independent of User and CCC, such party shall be deemed jointly a "User" for purposes of these terms and conditions. In any event, User shall be deemed to have accepted and agreed to all such terms and conditions if User republishes the Work in any fashion.
- 3. Scope of License; Limitations and Obligations.
 - 3.1. All Works and all rights therein, including copyright rights, remain the sole and exclusive property of the Rightsholder. The license created by the exchange of an Order Confirmation (and/or any invoice) and payment by User of the full amount set forth on that document includes only those rights expressly set forth in the Order Confirmation and in these terms and conditions, and conveys no other rights in the Work(s) to User. All rights not expressly granted are hereby reserved.
 - 3.2. General Payment Terms: You may pay by credit card or through an account with us payable at the end of the month. If you and we agree that you may establish a standing account with CCC, then the following terms apply: Remit Payment to: Copyright Clearance Center, 29118 Network Place, Chicago, IL 60673-1291. Payments Due: Invoices are payable upon their delivery to you (or upon our notice to you that they are available to you for downloading). After 30 days, outstanding amounts will be subject to a service charge of 1-1/2% per month or, if less, the maximum rate allowed by applicable law. Unless otherwise specifically set forth in the Order Confirmation or in a separate written agreement signed by CCC, invoices are due and payable on "net 30" terms. While User may exercise the rights licensed immediately upon issuance of the Order Confirmation, the license is automatically revoked and is null and void, as if it had never been issued, if complete payment for the license is not received on a timely basis either from User directly or through a payment agent, such as a credit card company.
 - 3.3. Unless otherwise provided in the Order Confirmation, any grant of rights to User (i) is "one-time" (including the editions and product family specified in the license), (ii) is non-exclusive and non-transferable and (iii) is subject to any and all limitations and restrictions (such as, but not limited to, limitations on duration of use or circulation) included in the Order Confirmation or invoice and/or in these terms and conditions. Upon completion of the licensed use, User shall either secure a new permission for further use of the Work(s) or immediately cease any new use of the Work(s) and shall render inaccessible (such as by deleting or by removing or severing links or other locators) any further copies of the Work (except for copies printed on paper in accordance with this license and still in User's stock at the end of such period).
 - 3.4. In the event that the material for which a republication license is sought includes third party materials (such as photographs, illustrations, graphs, inserts and similar materials) which are identified in such material as having been used by permission, User is responsible for identifying, and seeking separate licenses (under this Service or otherwise) for, any of such third party materials; without a separate license, such third party materials mote used.
 - 3.5. Use of proper copyright notice for a Work is required as a condition of any license granted under the Service. Unless otherwise provided in the Order Confirmation, a proper copyright notice will read substantially as follows: "Republished with permission of [Rightsholder's name], from [Work's title, author, volume, edition number and year of copyright]; permission conveyed through Copyright Clearance Center, Inc. " Such notice must be provided in a reasonably legible font size and must be placed either immediately adjacent to the Work as used (for example, as part of a by-line or footnote but not as a separate electronic link) or in the place where substantially all other credits or notices for the new work containing the republished Work are located. Failure to include the required notice results in loss to the Rightsholder and CCC, and the User shall be liable to pay liquidated damages for each such failure equal to twice the use fee specified in the Order Confirmation, in addition to the use fee itself and any other fees and charges specified.
 - 3.6. User may only make alterations to the Work if and as expressly set forth in the Order Confirmation. No Work may be used in any way that is defamatory, violates the rights of third parties (including such third parties' rights of copyright, privacy, publicity, or other tangible or intangible property), or is otherwise illegal, sexually explicit or obscene. In addition, User may not conjoin a Work with any other material that may result in damage to the reputation of the Rightsholder. User agrees to inform CCC if it becomes aware of any infringement of any rights in a Work and to cooperate with any reasonable request of CCC or the Rightsholder in connection therewith.
- 4. Indemnity. User hereby indemnifies and agrees to defend the Rightsholder and CCC, and their respective employees and directors, against all claims, liability, damages, costs and expenses, including legal fees and expenses, arising out of any use of a Work beyond the scope of the rights granted herein, or any use of a Work which has been altered in any unauthorized way by User, including claims of defamation or infringement of rights of copyright, publicity, privacy or other tangible or intangible property.

https://marketplace.copyright.com/rs-ul-web/mp/license/2e0845eb-bfe3-4e35-967e-5d4691709aae/0af06bde-1e25-415d-b0d2-0ffa17aefc94

2/3

7/27/2020 https://marketplace.copyright.com/rs-ui-web/mp/i/cense/2e0845eb-bfe3-4e35-967e-5d4691709aae/0af06bde-1e25-415d-b0d2-0ffa17aefc...

- 5. Limitation of Liability. UNDER NO CIRCUMSTANCES WILL CCC OR THE RIGHTSHOLDER BE LIABLE FOR ANY DIRECT, INDIRECT, CONSEQUENTIAL OR INCIDENTAL DAMAGES (INCLUDING WITHOUT LIMITATION DAMAGES FOR LOSS OF BUSINESS PROFITS OR INFORMATION, OR FOR BUSINESS INTERRUPTION) ARISING OUT OF THE USE OR INABILITY TO USE A WORK, EVEN IF ONE OF THEM HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES. In any event, the total liability of the Rightsholder and CCC (including their respective employees and directors) shall not exceed the total amount actually paid by User for this license. User assumes full liability for the actions and omissions of its principals, employees, agents, affiliates, successors and assigns.
- 6. Limited Warranties. THE WORK(S) AND RIGHT(S) ARE PROVIDED *AS IS*. CCC HAS THE RIGHT TO GRANT TO USER THE RIGHTS GRANTED IN THE ORDER CONFIRMATION DOCUMENT. CCC AND THE RIGHTSHOLDER DISCLAIM ALL OTHER WARRANTIES RELATING TO THE WORK(S) AND RIGHT(S), ETHER EXPRESS OR IMPLIED, INCLUDING WITHOUT LIMITATION IMPLIED WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE. ADDITIONAL RIGHTS MAY BE REQUIRED TO USE ILLUSTRATIONS, GRAPHS, PHOTOGRAPHS, ABSTRACTS, INSERTS OR OTHER PORTIONS OF THE WORK (AS OPPOSED TO THE ENTIRE WORK) IN A MANNER CONTEMPLATED BY USER, USER UNDERSTANDS AND AGREES THAT NEITHER CCC NOR THE RIGHTSHOLDER MAY HAVE SUCH ADDITIONAL RIGHTS TO GRANT.
- 7. Effect of Breach. Any failure by User to pay any amount when due, or any use by User of a Work beyond the scope of the license set forth in the Order Confirmation and/or these terms and conditions, shall be a material breach of the license created by the Order Confirmation and these terms and conditions. Any breach not cured within 30 days of written notice thereof shall result in immediate termination of such license without further notice. Any unauthorized (but licensable) use of a Work that is terminated immediately upon notice thereof may be liquidated by payment of the Rightsholder's ordinary license price therefor; any unauthorized (and unlicensable) use that is not terminated immediately for any reason (including, for example, because materials containing the Work cannot reasonably be recalled) will be subject to all remedies available at law or in equity, but in no event to a payment of less than three times the Rightsholder's ordinary license price for the most closely analogous licensable use plus Rightsholder's and/or CCC's costs and expenses incurred in collecting such payment.
- 8. Miscellaneous.
 - 8.1. User acknowledges that CCC may, from time to time, make changes or additions to the Service or to these terms and conditions, and CCC reserves the right to send notice to the User by electronic mail or otherwise for the purposes of notifying User of such changes or additions; provided that any such changes or additions shall not apply to permissions already secured and paid for.
 - 8.2. Use of User-related information collected through the Service is governed by CCC's privacy policy, available online here:https://marketplace.copyright.com/rs-ui-web/mp/privacy-policy
 - 8.3. The licensing transaction described in the Order Confirmation is personal to User. Therefore, User may not assign or transfer to any other person (whether a natural person or an organization of any kind) the license created by the Order Confirmation and these terms and conditions or any rights granted hereunder; provided, however, that User may assign such license in its entirety on written notice to CCC in the event of a transfer of all or substantially all of User's rights in the new material which includes the Work(s) licensed under this Service.
 - 8.4. No amendment or waiver of any terms is binding unless set forth in writing and signed by the parties. The Rightsholder and CCC hereby object to any terms contained in any writing prepared by the User or its principals, employees, agents or affiliates and purporting to govern or otherwise relate to the licensing transaction described in the Order Confirmation, which terms are in any way inconsistent with any terms set forth in the Order Confirmation and/or in these terms and conditions or CCC's standard operating procedures, whether such writing is prepared prior to, simultaneously with or subsequent to the Order Confirmation, and whether such writing appears on a copy of the Order Confirmation or in a separate instrument.
 - 8.5. The licensing transaction described in the Order Confirmation document shall be governed by and construed under the law of the State of New York, USA, without regard to the principles thereof of conflicts of law. Any case, controversy, suit, action, or proceeding arising out of, in connection with, or related to such licensing transaction shall be brought, at CCC's sole discretion, in any federal or state court located in the County of New York, USA, or in any federal or state court whose geographical jurisdiction covers the location of the Rightsholder set forth in the Order Confirmation. The parties expressly submit to the personal jurisdiction and venue of each such federal or state court. If you have any comments or questions about the Service or Copyright Clearance Center, please contact us at 978-750-8400 or send an e-mail to support@copyright.com.

v 1.1

https://marketplace.copyright.com/rs-ul-web/mp/license/2e0845eb-bfe3-4e35-967e-5d4691709aae/0af06bde-1e25-415d-b0d2-0ffa17aefc94

3/3